Complementary and Alternative Medicine Use in Australia: A National Population-Based Study

A thesis submitted in fulfilment of the requirements for the degree of

Doctor of Philosophy

Anthony Lin Zhang

(B.Med., M.P.H.)

Division of Chinese Medicine School of Health Sciences RMIT University August 2006

DECLARATION

I certify that except where due acknowledgement has been made, the work is that of the author alone; the work has not been submitted previously, in whole or in part, to qualify for any other academic awards; the content of the thesis is the result of work which has been carried out since the official commencement date of the approved research program; and, any editorial work, paid or unpaid, carried out by a third party is acknowledged.

Anthony Lin Zhang_____

Date: _____

ACKNOWLEDGEMENTS

First of all, I would like to express my sincere thanks to my senior supervisor Professor Charlie Changli Xue, and supervisors, Professor David Frederick Story and Professor Vivian Lin for their invaluable guidance, support and encouragement. I am forever grateful and thankful for having them supervise my study.

I would also like to thank Associate Professor Cliff Da Costa for his comments, stimulating ideas and generosity in helping with the statistical analyses. Thanks also to population health experts and complementary and alternative medicine researchers, in particular Ms. Anne-Louise Carlton, Associate Professor Chunguang Li, Professor Marc Cohen, Associate Professor Barbara Polus, Dr. Ray Myers, Dr. Henry Liang, Dr Sharon Willcox, Dr Michael Ackland, Mr. Adrian Serraglio, Ms. Jodie Avery, and many others who had provided advice on the design of the study. Thanks to the Wallis Consulting Group for conducting the telephone interviews for this study. Thanks also to all participants in the survey for sharing their experiences of CAM use. Special thanks to my RMIT colleagues (Drs. Thomas Cheung, Wenyu Zhou, Ye Shen and many others) and all friends for their friendship and support.

I thank the Australian Government for providing the scholarship for my research candidature (Australian Postgraduate Award), and the project funding provided by the Divisions of Chinese Medicine and Osteopathy at RMIT University, the Sydney Institute of Traditional Chinese Medicine, the Chiropractor's Association of Australia, and the Australian Acupuncture & Chinese Medicine Association.

Last but not the least, I am grateful and indebted to my parents and my brother for their endless support, care and love throughout the years.

TABLE OF CONTENTS

DEC	CLARATION.		i
ACI	KNOWLEDGH	EMENTS	ii
TAI	BLE OF CONT	TENTS	iii
PUE	BLICATIONS.		viii
ABI	BREVIATION	S	ix
LIS	T OF TABLES	 }	xi
LIS	T OF FIGURE	'S	xiv
SUN	MARY		1
СН	APTER 1 IN	TRODUCTION	5
1 1	Deckground		
1.1	Dackground.		
1.2	Kationale		
1.3	Objectives ar	nd Research Questions	9
1.4	Significance	of the Study	11
1.5	Organisation	of the Thesis	13
CHA	APTER 2. LI	TERATURE REVIEW	15
2.1	Definition of	Complementary and Alternative Medicine (CAM)	15
2.2	Classification	n of CAM Modalities	
2.3	Prevalence of	f CAM Use	
	2.3.1 Fore	word	
	2.3.2 CAN	1 Utilisation Worldwide: A Systematic Review	23
	2.3.2.1	Methodology	23
	2.3.2.2	Results	27
	2.3.2.3	Discussion	
	2.3.3 CAN	1 Utilisation in Australia: A Systematic Review	
	2.3.3.1	Methodology	<i>3</i> 8
	2.3.3.2	Results	
	2.3.3.3	Discussion	
	2.3.4 Chin	ese Medicine Utilisation Worldwide: A Systematic Review	57
	2.3.4.1	Methodology	57
	2.3.4.2	Results	61
	2.3.4.3	Discussion	
2.4	Summary of	the Literature Review	73
CHA	APTER 3. M	ETHODOLOGY	76
3.1	Foreword		76
3.2	Sampling Me	thods	77
	1 0 -		

	3.2.1 Sa	mple Size	77
	3.2.2 Sa	mple Selection	79
3.3	Developme	nt of Survey Instrument	
3.4	Computer-	Assisted Telephone Interview	
	3.4.1 Ra	ndom Digit Dialling	85
	3.4.2 Int	erviewer Training	86
	3.4.3 Qu	ality Control Procedures	87
	3.4.4 Int	erview Procedures	
3.5	Pilot Study		91
	3.5.1 For	reword	91
	3.5.2 Re	sults of the Pilot Study	
	3.5.3 Dis	scussions and Finalisation of the Survey Instrument	94
3.6	Data Analy	'sis	
	3.6.1 Da	ta Consolidation and Missing Data	96
	3.6.2 Re	liability and Validity of Survey Instrument	
	3.6.3 Da	ta Weighting Procedures	
	3.6.4 Sta	ttistical Analyses	104
	3.6.4.	1 Descriptive Analyses	104
	3.6.4.2	2 Inferential Analyses	
	3.6.4.	3 Multivariate Logistic Regression	
CHA	APTER 4.	NATIONAL PROFILE OF CAM UTILISATION	
4.1	Results		109
	4.1.1 For	reword	
	4.1.2 Su	rvey Response Rate	110
	4.1.3 Ch	aracteristics and Representativeness of Survey Participants	
	4.1.4 Us	e of Common Forms of CAM	115
	4.1.4.	1 Prevalence of CAM Use	115
	4.1.4.2	2 Characteristics of CAM Users	117
	4.1.5 Vis	sits to CAM Practitioners	
	4.1.5.	<i>1</i> Prevalence of Visiting CAM Practitioners	
	4.1.5.2	2 Frequency of CAM Practitioners Visits	126
	4.1.5	3 Characteristics of People Visiting CAM Practitioners	
	4.1.6 Us	e of Other Forms of CAM	
	4.1.6.	l Prayer	
	4.1.6.2	2 Other Forms of CAM	134
	4.1.7 Ec	onomic Considerations of CAM Use	
	4.1.7.	1 Overview	135
	417	2 Regional Differences in CAM Expenditure	137
	7.1./.2		
4.2	Discussion.		
4.2	Discussion. 4.2.1 Pre	evalence of CAM Use	 139 139

	4.2.1.2	Comparison with Overseas Studies	144
	4.2.2 Char	acteristics of CAM Users	148
	4.2.3 Econ	omic Considerations of CAM Use	151
CH	APTER 5. RI	EGIONAL DIFFERENCES IN CAM USE	154
5.1	Results		154
	5.1.1 Over	view	154
	5.1.2 Profi	le of CAM Users in New South Wales	158
	5.1.3 Profi	le of CAM Users in Victoria	160
	5.1.4 Profi	le of CAM Users in Queensland	162
	5.1.5 Profi	le of CAM Users in South Australia	164
	5.1.6 Profi	le of CAM Users in Western Australia	166
	5.1.7 Profi	le of CAM Users in Other States/Territories	168
	5.1.8 Geog	graphical Summary of CAM Use in Australia	170
5.2	Discussion		171
CH	APTER 6. US	SE OF FOUR SPECIFIC FORMS OF CAM: ACUPUNCTURE,	
CH	INESE HERBA	AL MEDICINE, CHIROPRACTIC AND OSTEOPATHY	173
6.1	Results		173
	6.1.1 Chin	ese Medicine (Acupuncture and Chinese Herbal Medicine)	173
	6.1.1.1	Overview	173
	6.1.1.2	Total and Specific Prevalence of Chinese Medicine Use	174
	6.1.1.3	Frequency of Practitioner Visits and Expenditure	177
	6.1.1.4	Acupuncture Treatment Provided by Medical Doctors	179
	6.1.1.5	Characteristics of Chinese Medicine Users	180
	6.1.1.6	Chinese Medicine Utilisation–Regression Analyses	182
	6.1.1.7	Rationale for Use of Acupuncture and Chinese Herbal Medicine	185
	6.1.1.8	Outcomes of Use of Acupuncture and Chinese Herbal Medicine	187
	6.1.1.9	Safety Concerns of Acupuncture and Chinese Herbal Medicine Use	e 191
	6.1.1.10	Referral Pattern and Future Use of Acupuncture	192
	6.1.1.11	Referral Pattern and Future Use of Chinese Herbal Medicine	195
	6.1.2 Chire	opractic	197
	6.1.2.1	Overview	197
	6.1.2.2	Rationale for the Use of Chiropractic	199
	6.1.2.3	Use of Chiropractic–Regression Analyses	201
	6.1.2.4	Outcomes and Safety Concerns of Use of Chiropractic	203
	6.1.2.5	Referral Pattern and Future Use of Chiropractic	205
	6.1.3 Oste	opathy	207
	6.1.3.1	Overview	207
	6.1.3.2	<i>Kationale for the Use of Osteopathy</i>	209
	6.1.3.3	Use of Usteopathy–Regression Analyses	210
	6.1.3.4	Outcomes and Safety Concerns of Use of Osteopathy	211
	0.1.3.3	Kejerral Pattern and Future Use of Osteopathy	212

6.2	Discussion	
	6.2.1 Chinese Medicine (Acupuncture and Chinese Herbal Medicine)	
	6.2.2 Chiropractic	
	6.2.3 Osteopathy	
CHA	APTER 7. FACTORS RELATED TO CAM USE	
7.1	Results	
	7.1.1 Rationale for CAM Use	
	7.1.2 Regression Analyses on CAM Use	
	7.1.2.1 Variables that Related to the Use of CAM	
	7.1.2.2 Variables that Related to Visiting CAM Practitioners	
	7.1.3 Use of CAM and Western Medicine	
	7.1.3.1 Use of Western Medicine	
	7.1.3.2 Concurrent Use of CAM and Western Medicine	
	7.1.3.3 Communication between Consumers and Practitioners	
	7.1.4 Health Insurance and the Use of CAM	
	7.1.4.1 National Health Insurance Coverage of CAM	
	7.1.4.2 Private Health Insurance Coverage of CAM	
	7.1.5 Regulation and the Use of CAM	
	7.1.5.1 CAM Products and Government Regulation	
	7.1.5.2 CAM Practitioner Registration	
	7.1.6 Lifetime Use of CAM	
	7.1.7 Future Use of CAM	
7.2	Discussion	
	7.2.1 Rationale for CAM Use	
	7.2.2 Regression Analyses on CAM Use	
	7.2.3 Impact of Health Insurance on CAM Use	
	7.2.4 Impact of Regulation on CAM Use	
	7.2.5 Risks Associated with CAM Use	
	7.2.6 Lifetime Use and Future Use of CAM	
CHA	APTER 8. GENERAL DISCUSSION, CONCLUSIONS AND POLICY	
IMP	PLICATIONS	
8.1	Overview of the CAM Survey	
8.2	Limitations of the CAM Survey	
8.3	General Conclusions	
8.4	Policy Implications	
8.5	Recommendations and Future Studies	
REF	FERENCES	
APP	PENDICES	

Appendix A Brief explanations of the 17 forms of CAM that used in the survey	311
Appendix B Survey questionnaire (sample)	313
Appendix C Computer programming scripts for the interview (sample)	315
Appendix D Initial telephone contact (with plain language statement)	317
Appendix E Plain language statement (print version)	318
Appendix F Human ethics study approval form	320
Appendix G Additional statistical outputs	321
Appendix G1 Variables employed in the survey	321
Appendix G2 Frequency of visits to CAM practitioners (unadjusted data)	325
Appendix G3 Statistics of frequency of CAM practitioner visits (adjusted data)	326
Appendix G4 Expenditure on CAM products (unadjusted data)	327
Appendix G5 Expenditure on CAM practitioners (unadjusted data)	328
Appendix G6 Expenditure on CAM-related items (unadjusted data)	329
Appendix G7 Statistics of expenditure on CAM (unadjusted data)	330
Appendix G8 Statistics of expenditure on CAM (adjusted data)	331
Appendix G9 Individual expenses on four forms of CAM (unadjusted data)	332
Appendix G10 Statistics of expenses on four forms of CAM (adjusted data)	333
Appendix G11 Inter-correlations of CAM use and potential predictor variables	334
Appendix G12 Statistics of a regression model on CAM use	336

PUBLICATIONS

Publications arising from this thesis

- Xue CCL, Zhang AL, Lin V, Da Costa C, & Story DF. Complementary and Alternative Medicine Use in Australia: Results of a National Population-Based Survey in 2005. *Journal of Alternative and Complementary Medicine*. (Submitted Nov 2006).
- Xue CCL, Zhang L, Lin V, & Story DF. The Use of Complementary and Alternative Medicine in Australia. *Health Issues*. Spring 2006, 88, p12-15.

Publications relevant to this thesis

• Xue CCL, Lee C, Karagiannis J, Li CG, Yang AWH, Zhang L, and Story DF. Public Attitudes to Chinese Medicine in Melbourne, Australia. *Journal of Complementary and Integrative Medicine*. 2005 Vol. 2: No.1, Article 8.

Conference proceedings

- Zhang L, Lin V, Story DF, Xue CCL. Usage of Complementary and Alternative Medicine by Elderly Australians: A National Population-Based Study in 2005. The 3rd International Conference on Healthy Ageing & Longevity, Melbourne. 13-15 October 2006.
- Xue CCL, Zhang L, Lin V, Story DF. Usage of Complementary and Chinese Medicine in Australia: A National Population-Based Study in 2005. International Conference & Exhibition of the Modernization of Chinese Medicine & Health Products (ICMCM) Conference, Hong Kong. 17-18 August 2006.
- Xue CCL, Zhang L, Lin V, Story DF. A National Population Study on Chinese Medicine Usage: 2004-2005. Australasian Acupuncture & Chinese Medicine Annual Conference, Adelaide. 19-21 May 2006.
- Xue CCL, Zhang L, Story DF, Lin V, Da Costa, C. Current Usage of Complementary Medicines in Australia: A National Study. Harmonisation of Traditional and Modern Medicine: An International Symposium, Melbourne. 12-14 December 2005.
- Zhang L, Xue CCL, Story DF, Lin V, Da Costa, C. Investigating Utilisation of Complementary Medicines in Australia: Methodological Approaches. Harmonisation of Traditional and Modern Medicine: An International Symposium, Melbourne. 12-14 December 2005.
- Zhang L, Xue CCL, Story DF. Development of a Population-Based Study to Investigate the Use of Complementary Medicine in Australia. The Inaugural School Research Conference, School of Health Science, RMIT University, Melbourne. 31st May 2005.

ABBREVIATIONS

Acronym	Description	
ABS	Australian Bureau of Statistics	
ACT	Australian Capital Territory	
ACU	Acupuncture	
AIHW	Australian Institute of Health and Welfare	
AROMA	Aromatherapy	
CAM	Complementary and Alternative Medicine	
CAMP	Complementary and Alternative Medicine Practitioner	
CATI	Computer Assisted Telephone Interview	
CAPI	Computer Assisted Personal Interview	
CI	Confidence Interval	
CHM	Chinese Herbal Medicine	
СНМР	Chinese Herbal Medicine Practitioner	
CHIRO	Chiropractic	
СМ	Chinese Medicine	
CMDT	Chinese Medicine Dietary Therapy	
CMRBVic	Chinese Medicine Registration Board of Victoria	
COPD	Chronic Obstructive Pulmonary Disease	
СТМ	Chinese Therapeutic Massage	
DoHA	Australian Department of Health and Ageing	
HM	Herbal Medicine	
HOMEO	Homeopathy	
DHS	Department of Human Services	
GP	General Practitioner	
MBS	Medicare Benefits Schedule	
NATURO	Naturopathy	
NCCAM	National Center for Complementary and Alternative Medicine (US)	
NHS	National Health System (Medicare)	
NHMRC	National Health and Medical Research Council (Australia)	

Acronym	Description
NSW	New South Wales
NT	Northern Territory
OR	Odds Ratio
OSTEO	Osteopathy
OTC	Over-the-counter
PAC	Percentage Accurate in Classification
PBS	Pharmaceutical Benefits Scheme
PHI	Private Health Insurance
QLD	Queensland
QGMATC	Qigong, Martial Act and Tai chi
RA	Rheumatoid Arthritis
RDD	Random Digit Dialling
RMIT	Royal Melbourne Institute of Technology
RSE	Relative Standard Error
SA	South Australia
SD	Standard Deviation
SE	Standard Error
SPSS	Statistical Package for Social Sciences
TAS	Tasmania
TGA	Therapeutic Goods Administration
UK	The United Kingdom
US	The United States of America
VIC	Victoria
WA	Western Australia
WHM	Western Herbal Medicine
WHMP	Western Herbal Medicine Practitioner
WHO	World Health Organisation
WM	Western Medicine
WTM	Western Therapeutic Massage

LIST OF TABLES

Table 2.1 CAM modalities classified in the National Center for Complementary and Alternative Medicine (NCCAM)	19
Table 2.2 CAM modalities and data collection method in Australian regional surveys	21
Table 2.3 Search strategy to identify CAM utilisation studies in MEDLINE	24
Table 2.4 Results of CAM utilisation studies in the general population	28
Table 2.5 An overview of CAM utilisation studies included in the systematic reviews	31
Table 2.6 Additional search strategy to identify Australian CAM utilisation studies	38
Table 2.7 CAM use among the Australian general population	43
Table 2.8 CAM use by children and women (patients) in Australia	46
Table 2.9 CAM use for cancer treatment in Australia	48
Table 2.10 CAM use for treatment of diseases other than cancer in Australia	51
Table 2.11 CAM use in other populations in Australia	55
Table 2.12 Search strategy to identify type I literature on Chinese medicine utilisation	58
Table 2.13 Chinese medicine use in general populations	62
Table 2.14 Chinese medicine use in defined populations	63
Table 2.15 Chinese medicine use in general populations (extracted from major CAM utilisation studies)	64
Table 3.1 Allocation of study samples in accordance with state and territory populations	79
Table 3.2 Gender and age quotas set in the study	80
Table 3.3 NCCAM classifications of the 17 selected forms of CAM in the survey	82
Table 3.4 List of "other forms of CAM" included in the survey	82
Table 3.5 Structure of the computer-assisted telephone interview questionnaire	83
Table 3.6 Basic demographic information on pilot study participants	92
Table 3.7 Item-total correlation matrix of the prevalence questions on 17 forms of	. 100
Table 3.8 Item-item correlation matrix of the prevalence questions on 17 forms of CAM	. 100
Table 3.9 Data weighting values applied in the data analyses	. 102

Table 3.10 An example of SPSS syntax for data weighting protocol	103
Table 4.1 Comparison of socio-demographic characteristics on survey participants and the Australian population	e 113
Table 4.2 The top ten countries of birth of survey participants	114
Table 4.3 Detailed education and employment background of survey participants	114
Table 4.4 Prevalence of the 17 investigated forms of CAM	116
Table 4.5 Socio-demographic characteristics on CAM users	118
Table 4.6 Socio-demographic characteristics on users of each of the 17 forms of CAM	121
Table 4.7 Percentage of CAM users that visited practitioners for 17 forms of CAM	124
Table 4.8 Percentage of participants that visited practitioners for 17 forms of CAM	125
Table 4.9 Frequency of visits to practitioners for 10 forms of CAM	127
Table 4.10 Characteristics of people who had visited CAM practitioners	129
Table 4.11 Characteristics of people who had visited practitioners for each of the 17 forms CAM	s of 130
Table 4.12 Socio-demographic characteristics on users of prayer	133
Table 4.13 The use of other forms of CAM	134
Table 4.14 National out-of-pocket expenditure on CAM	136
Table 4.15 Regional differences of out-of-pocket expenditure on CAM	137
Table 4.16 Comparison of CAM prevalence with previous Australian surveys	140
Table 4.17 Comparison of CAM prevalence with selected overseas CAM surveys	146
Table 5.1 Regional differences of use of 17 forms of CAM	156
Table 5.2 Regional differences of visits to practitioners for 17 forms of CAM	157
Table 5.3 Socio-demographic characteristics of CAM users in New South Wales	159
Table 5.4 Socio-demographic characteristics of CAM users in Victoria	161
Table 5.5 Socio-demographic characteristics of CAM users in Queensland	163
Table 5.6 Socio-demographic characteristics of CAM users in South Australia	165
Table 5.7 Socio-demographic characteristics of CAM users in Western Australia	167
Table 5.8 Socio-demographic characteristics of CAM users in other states/territories	169

LIST OF FIGURES

Figure 2.1	Prevalence of CAM use in seven countries	35
Figure 2.2	Prevalence of CAM use by children and women (patients) in Australia	45
Figure 2.3	Prevalence of CAM use by Australian cancer patients	49
Figure 2.4	Prevalence of CAM use for treatment of diseases other than cancer in Australia	53
Figure 2.5	Prevalence of Chinese medicine use in six countries	70
Figure 2.6	Prevalence of visits to Chinese medicine practitioners in three countries	71
Figure 3.1	Sample size calculation using Epi Info 6.0 software package	77
Figure 3.2	Survey development-process and quality-assurance protocols	37
Figure 3.3	Flowchart of computer-assisted telephone interview questions) 0
Figure 4.1	Flowchart of the final interview outcome	11
Figure 4.2	Frequency distribution of CAM expenditure data13	35
Figure 4.3	Prevalence of visits to CAM practitioners in the current survey and a recent South Australia survey	13
Figure 5.1	Prevalence of CAM use in Australia presented in a map17	70
Figure 6.1	Prevalence of Chinese medicine use in Australia presented in a map	76
Figure 7.1	Visits to CAM and Western medical practitioners	38
Figure 7.2	Concurrent use of CAM and Western medicine among different age groups 23	39
Figure 7.3	Gender difference in informing general practitioners about CAM use	40

SUMMARY

The use of various forms of complementary and alternative medicine (CAM) has been increasing steadily over the last a couple of decades in both developing and developed countries. Although there has been no national population-based study on CAM use in Australia, a number of regional studies have indicated the consistent trend of increasing CAM use. This has drawn attention from the community, health professionals and governments on issues associated with its use, such as quality, safety and the efficacy of CAM interventions. In addition, there have been questions about the validity of the reported prevalence data due to the heterogeneity of the CAM therapies and the way samples that were drawn for these studies.

This thesis presents findings of a series of systematic literature reviews of CAM usage internationally and in Australia, which provided the basis for the planning of a national population-based study. Most importantly, the thesis presents the key findings and detailed analyses of findings of this Australian CAM usage study based on a national representative sample.

The first systematic review considered provides comprehensive overview of the current CAM use in the general populations around the world. In the second review, concerning the situation in Australia, the usage of CAM by the general population in different regions and specific patient groups was evaluated. Finally, the use of all modalities of Chinese medicine worldwide was also systematically appraised. Although there are a number of methodological issues that might have an impact on the validity of the prevalence data of overall and specific CAM use, over the last decade, the use of the most common forms of CAM such as herbal medicine in a number of countries has increased markedly. In Australia, despite the fact that

1

there has been no national population-based study, existing literature on regional health studies suggests that, in general, a higher level of use of CAM therapies in Australia than in other Western countries.

The national population-based survey was planned to investigate the national profile of CAM use in Australia, and to identify possible regional differences. The survey employed a random digit dialling (RDD) sampling method and interviews were carried out using computer-assisted telephone interviews (CATI) for data collection. A sample size of 1,067 participants was calculated and allocated between the six Australian states and two territories in proportion to Australian census data. National quotas for age and gender were applied to enhance the representativeness of the study populations. After undertaking a pilot study to refine the survey instruments and the CATI program, the major survey was conducted between May and June 2005. Comparison of respondents' demographic statistics with the Australian Census data confirmed the representativeness of the study sample.

Through the three systematic reviews and consultation with experts in the field, a total of 17 CAM therapies were included in the study. The 17 forms of CAM consist of substance-based therapies, such as Chinese herbal medicine; manipulation therapies, such as chiropractic; nutrition therapies, such as multi-vitamins; exercise-based therapies, such as Qigong; and mind-body therapies, such as meditation.

The most striking finding of this study is that the prevalence of overall CAM use in Australia is considerably higher than estimated in previously regional studies. Over two thirds (68.9%, 95% CI: 66.1% to 71.7%) of participants had used at least one of the 17 CAM therapies in the previous 12 months. Of these CAM users, 64.0%, (95%CI: 60.5% to 67.5%) had visited a CAM practitioner, that is, 44.1% of the total survey participants. Concerning the specific use

of the 17 forms of CAM, the 10 most popular forms, in decreasing order, were clinical nutrition, Western massage therapy, meditation, Western herbal medicine, aromatherapy, chiropractic, yoga, naturopathy, acupuncture and Chinese herbal medicine.

The high prevalence of CAM may be partially explained by regional (state) variations. Overall, the state of New South Wales had the highest CAM use (72.1%), closely followed by Queensland (71.0%) and Victoria (69.8%), with South Australia being the lowest of prevalence (60.8%). In addition, significant regional differences were found for specific forms of CAM therapies. For example, the use of acupuncture and osteopathy was significantly lower in the states of Western Australia and South Australia. The prevalence of visits to acupuncturists, herbalists, naturopaths and osteopaths were also significantly lower in South Australia, compared to New South Wales and Victoria.

Consistent with the existing literature, the most common characteristics of CAM users were likely to be younger (aged from 18 to 34), female, employed, well educated, have private health insurance cover and have higher than average incomes. In terms of the frequency of practitioner visits and the expenditure related to CAM, the estimated number of visits to CAM practitioners by adult Australians in the 12 month period was comparable to the estimated number of visits to medical doctors. The annual "out of pocket" expenditure on CAM, nationally, was estimated as A\$4.13 billion.

Importantly, additional information was also sought about each of the four forms of CAM, which in Australia are subject to statutory regulation in one or more states: Chinese herbal medicine, acupuncture, chiropractic and osteopathy. The perceived and actual benefits and risks of using these four forms of CAM therapies and other matters related to its use, such as the attitudes towards the national and private health insurance coverage are presented.

In summary, this first national CAM survey in Australia gathered critical data concerning the usage of CAM, investigated potential regional differences and estimated expenditure associated with these therapies. It also suggested that while the concurrent use of Western medicine and CAM was common, the communication among the different health-care professions and the consumers had been grossly inadequate. These findings have a number of implications such as interpretation of the perceived benefits and risks associated with the frequent CAM use, the needs of product and practitioner regulation, economic impact on users, actions required from health insurance companies and related government agencies as well as research and education programs.

CHAPTER 1. INTRODUCTION

1.1 Background

The use of complementary and alternative medicine (CAM) is increasing in the Western world.¹⁻⁷ This trend is apparent from comparisons of the findings of population studies conducted in the United States (US) in 1990² and 1997³ and from national population health surveys in the US in 1999⁴ and 2002.¹ It is significant that a comprehensive national study is currently being planned by the US National Center for Complementary and Alternative Medicine (NCCAM).⁸ Surveys in other countries, including the United Kingdom,⁵ Canada⁶ and Japan⁷ have revealed a similar trend. In Australia, estimates of CAM usage have been based on the Omnibus Health Surveys conducted in the State of South Australia in 1993,⁹ 2000,¹⁰ and 2004.¹¹ The findings of these studies are generally in accord with those in the US.¹⁻⁴

Increasing CAM usage in the US, Australia and other Western countries is also reflected in the expenditure on CAM, which is of concern to consumers and governments. It has been estimated that a total of \$US27 billion was spent out-of-pocket on CAM in the US in 1997.³ Consistently, based on the South Australian Health Survey, ¹¹ an estimated \$AU1.8 billion was spent by Australians on CAM in 2004. It has also been estimated that there were approximately 1.9 million Western herbal medicine and naturopathy practitioner consultations in Australia in 2003.¹²

In recognition of the increasing use of CAM, state and national governments in Australia have recently introduced regulatory measures for certain forms of CAM. For example, the State of Victoria introduced statutory regulation of Chinese medicine in 2000,¹³ the first jurisdiction outside China to do so. Two other Australian states, Western Australia and New South Wales, are developing similar legislation. At the national level, the Office of Complementary Medicine was established within the Australian Therapeutic Goods Administration (TGA) in 1999 to align the regulation of some forms of CAM with that of conventional medicines.¹⁴ These actions are welcome; however, they relate only to certain specific forms of CAM and do not address a range of issues that are critical for public safety and confidence in CAM. Of particular concern are potential adverse interactions due to the concurrent use of CAM and conventional health-care services and products.

There has been no nationwide study to investigate CAM use in Australia, and the states of Queensland, New South Wales and Victoria have higher numbers of some types of CAM practitioners, such as Western herbal medicine practitioners and naturopaths, than other Australian states.¹² Thus, there is a view that there might be differences in the prevalence of use of different forms of CAM between states/territories of Australia, and, as mentioned above, Chinese herbal medicine and acupuncture have been subject to statutory regulation in Victoria since 2000, which may have influenced the use of Chinese medicine in this state.

1.2 Rationale

As mentioned above, in recent years numerous population studies have been conducted to determine the national prevalence and expenditure of CAM in many countries.^{1-7,9-11,15-17} In Australia, at least three regional studies were conducted between 1993 and 2004.⁹⁻¹¹ High levels of the use of CAM and expenditure on it were reported. However, systematic reviews ^{18,19} also suggested that there are considerable discrepancies in the findings of different studies on the use of various forms of CAM. In general, a clear definition of CAM and a system of classification of CAM modalities, an appropriated-designed survey instrument, as well as rigorous sampling and data collection methods are the major factors impacting on the internal and external validity of these studies. Considerable efforts have been made to address these issues in the current study.

Extrapolation of the findings of Australian regional studies ⁹⁻¹¹ to a national level could be problematic. As mentioned above, it is generally accepted and also observed in previous studies ¹² that some CAM practitioners are more concentrated in New South Wales, Victoria and Queensland. Other limitations of these regional studies include possible overlapping CAM modalities (e.g. Chinese medicine, herbal medicine and Ginseng) and also that the range of CAM modalities included were differed,⁹⁻¹¹ for example, the exclusion of some popular CAM modalities (e.g. massage, see Chapter 2, Table 2.2).

Although conventional medical therapies are still predominant in the treatment and management of diseases/conditions, there is evidence that many Australians are turning to various forms of CAM to supplement, and sometimes replace, conventional treatments. It is apparent that patients with chronic diseases, such as arthritis,²⁰ diabetes,²¹ asthma,²² and

chronic inflammatory conditions, commonly use CAM therapies.²³ A small number of highquality clinical trials have demonstrated that some forms of CAM have a marginal benefit for a narrow range of such chronic illnesses. These therapies include manipulation for low-back pain,²⁴ acupuncture for improving function and pain relief in osteoarthritis.²⁵ Thus, it is of interest to explore some common forms of CAM that being used by the general populations for a range of medical conditions.

Currently in Australia, limited studies were conducted to explore matters that related to CAM use. Thus, where relevant, matters such as the extent to which medical doctors are aware of their patients' use of CAM, the perceived and actual benefits and risks of CAM use, the relationship between CAM use and regulatory and health insurance coverage matters, and some related economic considerations, such as personal expenditure on CAM were also explored.

In summary, the need for a nationwide population-based study on CAM utilisation in Australia is warranted. This study was undertaken to investigate the national epidemiological profile of CAM use in Australia, and to reveal possible regional differences. A total of 17 CAM therapies were included in the study. They included substance-based therapies, such as Chinese herbal medicine; manipulation therapies, such as chiropractic; nutrition therapies, such as multi-vitamins; exercise-based therapies, such as Qigong; and mind-body therapies, such as meditation. Additional information was also sought about each of the four forms of CAM, which in Australia are subject to statutory regulation in one or more states: acupuncture, Chinese herbal medicine, chiropractic and osteopathy.

1.3 Objectives and Research Questions

The aims of the present study were to collect information about the use of CAM in the Australian general population, specifically, to investigate the prevalence of use, out-of-pocket expenditure on CAM and to evaluate other matters related CAM use such as health insurance coverage, regulation and, perceived and actual benefits and risks of using several specific forms of CAM.

The overall objectives and associated research questions are:

- 1. Delineate the current utilisation of CAM in Australia.
 - a. What is the total prevalence of use of CAM (practitioner visits and products)?
 - b. What are the relative popularities of different forms of CAM?
 - c. What is the prevalence and frequency of CAM practitioner visits?
 - d. Are there regional (state) differences in CAM use in Australia?
 - e. How is the public demand for CAM changing?
- 2. Estimate the expenditure associated with CAM products/services in Australia and assess the socio-economic characteristics of CAM users.
 - a. What is the out-of-pocket expenditure?
 - b. What are economic consequences and drivers for CAM use?
- 3. Explore the rationale for CAM use by consumers and their general concerns.
 - a. Are there differences in the forms of CAM used related to age, gender, national region or other socio-demographic characteristics?
 - b. What specific benefits and risks are perceived by the users of common forms of CAM?

- c. What benefits and risks do users of common forms of CAM consider they have actually received?
- d. What proportion of CAM users voluntarily discusses their use of CAM with their medical doctors? And what are the main reasons for non-disclosure of use?
- 4. Evaluate the extent, demand for and consequences of health insurance cover for CAM products/therapies.
 - a. What is the extent of health insurance coverage on CAM?
 - b. What proportion of users receives health insurance rebates for CAM?
 - c. What is the public demand for insurance coverage on CAM?
 - d. Does the availability of insurance cover affect the demand and utilisation of CAM?
- 5. Explore attitudes towards statutory CAM regulation.
 - a. Does statutory regulation have an impact on the demand/use of specific forms of CAM?
- 6. How do the survey results compare with those of other national and international studies?

1.4 Significance of the Study

The use of CAM therapies by Australians has increased markedly over the last two to three decades and it seems certain that this trend will continue. Given this situation, the present study was intended to address a broad range of issues concerning the use of CAM in Australia that have not been answered conclusively by previous studies. This study has created a national, population-based epidemiological profile of CAM use by the Australian adult population, including an investigation of differences between individual states and territories. Specifically, the study has provided reliable and comprehensive information on the prevalence of use of individual forms of CAM, rationales for using CAM and the perceptions of users of the risks and benefits of CAM therapies.

The findings also provide insights into the attitudes of CAM users and non-users of the provision of health insurance cover for CAM therapies and how the availability of insurance cover may influence the use of CAM products and services. A fuller understanding of such issues is essential for determining the need for, and the nature of initiatives by governments and health-care agencies to enhance public safety and confidence in CAM therapies. Such considerations should include adequate regulatory controls on CAM products and CAM practitioners, training of health-care professionals (CAM and conventional health-care providers), public education, and health services planning (including health economics). The study outcomes will also assist in defining additional research needs, ranging from socio-economic to scientific/clinical research related to safety and efficacy of CAM therapies.

The risks associated with the use of CAM are of growing concern. Therefore, important objectives of the proposed study were to investigate the level of understanding by users of the

potential for adverse effects of individual forms of CAM, the incidence and nature of adverse effects and, critically, the effectiveness of communication between patients and health-care practitioners. Thus, the study has provided information on the levels of awareness of healthcare professionals of their patients' use of CAM and the extent to which CAM users understand the risks and benefits of CAM. The outcomes reinforce the need for health-care professionals to be aware of their patients' therapeutic choices and the need for them to advise their patients about the appropriate and safe use of CAM.

The information collected in the study on the demographic characteristics of CAM users, on their rationales for deciding to use CAM, on the specific health conditions for which common forms of CAM are being used will provide a much-needed population data-set for health care researchers (for future population studies and basic and clinical research) and help identify priorities for further research on the use of CAM, including laboratory and clinical research on the efficacy and safety of CAM therapies. This is consistent with the recommendations of the Federal Government's Review of Complementary Therapies in the Australian Health-care System. ²⁶ In addition, the data collected will also guide government agencies in the development of further regulatory controls for CAM and provide consumers with relevant and timely advice on CAM use. The findings will also guide educational institutions concerned with both the training of conventional health-care practitioners and CAM practitioners in respect of designing curricula which recognise the broad range of therapies being used by the population.

1.5 Organisation of the Thesis

This thesis comprises eight chapters. Chapter 1 provides a brief overview of the background, rationale, objectives and details the specific research questions addressed by the study, as well as the significance of the study. Chapter 2 comprises three systematic reviews of CAM utilisation literature. The first review is concerned with global utilisation of CAM, the second summarises specifically CAM use among the general Australian population and by individuals with various diseases/conditions, and the third review exclusively covers Chinese medicine utilisation.

Chapter 3 discusses methodological issues associated with conducting a CAM population study in Australia. This chapter also addresses the approaches adopted to achieve a representative study population sample, the data collection process (computer-assisted telephone interviews), and the challenges of addressing commonly encountered problems, such as the definition and classification of CAM modalities.

The results and discussions of the national CAM population survey are divided into four sections: Chapter 4 presents a national profile of CAM utilisation in Australia through a multivariate analysis of the survey data. It examines CAM utilisation as a whole, including its prevalence of use and, prevalence and frequency of visiting CAM practitioners.

Chapter 5 presents detailed analyses of the regional differences in CAM use in Australia. Chapter 6 presents the findings on the four CAM modalities that are currently regulated in Australia, at either national or state level, that is, acupuncture, Chinese herbal medicine, chiropractic and osteopathy. Specific data related to these therapies such as expectations of the users, actual outcomes, and adverse events related to their use were sought from survey participants.

Chapter 7 presented additional results arising from the CAM survey including the expenditure on CAM, the rationales for CAM use, concurrent use of both CAM and Western medicine, and other matters such as, health insurance coverage and regulatory issues. The general discussion, conclusions, policy implications, limitations of the study and some possible solutions are included in Chapters 8.

CHAPTER 2. LITERATURE REVIEW

2.1 Definition of Complementary and Alternative Medicine (CAM)

There have been many debates surrounding a single definition of complementary and alternative medicine (CAM), but none have been conclusive.²⁷ Considering the diversity of CAM modalities, the definitions of CAM in the current literature are inconsistent, and vary significantly in different social contexts.

The term "complementary and alternative medicine" and its short form, "CAM", are terms most commonly used in the literature. "Complementary" generally mean that such medicine (including therapies and products) is being used in addition to conventional Western medicine. "Alternative" implies it is being used independently. Other terms are widespread, and are used interchangeably, to describe CAM.²⁸ They include: traditional, oriental, holistic, unconventional, non-conventional, unorthodox, essential and unproven medicine.

One of the major biologically based CAM therapies, "megavitamins", sometimes referred to dietary supplements is somewhat vaguely defined. A national CAM study in the US³ referred to these dietary supplements as vitamins not prescribed by a medical doctor. The US National Health Interview Survey¹ referred to these as vitamins used in excess of the recommended daily allowances (RDA) established by the National Academy of Sciences, Food and Nutrition Board.

Although the various definitions of CAM are debatable, after carefully examining the literature it is apparent that most of these definitions are made in relation to conventional (Western) medicine. This viewpoint is supported in a recent editorial published in the *Journal of Alternative and Complementary Medicine*.²⁹

In a US landmark study on CAM utilisation in 1990, the research team referred to CAM as "interventions neither taught widely in medical schools nor generally available in US hospitals".² At the same time, in an attempt to clarify the issue, the British Medical Association came up with the following:³⁰

"Both alternative and complementary types of medicine are described by the term "nonconventional". This is not an exclusionary term and illustrates the level of acceptance of these methods by the medical community. Complementary medicine is defined as those techniques which are used while the patient is using conventional ones. These include methods such as chiropractic and osteopathy. Alternative medicine includes techniques which are used to the exclusion of conventional medicine. These methods have been specified as incompatible with conventional practice (i.e. certain types of herbal remedies)."

In 1995, in a letter²⁷ sent to the *British Journal of General Practice*, Ernst and his colleagues commented that: "Complementary medicine is diagnosis, treatment and/or prevention which complements mainstream medicine by contributing to a common whole, by satisfying a demand not met by orthodoxy or by diversifying the conceptual frameworks of medicine". They further commented that an inclusive, positive approach, not defining what complementary medicine does not represent but what it actually means, would clearly be more constructive.

More recently, the definition of CAM appears to refer to a broad range of therapies and products. A document produced by the World Health Organisation (WHO)³¹ refers to CAM as "a broad set of health care practices that are not part of that country's own tradition and are not integrated into the dominant health care system". Similarly, the National Center for CAM (US) defined CAM as "a group of diverse health-care and medical practices and products that are not presently considered to be an integral part of conventional medicine".³² Perhaps a more encompassing definition of CAM was provided by the Office of Alternative Medicine (OAM) expert panel at the Conference on CAM Research Methodology in Bethesda, MD, in April 1995:³³

"A broad domain of healing recourses that encompasses all health systems, modalities, other than those intrinsic to the politically dominant health systems of a particular society or culture in a given historical period. CAM includes all such practices and ideas self-defined by their users as preventing or treating illness or promoting health and well-being. Boundaries within CAM and between the CAM domain and the domain of the dominant system are not always sharp or fixed".

2.2 Classification of CAM Modalities

No conclusive definition of CAM has been advanced. Debates continue in an attempt to classify CAM;²⁹ so do discussions on the CAM therapies that should be included to produce the overall CAM prevalence in various societies.³³ Most notably, there were critics of the abovementioned US landmark CAM survey in 1990.² Rosner ³⁴ argued for the inclusion of chiropractic as an alternative or complementary therapy to conventional medicine, as it has been largely accepted by the general public as a form of effective health care. Similarly, Fridman *et al.*³⁵ argued for the inclusion of relaxation techniques as a single form of CAM. Both views were published in the "*New England Journal of Medicine*", in the issue following publication of the 1990 CAM survey.²

Some so-called CAM modalities, such as Chinese medicine in China and ayurvedic medicine in India, have been used for thousands of years as medical systems in different countries. Some forms of CAM, such as osteopathy, homeopathy, and chiropractic, originated from, and to various extents have been incorporated into Western medical systems. However, some CAM may be considered outside the boundaries of current, well-established Western medical theory. An example is folk medicine which the medical community has rejected in the past, due to the lack of scientific evidence that meets mainstream medical standards. But to the extent that such therapies are better researched and produce convincing scientific evidence, they may become gradually accepted.

Excessively large numbers of modalities fall under the umbrella of the term CAM. Given the growing popularity of CAM use, the NCCAM³⁶ divided these into five major domains. Table 2.1 is a synthesis of these classifications.

Classifications	CAM therapies (NCCAM samples)
Alternative medical systems	 Homeopathic medicine Naturopathic medicine Traditional Chinese medicine Ayurvedic medicine
Mind-body interventions	• Meditation, prayer, mental healing, and therapies that use creative outlets such as art, music, or dance
Biologically based therapies	• Dietary supplements: vitamins, minerals, herbs or other botanicals, amino acids, and substances, such as enzymes, organ tissues, and metabolites
Manipulative and body-based methods	ChiropracticOsteopathicMassage
Energy therapies	• Qigong, reiki, therapeutic touch, pulsed fields, magnetic fields

Table 2.1 CAM modalities classified in the National Center for Complementary and Alternative Medicine (NCCAM)

Differences in the comprehensiveness of CAM modalities can be observed by making comparisons between two recent studies conducted in the US and Australia.^{1,11} The South Australian health survey¹¹ employed a prompt card in a face-to-face interview and asked participants whether they had used any one of seven forms of CAM or "other CAM". The 2002 US National Health Interview Survey¹ included questions on 27 types of CAM therapies, which had been considered to be commonly used in the US. Comparisons between an investigation of popular forms of CAM only, and a relatively comprehensive classification of CAM deserves further deliberation.

Most surveys on CAM prevalence used a questionnaire with a list of common therapies. The list has been adapted from previous studies and/or aimed to reflect the diverse cultures or make up of the population being studied. However, the current thesis does not provide judgement on how many on which CAM modalities comprise the definitive spectrum. In order to make comparisons with the existing classifications of CAM in Australian, a systematic search on published studies on CAM usage in Australia was conducted.

A total of five studies^{9-11,37,38} were identified as Australian regional surveys based on random or representative samples. The detailed methodologies and CAM modalities in these studies are summarised in Table 2.2. The CAM modalities used in these surveys, not surprisingly, varied considerably, and even changed over time in studies.⁹⁻¹¹ Nevertheless, this methodology comparison provided the essential data of CAM usage in the Australian context, and formed the basis of the list of CAM therapies used in the present study. Internationally, two systemic reviews on CAM utilisation in different countries that based on representative populations were referred,^{18,19} and a number of questionnaires that used in CAM utilisation surveys worldwide were consulted.

Author (year)	Place of study (year)	Sample (n)	Data collection method	Forms of CAM included in studies
				1. vitamins;
				2. mineral supplements;
				3. herbal medicines;
				4. aromatherapy oils;
				5. Chinese medicines;
		Representative	Omnibus Health	6. homeopathic medicines;
MacLennan	South Australia	South Australian	Survey: face-to-face	7. other;
<i>et al.</i> $(2006)^{11}$	(2004)	aged ≥ 15 (3,015)	interview	8. soy products.
				1-7 as for MacLennan <i>et al</i> . 2006 plus
MacLennan	South Australia			8. evening primrose oil;
<i>et al.</i> $(2002)^{10}$	(2000)	As above (3,027)	As above	9. ginseng.
				1-7 as for MacLennan <i>et al</i> . 2006 plus
				8. evening primrose oil;
				9. ginseng;
MacLennan	South Australia			10. PMTese;
<i>et al.</i> (1996) ⁹	(1993)	As above (3,004)	As above	11. Esten.
				1. vitamins/mineral & supplements;
				2. herbal therapy; 3. aromatherapy;
				4. traditional Chinese medicine; 5. homeopathy;
	Rural region of	Randomly selected		6. meditation; 7. iridology; 8. chiropractic;
Wilkinson	New South	from electoral		9. acupuncture; 10. reflexology; 11.massage;
<i>et al.</i> $(2001)^{37}$	Wales (1999)	database (300)	Postal questionnaire	12. naturopathy; 13. osteopathy; 14 other.
				1. nutritional (18 items listed);
	North coast of			2. plant medicine (2 items listed);
Kermode	New South	Randomly selected		3. homeopathic (4 items listed);
<i>et al.</i> (1998) ³⁸	Wales (1995)	adults (645)	Telephone interview	4. other.

Table 2.2 CAM modalities and data collection method in Australian regional surveys
2.3 Prevalence of CAM Use

2.3.1 Foreword

Despite the difficulties in defining CAM and describing the classifications of CAM modalities, the utilisation of CAM has been widely researched and there is extensive literature on the subject. In 2001, there were at least nine survey studies on the use of CAM among the general population of the US.³⁹ In Australia, CAM use among the general population has been investigated in sequential health surveys in the states of South Australia,⁹⁻¹¹ and New South Wales.^{37,38}

To design the current study, it was essential to conduct a systematic review of the relevant literature. Presented in this chapter are three systematic reviews covering the following topics: 1) CAM use in the general population around the world (mainly Western English-speaking countries, due to the limitation of getting published English literature in some countries); 2) CAM used specifically in Australia. This includes CAM use among the general population and in defined populations, such as cancer patients and adolescents; 3) a purposely-designed systematic review of Chinese medicine. This is considered important because of the extraordinary popularity of Chinese medicine use around the world, and the instigation of Chinese medicine statutory regulation in some states in Australia. All identified studies are presented in corresponding tables in this chapter but may not be fully discussed.

2.3.2 CAM Utilisation Worldwide: A Systematic Review

This review aimed to systematically evaluate findings from published surveys on the uses of CAM by the general population. The secondary aim was to determine the methodological quality of these CAM surveys, to identify the methodological strengths and weaknesses to inform the design of the present national population-based CAM survey.

2.3.2.1 Methodology

The identification of literature in the CAM field is challenging. Establishing the prevalence and utilisation of CAM is complicated and time-consuming, given that there is often no simple means of identifying such publications. Some databases (e.g. MEDLINE) have developed literature filters for different types of publications, such as clinical trials and reviews. Such a filter for epidemiological studies is currently not available. Optimal retrieval of reports in biomedicine literature depends on the appropriate use of Medical Subject Headings (MeSH), descriptors and keywords among authors and indexers.⁴⁰

A comprehensive list of key terminologies for the search strategy in MEDLINE was initially developed. Modified strategies were then used in other databases. Major MeSHs for CAM include "Complementary Therapies", "Medicine, Oriental Traditional" and "Medicine, Traditional". The MeSH search was used in conjunction with the keyword search. A systematic search of the literature was conducted on the following databases: MEDLINE (Ovid platform), EMBASE and CINAHL, covering all years since their inception up to September 2005. The search strategy for MEDLINE is presented in Table 2.3.

Table 2.3 Search strategy to identify CAM utilisation studies in MEDLINE

- 1. exp Complementary Therapies/
- 2. exp Alternative Medicine/
- 3. exp Medicine, Traditional/
- 4. exp Medicine, Oriental Traditional/
- 5. exp Drugs, Non-Prescription/
- 6. exp Plants, Medicinal/
- 7. exp Medicine, Herbal/
- 8. ((complementary or alternative or unconventional or traditional or unproven or integrat\$ or oriental or non-orthodox or holistic or essential) adj (therap\$ or medicine\$ or health or treatment\$ or remed\$)).tw.
- 9. or/1-8
- 10. exp Prevalence/
- 11. exp Health Surveys/
- 12. exp Questionnaire/
- 13. exp Data Collection/
- 14. exp Epidemiologic Studies/
- 15. (use\$ or usage\$ or utili?ation or utili?e\$ or survey\$ or prevalence or questionnaire\$ or pattern\$ or interview\$ or frequency).tw.
- 16. or/10-15
- 17.9 and 16
- 18. Randomized Controlled Trials/
- 19. Clinical Trials/
- 20. randomized controlled trial.pt.
- 21. clinical trial.pt.
- 22. or/18-21
- 23. 17 not 22
- 24. limit 23 to animal
- 25. limit 23 to human
- 26. 24 not 25
- 27. 23 not 26
- 28. limit 27 to English

[/ indicates Medical Subject Heading term, \$ = truncation, ? = substitution exp = exploded, tw = text word, pt = publication type, adj = adjacent; and/, or/, not/ are Boolean operators]

The databases were searched irrespective of publication status, although only articles published in English were included in the review. To qualify for inclusion, papers had to use specific survey methods, either questionnaires or interviews, to collect information relating to CAM use, and also meet one of the following criteria:

- 1. The study must have been based on a random or representative sample of a national general population.
- 2. The study must have been based on a random or representative sample of a general population for a defined regional area AND such a national study was not available.

Articles were excluded if one of the following criteria applied:

- The study was an utilisation study on CAM use by a subpopulation, such as a cohort of patients with particular conditions (cancer, diabetes etc.) or a specific group of people (e.g. older people or adolescents).
- 2. The study investigated a single form of CAM (e.g. chiropractic), rather than CAM as a whole.
- 3. The report did not describe the study methods, e.g. a review or a general article that did not mention how the data of CAM prevalence were collected, or a report on studies described elsewhere.
- 4. The full publication was not written in English.
- 5. It was not published in an academic journal or a formal report document (where a copy of the report was not supplied elsewhere).

In addition to the literature search across major databases, the references of all relevant papers were examined to identify further literature. This approach has been found particularly useful in identifying studies that are not included in the electronic databases. All identified publications were read in full, and data were extracted for their substantive information on the target population, study design, CAM modalities, prevalence of CAM use and other significant findings.

In order to access the most recent literature, new article-alert functions, available from the MEDLINE and EMBASE (through Science Direct) were activated in May 2004. These enable researchers to save search terms with databases when the searches are automatically performed over a defined period. Weekly searches were performed and article alters were produced continually. Literature reviews in this thesis include articles retrieved up to March 2006.

2.3.2.2 Results

The initial search identified over 1,200 articles. After reading titles and abstracts, a large proportion of the articles was excluded because they covered matters other than utilisation or prevalence (e.g. CAM regulation, integration approach and clinical trials) or duplicated other publications. The remaining 307 articles that presented data on CAM use were read in full. Of these, 290 articles were then excluded because they did not meet the inclusion criteria (see Chapter 2.3.2.1), for example, regional studies and studies on patients or health professionals. Two articles were identified by citation tracking of eligible articles. ^{15,41} Two additional articles were obtained in November 2005⁴² and January 2006¹¹ when being notified by new-article alerts. Thus a total of 21 studies are included in the current review. ^{1-7,9-11,15-17,41-48} The key data of included articles are summarised in Table 2.4.

The use of CAM in the general population was systematically reviewed by both Ernst¹⁸ and Harris *et al.*¹⁹ in 2000. In 2001, Wootton and her colleagues also conducted a review of CAM use among the US population.³⁹ An overview of these three reviews and the current review are presented in Table 2.5. The majority of studies that were selected in the three reviews are also included in the current review. Reasons for exclusion are presented in Table 2.5.⁴⁹⁻⁵⁸ As mentioned earlier, there have been many CAM utilisation studies in the past decade. 12 studies conducted after the publication of the three reviews mentioned above were identified and fulfilled the inclusion criteria.^{1,4-7,10,11,16,17,42,43,48}

Author (year)	Place of study (year)	Sample (n)	Data collection method (response rate, %)	General findings*
Eisenberg <i>et al.</i> $(1993)^2$	USA (1990)	Randomly selected adults $(1,539)^{\dagger}$	CATI (67%)	33.8% used one of 16 forms of CAM; [‡] 12.3% visited CAM providers; out-of-pocket CAM expenditure was about US\$10.3 billion.
Thomas <i>et al.</i> (1993) ⁴⁴	UK (1993)	Randomly selected adults from UK electoral registers (676)	Self completed questionnaire (73%)	8.5% used one of 6 forms of CAM: acupuncture; chiropractic; homeopathic; hypnotherapy; medical herbalist; and, osteopathy.
MacLennan et al. (1996) ⁹	South Australia (1993)	Random and representative samples aged 15+ (3,004)	South Australian Health Omnibus Survey, face-to-face interview (73%)	48.5% used one of 11 forms of CAM. 20.3% visited CAM practitioners; total CAM expenditure was A\$930 million.
Millar <i>et al.</i> (1997) ⁴⁵	Canada (1995)	Representative samples aged 15+ (17,626)	National Population Health Survey: personal interview (n/m)	15% had consulted some forms of CAM practitioner.
Paramore <i>et al.</i> (1997) ¹⁵	USA (1994)	Representative samples, all ages (3,450)	National Access to Care Survey: CATI (75%)	24.4% visited practitioners for one of 4 forms of CAM: chiropractic; therapeutic massage; relaxation techniques; and, acupuncture.
Astin (1998) ⁴⁶	USA (n/m)	Randomly selected adults (1,035)	National Family Opinion Survey: mail survey (69%)	40% used one of 17 forms of CAM.
Eisenberg <i>et al.</i> $(1998)^3$	USA (1997)	Randomly selected adults (2,055)	CATI (60%)	42.1% used one of 16 forms of CAM; 19.5% visited CAM providers; out-of-pocket CAM expenditure was about US\$27.0 billion.

Table 2.4 Results of	CAM	utilisation	studies in	the	general	po	pulation
----------------------	-----	-------------	------------	-----	---------	----	----------

CATI: computer assisted telephone interview. n/m: not mentioned.

* Unless otherwise defined, use, practitioner visit and expenditure were in the 12-month period preceding the survey.

† Adults: 18 years of age or older. ‡ refers to the use of at least one of the listed forms of CAM.

Author (year)	Place of study (year)	Sample (n)	Data collection method (response rate, %)	General findings*
Landmark Report (1998) ⁴¹	USA (1997)	Randomly selected adults with health insurance coverage ^{\dagger} (1,500)	Telephone interview (n/m)	42% used some forms of CAM.
Druss <i>et al</i> . (1999) ⁴⁷	USA (1996)	Randomly selected adults [†] (16,068)	The US Medical Expenditure Panel Survey: CAPI, (77.7%)	8.3% visited one of 12 types of CAM practitioner during 1996.
Ramsay <i>et al.</i> (1999) ⁶	Canada (1997)	Randomly selected adults ^{\dagger} (1,500)	CATI (25.7%)	50% used one of 22 forms of CAM; out-of- pocket CAM expenditure was about CAD\$3.8 billion.
Ernst <i>et al.</i> $(2000)^{17}$	UK (1999)	Randomly selected adults ^{\dagger} (1,204)	Telephone interview (no record was kept of those declined to participate)	20% used any CAM; CAM expenditure was £1.6 billion.
Millar <i>et al.</i> (2001) ⁴⁸	Canada (1998/99)	Representative adults [†] (14,150)	National Population Health Survey: personal interview with 1994/95 survey subjects (88.9%)	17% had consulted CAM practitioners (forms of CAM were not presented).
Thomas <i>et al.</i> $(2001)^5$	UK (1998)	Randomly selected adults ^{\dagger} (5,010)	Postal questionnaire (59%)	28.3% used one of 8 forms of CAM.13.6% visited CAM practitioners; out-of-pocket expenditure on visits to CAM practitioners was £580 million.
MacLennan <i>et al.</i> $(2002)^{10}$	South Australia (2000)	Random and representative samples aged 15+ (3,027)	South Australian Health Omnibus Survey: face-to-face interview (70.4%)	52.1 % used one of 9 forms of CAM. 23.3% visited CAM practitioners; total CAM expenditure was A\$2.3 billion.

 Table 2.4 Results of CAM utilisation studies in the general population (continued)

CATI: computer assisted telephone interview; CAPI: computer assisted personal interview. n/m: not mentioned.

* Unless otherwise defined, use, practitioner visit and expenditure were in the 12-month period preceding the survey.

† Adults: 18 years of age or older.

Author (year)	Place of study (year)	Sample (n)	Data collection method (response rate, %)	General findings*
Menniti- Ippolito <i>et al.</i> $(2002)^{43}$	Italy (1999)	Representative samples, all ages (70,898)	A national Institute of Health Survey: face-to-face interviews (n/m)	15.6% of Italians used CAM during 1997-1999.
Ni <i>et al.</i> $(2002)^4$	USA (1999)	Representative adults [†] (30,801)	National Health Interview Survey: CAPI (70%)	28.9% used one of 12 forms of CAM.
Yamashita <i>et al.</i> $(2002)^7$	Japan (2001)	Randomly selected samples aged 20+ (1,000)	CATI (23%)	76% used one of 11 forms of CAM; out-of- pocket expenditure was US\$153 per participant.
Barnes <i>et al.</i> $(2004)^1$	USA (2002)	Representative adults ^{\dagger} (31,044)	National Health Interview Survey: CAPI (74.3%)	62.1% used one of 27 forms of CAM; when prayer specific for health reasons was excluded, it was 36%.
Thomas $et al. (2004)^{16}$	UK (2001)	Representative samples aged 16+ (1,794)	Omnibus survey: face-to-face interview (65%)	10% visited one of 23 types of CAM practitioner.
Nielsen <i>et al.</i> (2005) ⁴²	Denmark (2000)	Representative samples aged 16+ (16,690)	Danish Health and Morbidity Survey: face-to-face interview (74.2%)	14% used CAM in the past 14 days (no individual forms of CAM were asked).
MacLennan <i>et al.</i> (2006) ¹¹	South Australia (2004)	Random and representative samples aged 15+ (3,015)	South Australian Health Omnibus Survey: face-to-face interview (71.7%)	52.2% used one of 8 forms of CAM; 26.5% visited CAM practitioners; total CAM expenditure was A\$1.8 billion.

Table 2.4 Results of CAM utilisation studies in the general population (continued)

CATI: computer assisted telephone interview; CAPI: computer assisted personal interview. n/m: not mentioned.

Unless otherwise defined, use, practitioner visit and expenditure were in the 12-month period preceding the survey.

† Adults: 18 years of age or older.

Reviewed by						
	Ernst	Harris	Wootton	This	Reason for not being	
Published studies	(2000)	(2000)	(2001)	thesis	reviewed in this thesis	
Haidinger (1988) ⁵³					Paper published in German	
Yung (1988) ⁵⁸					Regional study in UK	
Eisenberg $(1993)^2$	\checkmark			\checkmark		
Thomas (1993) ⁴⁴				\checkmark		
Vaskilampi (1993) ⁵⁵		\checkmark			Published as a book chapter	
Verhoef (1994) ⁵⁶		\checkmark			Rural study in Canada	
	I				Reported results from the	
Vickers (1994) ³⁷	V	1		1	1993 Thomas study	
MacLennan (1996) ⁹	V					
Emslie (1996) ³²					Regional study in UK	
Bernstein (1997) ⁴⁹					Subjects were aged 45-75	
Chi (1997) ⁵¹					Chinese medicine only	
Haussermann (1997) ⁵⁴					Paper published in German	
Millar (1997) ⁴⁵	\checkmark			\checkmark		
Paramore (1997) ¹⁵	\checkmark			\checkmark		
Astin (1998) ⁴⁶	\checkmark			\checkmark		
Burg (1998) ⁵⁰					Regional study in the US	
Eisenberg (1998) ³			\checkmark	\checkmark		
Landmark Report	.1	.1	.1	.1		
(1998)	N	N	<u> </u>	<u> </u>		
Druss (1999)*			V	V		
Ramsay (1999) ⁶						
Ernst (2000) ¹⁷						
Millar (2001) ⁴⁸						
Thomas (2001) ⁵						
MacLennan (2002) ¹⁰				\checkmark		
Menniti-Ippolito (2002) ⁴³						
Ni (2002) ⁴						
Yamashita $(2002)^7$				\checkmark		
Barnes (2004) ¹				\checkmark		
Thomas (2004) ¹⁶				\checkmark		
Nielsen (2005) ⁴²						
MacLennan (2006) ¹¹						
No. of studies	12	12	6	21		

Table 2.5 An overview of CAM utilisation studies included in the systematic reviews

All except the three Australian studies were conducted at a nationwide level. These 21 studies investigated CAM use by a total of 227,086 people in seven countries: eight studies were conducted in the US (and included a total of 87,492 participants), ^{1-4,15,41,46,47} four studies in the UK^{5,16,17,44} (8,684 participants); three studies in Canada^{6,45,48} (33,276 participants) and three studies in Australia⁹⁻¹¹ (9,046 participants); the remaining three studies were in Denmark⁴² (16,690 participants), Italy⁴³ (70,898 participants) and Japan⁷ (1,000 participants). The median sample size for all the studies was 3,015. However, the median sample size decreases to 1,500 when health surveys not specifically for CAM are excluded.

Over half (13 in total) of the studies were conducted as a supplement to a national health survey. In most studies, data were collected through postal questionnaires (two studies in the UK and one in the US), telephone interviews (seven studies), or face-to-face interviews (11 studies). In general, the sample selection methods used to ensure proper representation of the study populations were rigorous. The majority (90%) of the studies targeted the adult populations, defined in different countries as 15+, 16+, 18+, or 20+. The Italian study,⁴³ with a maximum sample size of 70,898, included all children and adults. A recent Australian study¹¹ also asked questions about CAM use by children (through their households' adult participants). In general, the populations represented in these studies are considered to be appropriately specified. The overall response rates range from 23% to 88.9%. However, the two highest response rates were obtained from sub-sample interviews with previous national surveys. When these are excluded, the response rates range from 23% to 75%, with a mean of 63%. Four studies did not provide a response rate.

The majority of studies (16 in total) employed a list of CAM to estimate overall prevalence. The types of CAM modalities ranged from four¹⁵ to 27 forms.¹ Similarly, in the UK, the number of CAM investigated were six, eight and 23 in 1993, 1998 and 2001, respectively.^{5,16,44} Thus, the prevalence presented in Figure 2.1 does not taken account of the comprehensiveness or totality of CAM, and should be interpreted with caution.

The overall prevalence of CAM varied between 8.5% in the UK in 1993⁴⁴ and 72% in Japan in 2001.⁷ Well over one in three Americans used CAM,³ and one in two Canadians and Australians were CAM users in the year preceding the relevant studies.⁶ Figure 2.1 suggests that the use of CAM, as well as the prevalence of visits to CAM practitioners, has been increasing over the past 15 years. In particular, marked increases were observed in sequential studies in the US (33.8% in 1990² to 42.1% in 1997³), and UK studies by Thomas *et al.* showed an increase from 8.5% in 1993⁴⁴ to 28.3% in 1998.⁵ However, only six forms of CAM were investigated in the 1993 UK study,⁴⁴ and the authors considered this to be a pilot study aimed at testing the feasibility of the methodologies employed.

The prevalence of visits to CAM practitioners was also reported in the US, UK, Canada and Australia. The lowest prevalence was reported in a 1996 US study (8.3%),⁴⁷ and the highest prevalence appeared in a 2004 Australian study (26.5%).¹¹ Overall prevalence on both CAM use and visits to CAM practitioners was commonly estimated over a one-year period (preceding to the survey), except the 1999 Italian study,⁴³ which obtained prevalence over a two-year period, and a Danish study in 2000,⁴² which were based on a 14-day prevalence.

Data on total costs and out-of-pocket expenditure have also been extracted from these studies, where available. The substantial costs drew considerable comment in some studies. For example, CAM expenditure in the US increased by 45%, reaching US\$ 21 billion in 1997.³ Moreover, from 1993 to 2000, the overall CAM costs in Australia had more than doubled to A\$2.3 billion; and in 2000, Australians paid out-of-pocket four times more for CAM than their contributions for all pharmaceuticals.¹⁰ In Canada, out-of-pocket expenditure was about \$3.8 billion Canadian dollar in 1997.⁶



Figure 2.1 Prevalence of CAM use in seven countries

Note: Years were at the time of conducting studies, which are different from the dates of publication

US: United States; UK: United Kingdom; CAN: Canada; AUS: Australia; ITA: Italy; DEN: Denmark; JAP: Japan.

US90, US97A (Eisenberg *et al.*^{2,3}); US94 (Paramore *et al.*¹⁵); US96 (Druss *et al.*⁴⁷); US97B (Landmark report⁴¹); US98 (Astin⁴⁶); US99; (Ni *et al.*⁴); US02 (Barnes *et al.*¹); UK93, UK98, UK01 (Thomas *et al.*^{5,16,44}); UK99 (Ernst *et al.*¹⁷); CAN95, CAN99 (Millar *et al.*^{45,48}); CAN97 (Ramsay *et al.*⁶); AUS93, AUS00, AUS04 (MacLennan *et al.*⁹⁻¹¹); ITA99 (Menniti *et al.*⁴³); DEN00 (Nielsen *et al.*⁴²); JAP01 (Yamashita *et al.*⁷).

2.3.2.3 Discussion

It is apparent that there are considerable uncertainties about the true prevalence of CAM among the general populations of countries. Uncertainties are even observed within countries. As discussed earlier, many of these uncertainties arise because of the complex problem of defining CAM and the modalities it includes. Nevertheless, when specific CAM questions were designed as part of a national health survey, it is likely that professional social scientists and statisticians were employed for the development of instruments and for data analyses.

Most studies provided estimates on the prevalence of use of individual forms of CAM, except for a 2000 Denmark study⁴² and a 1999 Canadian study.⁴⁸ Two therapies (acupuncture and chiropractic) were included in all studies and, perhaps, are the most frequently reported CAM modalities, together with massage, herbal medicine and multivitamins. The inclusion or exclusion of these individual therapies can potentially impact heavily on estimation of the overall prevalence of CAM.¹⁹ To avoid confusing survey participants it is also necessary that each specific CAM modality be defined. The use of a show-card in a face-to-face personal interview or the provision of a detailed list of all forms of CAM in a telephone interview or postal questionnaire would gather a more precise measurement on prevalence. In contrast, asking participants to voluntarily provide all the names of CAM they have used would be less likely to be comprehensive.

Two published reviews provided additional important information on CAM use in some other countries.^{59,60} For example, CAM prevalence data for a number of European countries, such as Germany and Austria.¹⁸ However, many of the papers cited were published in the local language, rather than English and therefore are not included in the current review. In one of the reviews, Fisher and Ward⁶⁰ summarised such European CAM utilisation data and

concluded that a high percentage of the general public in Belgium (31%), France (49%), Germany (46%), the Netherlands (20%), Denmark (23.2%), and Sweden (25%) were CAM users. However, it should be noted that these figures are somewhat higher than those given in Figure 2.1.

Since the pioneer research on CAM use published by the Harvard research group in 1993,³ the US has been leading the way in investigating CAM utilisation among the general public. More recently, employing the most rigorous data collection methods (i.e. computer-assisted personal interviews with representative residents), a number of studies were conducted as part of the US National Health Surveys.^{1,4} On the other hand, Australia is one of a few countries that has investigated CAM use in sequential surveys.⁹⁻¹¹ Yet, these were conducted at a state level, so a full picture of Australia's nationwide utilisation was not available. Thus, the current literature review includes CAM utilisation studies only at a national level, with the exception of Australian regional studies.

Additional common outcome measurements of these reviewed articles are the characteristics of CAM users, concurrent use of CAM and conventional medicine, attitudes towards such use and communication between patients, CAM practitioners and Western medicine practitioners. Substantial findings on these matters were also collected and are discussed in the relevant chapters in this thesis, in comparison with the results of the recently completed CAM Australia national study in 2005. In all, the most critical problem in producing the true prevalence of the study population is a clear definition of CAM, specifically, the classification of CAM modalities. Further high quality research is needed, preferably replicated studies supplementary to a national health survey.

2.3.3 CAM Utilisation in Australia: A Systematic Review

This review aimed to systematically examine the current status of use CAM in Australia, in particular, the profile of CAM use among the general population, among specific patients groups and among the health professionals.

2.3.3.1 Methodology

When identifying worldwide CAM utilisation studies, particular attention was paid to Australian literature. A modified version of the search strategy (see Table 2.3) was used to identify CAM utilisation studies that had been conducted in Australia with additional search limitations (Table 2.6). The two approaches are not mutually exclusive when identifying literature on CAM surveys. It was expected that the search limitation on "Australian studies only" did not capture all Australian studies. The final search and reference selection process took place in December 2005.

Table 2.6 Additional search strategy to identify Australian CAM utilisation studies

- 1. exp Australia/
- 2. exp South Australia/
- 3. exp Western Australia/
- 4. exp Victoria/
- 5. exp New South Wales/
- 6. exp Queensland/
- 7. exp Northern Territory/
- 8. exp Australian Capital Territory/
- 9. exp Tasmania/
- 10. (Australia or south Australia or western Australia or Victoria or New South Wales or Queensland or Northern territory or Australian Capital Territory or Tasmania).tw.
- 11. (Melbourne or Sydney or Brisbane or Adelaide or Perth or Darwin or Canberra or Hobart).tw.
- 12. or/1-11
- 13. 12 and 28 [of Table 2.3]

The new-article alert function and a reference cross-check were also included. Most of the previously mentioned inclusion and exclusion criteria (see Chapter 2.3.2.1) are applied to identify Australian studies, except that studies on CAM use on a subpopulation (e. g. cancer patients and the older population) are included in the Australian literature review. Studies on the general population and specific populations are presented separately.

2.3.3.2 Results

The initial search identified over 500 articles. After reading titles and abstracts (to clarify whether the study was conducted in Australia), a total of 122 articles were included for further examination and were read in full. These articles were then sorted in four categories to facilitate the article selection process: 1) CAM use in a general population; 2) CAM use in a defined population; 3) CAM used by other populations, such as health professionals, and 4) articles requiring further clarification, for which opinion from an independent researcher was sought. In addition, a study conducted in Western Australia⁶¹ and another in Melbourne⁶² were also identified. These studies^{61,62} also reported data on the use of CAM in the general population. However, data from both studies were collected from in-street participants; such a convenience sampling method did not satisfy the inclusion criteria for the current review.

Six articles were identified by citation tracking of relevant articles. One additional article was obtained in 2006 when notified by a new-article alert. Thus, a total of 38 studies were included in the current review. These included five studies on CAM use by the Australian general population, ^{9-11,37,38} seven studies on cancer patients, ⁶³⁻⁶⁹ eight studies on children and women (including patients), ⁷⁰⁻⁷⁷ 12 studies on people with other diseases other than cancer, ⁷⁸⁻⁸⁹ and six articles on other (non-clinical) populations. ⁹⁰⁻⁹⁵ The key data of included articles are summarised in Table 2.7 to Table 2.11

The 38 Australian studies represented CAM use by a subtotal of 9,991 of the general population, 2,572 cancer patients, 386 children patients, 54,708 women and women patients, 1,841 other patients (exclude Parslow's study; see Table 2.10 for the reason) and 2,729 other defined populations, in five states and one territory. Six studies were conducted on a nationwide basis (and included a total of 56,691 participants), 12 studies in New South Wales (NSW); eight in South Australia (SA); six in Victoria (VIC); three in Queensland (QLD), two in Western Australia (WA) and one in the Australian Capital Territory (ACT).

In terms of the data collection methods, the majority of these studies (25 in total) used traditional questionnaires (13 by mail, two completed by patients in the clinic waiting room and 10 completed by clinic patients in their own time). Eleven studies introduced face-to-face interviews (including one using a hand-held computer). Two studies employed telephone interviews. The use of CAM by General Practitioners (GP) was investigated by analysing Health Insurance Commission data (Medicare).

The qualities of sample selection methods varied. Studies on the general population targeted randomly selected samples, while studies on clinical or defined populations mainly recruited subjects through convenience sampling (e.g. patients attending clinics, and routine records/questionnaires for hospital inpatients). Questionnaires were also completed by GPs and nurses on their personal use of CAM and also on the CAM used for their patients. Nearly half (44%) the studies on a defined population did not state the ages of their subjects. This may be partially explained by the fact that no age limitation was applied when recruiting patients, particularly cancer patients. The sample size of the majority of the studies on defined

populations was less than 1,000, with the exception of two studies on women. The latter were supplemental analyses of large-scale national longitudinal studies.

It is not surprising that studies on clinical populations tended to achieve a higher response rate than studies on a non-clinical population. This is evidenced by the fact that response rates of four patient-based studies were higher than 90%. Thirteen studies, however, did not state their response rates.

Results of CAM Use among the General Population

As expected, the major results found in this category were Omnibus Health Surveys in a specific state of Australia.⁹⁻¹¹ Through extrapolation, these have been frequently quoted to represent the use of, and expenditure on, CAM in Australia. Two systematic reviews^{18,19} on CAM utilisation studies considered the methodology employed in the first 1993 survey⁹ to be rigorous. However, the CAM categories used in this survey was somewhat questionable.¹⁸ Such problem has also been observed in two more recent surveys.^{10,11}

Two regional studies in New South Wales were identified, with higher prevalence being reported than in the previous South Australian studies (Table 2.7). Kermode's New South Wales study in 1995³⁸ was based on nutritional products, including 24 vitamin and other supplements, herbal medicine, and homoeopathic medicine (see Table 2.2). Many of these items are not universally considered as CAM. It is also worth noting that the Wilkinson's study³⁷ was conducted in a rural community in New South Wales. An extraordinarily high prevalence of chiropractic (55.3% of the population had visited chiropractors at least once) was reported in this community. It is by far the highest prevalence of chiropractor visits documented in the international literature.

Common results from the three South Australian⁹⁻¹¹ and two New South Wales studies^{37,38} suggested that the use of CAM by the Australian general public is not less than 50%, and approximately one in four people have visited a CAM practitioner in a 12-month period. Wilkinson *et al.* reported the highest prevalence of both CAM use (70.3%) and CAM practitioner visits (62.7%).³⁷ The CAM use was specifically over a 12-month period; however, the period is unclear for CAM practitioner visits. A high prevalence in the use of herbal medicine and multivitamins was most commonly reported. In a recently published study,¹¹ the South Australian-based research team reported that 39.2% used vitamins, and an additional 13.6% used mineral supplements (excluding calcium, iron or vitamins prescribed by a medical doctor). For provider-based therapy, visits to chiropractors and massage therapists represented the majority of visits. ¹¹

Author (year)	Place of study (year)	Sample (n)	Data collection method (response rate, %)	General findings*
MacLennan et al. (1996) ⁹	South Australia (1993)	Random and representative samples, aged 15+ (3,004)	South Australian Health Omnibus Survey: face-to- face interview (73%)	48.5% used one of 11 forms of CAM; 20.3% visited CAM practitioners; total CAM expenditure was A\$930 million; 37.6% used vitamins; 9.9% used herbal medicine; 14.9% visited chiropractor.
Kermode <i>et al.</i> (1998) ³⁸	North coast of New South Wales (1995)	Randomly selected adults (645) [†]	Telephone interview (44%)	64% used CAM at the time of being surveyed; 25% visited CAM practitioner within the last year; while 75% had previously consulted CAM practitioners; 57% used nutritional products; 33% used herbal medicine; 50% visited chiropractor/osteopath.
Wilkinson <i>et al.</i> (2001) ³⁷	Rural region of New South Wales (1999)	Randomly selected adults from electoral database (300)	Postal questionnaire (31.4%)	70.3% used one of 14 forms of CAM; 62.7% visited a CAM practitioner; 68.7% used vitamins and minerals, 26.1% used chiropractic; 55.3% had previously visited a chiropractor.
MacLennan <i>et al.</i> (2002) ¹⁰	South Australia (2000)	Random and representative samples aged 15+ (3,027)	South Australian Health Omnibus Survey: face-to- face interview (70.4%)	52.1 % used one of 9 forms of CAM; 23.3% visited CAM practitioners; total CAM expenditure was A\$2.3 billion; 36.4% used vitamins; 13.4% used herbal medicine; 16.7% visited chiropractor.
MacLennan <i>et al.</i> (2006) ¹¹	South Australia (2004)	Random and representative samples aged 15+ (3,015)	South Australian Health Omnibus Survey: face-to- face interview (71.7%)	52.2% used one of 8 forms of CAM; 26.5% visited CAM practitioners; total CAM expenditure was A\$1.8 billion; 39.2% used vitamins; 20.6% used herbal medicine; 16.7% visited chiropractor.

Table 2.7 CAM use among the Australian general population

* Unless otherwise defined, use, practitioner visit and expenditure were in the 12-month period preceding the survey. Key results include two most popular forms of CAM and the highest prevalence of visits to CAM practitioner. † Adults: 18 years of age or older

CAM usage in children and women patients is another area in which the prevalence has frequently been investigated. Previous systematic reviews have concluded that there is a generally high and possibly increasing prevalence of CAM use among paediatric populations.⁹⁶ A number of national surveys have also consistently found that women are more likely to use CAM than men.³⁹ A total of ten studies were identified. Eight were on children and women with a medical condition other than cancer (Figure 2.2 and Table 2.8), two were on paediatric and women cancer patients^{68,69} (Table 2.9).

A considerably high prevalence of CAM was also found in Australian studies of children and women patients (Figure 2.2). The only study estimates CAM use at the time of study found a high prevalence of 35.9% among gastroenterology outpatient children.⁷¹ Three studies were concerned with the lifetime use of CAM, that is, use of CAM at least once previously. Over half (51.7%) of asthmatic children in Sydney⁷³ and approximately half (46%) of South Australian children with cancer had previously used CAM.⁶⁸ A lower lifetime-use rate (33%) was reported among non-surgical hospital inpatient children in Victoria.⁷² However, the above findings may need to be interpreted with extreme caution, as the sample size for all the studies was rather small (median sample size=106), and the measurement of lifetime use may have a higher chance of introducing great recall bias.

Female patients in Australia appeared to favour CAM products the most. A study by Gollschewski on 886 randomly selected menopausal women in Queensland found that over 82.5% had previously used CAM.⁷⁶ Importantly, and a safety concern, over 73% women used CAM products during pregnancy, according to an Adelaide-based study in 2000.⁷⁰ In this study, the most frequently used products were herbal medicine (56%) and aromatherapy oils

(40%). Specifically, chamomile, a form of herbal product, and lavender aromatherapy oil were most favoured among pregnant women.

CAM prevalence among healthy women was revealed in two large-scale national surveys. A 1996 study on 41,817 women aged 18-23, 45-50 and 70-75 found that 26.3% had visited CAM practitioners within a 12-month period. Later, in 1998, a follow-up study on the same group of women aged 45-50 further revealed that over 29% had consulted CAM practitioners in the 12 months preceding the survey.



Figure 2.2 Prevalence of CAM use by children and women (patients) in Australia

Author (year)	Place of study (year)	Sample (n)	Data collection method (response rate, %)	General findings*
Byrne <i>et al</i> . $(2002)^{70}$	A teaching hospital in Adelaide (2000)	Inpatients of an antenatal ward aged 20-43 (48)	Interview with a standard questionnaire (73%)	73% used CAM during current pregnancy.
Day <i>et al</i> . $(2002)^{71}$	South East Sydney (2001)	Gastroenterology outpatient children, aged <16 (92)	Self-completed questionnaire (80%)	35.9% were taking CAM at the time of the survey.
Fong <i>et al</i> . $(2002)^{72}$	Bendigo base hospital in Victoria (1999)	Non-surgical inpatient children $(120)^{\dagger}$	Structured questionnaire completed by parents (88%)	33% had previously used one of 15 forms of CAM.
Shenfield <i>et al.</i> (2002) ⁷³	A teaching hospital in Sydney (1998)	Asthmatic children, aged <19 (174)	Structured questionnaire (92%)	51.7% used CAM in their lifetime and 24.7% visited CAM practitioners in their lifetime.
Adams <i>et al.</i> $(2003)^{74}$	Australia (1996)	Women aged 18-23, 45-50 and 70-75 (41,817)	Australian Longitudinal Study on women's health: postal questionnaire (n/m) [‡]	Visited CAM practitioners were 19% (18-23 yo); 28% (45-50 yo), and 15% (70-75 yo); overall, 26.3% visited CAM practitioners.
Biddle <i>et al.</i> $(2003)^{75}$	Wollongong, New South Wales (n/m) [‡]	Women attended health practitioner services, aged 15- 60 (503)	Postal questionnaire (32%)	77% used CAM; 73% visited CAM practitioners.
Gollschewski <i>et al.</i> $(2004)^{76}$	Southeast Queensland (2001)	Randomly selected menopausal women, aged 47-67 (886)	Postal questionnaire (59%)	82.5% had previously used CAM.
Sibbritt <i>et al.</i> $(2004)^{77}$	Australia (1996-1998)	Women aged 45-50 (11,454)	Australian Longitudinal Study on women's health: postal questionnaire (87%)	29% consulted CAM practitioners.

*Unless otherwise defined, use, practitioner visit and expenditure were in the 12-month period preceding the survey. † Age was not state in the study. ‡ n/m: not mentioned

General Findings on CAM Use for Cancer Treatment

Studies on the use of CAM by cancer patients have been widely published.⁹⁷ This is also the case for Australian literature, with seven studies published in the last 10 years (Table 2.9). With the exceptions of the 1999 women's health longitudinal study and a study reviewing Queensland Cancer Registry records, the remaining five studies recruited patients from hospitals and clinics. Such a convenience sampling method is not necessarily a problem for studies on clinical patients, provided they are rigorously designed.

Data of CAM prevalence collected on cancer patient studies were varied. Most studies were interested in the use of CAM at any time in the past, rather than in its use over the preceding year. As shown in Figure 2.3, the prevalence of CAM use ranges from 12% by colorectal cancer patients⁶⁵ to 52% by a mixed group of cancer patients.⁶⁶ It is interesting to note that this highest prevalence was the only study that based on the estimated prevalence after diagnosis of cancer. Two studies reported on the prevalence of cancer sufferers seeking treatment from CAM practitioners. In one study on those with advanced cancer, 11% sought treatment.⁶⁹ Unfortunately, more than half the studies failed to investigate individual forms of CAM and, importantly, no follow-up studies were conducted to reveal the trend of CAM use by cancer patients over time.

Two studies reported the costs related to CAM use. Begbie's 1996 study⁶³ estimated that the median annual cost was \$530 per user, with a maximum of \$20,000 per annum per user. Two out of three (64.3%) users felt they were getting value for their money. Miller⁶⁶ also found that CAM users spent a substantial amount of money over \$1,000, on items such as herbal medicine, high-dose vitamins and aromatherapy, after being diagnosed with cancer. For most therapies, patients would recommend them to others and would take the same therapy again.

Author (year)	Place of study (year)	Sample (n)	Data collection method (response rate, %)	General findings*
Sawyer <i>et al.</i> (1994) ⁶⁸	A hospital in South Australia (1988-1992)	Children with cancer $(48)^{\dagger}$	Questionnaires completed by parents $(n/m)^{\ddagger}$	46% had previously used CAM.
Begbie <i>et al</i> . (1996) ⁶³	Two clinics at two hospitals in Sydney (1995)	Cancer patients, all ages (319)	Self-administered questionnaire (62%)	21.9% had previously used CAM.
Miller <i>et al</i> . (1998) ⁶⁶	A teaching hospital in Sydney (1998)	Cancer patients (156) [†]	Questionnaire in waiting room (90%)	52% used CAM after being diagnosed.
Correa-Velez et al. (2003) ⁶⁴	Queensland (n/m) [‡]	Advanced cancer patients, aged 20+ (111)	Cancer Registry records: follow up interview every four to six weeks (88%)	32% used CAM in one week preceding the survey; 11% visited CAM practitioners on a regular basis.
Sibbritt <i>et al.</i> (2003) ⁶⁹	Australia (1999)	A representative sample of women, aged 73-78 (9,375 includes 1,623 cancer patients)	Australian longitudinal study on women's health: postal questionnaire (n/m) [‡]	14.5% women with cancer consulted CAM practitioners; 11.6% women without cancer consulted a CAM practitioner.
McGorm <i>et al.</i> (2004) ⁶⁵	Two large public hospitals in Adelaide (n/m) [‡]	Patients with colorectal cancer $(100)^{\dagger}$	Postal questionnaire (43%)	12% had previously used CAM for colorectal cancer.
Salminen <i>et al.</i> (2004) ⁶⁷	A cancer institute in Melbourne (1999-2001)	Breast cancer patients $(215)^{\dagger}$	Questionnaire in waiting room $(n/m)^{\ddagger}$	17% had previously used CAM.

Table 2.9 CAM use for cancer treatment in Australia

* Unless otherwise defined, use, practitioner visit and expenditure were in the 12-month period preceding the survey.

† Age was not state in the study. ‡ n/m: not mentioned



Figure 2.3 Prevalence of CAM use by Australian cancer patients

Twelves studies on CAM use by patients other than cancer patients, children and women patients were found (Table 2.10). These represented CAM use by patients with eight different clinical conditions: HIV/AIDS,^{80,84} rheumatoid arthritis (RA),^{78,82} osteoarthritis,⁸⁹ depression,⁸⁸ chronic obstructive pulmonary disease (COPD),⁸⁷ diabetes,⁸⁶ psychiatric problems,⁸⁵ cystic fibrosis,⁸³ and among patients who were attended to two clinical settings: emergency department⁷⁹ and hospital.⁸¹ Again, a few studies reported lifetime CAM use (i.e. studies on HIV/AIDS, COPD and depression patients). An overview of all prevalence data is presented in Figure 2.4.

In seven studies, over 50% of those surveyed had previously used CAM on at least one occasion. The highest such prevalence was observed in RA patients in Victoria in 1982 (82% used CAM after being diagnosed, and 52% were using it at the time of the study).⁷⁸ A high prevalence among RA patients was further confirmed in a more recent study in 1998.⁸² Based on Melbourne RA patients this study found that 67.3% of patients were CAM users. On the other hand, the lowest prevalence (2.9% had previously used CAM) was found in a 2001 study in Canberra and its environment.⁸⁸ However, this prevalence refers to CAM use by those within a sample of the general population who suffered from depression and/or anxiety, the number of such users being expressed as a percentage of the entire sample. The true prevalence of CAM use by people with depressive or anxiety symptoms remains unclear.

The 1997 and 1999 studies found a similar prevalence of CAM use by HIV/AIDS subjects (56% and 55%, respectively). ^{80,84} These findings can be considered highly reliable, as studies were based on large samples of HIV/AIDS subjects in Australia. The authors concluded that CAM use was not significantly related to HIV positive individuals having been diagnosed with AIDS (p=0.06).

Author (year)	Place of study (year)	Sample (n)	Data collection method (response rate, %)	General findings*
Kestin <i>et al.</i> (1985) ⁷⁸	A teaching hospital in Victoria (1982)	Consecutive patients with rheumatoid arthritis, aged 15+ (90)	Interview in a structured questionnaire $(n/m)^{\ddagger}$	82% used CAM after being diagnosed; 52% were using CAM at the time of survey.
Kristoffersen et al. (1997) ⁷⁹	A Sydney teaching hospital (1994)	Patients presented to an emergency department, all ages (325)	Interviews guided by a structured questionnaire (98%)	52% used CAM; 30.5% used CAM in one week preceding the survey.
Visser <i>et al.</i> (2000) ⁸⁰	Non-clinical settings in Australia (1997)	Convenience sample of people living with HIV/AIDS (925)	Postal questionnaire (n/m) [‡]	56% had previously used CAM.
Welch <i>et al.</i> $(2001)^{81}$	A public hospital in Sydney (2000)	Hospital inpatients (511)	Survey as part of the routine medication history $(n/m)^{\ddagger}$	12% were using CAM at the time of study.
Buchbinder et al. (2002) ⁸²	A rheumatology clinic in Melbourne (1998)	Community clinic based rheumatoid arthritis patients (101)	Telephone administrated questionnaire (53%)	67.3% used CAM for RA and 31.7% visited CAM practitioners for RA.
Burrows <i>et al.</i> (2002) ⁸³	A hospital in Queensland (2001)	Cystic Fibrosis patients, aged 16+ (83)	Interview-based questionnaire (70%)	70% had previously used CAM and 61% had previously visited CAM practitioners; at the time of survey, 45% used CAM and 26% visited CAM practitioners.

Table 2.10 CAM use for treatment of diseases other than cancer in Australia

* Unless otherwise defined, use, practitioner visit and expenditure were in the 12-month period preceding the survey.

† Age was not state in the study. ‡ n/m: not mentioned

Author (year)	Place of study (year)	Sample (n)	Data collection method (response rate, %)	General findings*
Visser <i>et al.</i> (2002) ⁸⁴	Australia (1999)	A total of 8% of total Australian who were living with HIV/AIDS, aged 18-72 (924)	Self -administered survey (n/m) [‡]	55% had previously used CAM.
Alderman <i>et al.</i> (2003) ⁸⁵	A South Australian teaching hospital (n/m) [‡]	Psychiatry patients (52) [†]	Structured interviews (93%)	51.9% used CAM in the six months preceding the survey.
Clifford <i>et al.</i> $(2003)^{86}$	Western Australia (2001)	Patients with diabetes $(351)^{\dagger}$	Information obtained a diabetes study: interviews (n/m) [‡]	23.6% used CAM.
George <i>et al.</i> (2004) ⁸⁷	Melbourne (n/m)	Patients with chronic obstructive pulmonary disease [COPD] (173) [†]	Data obtained through a nested survey in a COPD randomised controlled trial $(n/m)^{\ddagger}$	41% had previously used CAM.
Parslow <i>et al.</i> (2004) ⁸⁸	Regional areas in the Australian Capital Territory (1999- 2001)	Subjects selected from electoral rolls from those aged 20-24 in 1999, aged 40-44 in 2000 and aged 60-64 in 2001 (7,485)	Longitudinal interviews by using a hand-held computer (n/m) [‡]	2.89% had previously used CAM to treat depression or anxiety (number of people with health conditions unclear).
Zochling et al. (2004) ⁸⁹	Sydney (1994-1999)	Osteoarthritis patients $(341)^{\dagger}$	Three prospective monthly-diaries and 12 month follow-up $(n/m)^{\ddagger}$	40% used CAM.

 Table 2.10 CAM use for treatment of diseases other than cancer in Australia (continued)

* Unless otherwise defined, use, practitioner visit and expenditure were in the 12-month period preceding the survey.

† Age was not state in the study. ‡ n/m: not mentioned



COPD: chronic obstructive pulmonary disease

Figure 2.4 Prevalence of CAM use for treatment of diseases other than cancer in Australia

CAM Use by Other Populations

With the large number of studies having investigated CAM use among the general population and clinical patients, researchers have also been interested in the personal use of CAM by health professionals and for their patients. With the global trend of increasing use and practice of alternative forms of medicine, CAM use by Western medical doctors is of particular interests. Studies were also carried out among other health professionals, such as nurses, midwives and university health sciences students (Table 2.11) The six studies identified were conducted in the states of New South Wales, Victoria, Western Australia, South Australia and nationwide by means of postal questionnaires and self-completed questionnaires.

The majority of GPs had referred their patients to a number of common forms of CAM practitioners, such as acupuncturists and massage therapists. Based on 488 randomly selected GPs in Victoria, Pirotta⁹¹ found 93% of them had referred their patients at least once to a CAM practitioner, and 82% had referred them at least a few times a year. A recent nationwide GP study⁹⁵ in Australia further revealed that 76% and 72% of GPs had referred their patients to acupuncturists or massage therapists, respectively, at least once a month.

The use of CAM by the GPs themselves was also common; 26% had used massage and 21% had used acupuncture.⁹⁵ A high prevalence of personal use was also reported among nurses (74% in the past 12 months)⁹³ and university nursing, pharmacy and biomedical sciences students (78% in the past 12 months).⁹² Among the individual forms of CAM that were found to be most popular were multivitamins and herbal medicine, a trend similar to CAM use by the general population and clinical patients. Moreover, an extremely high prevalence of aromatherapy use was found among nurses (44.1%) and students (nursing students: 32.5%, pharmacy students: 34.5% and biomedical sciences students: 35.1%).⁹²

Author (year)	Place of study (year)	Sample (n)	Data collection method (response rate, %)	General findings*
Hall <i>et al.</i> (2000) ⁹⁰	Perth metropolitan area (1998)	Randomly selected general practitioners [GPs] (282)	Cross-sectional survey: postal questionnaire (74.8%)	75% formally referred patients to one of 10 CAM practitioners in the previous 9 months.
Pirotta <i>et al</i> . (2000) ⁹¹	Victoria (1997)	Randomly selected GPs (488)	Postal questionnaire (64%)	93% referred at least once and 82% referred at lest a few times a year for a CAM therapy.
Wilkinson et al. (2001) ⁹²	New South Wales (1999)	University students in nursing, pharmacy and biomedical sciences (271)	Self completed questionnaire (72%)	78% used CAM. 56.3% visited CAM practitioners.
Wilkinson et al. (2002) ⁹³	New South Wales (2000)	Nurses (832)	Postal questionnaire (n/m) [‡]	74% used CAM for themselves. 38% used CAM for patients.
Gaffney <i>et al</i> . (2004) ⁹⁴	South Australia (n/m)	Obstetricians (75) and midwives (145)	Postal questionnaire (78%)	68% obstetricians and 78% midwives had formally referred patients to use CAM.
Cohen <i>et al</i> . (2005) ⁹⁵	Australia (2004)	GPs (636)	Postal questionnaire (33.2%)	The top three popular forms of CAM used by GP themselves were massage (26%), vitamin and mineral (13%) and acupuncture (12%). When using in their practices, these were acupuncture (18%), meditation (18%) and massage (11%). 76% and 72% GP referred patients to acupuncturists or massage therapists, respectively, at least once a month.

 Table 2.11 CAM use in other populations in Australia

* Unless otherwise defined, use, practitioner visit and expenditure were in the 12-month period preceding the survey. ‡ n/m: not mentioned

2.3.3.3 Discussion

The use of CAM among the general Australian population had not been comprehensively reviewed at the time of writing this thesis. Utilisation studies on clinical conditions, in particular, cancer patients, children and women patients were the most popular.

The methodological quality of the Australian studies is varied. In general, there is a lack of a standard approach to collect data. As with CAM utilisation studies outside Australia, the extent of CAM modalities included in the totality of CAM prevalence is the key challenge for a comparable conclusion to be drawn.

Despite these shortcomings, the prevalence of CAM use in Australia, particularly aromatherapy, herbal medicine and natural products, is higher than in other Western countries, noticeably among cancer patients and those with other defined clinical conditions.

It is important to note that in Australia common forms of CAM therapy are covered, to various levels by private health insurance funds. When conventional medicine cannot offer an effective cure for a disease, patients are turning to other CAM especially as more scientific evidence is becoming available for certain CAM therapies. This is evidenced by the fact that over half (58%) of patients with colorectal cancer believed that a hospital should provide information about CAM, although they did not necessarily use it.⁶⁵ In general, there is a lack of detailed information on the forms of CAM being used by Australians, and the efficacy of these treatments for specific clinical conditions. In this respect, poor communication between patients and practitioners,⁹⁸ is of safety concern.

2.3.4 Chinese Medicine Utilisation Worldwide: A Systematic Review

This review aimed to systematically examine published surveys on the use of Chinese medicine (CM) by the general population. Specific information about CM was extracted from identified CAM surveys (see Chapter 2.3.2.2) that included CM as one of the CAM modalities.

2.3.4.1 Methodology

Chinese medicine (CM) is classified as an alternative medical system by the US National Center for Complementary and Alternative Medicine (NCCAM).³⁶ It includes not only the commonly known acupuncture, Chinese herbal medicine and Chinese therapeutic massage but other modalities, such as Qigong and Chinese medicine dietary therapy. Many Western countries, including Australia, have experienced tremendous growth in the use of Chinese medicine, particularly acupuncture and some popular herbs, over the past decades. However, very few studies have focused exclusively on the totality of all CM modalities. The use of CM as a defined health-care system has not been specifically investigated. Thus, the current review was aimed to systematically extract CM data from surveys.

For the purpose of this review, CM was categorised into five major domains: 1) acupuncture, electro-acupuncture and moxibustion; 2) Chinese herbal medicine; 3) Chinese therapeutic massage; 4) Qigong, Tai Chi and martial art; and 5) Chinese medicine dietary therapy.

The database literature search looked for two types of studies: 1) those that explicitly stated the prevalence of any one form of CM or CM as a whole [Type I literature], and 2) those in which CM information was embedded in utilisation studies on general CAM use among
random or representative samples of the general population [Type II literature]. The search strategy to identify Type I CM utilisation literature is illustrated below (Table 2.12). This was used in combination with the search strategy for general CAM as presented in Table 2.3.

Table 2.12 Search strategy to identify type I literature on Chinese medicine utilisation

- 1. exp Medicine, Chinese Traditional/
- 2. exp Drugs, Chinese Herbal/
- 3. exp Medicine, Oriental Traditional/
- 4. exp Plants, Medicinal/
- 5. exp Drugs, Non-Prescription/
- 6. (medicin\$ adj05 (chines\$ or oriental\$ or tibetan\$)).tw.
- 7. (herb\$ medicin\$ or medicin\$ herb\$).tw.
- 8. (plant\$ medicin\$ or medicin\$ plant\$).tw.
- 9. herb\$.tw.
- 10. or/1-9
- 11. Acupuncture/
- 12. exp Acupuncture Analgesia/
- 13. exp Acupuncture, Ear/
- 14. exp Acupuncture Points/
- 15. exp Acupuncture Therapy/
- 16. exp Electroacupuncture/
- 17. exp Meridians/
- 18. (acupuncture\$ or electroacupuncture or electro-acupuncture or acupressure or acupoint\$ or ((meridian or non-meridian or trigger) adj05 point\$)).tw.
- 19. or/11-18
- 20. exp Moxibustion/
- 21. (moxibustion or moxa or Ai Jiu).tw.
- 22. exp Massage/
- 23. (chines\$ adj05 massage).tw.
- 24. (Tuina or Tui na or An mo).tw.
- 25. exp Tai Ji/
- 26. (Tai Ji or Taiji Tai Chi or Taichi).tw.
- 27. exp Martial Arts/
- 28. (Gong Fu or Gongfu).tw.
- 29. (Qi Gong or Qigong).tw.
- 30. (chines\$ adj05 (diet\$ therap\$ or nutrition\$ therap\$)).tw.
- 31. or/20-30
- 32. 10 or 19 or 31

The language of the publications included was limited to English or Chinese. To qualify for inclusion, papers had to use survey methods: either a questionnaire or interview to collect the information of CM use, and they had to meet one of the following selection criteria:

- 1. The study estimated explicitly the prevalence of CM as a whole or any one form of CM and, based on a random or representative sample of a national or regional general population or a clinical population.
- 2. The study was an utilisation study on CAM use among random or representative samples of a national population or a regional population, if such a national study was not available, and explicit data on the prevalence of CM as a whole or any one form of CM was available in the CAM study.

Papers were excluded if one of the following criteria applied:

- 1. The study was an utilisation study on CAM use in a clinical population.
- 2. The study did not describe the study methods, e.g. a review, an article that did not mention how the data of CM prevalence were collected, or a report describing studies elsewhere.
- 3. The full publication was not written in English or Chinese.
- 4. The study was not published in an academic journal or a formal report document.
- 5. Data addressed forms of herbal medicine other than Chinese herbal medicine.

Again, the references of all relevant papers were examined to identify further literature. Furthermore, the following publications, which are not available in electronic databases or partially admitted to electronic databases, were hand-searched: "*Journal of Complementary Medicine* (Australia, 2002-2006)"; "*Journal of Complementary and Integrative Medicine* (2004-2006); "*Journal of Traditional Chinese Medicine* (1991-2005)"; "*The Journal of Chinese Medicine* (2003-2005)"; "*Chinese Journal of Integrative Medicine* (2003-2005)". A search was also conducted on one of the major Chinese literature databases (CQVIP)

Information, a fee-paying database in China⁹⁹) and through Google Scholar, using the key words listed in Table 2.12 to screen for further literature. Using the same data extraction protocol, all identified publications were read in full and searched for substantive information that specifically related to CM.

2.3.4.2 Results

Methodological Quality and Study Design

Twelve Type I articles (i.e. exclusively Chinese medicine studies) met the criteria and were included in the review. Among these, five articles^{51,100-103} were based on general populations and another seven¹⁰⁴⁻¹¹⁰ were based on defined populations. The key data of these 12 articles are summarized in Table 2.13 and Table 2.14.

CM use among the general population was investigated in Taiwan,^{51,102} Hong Kong^{101,103} and New York City's Chinatown.¹⁰⁰ Except for Chou's 1991 acupuncture study, all other studies investigated CM as a whole.¹⁰² Similar to the data collection method for CAM studies on clinical patients or a defined population, CM results for a defined population were also based on convenience-based participant recruitment (Table 2.14). The sample sizes for all studies were large (median size for 11 studies=1,145). Studies on the general population mainly comprised data collected through a structured interview or telephone interview, while studies on a defined population were predominantly conducted by paper-based questionnaire.

A further 21 Type II literature (i.e. CAM studies with information on CM) met the criteria for this review (Table 2.15). The methodological quality of these studies has been discussed earlier. Individual forms of CM were available in acupuncture, herbal medicine, moxibustion and massage. Data on the use of Qigong and Tai Chi was available only in a US CAM study,¹ while data on Chinese medicine dietary therapy and martial art were not available.

Author (year)	Place of study (year)	Sample (n)	Data collection method (response rate, %)	General findings*
Chan <i>et al.</i> $(1976)^{100}$	New York City's Chinatown (n/m) [‡]	Randomly selected Chinatown residents aged 16+ (87)	Face-to-face interviews using a questionnaire (n/m) [‡]	17.2% had seen Chinese medicine (CM) practitioners in the past two years.
Wong <i>et al.</i> (1995) ¹¹¹	Territory-wide in Hong Kong (HK) (1989-1990)	Randomly selected ethnics Chinese HK residents aged 18+ (2,822)	Telephone interviews (76%)	48.9% had previously consulted an herbalist; 9% consulted herbalist for their most recent illness; 5.8% had previously consulted an acupuncturist.
Chi <i>et al.</i> (1997) ⁵¹	Taiwan (1991-1992)	Randomly selected adults in Taiwan (1,358)	A 4-week diary and an interview (52.9%)	9.6% used CM as the major source of medical care; the top three conditions to use CM were diseases of 1) respiratory system, 2) musculoskeletal system and 3) injuries and poisoning.
Chou <i>et al.</i> (1998) ¹⁰²	Taiwan (1991)	Randomly selected from the voter registration, aged 20+ (5,805)	Structured questionnaires used during home visits (70.1%)	12.4% had previously received acupuncture treatment.
Lau <i>et al.</i> $(2001)^{103}$	A district in HK with 0.6 million population (1995)	Randomly selected 1,183 households with 4,339 participants	Face-to-face interviews using a questionnaire (54.2%)	3.3% visited a CM practitioner within 3 months prior to the survey; 13.5% had previously used CM.

Table 2.13 Chinese medicine use in general populations

* Unless otherwise defined, use, practitioner visit and expenditure were in the 12-month period preceding the survey. ‡ n/m: not mentioned

Author (year)	Place of study (year)	Sample (n)	Data collection method (response rate, %)	General findings*
Wong <i>et al</i> . (1997) ¹⁰⁴	A small town in Hong Kong (HK) (1992-93)	People reported suffering from acute illness aged 18+ (847)	Subgroup analysis of a survey with 7,570 subjects: telephone interview (82.8%)	14% had previously consulted Chinese medicine (CM) practitioners when ill.
Cassidy <i>et al.</i> (1998) ¹⁰⁵	Six purposively selected clinics in the US (1995)	Patients used general service clinics and Chinese medicine care, aged 11+ (575)	Anonymous self- completed questionnaire (45.9%)	99% had previously used acupuncture; 59.7% received moxibustion and 35.5% used Chinese herbal medicine (CHM); the top three reasons of seeking CM were relief of pain, unstable mood and, maintaining well being.
Wong <i>et al.</i> (1998) ¹⁰⁶	Family practices in Vancouver (n/m) [‡] .	Patients and their family members, all ages (932)	Waiting room questionnaire (n/m) [‡]	28% used CHM and 7% used acupuncture in the past year, 46% ever used CHM and 15% had previously used acupuncture.
Lau <i>et al</i> . (2000) ¹⁰⁷	12 randomly selected schools in HK (1996)	Secondary school students (3,355)	Structured self- administrated questionnaire (89%)	8.6% had consulted a CM practitioner over a three months period. Injury suffered from falls was the most popular reason for seeking CM practitioner (66.5%).
Hon <i>et al.</i> $(2004)^{108}$	A school of pharmacy in HK (2002)	Pharmacy students (91)	Anonymous self- completed questionnaire (100%)	96% had previously tried CM; 38% had used CM at least once in the past year; >40% of CM users had visited a CM practitioner; upper respiratory tract infections were the most common condition for CM use.
Ng <i>et al.</i> (2004) ¹⁰⁹	A community in Singapore (1999)	Non-institutionalised elderly aged 65+ (2,010)	Door-to-door residential survey by using a questionnaire (85%)	25.3% used CHM in the past year, among whom 12.1% used CHM only. The use of CHM was strongly associated with arthritis and cancer (odds ratio=2.43 and 2.57).
Wong <i>et al.</i> (2005) ¹¹⁰	A hospital clinic in HK (2004)	Chronic hepatitis B patients aged 18+ (362)	Self-completed questionnaire (93%)	32% had previously used CM and 16% used CM at the time of the study.

 Table 2.14 Chinese medicine use in defined populations

* Unless otherwise defined, use, practitioner visit and expenditure were in the 12-month period preceding the survey. ‡ n/m: not mentioned

Table 2.15 Chinese medicine use in	general populations	(extracted from major	CAM utilisation studies)
------------------------------------	---------------------	-----------------------	--------------------------

Author (year) & country	General findings in Chinese medicine
	Chinese medicine (CM) use: acupuncture (0.4%), massage (7%), herbal medicine (3%);
	Visited practitioners: acupuncturist (0.37%); massage therapist (2.9%) herbal therapist (0.3%);
Eisenberg <i>et al.</i> (1993), USA ²	The mean number of visits to an acupuncturist per user was 38.4 per year.
Thomas <i>et al.</i> (1993), UK ⁴⁴	0.5% adult populations used acupuncture.
MacLennan et al. (1996),	CM use: herbal medicines (9.9%), Chinese medicine (1.8%), Ginseng (3.0%);
South Australia ⁹	Visited practitioners: acupuncturist (2.0%), herbal therapist (0.4%).
Millar et al. (1997), Canada ⁴⁵	N/A
Paramore et al. (1997), USA ¹⁵	0.4% visited acupuncturists
Astin et al. (1998), USA ⁴⁶	CM use: acupuncture (1.4%), massage (5.7%), herbal remedies (4.2%).
	CM use: acupuncture (1.0%), massage (11.1%), herbal medicine (12.1%);
	Visited practitioners: acupuncturist (0.9%); massage therapist (6.8%) herbal therapist (1.8%);
Eisenberg <i>et al.</i> (1998), USA ³	The mean number of visits to an acupuncturist per user was 3.1 per year.
Landmark Report (1998), USA ⁴¹	CM use: acupuncture (2%), massage (14%), herbal medicine (17%), acupressure (5%).
Druss <i>et al.</i> (1999), USA ⁴⁷	Visited practitioners: acupuncture (0.6%), massage (2.0%), herbal remedies (1.8%).
	CM use: massage (12%), herbal therapies (12%);
Ramsay et al. (1999), Canada ⁶	Lifetime use: acupuncture (12%), massage (23%), herbal therapies (17%).
Ernst <i>et al.</i> (2000), UK ¹⁷	CM use: acupuncture/acupressure (2.8%), massage (1.2%), herbal medicine (6.8%).

* Unless otherwise defined, use, practitioner visit and expenditure were in the 12-month period preceding the survey.

General findings in Chinese medicine
N/A
1.6% visited acupuncturists, 19.8% purchased over-the-counter herbal remedy;
Over 3 million visits were made to acupuncture providers in 1998;
The estimated annual out-of-pocket expenditure on acupuncture was 47 million pounds.
CM use: herbal medicine (13.4%), Chinese medicine (3.2%), Ginseng (5.0%);
Visited practitioners: acupuncturist (2.8%), herbal therapist (0.9%).
4.8% used herbal medicine, 2.9% used acupuncture during 1997-1999.
CM use: acupuncture (1.4%), massage (6.4%), herbal medicine (9.6%).
CM use: acupuncture and moxibustion (6.7%), massage and acupressure (14.8%); herbs and over-the-counter Kampo (17.2%).
CM use: acupuncture (1.1%), Tai Chi (1.3%), Qigong (0.3%), Ginseng (24.1%).
Visited practitioners: acupuncturist (1.6%), herbal therapist (0.8%), massage therapist (2.1%); traditional Chinese medicine (0.4%).
N/A
CM use: herbal medicine (20.6%), Chinese medicine (2.3%), Ginseng (5.0%);
Visited practitioners: acupuncturist (2.1%), herbal therapist (1.9%).

Table 2.15 Chinese medicine use in general population (extracted from major CAM population studies, continued)

* Unless otherwise defined, use, practitioner visit and expenditure were in the 12-month period preceding the survey.

CM prevalence was first described in a 1976 study in New York City's Chinatown.¹⁰⁰ It found that 17.2% of the population had seen CM practitioners in the previous two years. However, the sample size was rather small (87 participants) and the survey did not differentiate between visits for acupuncture, massage or other services. Studies in Hong Kong (HK)^{101,103} and Taiwan⁵¹ were based on data over 10-15 years ago, the representation of the current status of CM use is not clear.

The highest prevalence (48.9% had previously consulted an herbalist) was reported in a Hong Kong territory-wide survey conducted by Wong *et al.* in 1990.¹⁰¹ In contrast, a low prevalence of previous use of CM (13.5%) was reported in another Hong Kong survey in 1995,¹⁰³ based on randomly selected household residents. Study subjects for the latter survey were selected in the Kwun Tong district of Hong Kong, and females in the households were chosen to answer questions on behalf of all family members. Thus, this 1995 survey included utilisation data from all age ranges and may influence the overall prevalence.

A study in Taiwan employed an unusual data collection method.⁵¹ Based on a four-week diary and a retrospective interview, the researcher found that nearly one in 10 people used CM as their major form of medical care. Moreover, a large-scale study on acupuncture conducted at the same time also found just over one tenth (12.4%) of the Taiwanese population had ever received acupuncture treatment.¹⁰² Seven studies on CM use by patients, students or other defined populations were found (Table 2.14). These studies were conducted in the US,¹⁰⁵ Canada,¹⁰⁶ Singapore¹⁰⁹ and Hong Kong.^{104,107,108,110} Similar to the findings on CM use by the general population, CM prevalence in defined populations varied markedly.

The first in-depth, large-scale survey of Chinese medicine use in the US¹⁰⁵ found that 99% of the survey participants had previously received acupuncture. The use of Chinese herbal medicine and moxibusion by the study's participants was also substantial. However, the study was conducted in six selected clinics, including Chinese medicine clinics. It is highly likely that study participants attended these clinics for the purpose of a specialised Chinese medicine service. Thus, the results of this study on CM use have highly to be much higher than its actual use by the general population (Table 2.14).

A Canadian study by Wong *et al.*¹⁰⁶ recruited patients who had visited medical practitioners without Chinese medicine expertise. The prevalence of 28% of patients and their family members who had used Chinese herbal medicine in the past year was similar to a Singapore study¹⁰⁹ (25.3%) on elderly people aged 65 or older.

The use of CM in Hong Kong, a specially administrated region of China, had been considered more popular than in most, if not all, Western countries. However, this was not clearly confirmed by four Hong Kong studies. ^{104,107,108,110} A relatively low prevalence was reported for secondary school students (8.6% consulted CM practitioners over a three-month period),¹⁰⁷ for people suffering from acute illness (14% lifetime consulting rate),¹⁰⁴ and for chronic hepatitis B patients (16% at the time of the survey).¹¹⁰

Acupuncture and herbal medicine are popular forms of CAM in many countries. Although very few studies have been conducted to collect data exclusively on acupuncture and other forms of CM, promising data are available from well-designed CAM studies. Such relevant data were extracted and are discussed below. With a lack of specific studies on CM use in Western countries, the extracted data are of particular value.

Of 21 identified CAM studies, three did not provide information on Chinese medicine in any form. Information on acupuncture was specifically investigated in the majority of studies, however, limited information is available concerning overall CM prevalence. Moreover, no studies included a classification of Chinese-style massage and herbal medicine. Prevalence data on acupressure and moxibusion have been occasionally reported (Figure 2.5).

Overall CM prevalence was determined in three Australian studies⁹⁻¹¹ and a UK study.¹⁶ In Australia, the prevalence increased markedly from 1993 to 2000, from 1.8% to 3.2%, but then deceased to 2.3% in 2004. The prevalence of visits to CM practitioners was estimated in a UK study. ¹⁶ It was estimated that 0.4% of the total population had consulted CM practitioners in a 12-month period but this excluded visits to acupuncturists, herbalists and massage practitioners.

The prevalence of herbal medicine use in the US ranged between 3.0% in 1990^2 and 17% in 1997.³ Based on four studies between 1997 and 1999, the average prevalence of herbal medicine use was 10.8%.^{3,4,41,46} A high prevalence of herbal medicine use was also reported by studies in the UK (19.8%),⁵ Australia (20.6%)¹¹ and Japan (17.2%).⁷ Similar to the prevalence of herbal medicine use, approximately one in ten Americans used massage

therapy.^{3,41} This prevalence was similar to that in Canada and Japan. However, a 1999 UK study suggested that only 1.2% of the population used massage.¹⁷ For acupuncture, except for a higher rate (6.7%) reported in Japan,⁷ most studies estimated the prevalence at less than 3%.

In terms of practitioner visits, the prevalence of visits to acupuncturists, massage therapists and herbal medicine practitioners was reported in the US, the UK and Australia (Figure 2.6). Although the use of acupuncture treatment was frequently reported, less than 3% of the populations was estimated to have visited such a practitioner in the previous 12 months, based on the utilisation studies in six countries (Figure 2.6). These studies also estimated that less than 2% had consulted herbal medicine practitioners.



Figure 2.5 Prevalence of Chinese medicine use in six countries

CM: Chinese medicine; HM: Herbal medicine

Note: Years were at the time of conducting studies which are different from the dates of publication. US90, US97A (Eisenberg *et al.*^{2,3}); US97B (Landmark report⁴¹); US98 (Astin⁴⁶); US99 (Ni *et al.*⁴); US02 (Barnes *et al.*¹); UK93, UK98 (Thomas *et al.*^{5,44}); UK99 (Ernst *et al.*¹⁷); Canada99 (Millar *et al.*⁴⁸); AUS93, AUS00, AUS04 (MacLennan *et al.*⁹⁻¹¹);

Italy99 (Menniti et al.⁴³); Japan01 (Yamashita et al.⁷).



Figure 2.6 Prevalence of visits to Chinese medicine practitioners in three countries

CM: Chinese medicine; HM: Herbal medicine

Note: Years were at the time of conducting studies which are different from the dates of publication. US90, US97A (Eisenberg *et al.*^{2,3}); US94: (Paramore *et al.*¹⁵), US96 (Druss *et al.*⁴⁷); UK98, UK01 (Thomas *et al.*^{5,16}); AUS93, AUS00, AUS04 (MacLennan et al.⁹⁻¹¹).

2.3.4.3 Discussion

In the US, Chinese medicine (CM) is an integral part of the growing field of CAM.²⁸ The results of studies drawn from the past three decades also suggest that CM has been used by a large part of the population of several countries, and that it is one of the most popular forms of alternative treatment to deal with diseases and maintain general health. Based on the current review, the prevalence of CM use may not be conclusive for most countries. In addition, in some countries, at least in China, CM utilisation, including public use and practices, may have been investigated and reported in some other forms, such as government reports.

Although an eligible CM utilisation study has not been identified in Mainland China in the current review, data on Chinese medicine services, including specialised Chinese medicine hospitals and clinics are widely available in China. An examination of data from the State Administration of Traditional Chinese Medicine, P. R. China (SATCM)¹¹³ suggested that, in 2004, there were 3,716 Chinese medicine hospitals in China, including hospitals where CM was integrated with Western medicine (WM) and hospitals of national minority medicine. The SATCM also estimated that, in 2003, in China, nationwide, there were more than 171,419 CM physicians and 30,934 assistant CM physicians.¹¹³ As such, if rigorous utilisation study was designed to estimate the total prevalence of CM in China, substantial proportion of CM usage is highly to be estimated.

When conducting the literature search for CM, a considerable number of studies on general herbal medicine, rather than specific Chinese herbal medicine, were also identified. However, herbal medicine/products presented in such studies were mainly Western-type medicines (e.g. St. John's Wort, evening primrose and Echinacea). Detailed analyses of the extremely broad range of herbal medicine modalities are considered beyond the scope of the current thesis.

2.4 Summary of the Literature Review

These three systematic reviews considered to provide a comprehensive overview of current CAM utilisation in general populations around the world. CAM use in different regional and clinical populations in Australia, as well as the use of all forms of Chinese medicine around the world, have been systematically summarised. Due to the heterogeneity of the studies included, data from these reviews should be interpreted with caution, despite the fact that the reviews have been completed adhering to rigorous methodology. Nevertheless, the researcher performed these tasks through an objective protocol: an explicit statement of the literature search strategy (see Table 2.3), a detailed inclusion and exclusion criteria (see Chapter 2.3.2.1) and critically, a standardised, predefined data extraction form (see Table 2.4).

As expressed in Chapters 2.1 and 2.2, it appeared that there was little capacity to compare international literature in CAM utilisation surveys due to the substantive differences in its definition and the totality of CAM modalities. As might be expected, similar obstacles were encountered in the current review, particularly with respect to the systemic comparison between countries/regions. Nevertheless, the important function of these reviews was to summarise the existing literature to assist the design of a population-based CAM survey in Australia. The methodological issues identified were discussed in details (see Chapter 8.2).

Although there are considerable uncertainties about the true prevalence of overall CAM and CM use, the prevalence of most forms of CAM (e.g. herbal medicine) in many countries was found to be increasing markedly. On the basis of the data outlined for Australia, it is apparent that overall CAM prevalence is generally higher than in most other Western countries. Specifically, the highest prevalence of a single form of CAM (aromatherapy) was found in Australia amongst the majority (if not all) Western countries.^{10,11} This therapy, surprisingly,

was not included in previous US national health interviews,^{1,4} nor has it been reported as part of overall CAM prevalence in the US.^{2,3}

In 2005, the Kobe Centre for Health Development of the World Health Organisation (WHO) published a comprehensive global atlas of CAM use.¹¹⁴ It was designed to record details of policy, regulation, education, research practices and use of CAM around the world. CAM use in six WHO regional areas was outlined.

In the WHO African region, it has been suggested that CAM, also known as African Traditional Medicine (ATRM) is still the only form of health care in some communities.¹¹⁵ In addition, it has been estimated that more than 80% of Africans used ATRM, whereas only about 50% of the African population had regular access to essential pharmaceuticals.¹¹⁶

In the WHO region of the Americas, a review was conducted across 35 member states/countries. Apart from the US and Canada (see Table 2.4), a CAM survey was only available for the Argentine Republic.¹¹⁷ It revealed that more than half (54.4%) of the population of Argentina used CAM, among which homeopathy (40%), herbal medicine (36.7%), chiropractic (21.4%), acupuncture (12.9%) and bath flowers (11.6%) were most popular.

No consolidated data are available on utilisation and practices of CAM in South-East Asian region and the Eastern Mediterranean region, despite the belief that various modalities of CAM are being used in this region.^{118,119} Other than some general data on the use of Ayurveda and Unani in India, Bangladesh and Nepal; Thai therapy in Thailand; and Jamu in Indonesia, other forms of CAM have rarely been investigated.

There is a wide range of CAM within the European region, particularly in Western European countries, such as Germany, Denmark and the UK. More importantly, nearly all countries in this region have some form of government subsidisation of CAM.¹²⁰ Population surveys are available in the UK and Denmark to show that the trend of CAM use by the general public is increasing over time (see Table 2.4).

Perhaps, more comprehensive results are available in the Western Pacific region.¹²¹ Traditional Chinese medicine (including acupuncture, herbal medicines and massage therapy) is officially recognised in a number of countries, for example, China, Japan, Singapore and the Philippines. As mentioned in Chapter 2.3.4 (a systematic review of the specific prevalence of Chinese medicine), information available for these countries is mainly on the profile of CAM services and practices, other than the use by the general population. A profile of CAM usage in Australia was not included in the WHO report.

In summary, as discussed in most published CAM surveys, rigorous survey research is required to exclude the drawbacks of all aspects of CAM research, in particular the classification and definition. In addition, future research should focus on the use of listed, commonly used forms of CAM over a defined period, preferably the previous year. Other critical matters related to CAM use, are why people use CAM, and what are the perceived and actual benefits and risks of using CAM. Presented below is the methodology and results of a recently completed CAM national study in Australia, in which considerable efforts were made to answer the research questions and objectives according to rigorous standards.

CHAPTER 3. METHODOLOGY

3.1 Foreword

In general, a clear CAM definition, a reliable survey instrument, as well as rigorous sampling and data collection methods are vital in designing a reliable CAM survey. Based on the existing literature, with specific consideration to the Australian health-care context, the identified methodological concerns in conducting a national population CAM study in Australia are discussed below.

Extensive consultations were conducted with experienced CAM researchers, CAM academics and population health researchers to assist in the survey design. These consultations were extremely helpful in confirming the issues faced by researchers conducting CAM surveys. Data collected through computer-assisted telephone interviews (CATI) with randomly selected individuals using random-digit dialling (RDD) seem to be the most appropriate methods in the Australian context. Seventeen (17) forms of CAM were included in the survey on the basis of a review of existing literature on CAM utilisation, in Australia and overseas. Since a comprehensive list of CAM modalities used by the Australian population is not readily available, a pilot study was conducted. The initial draft of the survey instrument was developed from similar considerations.

The detailed sampling methods, the process of developing an appropriate survey instrument, the data collection protocol, the quality assurance process and the data analyses approaches are presented in this chapter.

3.2 Sampling Methods

3.2.1 Sample Size

In general, the sample size required to represent a population depends on the size of the population, the desired level of statistical significance and the acceptable level of margin of error. The Australian adult population at the time of conducting this study was approximately 15.3 million.¹²² CAM prevalence estimation in South Australia in 2000 was $52.1\%^{10}$ (confirmed in 2004 with $52.2\%^{11}$). With 95 percent confidence intervals of plus/minus 3%, it was determined that 1,067 interviews would be appropriate for the proposed study (Figure 3.1,calculated by Epi Info 6.0 package, Centers for Disease Control and Prevention, USA).¹²³

Enilnín Version ó	Stateale		November 1993		
Population Survey or Descriptive Study Using Random (Not Cluster) Sampling					
	Population Size	: 15,300,000			
	Expected Frequency: 50.00 %				
	Worst Acceptable	: 47.00 %			
	Confidence Level	Sample Size			
	80 %	456			
	90 %	752			
	95 %	1,067			
	99 %	1.843			
Fl-Heh	F5-Print	F6-Onen File	F10-Drne		

Figure 3.1 Sample size calculation using Epi Info 6.0 software package

The appropriate sample size can also be calculated through the modification of a common statistical formula:

Because an estimated margin of error can be calculated as: $e = z \sqrt{\frac{p(1-p)}{n}}$

The estimated number of interviews: $n = \frac{z^2 p(1-p)}{e^2} = \frac{1.96^2 * 0.5 * (1-0.5)}{0.03^2} = 1,067$

Where p = Value used to represent the population proportion (prevalence of outcome measurement, when p=0.5 it maximises the required sample size)

- e = Desired margin of error
- z = Critical value from the standard normal distribution table
- n = Target population

This calculation is based on the assumption that random samples will be drawn from an extremely large general population. In the current case, samples will be drawn from 15.3 million adult Australians, which can be considered as infinitely large in statistical terms.

3.2.2 Sample Selection

Target Population

The participants of the study were adults aged 18 years or older who were randomly selected from all states and territories around Australia. As all interviews were conducted in English only, only participants who were fluent in English were recruited. In addition, household members who had physical or mental handicaps, and those had not given oral consent to interviewers were excluded.

Stratified Household Sampling Frames

This study was conducted concurrently in six states and two territories: New South Wales, Victoria, Queensland, South Australia, West Australia, Tasmania, Australian Capital Territory and Northern Territory. Sample distributions were allocated at state level (Table 3.1). The populations of Tasmania, Northern Territory and the Australian Capital Territory are rather small. Thus, although representative proportions of interviews were achieved for all states/territories, it was decided that state-level comparisons would only be done for New South Wales, Victoria, Queensland, South Australia and Western Australia.

State/Territory	ABS (%)	Targeted sample size (n)
New South Wales	33.5	358
Victoria	24.9	266
Queensland	19.1	204
South Australia	7.8	83
Western Australia	9.8	104
Tasmania	2.4	25
Northern Territory	0.9	10
Australian Capital Territory	1.6	17

Table 3.1 Allocation of study samples in accordance with state and territory populations

In an effort to achieve a representative sample through the random dialling process, using the 2004 ABS population data percentages on age and gender,¹²⁴ interviews for each gender and each age range were then allocated (i.e. 18-24, 25-34, 35-44, 45-54, 55-64 and 65+) (Table 3.2). The percentages of interviews in the various age groups for each gender in the current samples matched the Australian population data. Participants were accepted in each defined category until the quotas were full. These protocols were to ensure the random digit dialling process (see below) and stratified sampling frames were practical. However, the use of quotas limited the randomised selection process, and the need to overcome this possible limitation will be discussed in this chapter.

	Male		Female	
Age range	ABS (%)	Quota set in the study	ABS (%)	Quota set in the study
18-24	6.57%	60-70	6.29%	60-70
25-34	9.39%	90-100	9.40%	90-100
35-44	9.75%	100-110	9.85%	100-110
45-54	8.94%	90-100	9.04%	90-100
55-64	6.94%	70-80	6.82%	70-80
65+	7.61%	75-85	9.40%	95-105

Table 3.2 Gender and age quotas set in the study

Selection of Participants within Households

The methodology used to select participants within households was the "next birthday" technique to minimise volunteer bias. Only one participant was selected from each household. Once identified, the participant would not be substituted with another member of the household.

3.3 Development of Survey Instrument

A definite CAM definition is not possible, yet CAM usually refers to therapies/products that purport to offer diagnostic and therapeutic alternatives or to complement conventional medicine. Due to the complexity of CAM modalities, it was decided to include a list of therapies based on the previously discussed systematic reviews of the literature (Chapter 2), which were commonly being used in Australia.

To address the issue of a potential overlap of CAM modalities, and to provide an appropriate classification of CAM, with consideration of the Australian health-care context and previous CAM utilisation studies nationally and internationally,^{1-5,9-11} 17 forms of CAM were included in the main survey (Table 3.3) (Appendix A). In addition, based on a previous study from our research group,⁶² it was decided to include the specific forms of Chinese medicine: Chinese herbal medicine, acupuncture, Chinese therapeutic massage, Chinese dietary therapy, Qigong, martial art and Tai Chi. This was done to ensure that all major forms of Chinese medicine were covered by the survey, and to produce explicit utilisation information on Chinese medicine as a whole within a general CAM study. This will allow comparison with previous Australian regional studies,^{10,11} which collected CAM prevalence by using different terms of individual CAM modalities, for example, "Chinese medicine" and "herbal medicine" were researched independently.

In addition to the 17 selected forms of CAM, the contracted professional survey company suggested that a list of "other CAM" should be included in the survey. This was to assist the interviewers when a participant indicated that they had used CAM other than, or in addition to any of the 17 selected forms of CAM. The list of "other CAM" modalities was available to interviewers through an auto-dropdown list on the computer interview screen (Table 3.4).

81

NCCAM Classifications*	17 CAM therapies
Alternative medical	1. Acupuncture 2. Homeopathy
systems	3. Naturopathy
Mind-body interventions	4. Meditation
	5. Yoga
Biologically based	6. Aromatherapy
therapies	7. Chinese herbal medicine
:	8. Chinese medicine dietary therapy
	9. Clinical nutrition (e.g. multivitamins and minerals)
	10. Western herbal medicine
Manipulative and body-	11. Chinese therapeutic massage
based methods	12. Chiropractic
	13. Osteopathy
	14. Reflexology
	15. Western therapeutic massage
Energy therapies	16. Energy healing (e.g. reiki and therapeutic touch)
	17. Qigong, martial art and Tai Chi

Table 3.3 NCCAM classifications of the 17 selected forms of CAM in the survey

* NCCAM: National Center for Complementary and Alternative medicine (US)

Table 3.4 List of "other forms of CAM" included in the survey

Alexander technique
Art/music/dance therapy
Ayurveda
Bowen technique
Hypnotherapy
Iridology
Kinesiology
Magnetic field therapy
Myotherapy
Prayer
Other forms of relaxation techniques (eg. Pilates)
Shiatsu
Otherspecify

A questionnaire was developed to answer the research questions which were designed to be completed in an average of 15 minutes. The questionnaire was divided into six sections (Table 3.5) and contained some 60 questions: 40 yes/no questions, 15 multiple-choice questions and five open-ended questions. In addition, groups of questions on four specific forms of CAM: acupuncture, CHM, chiropractic and osteopathy were developed by relevant experts in these fields, with particular emphasis on the benefits and risks, insurance coverage and regulation matters. These questions were later re-formatted in a coherent manner.

Table 3.5 Structure of the computer-assisted telephone interview questionnaire

- Section 1: General health and the use of conventional medical services
- Section 2: Prevalence and frequency of use, and expenditure on CAM
- Section 3: Rationales for CAM use and attitudes toward CAM regulation
- Section 4: CAM and private health insurance reimbursement and Medicare coverage
- Section 5: Questions on four CAM therapies that are regulated at national or state level
- Section 6: Socio-demographic information on participants

A total of eight revisions were made to the instrument between December 2004 and April 2005. The instrument was then programmed for use in a CATI telephone interviewing system. A sample of the final version of the questionnaire and a sample of the CATI programming script are attached (Appendix B C). Finalisation of the instrument was achieved after conducting the pilot study.

Participants were first asked about their use of the 17 CAM therapies and any other forms of CAM over the 12 months preceding the survey. Thus were asked whether or not they had visited a practitioner for any of the 17 forms of CAM. Participants were not required to answer all questions. Depending on each participant's historical CAM use, specific questions were customised by the CATI system. In an attempt to enhance the validity, the sequence of the 17 forms of CAM therapies was offered randomly for each interview session.

3.4 Computer-Assisted Telephone Interview

When determining whether to use a mail or telephone survey to collect information, it is important to evaluate the advantages and disadvantages of each method in relation to the research objectives. Mail surveys are frequently used and are relatively inexpensive. However, they tend to have a low response rate and are time-consuming to implement. Mail surveys are suitable for the investigation of sensitive issues, such as illicit drug use, or for surveys requiring participants to fill in complex and lengthy questionnaires. However, unlike telephone surveys, there is less opportunity for mail survey participants to provide explanations of their perceptions, beliefs and attitudes towards the research questions.

Computer-assisted telephone interview (CATI) involves electronic data collection, capture, and associated tasks.¹²⁵ It is commonly used in large-scale health surveys, such as the Victorian Population Health Surveys and the New South Wales Health Surveys. In Australia, the percentage of households with fixed telephones is 97.5%,¹²⁶ hence the CATI surveys methodology is feasible. The advantage of using CATI for a survey such as the one proposed, is that it is relatively cost effective, providing standardised prompts, thereby ensuring data consistency. It also offers direct data input and data output.^{127,128} For the reasons outlined above, it was decided that the study would employ the CATI method of data collection.

A CATI facility is currently not available at RMIT University, and the cost of setting up a CATI centre is substantial. State health authorities in Australia conduct CATI surveys, either through their own call rooms or by contracting with commercial providers. Thus, a survey research company was contracted to conduct the telephone interviews. The use of professional trained telephone interviewers also ensured that reliable data would be obtained. A similar approach has been used by other researchers for well cited reports.^{2,3}

3.4.1 Random Digit Dialling

Telephone surveys, based on the selection of random telephone numbers, were used in the early 1960s.¹²⁹ Most importantly, telephone numbers should be generated by a random digitdialling (RDD) method. Such a method had been used in previous similar CAM utilisation studies in the US.^{2,3} Unlike drawing numbers from the phone book listings, RDD allows for the inclusion of silent, unlisted and recently allocated telephone numbers. Thus, telephone numbers used in the interviews did not come from any specific source but were randomly generated by computer to ensure all household members had an equal probability of being contacted.

In an attempt to minimise company or other business numbers, existing residential-prefix telephone numbers was used. For the current study, the first step of the RDD sampling was to select telephone numbers from the current available Telstra (a national telephone company) telephone exchanges. Randomly generated telephone numbers with six-digit prefixes for each study area in different states/territories were produced. Two digit suffixes were then randomly appended at the end of the prefixes (e.g. 9666 55** might become a telephone number between 9666 5501 and 9666 5599).

A total of 20,000 telephone numbers was initially generated; the CATI computer system then screened out calls to facsimile machines, data lines, and unassigned and disconnected numbers, which meant only live calls were assigned to interviewers. Additional random telephone numbers were generated to recruit sufficient participants. A minimum of five call attempts was made to contact the household member. Calling proceeded, and a record of all calls was made, including non-reachable numbers, confirmed call-back numbers and household members who declined to participate.

3.4.2 Interviewer Training

In an effort to produce the precise prevalence of CAM users and non-users, interviewers with experience in health surveys were selected from the survey company to conduct the interviews for the current survey. All interviewers attended a four-hour training and debriefing session in which the fieldwork manager and study investigator went through each interview question. Interviewers were given explanations of survey background information and all listed CAM modalities to be investigated.

In addition, a brief interview memorandum with a short definition of each form of CAM was also provided to all interviewers, and used during interview sessions. Samples of common forms of CAM were highlighted on the computer interview screen, and were provided to the survey participants if requested. Other interviewer instructions included, "Australian common allied health professions such as physiotherapy and occupational therapy are not considered as CAM in this study", "please distinguish between CAM use under the supervision of a practitioner and use without such supervision"; "please confirm it was out-of-pocket expenditure excluding money you [the survey participant] have obtained rebates from any source", "please only provide expenditure and frequency of visits to CAM practitioners in the past 12 months"; "please differentiate the use of Chinese herbal medicine and Western herbal medicine, and Chinese therapeutic massage in compared to other massage therapies (i.e. Western therapeutic massage)".

3.4.3 Quality Control Procedures

For quality control purposes, two interview fieldwork supervisors randomly checked interview conversations simultaneously. The researcher was also on site, and randomly listened to most of the interview sessions. Interviewers had the opportunity to refer any questions from participants to the researchers during the interview sessions. The researchers also had the opportunity to correct or make comments on participants' responses. The detailed survey development process, including the quality assurance strategy, is summarised in the following flowchart (Figure 3.2).



Figure 3.2 Survey development-process and quality-assurance protocols

3.4.4 Interview Procedures

The procedure of conducting the CATI telephone interview is described below. The automated dialling CATI system makes the initial phone calls from the monitoring stations; all working telephones then transfer to the interviewers. The interviewers then initiate the interview procedures, and the computer program controls questions branching or skipping. The interviewers then conduct the tasks of interviewing and entering the data into the computer system simultaneously. Where appropriate, the CATI system will also perform data checking while an interview is in progress. The subjects' responses are then transferred directly into a CATI program linked to a database. These procedures ensured accurate and consistent answers were given by the participants.

The interactive statistical software in the CATI system permits rapid preliminary analysis of the survey data. The final output of the data will be produced in different formats: raw, uncoded preliminary data files, Excel and SPSS data files and partially coded qualitative data files. The project managers in the research company will also audit a minimum of 10% of the interviews.

The process of CATI auto branching is very important to collect reliable information. For instance, if a participant responded that they had never used any form of CAM, the CATI system would skip to the relevant questions instead of asking them specific questions on using CAM. Similarly, if a participant had used a specific form of CAM, the CATI system would continue to ask questions relevant to that type of CAM (e.g. practitioner visits and frequency). As a result of this automatic branching or skipping of questions, interview time was minimised.

Having complied with the ethical requirements, potential participants were given an abbreviated statement in plain language with background information about the survey to obtain their consent to participate (Appendix D). A full, printed version of the statement was provided if a potential participant requested one (Appendix E). Participants were informed prior to participating in the interview that this survey was intended to obtain accurate information about CAM utilisation to determine national prevalence data.

The CATI interviews were conducted in May and June 2005. A maximum of 15 attempts were made to establish contact with an individual. People who normally stay at home during the day may have different characteristics, compared to those normally at work during the day. Thus, the telephone calls were made from nine o'clock in the morning to nine o'clock in the evening, seven days a week, to get a balance of different groups of participants. Appointments were made during the duration of the survey period, and an additional 10 attempts were made to contact each identified individual with an appointment. No financial incentive was provided.

In order to obtain accurate information on CAM prevalence, the following questions were asked: "In the past 12 months, have you used any one of the following types of complementary medicine? Please say yes or no when it is named". If a participant used one specific form of CAM, the following questions were asked: "Have you visited a practitioner for [name of automated CAM appeared in the screen based on previous question] in the past 12 months?" and "how many times in total did you visit a practitioner for [name of CAM] in the past 12 months?" Due to the limitation on the time of interviews, the frequency question was asked only if a participant used one of ten types of CAM (mainly provider-based CAM therapies, see result section Table 4.9). The flowchart below summarises the process of the interview (Figure 3.3).



Figure 3.3 Flowchart of computer-assisted telephone interview questions

3.5 Pilot Study

3.5.1 Foreword

Ethics approval (Appendix F) was obtained for the study from RMIT University's Human Research Ethics Committee. A pilot survey with 30 randomly selected individuals in the state of Victoria was conducted in April 2005. The responses from the pilot study helped to clarify the understanding on the 17 CAM therapies to be included in the main study. The pilot study was also used to validate and refine the survey questions, the flow of the interview, and to assess the quality assurance procedures of the research company contracted to conduct the fieldwork. As a result of the pilot study, changes were made to the questionnaire programming, the final version of the questionnaire and the interview schedules. Interview notes were provided to interviewers to clarify some frequently asked questions that arose from the pilot study.

3.5.2 Results of the Pilot Study

Telephone interviews were successfully conducted with 30 randomly selected Victorians. Their age ranges, gender and country of birth are presented in Table 3.6 below, and are compared with the 2001 Victorian population data.

С	haracteristic	Results of pilot study %	ABS 2001 Victoria %
	18-24	16.7	12.7
	25-34	20.0	19.8
	35-44	30.0	20.4
	45-54	10.0	18.0
	55-64	13.3	12.2
Age range	65+	10.0	16.9
	Male	33.3	49.1
Gender	Female	66.7	50.9
	Australia	80.0	71.1
Birthplace	Overseas	20.0	28.9

Table 3.6 Basic demographic information on pilot study participants

In relation to CAM utilisation, 22 out of 30 (73.3%, unweighted percentage) participants had used at least one of the 17 forms of CAM in the 12 months preceding the survey, while 86.7% participants had visited a medical doctor in the same period. The most popular forms of CAM were: aromatherapy, multivitamins or nutritional supplements, and yoga (each modality, >20%). Other interesting findings include:

 Fifty-nine percent considered Western medicine has a relatively higher risk of side effects, while 77% considered CAM to be relatively safe.

- 2. One in four participants who used CAM had never informed their medical doctors.
- 3. Out-of-pocket expenditure on CAM products was between \$0 and \$350 per annum.
- 4. Out-of-pocket expenditure on CAM consultations was between \$0 and \$1,000 per annum.
- 5. Additional out-of-pocket expenditure on CAM-related items was between \$15 and \$1,500 per annum.
- 6. If a private health insurance offered CAM for no or little additional cost, 60% participants would buy insurance from that company.
- 7. Medicare should or should not cover visits to practitioners of selected forms of CAM:
 - 80% suggested it should cover visits to an acupuncturist in addition to a registered GP who provided acupuncture treatment.
 - 63.3% suggested it should cover visits to Chinese herbal medicine practitioners
 - 66.7% suggested it should cover visits to chiropractors
 - 70% suggested it should cover visits to osteopaths

Due to the small size of the pilot sample, the above data were collected to assist the design of the main survey only and must not be generalised.

In the pilot study, the researcher randomly listened to some interview sessions, and established that the vast majority of participants were able to follow the sequence of the interview questions. However, participants commented that the interview was lengthy and that some questions were repetitive (specific research questions on four forms of CAM: acupuncture, etc). Importantly, some participants were unsure about some forms of CAM and queried them. A clear explanation was desirable. In addition, two interviewers mentioned that they were unable to inform survey participants what homeopathy and osteopathy were. These matters were resolved before the main survey.
3.5.3 Discussions and Finalisation of the Survey Instrument

Perhaps the most important outcome of the pilot study was that it indicated that over 70% of the participants had used at least one form of complementary medicine in the past 12 months; this provided further confirmation of the adequacy of the target sample size (i.e. 1,067).

All participants were also asked whether they had used any other forms of CAM other than the 17 surveyed. Only three mentioned such use, with none of them saying they had used other CAM but not any of the 17 forms of CAM. Thus, the pilot study also indicated that these 17 forms of CAM therapies are likely to be the most commonly used CAM in Australia. Therefore, the pilot study to test CAM modalities used in Australia was most valuable for finalising the survey instrument.

Other problems encountered during the pilot study included the definition of multivitamins and minerals. This was expected and has been commented by researchers in CAM utilisation studies.¹ However, there is no easy way for the general public to define "non-CAM vitamins" and "CAM vitamins". It was decided, therefore, for the purpose of this survey, the term "clinical nutrition" would include multivitamins and minerals. The term clinical nutrition is more commonly known in Australia. The use of the prefix "multi-" also indicated that the current study was not referring to a single vitamin consumed daily. Interviewers were expected to clarify this with participants while conducting the main survey.

In addition, a few questions were designed to capture the reason(s) why participants used a particular form of complementary medicine. For instance, the original question on chiropractic was: "you mentioned earlier that you have sought chiropractic treatment in the last 12 months. Would you say the main reasons were: [read out by interviewers] 1) To improve general

health and well-being; 2) To improve sporting performance; 3) To improve [the] ability to undertake normal daily activities; 4) For relief of symptoms; 5) None of above–other specify." The question was designed with multiple choices, as participants may have been using chiropractic for multiple purposes. The computer program, however, captured only one answer then skipped to the next question, so interviewers lost the opportunity to gather additional answers. Similar programming problems also occurred with another three questions. Such problems were resolved in consultation with the CATI script programmer before the main survey.

Finally, as would have been anticipated, the recruitment from certain cohorts of participants was difficult. This was particularly observed in young males aged 18-34. This reinforces that the use of a national quota (see Table 3.2) in the main survey is necessary to recruit a nationally representative study population.

3.6 Data Analysis

3.6.1 Data Consolidation and Missing Data

Abandoned interviews (four in total) did not count toward the total of 1,067 successful interviews. Information on them is not available, even though the interviews were partially completed, as the temporary records were deleted immediately after the interviews had been discarded.

The final dataset produced by the survey company was a SPSS (Statistical Package for Social Sciences¹³⁰) data file. Because of the abovementioned quality assurance protocol, and the fact that the information was recorded by using a computer-assisted telephone interview system, missing data was minimal throughout the data set. The number of missing data in each research question is no more than 5%; thus, according to Korn and Graubard, this is fully acceptable.¹³¹

There are two types of missing data: the purposively designed non-relevant items and items to which participants did not respond. Non-relevant items are those that were not relevant to a particular participant; for example, a question about expenditure on CAM was not put to a non-CAM user. A negative value "-1" was assigned to these responses. Such results do not have any impact on the data analyses. While the "real" missing data (e.g. a CAM user who did not state CAM expenditure for no given reason), was assigned an extreme value of "999999" and was clearly defined in the data property in the SPSS file.

The task of identifying missing data and distinguishing it from non-relevant data was carried out in two steps. Firstly, all data were exported to a Microsoft ACCESS database, using purposively designed queries, those participants who stated "don't know" or "refused to answer" were identified. These mainly occurred in questions about CAM expenditure, number of visits to CAM practitioners, participants' age, income and personal demographic information. In the second step, a new function available from the SPSS software 14.0¹³² was employed. The "data validation" function was particularly useful in customising a particular rule to check whether an individual value was within a valid range and to check the minimum and maximum data values in each data cell. It also flagged the identification of invalid cases for further investigation. A number of outliers of particular values had been flagged in this step for future notice. The overall cases of missing data in any variable are small. Thus, a missing value analysis was not conducted.

The original dataset consists of 236 variables (Appendix G1). However, qualitative data (e.g. medical conditions for which a form of CAM was used) and "other-specify" questions (e.g. other forms of CAM and country of birth) were recorded in a separate Excel file. Such data were merged with the original dataset. More importantly, before any analysis could be carried out, a series of variables was created. These included "CAM user", "those who visited CAM practitioners", "acupuncture user", "chiropractic user" etc. Questions with multiple responses were defined as a separate set of data in order to produce multiple-response frequency tables. Finally, most demographic data were recoded in accordance with relevant classifications used in government reports.

3.6.2 Reliability and Validity of Survey Instrument

Although questions contained in the current survey had been tested by a group of researchers and re-tested in a pilot study, it should be noted that the newly developed instrument has not been widely tested and its reliability has not been formally and statistically evaluated. Also, little information is known about its validity.

The relationship between reliability and validity requires further consideration. Outcome data can be valid but not reliable. Also, a measure can be reliable but invalid.¹³³⁻¹³⁵ Very often, the reliability, which measures the consistency of obtaining data from a scale, can be measured by examining how people answer each item of a set of questions. Mathematically, Cronbach's alpha (α) value can be calculated for each section of the questionnaire with a value that ranges from 0 to 1. In general, an alpha coefficient of 0.70 or higher is considered acceptable for most instrumental design purpose.¹³⁶ Unfortunately, for validity, a conclusive method of assessment is not readily available. Most social science analyses rely on a scale that produces results that are expected on the basis of well-established theories.¹³⁶ Most often face validity is ascertained via experts within the particular area that is researched.

Therefore, scale reliability analyses (through SPSS) were carried out for different sections within the instrument to assess internal consistency of items measured by the scale. This is to ensure that answers from participants will not differ because the survey is confusing or has multiple interpretations.

In the current survey, prevalence questions and attitudes towards a series of pre-defined questions on CAM-related matters are of major concern from a reliability perspective. The key

results of the reliability test on the 17 CAM prevalence questions confirm that the scale is reliable. The Cronbach's alpha value is calculated as 0.759, which suggests that the scale is reliable (i.e. a value higher than 0.7).¹³⁷ More specifically, Table 3.7 shows the item-total correlation of these 17 questions. It indicates that when removing any of the items, the Cronbach's alpha value would not increase markedly, from the calculated reliability of the entire scale, 0.759. Therefore, the reliability coefficient for all items may be considered appropriate. In addition, Table 3.8 shows the inter-item correlation for the 17 CAM prevalence questions; the average correlation value is 0.163 (standard deviation (SD)=0.08) with a maximum of 0.404. It reveals that responses to these questions are not highly correlated. In addition, the item-total correlations of these questions range between 0.199 to 0.456 (Mean=0.354, SD=0.07). According to Briggs and Cheek,¹³⁷ this falls within the optimal range for the inter-item correlation (0.2-0.4).

Similarly, when examining the questions on whether or not participants visited a CAM practitioner, Cronbach's alpha value of 0.760 and an average inter-item correlation of 0.171 (SD=0.09) were obtained. Moreover, considerably high Cronbach's alpha values were also achieved for the items on attitudes towards CAM and Western medicine (α =0.954); the reasons for informing medical practitioners about CAM use (α =0.893); the reasons for not informing medical practitioners about CAM use (α =0.970); the reasons for using acupuncture, Chinese herbal medicine, chiropractic or osteopathy (α =0.924); the recommendations received for using these four therapies (α =0.908); and questions on regulations and private health insurance related to CAM use (α =0.763).

	Scale Mean if Item Deleted	Scale Variance if Item Deleted	Corrected Item-Total Correlation	Cronbach's Alpha if Item Deleted
1 Use ACU	2.07	5.422	.340	.748
2 Use CHM	2.09	5.415	.406	.744
3 Use CTM	2.11	5.554	.349	.748
4 Use CMDT	2.14	5.724	.290	.753
5 Use QG MA TC	2.10	5.612	.269	.753
6 Use WHM	1.99	5.150	.398	.742
7 Use WTM	1.89	4.876	.456	.737
8 Use Chiropractic	2.00	5.488	.199	.762
9 Use Osteopathy	2.12	5.678	.244	.755
10 Use Homeopathy	2.10	5.437	.417	.743
11 Use Naturopathy	2.06	5.266	.428	.740
12 Use Meditation	1.99	5.131	.400	.742
13 Use Aromatherapy	2.00	5.138	.417	.740
14 Use Clinical Nutrition	1.71	5.033	.309	.757
15 Use Energy Healing	2.09	5.434	.393	.745
16 Use Reflexology	2.12	5.609	.338	.750
17 Use Yoga	2.04	5.311	.370	.745

Item-Total Statistics

Table 3.7 Item-total correlation matrix of the prevalence questions on 17 forms of CAM

Inter-Item Correlation Matrix																	
	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17
1 Use ACU	1.000																
2 Use CHM	.318	1.00															
3 Use CTM	.355	.324	1.000														
4 Use CMDT	.207	.385	.303	1.000													
5 Use QG MA TC	.067	.143	.153	.107	1.00												
6 Use WHM	.160	.239	.139	.159	.074	1.00											
7 Use WTM	.219	.187	.214	.110	.147	.175	1.000										
8 Use Chiropractic	.175	.052	.043	.022	.047	.080	.232	1.00									
9 Use Osteopathy	.153	.158	.125	.101	.073	.120	.159	.054	1.00								
10 Use Homeopathy	.230	.313	.197	.245	.063	.269	.159	.101	.193	1.00							
11 Use Naturopathy	.115	.218	.162	.132	.103	.324	.215	.147	.177	.404	1.00						
12 Use Meditation	.134	.145	.099	.107	.244	.219	.216	.051	.125	.210	.178	1.00					
13 Use Aromatherapy	.136	.143	.159	.087	.199	.206	.313	.085	.068	.135	.216	.261	1.000				
14 Use Clinical Nutrition	.070	.137	.063	.061	.062	.275	.220	.140	.077	.169	.215	.169	.194	1.00			
15 Use Energy Healing	.147	.176	.196	.137	.252	.155	.248	.083	.092	.181	.174	.290	.323	.068	1.000		
16 Use Reflexology	.123	.230	.222	.170	.124	.116	.248	.083	.084	.174	.153	.158	.211	.098	.304	1.00	
17 Use Yoga	.107	.142	.120	.084	.210	.185	.247	.007	.134	.163	.216	.356	.268	.142	.178	.123	1.00

Table 3.8 Item-item correlation matrix of the prevalence questions on 17 forms of CAM

ACU: Acupuncture, CHM: Chinese herbal medicine, CTM: Chinese therapeutic massage, CMDT: Chinese medicine dietary therapy, QGMATC: Qigong, martial art and Tai Chi, WHM: Western herbal medicine, WTM: Western therapeutic massage.

3.6.3 Data Weighting Procedures

The purpose of the data weighting process was to adjust for representativeness with the national population. The weighting procedure used in this thesis is the most conventional, post-stratified direct weighting method.¹³⁸ The quota used in the CATI interview was based on nationwide statistics; a reasonable distribution of the study population was achieved at the end of the recruitment to represent Australia's overall population. However, the final make-up of the study population does not match the national population distribution exactly, particularly on a state/territory level. In addition, at the time of developing the national quota, the available ABS population data were for June 2004, while the study participants were recruited between May and June 2005. A post-stratified weighting process was required.

As a result of the age/gender/region weighting process, a total of 89 weighting values were applied to the original dataset (Table 3.9). The method used to calculate the weighting value is in accordance with the number of interviews in this survey and the proportion of sample required to represent corresponding adult Australians.¹³⁸ For instance, a total of 28 interviews were conducted with male participants aged 18-24 in New South Wales; this represented 2.63% of the total 1,067 interviews in the current survey (three participants who did not state their ages were weighted differently, see Table 3.9). Based on the ABS data, there was a population of 330,628 adults in this division, or 2.15% of the total adult Australian population. The actual interview numbers exceeded the total number of people required to represent the overall national population. Thus, the assigned weighting value for people within this division is less than 1.0. This can be calculated as a ratio of approximately 2.15% over approximately 2.63%, which equals 0.81620. The sum of the total weights equals the total number of adult Australians interviewed (1,067).

Weighting values*											
Regions/age	18-24	25-34	35-44	45-54	45-54	65+					
	Male										
New South Wales	0.81620	0.83725	0.88032	1.09447	1.40855	0.75336					
Victoria	1.44342	1.77789	1.42955	0.83317	0.87099	1.88991					
Queensland	1.00671	0.83976	1.04984	0.93082	0.72145	1.17948					
South Australia	1.73776	1.39896	0.97004	0.93170	2.01745	3.56656					
Western Australia	1.20118	1.40761	0.87729	1.09573	0.94109	0.93408					
Tasmania	0.79241	1.92156	0.57844	1.20164	0.49743	0.72875					
Northern Territory	0.82503	1.24432	0.60585	0.48744	0.63889	0.35148					
ACT	N/A^{\dagger}	0.88589	0.33046	0.77136	0.57426	0.96805					
		Fem	ale								
New South Wales	1.55654	1.07618	0.97995	1.27818	0.92574	1.26370					
Victoria	0.83665	1.25713	0.82253	0.85538	0.80423	0.74546					
Queensland	1.33450	0.76631	1.57041	0.81622	1.33693	1.27532					
South Australia	0.70255	1.66628	0.70077	0.54470	0.68234	0.69950					
Western Australia	1.13337	1.06591	1.04641	1.41271	1.20283	1.11831					
Tasmania	0.49671	1.00738	1.21232	1.22504	N/A^{\dagger}	2.68163					
Northern Territory	N/A^{\dagger}	N/A^{\dagger}	N/A^{\dagger}	N/A^{\dagger}	0.50673	N/A^{\dagger}					
ACT	1.29250	0.43840	0.28479	1.65587	N/A^{\dagger}	0.59783					

Table 3.9 Data	weighting	values	applied	in	the data	analyse	S
----------------	-----------	--------	---------	----	----------	---------	---

ACT: Australian Capital Territory *Three participants did not state their age, a weighting value of 1.00000 was applied. [†] N/A: Interviews did not conduct in that regional areas, a weighting value was not applicable.

The method used to define the weighting values for each participant was through the SPSS syntax technique (Table 3.10).

Table 3.10 An example of SPSS syntax for data weighting protocol

COMPUTE Wgt=0.

IF ((Region=NSW) and (Age=18-24) and (Gender=Male)) Wgt=0.81620.

IF ((Region=VIC) and (Age=18-24) and (Gender=Male)) Wgt=1.44342.

IF ((Region=OLD) and (Age=18-24) and (Gender=Male)) Wgt=1.00671.

IF ((Region=SA) and (Age=18-24) and (Gender=Male)) Wgt=1.73776.

IF ((Region=WA) and (Age=18-24) and (Gender=Male)) Wgt=1.20118.

IF ((Region=TAS) and (Age=18-24) and (Gender=Male)) Wgt=0.79241.

EXECUTE .

Alternatively, a weight can also be based on how many Australian each participant represents. On average, each of the 1,067 participants in the current survey represented a total of 14,544.37 adult Australians (based on the ABS June 2005 data,¹²² the total number of adult Australians was 15,518,843). More specifically, for the abovementioned 28 male adults aged 18-24 in New South Wales, each participant will represent a total of 11,808.11 persons (the total adult population in that division was 330,628¹²²). When applying these weighting factors, the sum of the weights was the total number of adult Australians aged 18 or over at the time of the survey, which was approximately 15.5 million people. However, this method produces large weighting values and is more difficult to manage when performing SPSS data analyses. Consequently, the method mentioned earlier (direct weighing method), with much smaller weighting values, was used in this study.

3.6.4 Statistical Analyses

3.6.4.1 Descriptive Analyses

A general descriptive analysis was initially carried out to examine the distribution of the data in terms of their central tendency and spread, such as mean, median, standard deviation, skewness and kurtosis. A symmetric distribution of a number of variables (e.g. the CAM cost and number of visits to CAM practitioners) is of particular importance before conducting correlations or regressions analyses (see below). When calculating prevalence and other major results, 95% confidence intervals are also calculated and, where appropriate, the exact level of statistical significance and the corresponding numbers are also provided.

Standard errors (SE) for all proportions are calculated as: $SE=\sqrt{p^*(1-p)/n}$ where p is the proportion, and n is the total number of participants investigated. Consistent with some ABS reports, relative standard error (RSE) is also calculated with major prevalence. The formula is: RSE (as a percentage) = (Standard error of the percentage/actual estimation)*100. All estimations with a RSE greater than 25% are recommended to be used with caution, while a RSE greater than 50% is considered unreliable, as it does not meet the standard of reliability.

3.6.4.2 Inferential Analyses

The independent t-test, and one-way and two-way ANOVA tests were used to test for differences between groups. These were conducted via the SPSS program. Importantly, for prevalence data, the Z-test was used to determine the significance of differences between two percentages (through the statistical formula below (calculated in Microsoft Excel) and also confirmed in the MINITAB–a specialised statistical software package).

$$Z = \frac{|X_a - X_b|}{\sqrt{S_a^2 - S_b^2}} = \frac{|X_a - X_b|}{\sqrt{X_a(1 - X_a)/N_a - X_b(1 - X_b)/N_b}}$$

Where X_a and X_b are two percentages being compared

 S_a and S_b are the standard errors of those percentages N_a and N_b are the total number of participants investigated The critical z value at the 0.05 significance level is ±1.96

Cross-tabulation analysis was used to reveal the frequency and common characteristics of the outcome measurements: CAM users and related variables. Because participants' demographic information is mainly in the form of categorical variables, the results can be displayed in customised multi-way frequency tables. The chi-square-based statistics (phi, Cramér's V, and contingency coefficient) are obtained depending on the nature of the variables (nominal, ordinal or interval). In the result section, the weighted percentages are presented rather than the actual frequencies of cases together with their chi-square value and probability (p).

Evidence exists to explain the characteristics that account for the use of CAM and visits to CAM practitioners. These include being female, having a better socio-economic status, holistic personal beliefs and experiences with conventional medicine. Thus, hypotheses on factors that contribute to the use of CAM were tested to confirm whether an association exists. These hypotheses will be discussed in the logistic regression model section.

3.6.4.3 Multivariate Logistic Regression

A regression model is commonly used to assess the relationship between a dependent variable and a set of independent variables (predictors). In the case of the current study, the dependent variables of interest are mainly dichotomous, (e.g. whether or not CAM was used) with more than one explanatory independent variable. According to the principles of correlation and regression, a binary logistic regression is most appropriate.¹³⁹ Mathematically, in general, the overall effect of the independent variables (known as predictors in the regression) on the dependent variable (the outcome measure) is expressed by the square of the correlation coefficient (r^2), which can be used to predict the proportion of the variance in the dependent variable which is explained by its relationship with the independent variables.¹⁴⁰

In the current study, the major dependent variables are "CAM user" and "visited CAM practitioners" which represents the use of any of the 17 forms of CAM and visits to a practitioner for any one of the 17 forms of CAM, respectively. Separate regression analyses incorporating predictor variables were carried out within four specific forms of CAM. Guided by a previous US national study on why people use CAM,⁴⁶ a list of the hypothesised predictor variables that used to assess the potential correlation with CAM use are generated (details below). However, the use of the existing classification may eliminate and/or include potential factors, both unintentionally and intentionally. Thus, the maximum number of independent variables was included in the initial model-developing stage. Final regression outputs presented in this dissertation are variables that had been entered via a step-wise method and which contributed significantly to the model.

Although logistic regression does not assume a linear relationship between the dependent and independent variables, and the dependent variables need not be normally distributed, there are

106

a few assumptions for the data that may restrict the application of logistic regression analysis. As logistic regression is based on the principle of maximum likelihood estimation (MLE), a considerably large sample is required. Peduzzi *et al.* suggested that a minimum of 10 observations per parameter should be included in the model.¹⁴¹

As with most statistical procedures, outliers can substantially affect all results. As discussed in the data consolidation process above, a thorough data check was conducted before performing data analysis. In addition, for the purpose of logistic regression, the residuals produced alongside the regression model were inspected to identify outlying cases that may have affected the model.

Moreover, as required for general multiple regression, logistic regression also requires no multicollinearity between variables.¹⁴² In simple terms, this means that high inter-correlations among independent variables will change the estimation of the coefficient and will be a problem. As Pallant suggested, ideally the predictor variables should be strongly related to the dependent variables but not strongly related to each other.¹⁴² In the current study, a collinearity diagnostic was conducted, and variables with the problem of multicollinearity was detected by examining the correlation matrix and tolerance value. For example, a variable with a tolerance value less than 0.1 was not included in the regression model.

A few technical terms that are presented in the SPSS output format should be noted, as they are not exactly the same as conventional regression values. First, as stated in the SPSS Regression Models 12.0 Manual,¹⁴³ the Hosmer and Lemeshow Test provides the most reliable test on a model performance (to support the model, the *p* value should be greater than 0.05). The significance value is in contrast to the significant results in the Omnibus Tests of

Model Coefficients. In the latter case, a p value less than 0.05 indicates the final model is better than the original guessing model in Block 0.¹⁴²

For easy interpretation, the original beta coefficients in the logistic regression are in a log format, and have been transformed into an odds ratio by taking the exponential (presented as Exp (B) in the SPSS output). According to Tabachnick and Fidell,¹⁴⁴ the odds ratio is "the increase (or decrease if the odds ratio is less than one) in odds of being in one outcome category when the value of the predictor increases by one unit" while keeping all the other predictors constant. For example, in the current study, this may refer to the odds of a participant with a particular characteristic using CAM a greater or lesser number times than someone who does not have that particular characteristic, all other factors being equal. All analyses in this thesis used SPSS to compute regression statistics, and will be presented in comparison to a reference category for all categorical variables.

A Classification Table is also provided in the SPSS output as an indication of how well the model predicted the correct category (CAM use or visits to CAM practitioner). Values in the table correspond to the conventional statistics of sensitivity (the true positives) and specificity (true negatives) of the model.¹⁴³ Thus, the summarised statistics of the classification table (the percentage accurate in classification (PAC) statistics) are presented in the current thesis.

Lastly, regression results presented in the current thesis were based on the original unweighted data. This is on the basis of the recommendation by Winship and Radbill that population weights do not make a difference in the results and therefore are not necessary for the regression model.¹⁴⁵ It is important to note that this method was also employed in a previous CAM utilisation study in South Australia.⁹ Thus, comparison of results between the current study and those previous studies can be much more straightforward.

108

CHAPTER 4. NATIONAL PROFILE OF CAM UTILISATION

4.1 Results

4.1.1 Foreword

The intended number of participants (1,067) completed the survey interview. Ninety percent of the interviews ranged between 7.5 and 22.5 minutes, with an average of 13.5 minutes. When considering the downtime in filling appointments, the total average time taken to complete the whole interview was just over 22 minutes. The minimum interview time was four minutes, and the maximum was 37 minutes.

The representativeness of the study population can be evaluated through a comparison with existing independent sources (e.g. the ABS data and recent government surveys), although some data collected from other sources are not readily available for comparison (e.g. some ABS data are only applicable to people aged 15 years and older). Nevertheless, after comparing with a variety of publications (details below), the study population can be considered as a representative sample of the entire Australian adult populations.

Detailed results of national profile of CAM use are presented in this chapter. A regional (state) profile of CAM utilisation is presented in Chapter 5, while Chapter 6 presents specific analyses on some popular forms of CAM in Australia, such as acupuncture, Chinese herbal medicine, chiropractic and osteopathy. Matters related to CAM use, such as rationale for using CAM, insurance coverage and regulation are presented in Chapter 7.

4.1.2 Survey Response Rate

For the main survey 26,367 telephone numbers were generated for dialling. Of these, 28% were not functioning, 10% were not assigned to households or were facsimile/modem numbers, 15% did not answer after a maximum of 18 call attempts. Among the remaining 7,260 participants (the eligibility of most of whom has not been established), 1,067 completed the interview. This corresponds to a 14.7% overall response rate. The response rate was calculated using the formula in the flowchart below (Figure 4.1), which is comparable to the method suggested by the American Association for Public Opinion Research.¹⁴⁶

Due to the relatively short period of the survey (two months), 320 participants were not available during the study period, and 76 interview appointments remained unsolved at the end of the survey. These figures were not taken into consideration when calculating the response rate. On the other hand, the estimated proportion of cases of unknown eligibility was not reported by the survey company. Thus, an adjusted response rate may be reasonably higher than the current estimation. The majority of people who refused to participate did not provide a reason. Of those who gave a detailed explanation, "don't have time" and "never do surveys" were the most common responses.



*Telephone numbers not functioning (7,241), fax/internet/data numbers (1,308), Business and other non-households numbers (1,258), Phone did not answer, engaged after five attempts or was an answering machining (4,403).

Figure 4.1 Flowchart of the final interview outcome

4.1.3 Characteristics and Representativeness of Survey Participants

Every effort was made to ensure that the survey participants were representative of the Australian national population. National aggregated quotas for gender, age ranges, and state/territory were set on the basis of 2004 ABS data,¹²⁴ and the proportions recruited for each of these did not differ significantly from the relevant ABS data for 2005^{122} (p>0.05) (Table 4.1). The survey sample was also representative of the Australian population in terms of self-reported health status,¹⁴⁷ place of birth (born in Australia or overseas), educational status, employment status and household income range¹⁴⁸ (Table 4.1).

The sample representativeness in this survey was extended when examining the details of participants' country of birth (Table 4.2 shows the top 10 countries of birth of survey participants). It suggested that participants' countries of birth were reasonably distributed across a wide range of different nations similar to the Australian national data,¹⁴⁹ although the population of the state of Tasmania, the Northern Territory and the Australian Capital Territory are relatively small, and the corresponding numbers of interviews were small.

In most cases, participants were asked about their detailed demographic information rather than the combined categories presented in Table 4.1. For example, educational background was categorised as "did not complete high school, completed high school, enrolled at TAFE or university, TAFE or university degree, postgraduate degree" (Table 4.3). These categorical data were merged for easy interpretation, and enhance the statistical power when performing sub-group analyses. In addition to the characteristics presented in Table 4.1, participants were also asked about their status of health insurance coverage. Overall, more than half (55.6%) of the participants were covered by a private health insurance.

	Survey	Australian	n value [†]
Characteristic*	participants (%**)	population (%)	(chi square)
Gender	rai rai (a)	F	
Male	49.0	49.7	0.65 (0.20)
Female	51.0	50.3	~ /
Age range (year)			
18-24	11.9	12.8	0.83 (2.14)
25-34	17.7	18.5	
35-44	20.3	19.4	
45-54	18.8	18.0	
55-64	14.6	14.1	
65+	16.7	17.2	
Region			
New South Wales	33.1	33.4	0.57 (5.74)
Victoria	24.7	24.9	
Queensland	19.3	19.3	
South Australia	8.2	7.7	
Western Australia	9.0	9.8	
Tasmania	2.4	2.4	
Northern Territory	0.8	0.9	
Australian Capital Territory	2.4	1.6	
Self-reported health			
Excellent	18.2	17.7	0.06 (9.16)
Very good	35.2	32.6	
Good	31.2	30.9	
Fair	11.5	13.8	
Poor	3.8	5.0	
Country of birth [‡]			
Australia	76.2	76.8	n/a
Overseas	23.8	23.2	
Post-secondary education [‡]			
No	43.8	49.1	n/a
Yes	56.2	50.9	
Employment [‡]			
Employed	65.7	61.4	n/a
Not in labour force	34.3	38.6	
Annual household income A\$ [‡]			
<20,000	17.7	20.0	n/a
20,001-40,000	21.1	24.1	
40,001-60,000	21.5	19.2	
60,001+	39.7	36.7	

 Table 4.1 Comparison of socio-demographic characteristics on survey participants and the Australian population

* Except for self-reported health, population data are from the ABS 2005 collection. Self-reported health data are from the AIHW Health Report 2004.

** Percentages are of those who provided characteristic information in each category.

[†] Probability of a difference in the frequency of each categorical variable between the survey and ABS data (chi square).

‡ n/a: Data not directly comparable; survey sample data are for 18 years and older, while corresponding population data are for 15 years and older.

	2005 CA	Australian Bureau of		
Country of Birth	Ν	%*	Statistics [†] (%)	
Australia	808	76.2%	76.8%	
UK & Ireland	91	8.6%	6.0%	
New Zealand	25	2.4%	2.1%	
Germany	11	1.0%	0.6%	
USA	9	0.8%	0.3%	
India	9	0.8%	0.6%	
Indonesia	7	0.7%	n/a	
South Africa	7	0.7%	0.5%	
Netherlands	6	0.6%	0.5%	
Poland	5	0.5%	0.3%	
Others	82	7.7%	12.4%	

Table 4.2 The top ten countries of birth of survey participants

* Exclude seven participants who refused to provide information on country of birth

† Source: Australian Year Book 2005 Page 122

Characteristics	Ν	%
Post-secondary education*		
<high school<="" td=""><td>204</td><td>19.2%</td></high>	204	19.2%
Completed high school	260	24.5%
Enrolled at TAFE or university	72	6.8%
TAFE or university degree	341	32.2%
Postgraduate	183	17.3%
Employment [†]		
Full time	435	41.3%
Part time or casual	181	17.2%
Self employed	75	7.1%
Home duties	68	6.5%
Student	36	3.4%
Retired or pensioner	221	21.0%
Unemployed	36	3.4%

Table 4.3 Detailed education and employment background of survey participants

* Exclude seven participants who refused to provide information on education

† Exclude 15 participants who refused to provide information on employment

4.1.4 Use of Common Forms of CAM

4.1.4.1 Prevalence of CAM Use

Unless otherwise indicated, the results shown in this chapter were obtained by weighting the sample data to reflect population norms. Table 4.4 shows the prevalence of use of the 17 forms of CAM included in the survey. Over two thirds (68.9%, 95% CI: 66.1% - 71.7%) of participants had used at least one of the 17 CAM therapies in the previous 12 months.

Of the total 17 forms of CAM, the use of clinical nutrition (including multivitamins and minerals) was found to be the most popular (45.8%, 95%CI: 42.8% – 48.8%), while Chinese medicine dietary therapy had the smallest following at 2.3% (95%CI: 1.4% - 3.2%). Low prevalence was also observed in the use of reflexology (4.1%, 95%CI: 2.9% - 5.3%) and of osteopathy (4.6%, 95%CI: 3.3% - 5.9%). In summary, the 10 most popular forms of CAM, in decreasing order, were clinical nutrition (including multivitamins and minerals), Western massage therapy, meditation, Western herbal medicine, aromatherapy, chiropractic, yoga, naturopathy, acupuncture and Chinese herbal medicine. None of the relative standard errors of the prevalence was higher than 25%.

	Prevalence					
	%	95% CI	RSE * (%)			
Clinical nutrition	45.8	(42.8 - 48.8)	3.3			
Western massage therapy	27.2	(24.5 – 29.9)	5.0			
Meditation	17.5	(15.2 – 19.8)	6.6			
Western herbal medicine	16.3	(14.1 – 18.5)	6.9			
Aromatherapy	16.1	(13.9 – 18.3)	7.0			
Chiropractic	16.1	(13.9 – 18.3)	7.0			
Yoga	12.0	(10.1 – 13.9)	8.3			
Naturopathy	10.7	(8.8 – 12.6)	8.8			
Acupuncture	9.2	(7.5 – 10.9)	9.6			
Chinese herbal medicine	7.0	(5.5 - 8.5)	11.2			
Energy healing	7.0	(5.5 - 8.5)	11.2			
Homeopathy	6.0	(4.6 - 7.4)	12.1			
Qigong, martial art and Tai Chi	6.0	(4.6 - 7.4)	12.1			
Chinese therapeutic massage	5.1	(3.8 - 6.4)	13.2			
Osteopathy	4.6	(3.3 - 5.9)	13.9			
Reflexology	4.1	(2.9 - 5.3)	14.8			
Chinese medicine dietary therapy	2.3	(1.4 – 3.2)	20.0			
At least 1 of the above forms of CAM	68.9	(66.1 – 71.7)	2.1			

Table 4.4 Prevalence of the 17 investigated forms of CAM

* RSE: relative standard error = (standard error/estimated prevalence)*100

4.1.4.2 Characteristics of CAM Users

A total of 737 participants were classified as CAM users, 45.5% of whom were male, and 54.5% female. Key socio-demographic characteristics of the CAM users are summarised in Table 4.5. A higher proportion of females were CAM users than males, and fewer of those aged 65 and above were users than were younger adults. However, it is worth noting that 60% of those aged 65 and above used CAM (Table 4.5). It appears that CAM users are more likely to have higher levels of education, to have private health insurance, and to be in households with higher than average incomes (Table 4.5). There was little difference in the use of CAM between those born in Australia and those born overseas. In contrast, there was considerable variation in the prevalence of the use of CAM between Australian states (details Chapter 5).

Table 4.6 is the detailed cross-tabulation analysis that reveals the proportions of survey participants who had used each of the 17 forms of CAM, and the participants' specified sociodemographic characteristics. The largest gender differentials were in the use of aromatherapy, Western herbal medicine, Western massage, naturopathy, energy healing and yoga (all modalities with female users being higher than male users, p<0.001). In particular, the use of aromatherapy by females (24.0%) was almost three times that by males (8.1%) (p<0.0001). Gender differences were also observed in osteopathy and reflexology users, but less significant (p<0.05).

As mentioned earlier, the youngest group (18 to 34 years of age) was more likely to use CAM therapies than the older cohorts. However, this may be largely due to the relatively high prevalence of use of yoga, Qigong, martial art, Tai Chi and clinical nutrition by the youngest group (18-34 age group, compared with the 35-64 and 65 and older age groups, p<0.05). Not surprisingly, older people also had a low prevalence of use of Chinese herbal medicine,

	Were CAM users					
Characteristics	% (95%CI)	p value [†]				
Gender						
Male	63.4 (59.3 - 67.5)					
Female	74.4 (70.7 - 78.1)	< 0.001				
Age (year)						
18-34	75.3 (70.7 – 79.9)					
35-64	69.0 (65.1 - 72.9)	< 0.05				
65+	57.8 (50.7 - 64.9)	< 0.001				
Country of birth						
Australia	70.2 (67.1 – 73.3)					
Other	65.0 (59.1 - 70.9)	>0.05				
Region						
New South Wales	72.1 (67.5 – 76.7)					
Victoria	69.8 (64.3 - 75.3)	>0.05				
Queensland	71.0 (64.8 - 77.2)	>0.05				
South Australia	60.8 (50.3 - 71.3)	>0.05				
Western Australia	66.6 (57.6 - 75.6)	>0.05				
Self-reported health status						
Excellent/very good/good	68.4 (65.4 - 71.4)					
Fair/poor	71.7 (64.7 – 78.7)	>0.05				
Post-secondary education						
No	61.6 (57.1 – 66.1)					
Yes	74.7 (71.2 - 78.2)	< 0.001				
Employment						
Employed	72.5 (69.2 - 75.8)					
Unemployed or not in labour force	62.9 (57.9 - 67.9)	< 0.005				
Private health insurance						
Yes	72.2 (68.6 - 75.8)					
No	64.9 (60.6 - 69.2)	< 0.05				
Annual household income (A\$)						
<20,000	60.1 (52.5 - 67.7)					
\$20,000 - 40,000	65.5 (59.0 - 72.0)	>0.05				
>40,000	72.8 (69.1 – 76.5)	< 0.005				

Table 4.5 Socio-demographic characteristics on CAM users

Note: significance levels were adjusted to allow for multiple comparisons

* Excludes subjects who did not provide socio-demographic characteristics

 Probability of proportion being significantly different from the first value in each demographic category (Z-test) aromatherapy, Western herbal medicine, (all modalities, 65 and above age group, compared with 18-34 age group, p<0.01), and Chinese therapeutic massage (p<0.05). In contrast, the highest proportions of users of osteopathy and reflexology were those over 65 years of age.

When examining participants' country of birth, it is clear that a higher prevalence in the use of Chinese herbal medicine (p<0.01), acupuncture (p<0.05), Chinese medicine dietary therapy (p<0.05) and reflexology (p<0.05) was observed in those born overseas than those born in Australia. However, the use of chiropractic by those people born in Australia (18.1%) was almost twice that of those born overseas (9.7 %) (p<0.001).

Similar to the finding of overall CAM use, self-reported health status did not seem to significantly affect the use of any specific form of CAM. However, a clearly higher use of some forms of CAM (yoga, Western herbal medicine, aromatherapy and naturopathy) was observed in those with an excellent/very good or good health status. In addition, those who rated their health status as fair or poor tended to have a higher (but not statistically significant) use of acupuncture, Qigong, martial art, Tai Chi, chiropractic, osteopathy, meditation, clinical nutrition and energy healing.

With respect to educational background, for each of the 17 forms of CAM higher use was observed in those with post-secondary education, although a statistically significant difference was only observed in the use of acupuncture, Chinese herbal medicine, Western herbal medicine, Western therapeutic massage, homeopathy, naturopathy, meditation, aromatherapy, clinical nutrition and yoga. Very similar findings were found in participants' employment status, that is, people who were employed at the time of the survey also had a higher rate of CAM use, except for osteopathy and energy healing. For the latter two categories of users, those unemployed or not in the labour force, had a slightly higher rate of CAM use (p>0.05).

Private health insurance funds in Australia provide limited coverage of a number of CAM therapies, such as acupuncture, chiropractic and massage therapy. The common view of the correlation between CAM use and insurance cover is that those covered by private health insurance would be more likely to choose some specific forms of CAM, as they can claim for limited reimbursements for those types of CAM therapies. This hypothesis was confirmed by most users but not users of aromatherapy, Chinese medicine dietary therapy, as well as Qigong, Tai Chi and martial art. Among all CAM therapies, there was a statistically significant difference between those covered by private health insurance and those who are not on the use of chiropractic (p<0.01) and Western massage therapy (p<0.01).

As for most Western countries, the cost related to seeing a CAM practitioner and/or the use of CAM products are mainly borne by the consumers themselves. Thus, the use of CAM may be partially determined by the affordability of that CAM therapy/product. This hypothesis is generally acceptable for most forms of CAM used by the current survey participants. The most significant income differential was for chiropractic and aromatherapy users. Only approximately one tenth of the survey participants with an overall household income of less than \$20,000 had used chiropractic (7.5%) or aromatherapy (10.1%), whereas the prevalence of use of these therapies doubled to approximately 20% among those with an income between \$20,000 and \$40,000, and among those with incomes exceeding \$40,000.

The tests of significance mentioned above between categories were conducted without controlling the potential cross-effects (referred to as interaction in statistical analyses) among predictor variables. For example, those employed may also have higher household incomes and be covered by private health insurance. Due to this, a multivariate logistic regression analysis that considers the correlation between variables was also carried out.

				Percents	Percents (standard error) of use of -					
Characteristics	Acupuncture	СНМ	СТМ	CMDT	QGMATC	WHM	WTM	Chiropractic	Osteopathy	
Gender										
Female	9.6 (1.27)	7.7 (1.15)	5.8 (1.01)	3.0 (0.73)	7.2 (1.12)	20.2 (1.73)	33.3 (2.03)	17.1 (1.62)	5.8 (1.01)	
Male	8.9 (1.24)	6.1 (1.04)	4.3 (0.89)	1.7 (0.57)	4.6 (0.91)	12.3 (1.43)**	21.0 (1.77)**	15.0 (1.55)	3.3 (0.78)*	
Age (year)										
18-34	7.6 (1.46)	8.1 (1.49)	5.7 (1.27)	2.3 (0.82)	9.5 (1.61)	16.3 (2.03)	33.8 (2.59)	15.0 (1.96)	3.1 (0.96)	
35-64	10.3 (1.30)	7.8 (1.15)	5.6 (0.98)	2.8 (0.71)	4.7 (0.90)**	17.7 (1.63)	27.7 (1.91)	16.7 (1.60)	5.0 (0.93)	
65+	9.2 (2.13)	2.6 (1.16)**	2.4 (1.12)*	1.1 (0.78)	2.9 (1.23)**	12.3 (2.42)	13.9 (2.55)**	15.7 (2.68)	5.9 (1.74)	
Country of birth										
Australia	10.2 (1.06)	5.5 (0.80)	4.7 (0.74)	1.7 (0.45)	6.2 (0.84)	15.9 (1.28)	28.2 (1.58)	18.1 (1.35)	4.8 (0.75)	
Overseas	6.5 (1.56)*	11.1(2.00)**	6.4 (1.55)	4.7 (1.34)*	5.2 (1.40)	17.7 (2.42)	25.0 (2.75)	9.8 (1.89)**	3.8 (1.21)	
Self-reported health status										
Excellent/very good/good	8.6 (0.93)	6.9 (0.85)	5.1 (0.73)	2.8 (0.55)	5.8 (0.78)	17.1 (1.25)	27.3 (1.48)	15.2 (1.19)	4.5 (0.69)	
Fair/poor	12.1 (2.57)	6.4 (1.93)	4.9 (1.70)	0.0 (0.00)**	6.4 (1.93)	11.8 (2.54)	26.3 (3.47)	20.6 (3.19)	4.8 (1.68)	
Post-secondary education										
No	6.4 (1.15)	4.7 (0.99)	4.7 (0.99)	1.9 (0.64)	4.9 (1.01)	13.0 (1.58)	20.8 (1.90)	15.4 (1.70)	3.6 (0.87)	
Yes	11.3 (1.29)**	8.7 (1.15)**	5.4 (0.92)	2.7 (0.66)	6.8 (1.03)	18.8 (1.59)**	32.2 (1.90)**	16.7 (1.52)	5.3 (0.91)	
Employment										
Employed	9.8 (1.13)	8.1 (1.04)	6.0 (0.90)	2.9 (0.64)	6.4 (0.93)	17.4 (1.45)	29.7 (1.74)	16.8 (1.42)	4.5 (0.79)	
Unemployed/not in labour force	7.8 (1.40)	4.6 (1.10)*	3.5 (0.97)	1.3 (0.60)	5.1 (1.16)	14.4 (1.84)	22.5 (2.19)**	14.5 (1.85)	4.7 (1.11)	
Private health insurance										
Yes	10.8 (1.27)	7.3 (1.07)	5.7 (0.95)	2.3 (0.62)	5.8 (0.96)	17.2 (1.55)	32.5 (1.92)	19.0 (1.61)	5.6 (0.94)	
No	7.4 (1.21)	6.6 (1.15)	4.3 (0.94)	2.4 (0.71)	6.3 (1.12)	14.8 (1.65)	20.9 (1.88)**	12.4 (1.53)**	3.3 (0.83)	
Annual household income										
<a\$20,000< td=""><td>8.1 (2.16)</td><td>5.9 (1.88)</td><td>3.7 (1.49)</td><td>2.8 (1.32)</td><td>6.2 (1.91)</td><td>12.1 (2.59)</td><td>18.3 (3.07)</td><td>7.3 (2.07)</td><td>6.7 (1.99)</td></a\$20,000<>	8.1 (2.16)	5.9 (1.88)	3.7 (1.49)	2.8 (1.32)	6.2 (1.91)	12.1 (2.59)	18.3 (3.07)	7.3 (2.07)	6.7 (1.99)	
A\$20,000 – A\$40,000	7.0 (1.79)	7.6 (1.85)	4.0 (1.37)	2.0 (0.98)	7.0 (1.78)	19.3 (2.76)	23.6 (2.97)	19.0 (2.74)**	1.3 (0.79)*	
>A\$40,000	10.1 (1.27)	6.9 (1.07)	5.5 (0.96)	2.2 (0.62)	5.3 (0.94)	17.2 (1.59)	32.2 (1.97)**	18.4 (1.63)**	5.5 (0.96)	

	Table 4.6 S	Socio-demogra	hic characteristics	on users of each	of the 17	/ forms of CAM
--	-------------	---------------	---------------------	------------------	-----------	----------------

Note 1: probability of proportion being significantly different from the first value in each demographic category * p<0.05, ** p<0.01; significance levels were adjusted to allow for multiple comparisons. Note 2: CHM: Chinese herbal medicine; CTM: Chinese therapeutic massage; CMDT: Chinese medicine dietary therapy; QGMATC: Qigong, Martial art and Tai Chi; WHM: Western herbal medicine, WTM: Western therapeutic massage.

	Percents % (standard error) of use of –								
Characteristic	Homeopathy	Naturopathy	Meditation	Aromatherapy	Clinical nutrition	Energy healing	Reflexology	Yoga	
Gender									
Female	6.6 (1.07)	14.2 (1.50)	19.3 (1.70)	24.0 (1.84)	51.5 (2.15)	9.9 (1.29)	5.5 (0.98)	16.6 (1.61)	
Male	5.5 (0.99)	7.1 (1.11)**	15.6 (1.58)	8.1 (1.19)*	40.1 (2.13)**	4.0 (0.85)**	2.7 (0.71)*	7.3 (1.13)**	
Age (year)									
18-34	5.9 (1.30)	11.1 (1.72)	18.1 (2.11)	19.9 (2.19)	53.6 (2.73)	7.2 (1.42)	3.2 (0.97)	20.4 (2.21)	
35-64	6.6 (1.06)	11.5 (1.36)	18.3 (1.65)	17.8 (1.63)	44.9 (2.13)*	7.3 (1.11)	3.9 (0.83)	8.4 (1.18)**	
65+	4.6 (1.55)	7.5 (1.94)	13.6 (2.53)	4.2 (1.48)*	35.4 (3.52)**	5.0 (1.61)	5.7 (1.71)	7.9 (1.99)**	
Country of birth									
Australia	5.0 (0.76)	10.6 (1.08)	17.0 (1.32)	17.2 (1.32)	45.4 (1.75)	7.5 (0.92)	3.3 (0.63)	11.7 (1.13)	
Overseas	8.8 (1.80)	11.0 (1.99)	18.4 (2.46)	13.2 (2.15)	47.3 (3.17)	5.5 (1.45)	7.0 (1.62)*	13.5 (2.16)	
Self-reported health status									
Excellent/very good/good	6.0 (0.79)	10.9 (1.04)	16.7 (1.24)	16.6 (1.24)	45.6 (1.66)	6.1 (0.80)	4.1 (0.66)	12.9 (1.11)	
Fair/poor	5.9 (1.85)	9.4 (2.30)	21.1 (3.21)	13.5 (2.69)	46.7 (3.93)	11.2 (2.48)	3.8 (1.51)	6.8 (1.98)**	
Post-secondary education									
No	4.0 (0.92)	8.2 (1.29)	11.9 (1.52)	12.7 (1.56)	38.1 (2.28)	5.7 (1.09)	3.6 (0.87)	7.4 (1.23)	
Yes	7.4 (1.07)*	12.5 (1.34)*	21.7 (1.67)**	18.9 (1.59)**	52.0 (2.03)**	8.0 (1.10)	4.6 (0.85)	15.6 (1.48)**	
Employment									
Employed	6.9 (0.96)	12.4 (1.25)	17.8 (1.46)	18.4 (1.48)	50.2 (1.91)	6.8 (0.96)	4.1 (0.75)	13.1 (1.28)	
Unemployed or not in labour force	4.7 (1.11)	7.5 (1.38)**	15.7 (1.91)	12.1 (1.71)**	39.1 (2.56)**	6.8 (1.32)	4.1 (1.04)	10.6 (1.61)	
Private health insurance									
Yes	6.5 (1.01)	11.3 (1.30)	18.0 (1.58)	15.7 (1.49)	48.5 (2.05)	7.0 (1.04)	4.5 (0.85)	12.6 (1.36)	
No	5.4 (1.04)	9.8 (1.38)	16.8 (1.73)	16.9 (1.73)	42.8 (2.29)	7.0 (1.18)	3.7 (0.88)	11.4 (1.47)	
Annual household income									
<a\$20,000< td=""><td>4.8 (1.69)</td><td>6.9 (2.01)</td><td>17.5 (3.01)</td><td>10.1 (2.40)</td><td>36.3 (3.81)</td><td>6.5 (1.96)</td><td>2.5 (1.25)</td><td>6.9 (2.01)</td></a\$20,000<>	4.8 (1.69)	6.9 (2.01)	17.5 (3.01)	10.1 (2.40)	36.3 (3.81)	6.5 (1.96)	2.5 (1.25)	6.9 (2.01)	
A\$20,000 - A\$40,000	6.3 (1.70)	11.7 (2.25)	16.9 (2.62)	20.4 (2.82)**	42.9 (3.46)	7.8 (1.87)	7.0 (1.79)*	9.9 (2.09)	
>A\$40,000	5.9 (0.99)	11.0 (1.32)	18.6 (1.64)	17.2 (1.59)*	49.7 (2.11)**	7.2 (1.09)	3.7 (0.79)	14.2 (1.47)**	

Table 4.6 Socio-demographic characteristics on users of each of the 17 forms of CAM (continued)

Note 1: probability of proportion being significantly different from the first value in each demographic category p<0.05, p<0.01 Note 2: significance levels were adjusted to allow for multiple comparisons.

4.1.5 Visits to CAM Practitioners

4.1.5.1 Prevalence of Visiting CAM Practitioners

Almost two thirds (64.0%, 95%CI: 60.5% - 67.5%) of CAM users had visited a CAM practitioner in the previous 12 months (Table 4.7). Not surprisingly, there were considerable differences between the 17 forms of CAM in the proportions of practitioner visits. For example, in the 12 months preceding the survey, only about one sixth (16.4%) of clinical nutrition users had consulted a practitioner, whereas about one third (32.9%) of Chinese herbal medicine users and 29.1% of Western herbal medicine users had visited a practitioner. As might have been expected, the highest proportions of practitioner visits were by users of provider-based CAM therapies, such as chiropractic (90.6%), acupuncture (81.1%), osteopathy (76.4%) and Western massage therapy (73.7%) (Table 4.7). It is a safety concern and perhaps somewhat puzzling that nearly one in 10 chiropractic users and approximately one in five users of acupuncture, osteopathy or Western massage therapy did not visit a relevant practitioner.

The actual proportion of participants who had visited a CAM practitioner in the preceding 12 months was 44.1% (95%CI: 41.1% – 47.1%) of the total survey participants (Table 4.8). Of the total practitioner visits, Western massage therapist visits was found to be the most popular (20.1%, 95%CI: 17.7% – 22.5%); high prevalence was also observed in visits to chiropractors (14.6, 95%CI: 12.4% – 16.7%) and acupuncturists (7.5%, 95%CI: 5.9% – 9.15).

Type of CAM*	Percentage**		
Type of CAM	%	95% CI	$RSE^{\dagger}(\%)$
Clinical nutrition	16.4	(13.1 – 19.7)	10.2
Western massage therapy	73.7	(68.6 – 78.8)	3.5
Meditation	23.7	(17.6 – 29.8)	13.2
Western herbal medicine	29.1	(22.4 - 35.8)	11.8
Aromatherapy	19.3	(13.4 – 25.2)	15.6
Chiropractic	90.6	(86.2 - 95.0)	2.5
Yoga	56.9	(48.3 - 65.5)	7.7
Naturopathy	55.8	(46.7 – 64.9)	8.3
Acupuncture	81.1	(73.4 - 88.8)	4.9
Chinese herbal medicine	32.9	(22.2 - 43.6)	16.6
Energy healing	46.7	(35.3 – 58.1)	12.4
Homeopathy	47.7	(35.5 - 59.9)	13.1
Qigong, martial art and Tai Chi	53.3	(41.1 – 65.5)	11.7
Chinese therapeutic massage	62.2	(49.3 – 75.1)	10.6
Osteopathy	76.4	(64.5 - 88.3)	7.9
Reflexology	51.5	(36.7 – 66.3)	14.6
Chinese medicine dietary therapy	24.8	(7.9 – 41.7)	34.8 ^{††}
At least 1 of the above forms of CAM	64.0	(60.5 - 67.5)	2.8

Table 4.7 Percentage of CAM users that visited practitioners for 17 forms of CAM

* Types of CAM are listed in descending order of the type of CAM prevalence

** Percentages are of users that visited a practitioner of a specific type of CAM

† RSE: relative standard error = (standard error/estimated prevalence)*100

^{$\dagger\dagger$} Relative standard error >25%, results to be interpreted with caution

Type of CAM*	Percentage**		
Type of CAM	%	95% CI	$RSE^{\dagger}(\%)$
Western massage therapy	20.1	(17.7 – 22.5)	6.1
Chiropractic	14.6	(12.4 – 16.7)	7.4
Acupuncture	7.5	(5.9 - 9.1)	10.8
Clinical nutrition	7.5	(5.9 - 9.1)	10.8
Yoga	6.8	(5.3 – 8.4)	11.3
Naturopathy	5.9	(4.5 - 7.4)	12.2
Western herbal medicine	4.7	(3.5 - 6.0)	13.7
Meditation	4.1	(2.9 - 5.3)	14.7
Osteopathy	3.5	(2.4 - 4.6)	16.1
Energy healing	3.2	(2.2 - 4.3)	16.7
Qigong, martial art and Tai Chi	3.2	(2.1 – 4.2)	16.9
Chinese therapeutic massage	3.2	(2.1 – 4.2)	17.0
Aromatherapy	3.1	(2.1 – 4.2)	17.1
Homeopathy	2.9	(1.9 – 3.9)	17.8
Chinese herbal medicine	2.3	(1.4 – 3.2)	20.0
Reflexology	2.1	(1.3 – 3.0)	20.8
Chinese medicine dietary therapy	0.6	(0.1 – 1.0)	$40.1^{\dagger\dagger}$
At least 1 of the above forms of CAM	44.1	(41.1 – 47.1)	3.5

Table 4.8 Percentage of participants that visited practitioners for 17 forms of CAM

* Types of CAM are listed in descending order of the prevalence of CAM practitioner visits

** Percentages are of survey participants that visited a practitioner of a specific type of CAM

† RSE: relative standard error = (standard error /estimated prevalence)*100

^{††} Relative standard error >25%, results to be interpreted with caution

4.1.5.2 Frequency of CAM Practitioners Visits

The number of practitioner visits by survey participants during the 12 months preceding the survey was determined for the 10 most common provider-based CAM therapies (Table 4.9). Thirty seven percent (37.0%, 95%CI: 34.2% - 40.0%) of the participants had visited a practitioner of at least one of the 10 forms of CAM listed in Table 4.9, on at least one occasion in the 12-month period. The highest mean number of visits was for users of acupuncture (8.8), closely followed by users of chiropractic (8.4) (see Appendix G2 for the original unweighted data and Appendix G3, statistical summary of weighted data).

The average number of visits to one or more of the types of CAM practitioners in Table 4.9, over the 12-month period was 12.1, which when extrapolated to the national adult population, equates to 69.2 million visits. Almost one third of these (32.1%) were to a massage therapist (Western and Chinese therapeutic massage) and over one quarter (27.5%) were to chiropractors, that is, adult Australians made 22.2 million visits to massage therapists and 19.0 million visits to chiropractors in the 12-month period.

	Survey participants	Estimate for national population [*]	
Type of CAM	Mean (95%CI)	Million (95%CI)	
Acupuncture	8.75 (5.2 – 12.3)	10.16 (6.0 – 14.3)	
Chiropractic	8.44 (6.5 – 10.4)	19.05 (14.6 – 23.5)	
Homeopathy	6.65 (1.9 – 11.4)	2.97 (0.8 - 5.1)	
Western massage	6.29 (5.1 – 7.5)	19.55 (15.9 – 23.2)	
Chinese herbal medicine	5.98 (3.5 - 8.5)	2.12 (1.2 - 3.0)	
Osteopathy	5.69 (3.8 - 7.6)	3.08 (2.0 – 4.1)	
Chinese therapeutic massage	5.43 (3.2 - 7.7)	2.66 (1.6 – 3.8)	
Naturopathy	5.32 (2.9 - 7.7)	4.87 (2.7 – 7.0)	
Aromatherapy [†]	3.98 (1.8 - 6.2)	1.93 (0.9 – 3.0)	
Western herbal medicine	3.75 (2.7 – 4.8)	2.76 (2.0 - 3.5)	
Visits to one or more of the above CAM practitioners	12.1 (10.0 –14.2)	69.15 (57.6 - 81.4)	

Table 4.9 Frequency of visits to practitioners for 10 forms of CAM

* Based on an adult Australian population of 15.5 million and corresponding prevalence of CAM practitioner visits as estimated in the current study (see Table 4.8).

[†] Adjusted mean after excluding two obvious outliers

4.1.5.3 Characteristics of People Visiting CAM Practitioners

Table 4.10 shows the socio-demographic profiles of survey participants who had visited a CAM practitioner during the preceding 12 months. In general, the findings for practitioner visits were in accord with those for CAM users. That is, higher proportions of those who visited CAM practitioners were female, younger adults, people with higher levels of education, those with private health insurance, and those who had higher household incomes. However, it is of interest that there was less variation between Australian states in the proportions of participants who had visited a CAM practitioner than in the proportions of CAM users. Detailed analyses of the possible regional differences are presented in Chapter 5.

Despite the similarities in the socio-demographic profiles of CAM users and those who had visited CAM practitioners, there were some differences. Thus, there was no statistically significant difference between the proportions of employed and unemployed participants that visited CAM practitioners or in the proportions in the two younger age ranges (Table 4.11). On the other hand, unlike the findings for CAM users, significantly more participants who visited CAM practitioners were born in Australia than born overseas and a higher proportion had household incomes of \$20,000-\$40,000 than less than \$20,000 (Table 4.11).

Table 4.11 presents a detailed cross-tabulation analyses that reveals the proportions of survey participants with the specified socio-demographic characteristics who had visited practitioners for each of the 17 forms of CAM. The overall pattern of the characteristics' differentials was similar to the findings discussed above relating to the users of each of the 17 forms of CAM, although a statistically significant difference was more commonly detected in visiting provider-based CAM practitioner, such as acupuncture, massage therapy and osteopathy.

128

	Visited CAM practitioners		
Socio-demographic characteristic	% (95%CI)	p value [†]	
Gender			
Male	38.9 (34.7 - 43.1)		
Female	49.2 (45.0 - 53.4)	< 0.001	
Age range (year)			
18-34	48.1 (42.7 – 53.5)		
35-64	45.0 (40.8 - 49.2)	>0.05	
65+	34.9 (28.0 - 41.8)	< 0.01	
Country of birth			
Australia	46.5 (43.1 - 49.9)		
Other	36.7 (30.7 - 42.7)	< 0.01	
Region			
New South Wales	45.0 (39.9 - 50.1)		
Victoria	48.0 (42.0 - 54.0)	>0.05	
Queensland	41.4 (34.7 – 48.1)	>0.05	
South Australia	44.5 (33.8 - 55.2)	>0.05	
Western Australia	42.0 (32.6 - 51.4)	>0.05	
Self-reported health status			
Excellent/very good/good	43.6 (40.4 - 46.8)		
Fair/poor	46.7 (39.0 – 54.4)	>0.05	
Post-secondary education			
No	35.7 (31.3 – 40.1)		
Yes	50.8 (46.8 - 54.8)	< 0.001	
Employment			
Employed	46.3 (42.6 - 50.0)		
Unemployed or not in labour force	40.1 (35.1 – 45.1)	>0.05	
Private health insurance			
Yes	49.3 (45.3 – 53.3)		
No	37.9 (33.5 - 42.3)	< 0.001	
Annual household income (A\$)			
<20,000	30.3 (23.2 - 37.4)		
\$20,000 - 40,000	43.2 (36.4 - 50.0)	< 0.05	
>40,000	49.0 (44.9 - 53.1)	< 0.001	

Table 4.10 Characteristics of people who had visited CAM practitioners

Note: significance levels were adjusted to allow for multiple comparisons

* Excludes subjects who did not provide socio-demographic characteristics

Probability of proportion being significantly different from the first value in each demographic category (Z-test)
	Percentages (standard error) of people of visiting a practitioner for –								
Characteristic	ACU	СНМ	СТМ	CMDT	QGMATC	WHM	WTM	Chiropractic	Osteopathy
Gender									
Female	6.9 (1.09)	2.1 (0.62)	3.6 (0.81)	0.6 (0.34)	3.3 (0.77)	5.9 (1.02)	23.0 (1.81)	15.3 (1.55)	4.0 (0.84)
Male	8.1 (1.18)	2.5 (0.67)	2.6 (0.70)	0.5 (0.32)	3.1 (0.75)	3.5 (0.80)	17.1 (1.64)*	13.8 (1.50)	3.0 (0.74)
Age (year)									· · · ·
18-34	6.2 (1.32)	2.6 (0.87)	3.9 (1.06)	0.2 (0.26)	5.9 (1.29)	5.7 (1.27)	26.0 (2.40)	13.7 (1.88)	2.7 (0.90)
35-64	8.6 (1.20)	2.7 (0.69)	3.1 (0.74)	0.9 (0.39)	2.1 (0.62)**	5.0 (0.93)	20.2 (1.72)	15.1 (1.53)	4.1 (0.84)
65+	6.7 (1.84)	0.7 (0.61)	2.0 (1.03)	0.4 (0.47)	1.5 (0.90)**	2.2 (1.09)*	9.5 (2.16)**	14.6 (2.60)	3.1 (1.28)
Country of birth									· · · ·
Australia	8.1 (0.96)	2.2 (0.52)	3.2 (0.62)	0.5 (0.24)	3.7 (0.66)	4.2 (0.70)	21.1 (1.43)	16.2 (1.30)	3.7 (0.67)
Other	5.8 (1.49)	2.6 (1.01)	3.2 (1.11)	0.9 (0.60)	1.7 (0.82)	6.8 (1.59)	17.2 (2.40)	9.5 (1.86)**	2.7 (1.04)
Self-reported health status									
Excellent/very good/good	7.1 (0.85)	2.3 (0.50)	2.8 (0.55)	0.7 (0.27)	3.2 (0.58)	4.9 (0.72)	19.9 (1.33)	14.1 (1.16)	3.6 (0.62)
Fair/poor	9.2 (2.28)	2.2 (1.14)	4.9 (1.70)	0.0 (0.00)**	2.5 (1.23)	3.9 (1.53)	20.6 (3.19)	16.5 (2.92)	3.0 (1.34)
Post-secondary education		· · ·							i
No	4.7 (0.99)	1.5 (0.57)	2.8 (0.78)	0.4 (0.30)	1.9 (0.63)	3.3 (0.84)	12.8 (1.57)	13.4 (1.60)	2.2 (0.69)
Yes	9.5 (1.19)**	2.9 (0.68)	3.4 (0.74)	0.7 (0.34)	4.2 (0.81)*	5.8 (0.95)*	25.7(1.78)**	15.6 (1.48)	4.5 (0.84)*
Employment	· · ·								· · · ·
Employed	8.4 (1.06)	2.7 (0.62)	3.6 (0.71)	0.7 (0.31)	3.6 (0.71)	5.4 (0.86)	22.0 (1.58)	15.3 (1.37)	4.0 (0.74)
Unemployed or not in labour force	5.3 (1.17)	1.5 (0.64)	2.5 (0.81)	0.4 (0.35)	2.5 (0.82)	3.8 (1.00)	16.6 (1.95)*	13.4 (1.78)	2.7 (0.85)
Private health insurance									
Yes	9.3 (1.19)	2.9 (0.69)	3.6 (0.76)	0.6 (0.32)	3.2 (0.72)	4.4 (0.84)	24.9 (1.77)	17.7 (1.57)	4.9 (0.89)
No	5.3 (1.03)**	1.5 (0.57)	2.7 (0.75)	0.6 (0.35)	3.2 (0.81)	5.2 (1.03)	14.2 (1.62)**	10.6 (1.43)**	1.7 (0.60)**
Annual household income (A\$)									
<20,000	4.4 (1.64)	1.0 (0.80)	1.5 (0.95)	1.5 (0.97)	3.0 (1.36)	4.3 (1.61)	9.3 (2.30)	6.3 (1.93)	4.3 (1.61)
\$20,000 - 40,000	6.3 (1.69)	2.9 (1.18)	3.2 (1.23)	0.4 (0.45)	3.3 (1.25)	7.2 (1.80)	15.1 (2.50)	17.2 (2.64)**	0.9 (0.65)*
>40,000	87(119)*	2.1 (0.60)	37(079)	0.4(0.26)	30(0.72)	46(0.88)	26.2 (1.85)**	17 3 (1 59)**	47(0.89)

Table 4.11 Characteristics of people who had visited practitioners for each of the 17 forms of CAM

Note 1: probability of proportion being significantly different from the first value in each demographic category (Z test) * p<0.05, ** p<0.01; Note 2: significance levels were adjusted to allow for multiple comparisons; Note3: CHM: Chinese herbal medicine; CTM: Chinese therapeutic massage; CMDT: Chinese medicine dietary therapy; QGMATC: Qigong, Martial art and Tai Chi; WHM: Western herbal medicine, WTM: Western therapeutic massage.

	Percentages (standard error) of people of visiting a practitioner for –									
Characteristic	Homeopathy	Naturopathy	Meditation	Aromatherapy	Clinical nutrition	Energy healing	Reflexology	Yoga		
Gender										
Female	3.9 (0.83)	9.0 (1.23)	3.8 (0.82)	4.5 (0.89)	10.1 (1.30)	4.6 (0.91)	2.6 (0.69)	9.2 (1.25)		
Male	1.9 (0.59)*	2.9 (0.73)**	4.5 (0.90)	1.7 (0.56)**	4.9 (0.94)**	1.8 (0.58)**	1.6 (0.55)	4.4 (0.90)**		
Age (year)										
18-34	2.1 (0.78)	6.2 (1.32)	6.3 (1.34)	4.3 (1.11)	9.7 (1.62)	4.5 (1.13)	3.2 (0.97)	12.5 (1.81)		
35-64	3.5 (0.78)	6.5 (1.05)	3.5 (0.79)	3.5 (0.78)	6.5 (1.06)	3.3 (0.76)	1.8 (0.57)	4.4 (0.88)**		
65+	2.6 (1.17)	4.0 (1.44)	2.0 (1.04)*	0.0 (0.00)**	6.5 (1.81)	1.1 (0.77)*	1.1 (0.77)	4.1 (1.46)**		
Country of birth										
Australia	2.4 (0.53)	5.6 (0.81)	4.5 (0.73)	3.0 (0.60)	7.4 (0.92)	3.0 (0.60)	1.5 (0.43)	6.4 (0.86)		
Overseas	3.8 (1.21)	7.3 (1.65)	3.1 (1.10)	3.5 (1.17)	7.1 (1.63)	4.1 (1.26)	4.2 (1.27)*	8.4 (1.76)		
Self-reported health status										
Excellent/very good/good	2.7 (0.54)	6.0 (0.79)	4.2 (0.67)	3.4 (0.60)	6.8 (0.84)	2.9 (0.56)	2.2 (0.49)	7.8 (0.89)		
Fair/poor	4.0 (1.55)	5.7 (1.82)	4.0 (1.54)	1.7 (1.01)	11.5 (2.51)	4.4 (1.61)	0.8 (0.72)	1.1 (0.83)**		
Post-secondary education										
No	1.9 (0.64)	4.8 (1.01)	2.6 (0.75)	2.4 (0.71)	6.0 (1.12)	2.1 (0.68)	1.2 (0.52)	4.0 (0.92)		
Yes	3.7 (0.76)	6.8 (1.03)	5.4 (0.92)*	3.7 (0.77)	8.7 (1.14)	4.1 (0.81)	2.8 (0.67)	9.0 (1.17)**		
Employment										
Employed	3.1 (0.66)	7.2 (0.99)	4.5 (0.79)	3.4 (0.69)	8.8 (1.08)	3.2 (0.67)	2.3 (0.57)	7.4 (1.00)		
Unemployed or not in labour force	2.6 (0.83)	3.8 (1.00)*	2.5 (0.82)	2.5 (0.81)	5.4 (1.18)*	3.5 (0.97)	1.8 (0.70)	6.1 (1.25)		
Private health insurance										
Yes	3.3 (0.74)	6.7 (1.02)	4.0 (0.81)	3.4 (0.75)	8.8 (1.16)	3.4 (0.74)	2.4 (0.62)	7.8 (1.10)		
No	2.1 (0.67)	4.9 (1.00)	4.4 (0.95)	2.7 (0.76)	5.8 (1.08)	3.1 (0.81)	1.8 (0.62)	5.7 (1.07)		
Annual household income (A\$)										
<20,000	2.5 (1.23)	2.3 (1.19)	3.0 (1.34)	2.4 (1.22)	5.4 (1.79)	3.3 (1.42)	0.0 (0.00)	2.7 (1.30)		
\$20,000 - 40,000	3.3 (1.24)	6.8 (1.75)*	3.0 (1.19)	2.2 (1.02)	7.7 (1.87)	4.0 (1.36)	3.0 (1.20)*	5.1 (1.54)		
>40,000	2.7 (0.68)	6.4 (1.03)**	5.4 (0.95)	4.1 (0.83)	7.0 (1.07)	3.4 (0.76)	2.4 (0.65)**	9.2 (1.22)**		

Table 4.11 Socio-demographic characteristics and	visits to practitioners of each	1 of the 17 forms of the table of tabl	CAM (continued)
--	---------------------------------	--	-----------------

Note 1: probability of proportion being significantly different from the first value in each demographic category (Z test) * p<0.05, ** p<0.01 Note 2: significance levels were adjusted to allow for multiple comparisons

4.1.6 Use of Other Forms of CAM

4.1.6.1 Prayer

In addition to the 17 major forms of CAM investigated, participants were asked about their use of prayer and "other forms of CAM not mentioned elsewhere in the survey". In total, one in five (20.8%, 95%CI: 18.4% – 23.2%) surveyed participants had prayed in the 12 months prior to the survey. Of these, over a quarter (25.7%, 95%CI: 20.0% – 31.4%) had visited a prayer practitioner, such as a priest, in the 12-month period. A proportion of participants also mentioned that they prayed or visited a professional to pray for their health or someone else's health. Detailed percentages of such use, however, were not recorded.

Presented in Table 4.12 is the demographic information related to prayer users and people who had visited a practitioner for the purpose of praying. Importantly, the answers recorded on prayer did not differentiate between "prayer for personal health purposes" and "prayer for religious purposes". The latter is not normally regarded as complementary or alternative health care.

As opposed to the overall use of the 17 specific forms of CAM, the proportion (26.7%) of adults aged 65 or older who had prayed for any reason was much higher than the proportion among younger adults aged 18-34 (17.9%, p<0.05) who had prayed. It is interesting to note that a statistically significant difference was found between those having a lower household income (less than \$20,000, with a prevalence of 26.7%) and those having a relatively high household income (more than \$40,000, with a prevalence of 17.2%, p<0.05).

Characteristic*	Pray	er	Visited prayer practitioner		
	% (SE %)	P value [†]	% (SE %)	p value [†]	
Gender					
Female	25.3 (1.87)		6.1 (1.03)		
Male	16.1 (1.60)	< 0.001	4.5 (0.90)	>0.05	
Age (year)					
18-34	17.9 (2.10)		3.7 (1.04)		
35-64	20.2 (1.72)	>0.05	5.4 (0.97)	>0.05	
65+	26.7 (3.26)	< 0.05	7.4 (1.93)	>0.05	
Country of Birth					
Australia	20.1 (1.41)		5.3 (0.79)		
Non-Australia	21.9 (2.62)	>0.05	4.7 (1.35)	>0.05	
Region					
New South Wales	19.2 (2.08)		4.9 (1.14)		
Victoria	22.3 (2.54)	>0.05	6.3 (1.48)	>0.05	
Queensland	23.4 (2.95)	>0.05	6.7 (1.73)	>0.05	
South Australia	19.6 (4.36)	>0.05	5.3 (2.46)	>0.05	
Western Australia	15.7 (3.54)	>0.05	2.9 (1.65)	>0.05	
Self-reported health status					
Excellent/very good/good	19.5 (1.32)		5.0 (0.73)		
Fair/poor	27.9 (3.53)	< 0.05	6.9 (2.00)	>0.05	
Post-secondary education					
No	18.3 (1.81)		4.2 (0.94)		
Yes	22.6 (1.70)	>0.05	6.2 (0.98)	>0.05	
Employment					
Employed	19.4 (1.51)		5.4 (0.86)		
Unemployed or not in labour force	24.0 (2.24)	>0.05	5.4 (1.18)	>0.05	
Private health insurance					
Yes	20.1 (1.64)		6.5 (1.01)		
No	20.9 (1.88)	>0.05	3.7 (0.87)	< 0.05	
Annual household income					
<a\$20,000< td=""><td>26.7 (3.51)</td><td></td><td>4.7 (1.68)</td><td></td></a\$20,000<>	26.7 (3.51)		4.7 (1.68)		
A\$20,000-A\$40,000	22.8 (2.93)	>0.05	5.6 (1.61)	>0.05	
>A\$40,000	17.2 (1.59)	< 0.05	5.3 (0.94)	>0.05	

Table 4.12 Socio-demographic characteristics on users of prayer

SE: standard error

Note: significance levels were adjusted to allow for multiple comparisons

* Excludes subjects who did not provide socio-demographic characteristics
 [†] Probability of proportion being significantly different from the first value in each demographic category (Z-test)

4.1.6.2 Other Forms of CAM

Among all survey participants, less than 5% mentioned the use of forms of other CAM than the 17 named forms of CAM or prayer. Only three participants mentioned that they used other forms of CAM but did not use any of the 17 named CAM therapies or prayer. Diverse answers were provided and none of these therapies had a prevalence of higher than 2% among all participants (Table 4.13).

	Used in Past 12 Months						
Type of CAM	Original (N)	Weighted (%)	RSE (%)*				
Other forms of relaxation techniques ^{\dagger}	13	1.24	27.3				
kinesiology	6	0.60	39.5				
Iridology	4	0.36	51.0				
Bowen technique	4	0.33	53.2				
Bioelectromagnetic-based therapies	3	0.32	54.2				
Shiatsu	3	0.25	60.6				
Exercise	2	0.22	64.8				
Cranio sacral therapy	2	0.17	75.0				
Indigenous and aboriginal medicine	2	0.16	76.8				
Hypnotherapy	2	0.15	80.2				
Ayuveda	1	0.07	119.4				
Colonic irrigation	1	0.12	88.2				
Ear candling	1	0.11	91.5				
Hot bath	1	0.07	113.0				
Juice therapy	1	0.08	105.5				
Level three reconnection therapy	1	0.07	116.1				
Silver clove	1	0.08	105.1				
Tae Bo	1	0.07	113.6				
Other ^{††}	5	0.44	46.0				
Total other CAM other than the 17 major							
CAM and prayer	51	4.6	14.0				

Table 4.13 The use of other forms of CAM

* RSE: relative standard error = (standard error /estimated prevalence)*100; a relative standard error >25% indicates that result should be interpreted with caution

† Examples include Pilates and weight-loss exercise

†† Participants did not remember the actual name of the CAM or had stated other forms of CAM, but the interviewers were not able to record the names.

4.1.7 Economic Considerations of CAM Use

4.1.7.1 **Overview**

CAM users were asked about their "out-of-pocket" annual expenditure on CAM (products, practitioner visits and other related items). Thirty-five CAM users, who did not respond to one or more of the three questions on expenditure, were excluded from the expenditure analysis. The crude expenditure estimations on CAM were skewed toward the lower values (Figure 4.2, raw data). The maximum expenditure reported by CAM users was \$10,000, either on CAM products or on CAM practitioners, and \$6,000 on other related CAM. Some of the high amounts (e.g. \$5,000 spent on visiting CAM practitioners) were recorded by interviewers and verified by the researchers during the CATI telephone interviews. Such values may be considered as the actual expenditure of the users. However, when examining the raw data, a total of 95% users provided an amount of less than \$1,500 (see Appendix G4-G6 for the original data and G7 for a statistical summary). Thus, there is a need to adjust for the average expenditure among all CAM users, in order to extrapolate these values to a national level.



Figure 4.2 Frequency distribution of CAM expenditure data

Since the expenditure reported by a small number of participants appeared to be extraordinarily high, in an attempt to exclude aberrant values in total expenditure estimates, those extremely high values (e.g. in general, a value that is higher than the mean±2SD can be considered as a potential outlier) and the same numbers of the lowest value were excluded. To be consistent, these numbers, corresponding approximately to 2.5 % of the highest and the lowest expenditure values were ignored. (presented as adjusted means below).

Using this correction, the estimated annual expenditure by survey participants on CAM products, CAM practitioners and other CAM-related items is summarised in Table 4.14. Thus, based on a total of 15.5 million adult Australians in 2005, together with the estimated CAM prevalence arising from the survey, the Australian national extrapolated total expenditure in the 12 months preceding the survey is estimated as A\$4.13 billion (Table 4.14) (see Appendix G8 for the detailed statistics). On the other hand, as mentioned above, it was estimated that adult Australians had made a total of 69.15 million CAM practitioner visits (see Chapter 4.1.5.2). This suggests that the average out-of-pocket expenditure to CAM practitioner visits is estimated at A\$25 per visit (i.e. A\$1.73 billion divided by 69.15 million visits).

Expenditure by survey participants	CAM products (65.8%)*	CAM practitioners (42.3%)*	Other items related to CAM (16.3%)*
Adjusted mean $(SE)^{\dagger}$	182.3 (9.3)	263.6 (15.4)	212.3 (23.6)
95% CI	164-200	233-294	166-259
Median	100	150	100
Range	0-1400	0-2000	12-2000
Estimated national expenditure [#]	1.86 billion	1.73 billion	0.54 billion

 Table 4.14 National out-of-pocket expenditure on CAM

* % of participants that provided estimates

[†] Lowest and highest 2.5% of estimates ignored (see text)

[#] Based on an adult Australian population of 15.5 million

4.1.7.2 Regional Differences in CAM Expenditure

While CAM prevalence and the prevalence of visiting CAM practitioners in each state in Australia varied considerably, it is also of interest to compare the average expenditure in different regions. Presented in Table 4.15 are adjusted means expenditure for CAM products, CAM practitioners and CAM-related items (weighted), as well as state estimations based on the corresponding state population.

State expenditure (A\$) Products NSW VIC SA WA QLD Adjusted mean 186.7 172.7 187.1 187.4 163.2 SE 16.2 28.6 18.7 19.8 33.5 State estimation (million) 664.0 449.4 382.6 128.2 150.9 F (4, 638)=0.20, p=0.938 Test for significant difference via ANOVA: **Practitioners NSW** VIC SA WA QLD 200.3 Adjusted mean 285.9 258.6 281.1 204.1 SE 27.5 28.9 39.9 39.0 38.8 State estimation (million) 644.9 449.2 344.7 106.8 121.7

Table 4.15 Regional differences of out-of-pocket expenditure on CAM

Test for significant difference via ANOVA: F (4, 410)=0.952, p=0.434

Other items related to CAM	NSW	VIC	QLD	SA	WA
Adjusted mean	211.1	176.0	321.4	235.9	171.3
SE	26.1	45.3	90.2	83.5	51.8
State estimation (million)	181.5	122.7	111.0	52.5	48.0
Test for significant difference vi	F (4, 156)	=0.964, p=0	.429		

SE: standard error; NSW: New South Wales; VIC: Victoria; QLD: Queensland; SA: South Australia; WA: Western Australia.

For CAM products, it appeared that the average expenditure per user in each state is similar (ANOVA, F (4, 638)=0.20, p>0.05). In addition, the average amount spent on CAM products in most states is similar to the national expenditure level (A\$182.3), although Western Australia users reported a relatively lower average expenditure (A\$163.2).

On average, South Australia and Western Australia users appeared to spend less on visiting CAM practitioners than the estimated national average (approximately A\$60 less per person per year), although, again, the differences between states were not found to be statistically significant (ANOVA, F (4,410)=0.952, p>0.05).

A much bigger variation (although not significantly different, ANOVA, F (4, 156)=0.964, p>0.05) was found between the costs related to CAM use, such as books, equipment and classes. That is, New South Wales and South Australia users had similar costs to the national average; Queensland users had a higher cost, whereas the costs estimated by Victoria and Western Australia users appeared to be relatively low (Table 4.15).

In all, at least one third (36.1%) of the total CAM costs in Australia was spent in New South Wales, a further quarter (24.7%) was spent in Victoria, 20.3% in Queensland, 7.0% in South Australia, 7.8% in Western Australia, and only approximately 4.2% was spent in the remaining states and territories. Rather than the proportions distributed to each state for total CAM costs, individual state expenditure on the costs of each of the CAM products, CAM practitioners or other related CAM items also had similar distributions. For example, at least one third (35.7%) of the total national costs on CAM products was spent in New South Wales, and approximately a quarter (24.2%) was spent in Victoria (Table 4.15).

4.2 Discussion

4.2.1 Prevalence of CAM Use

4.2.1.1 Comparison with Other Australian Studies

Perhaps the most important finding of the present study is that the prevalence of CAM use in Australia is considerably higher than previously estimated. The current study indicates that 68.9% of the Australian population are CAM users, whereas the figure usually quoted is 52.2%.¹¹ The previous estimate is, in fact, an extrapolation from the findings of a survey conducted within a single Australian state, South Australia, in 2004.¹¹ Although the types of CAM included in this national study were not identical to those in the previous regional study, it is probable that regional differences account for the higher national estimate of CAM use. Unfortunately, within the national quota, the regional quotas were insufficient to demonstrate what might be real differences in the prevalence of CAM use between Australian states. The current study suggests that only 60.8% of the South Australian cohort were CAM users, whereas for the most populous state, New South Wales, the prevalence of use was 72.1%, and only slightly lower in Queensland (71.0%) and Victoria (69.8%). However, as noted above, statistically significant differences between these regional figures were not demonstrated in the current study.

The findings of the high prevalence of CAM use in several defined states are consistent with previous small-scale regional studies, which were based on randomly selected participants in New South Wales. It was suggested that between 67% and 70.3% of participants used CAM during a 12-month period.^{37,38} In line with the current global trend of an increase in the use of CAM over the past decade, the relatively higher prevalence of CAM use as estimated in the present study appears accurate.

It is worth noting that CAM modalities investigated in the current and previous studies varied greatly. Thus, comparisons of the prevalence of individual forms of CAM can only be made with limited forms of CAM. In addition, the previous surveys were conducted in the states of South Australia and New South Wales. Thus, state and national prevalence of CAM use in the current survey are presented (Table 4.16).

	Current survey prevalence			Regional survey prevalence			
САМ	National	NSW	SA	NSW 95 ³⁸	NSW 99 ³⁷	SA 04 ¹¹	
Overall	68.9%	72.1%	60.8%	67.0%	70.3%	52.2%	
Clinical nutrition*	45.8%	47.3%	43.6%	$57.0\%^\dagger$	68.7%	39.2%**	
Herbal medicine [#]	19.8%	20.3%	14.2%	32.0% ^{††}		20.6%	
Chiropractic	16.1%	17.0%	17.0%		26.1%		
Massage therapy [#]	28.8%	28.7%	25.3%		25.1%		
Homeopathy	6.0%	5.0%	8.9%	$3.0-8.0\%^{\ddagger}$		2.2%	
Aromatherapy	16.1%	13.8%	10.0%			$11.2\%^{\ddagger\ddagger}$	
Soy products						3.8%	
Chinese medicine	19.3%	17.1%	17.2%		##	2.3%	

 Table 4.16 Comparison of CAM prevalence with previous Australian surveys

NSW: New South Wales; SA: South Australia

- ... Indicates prevalence was not provided in that survey
- * Including multivitamins and mineral supplements
- # Includes different forms of massage therapy or herbal medicine
- ** Prevalence was for vitamins only; the prevalence for mineral supplement was 13.6%, combined prevalence was not presented in the article.
- † Includes 18 different nutritional items
- †† Western herbal medicine only; total herbal medicine was 33.0%
- ‡ General homeopathies, 3%, when including other homeopathic remedies, 8.0%
- ‡‡ Refers to aromatherapy oil
- ## Categories of acupuncture and Chinese medicine were used in the survey, details did not present.

In general, previous regional surveys appear to show a higher individual prevalence of CAM use. For example, the use of clinical nutrition in New South Wales, estimated in the current survey, was 47.3%, while the prevalence in the 1995³⁸ and 1999³⁷ New South Wales surveys was higher. However, these regional surveys included a broader range of vitamins and minerals, whereas the current survey specified multivitamins and minerals, rather than daily consumer vitamin (e.g. vitamin A prescribed by a medical doctor). On the other hand, the 39.2% of vitamin prevalence in South Australia was estimated separately from mineral supplements use (13.6%).¹¹ The combined prevalence is considered no less than the South Australia prevalence as estimated in the current study (43.6%). Similarly, the previously estimated prevalence of chiropractic and herbal medicine in New South Wales^{37,38} and the prevalence of herbal medicine in South Australia¹¹ were also higher than the current estimated prevalence in each of these corresponding states (Table 4.16).

The prevalence of massage therapy use in New South Wales^{37,38} and aromatherapy use in South Australia¹¹ were similar to estimations in the current survey. However, a great variation in the prevalence of homeopathy has been reported. The New South Wales prevalence (5.0%) estimated in the current survey was within the broad range of that from the New South Wales survey in 1995 (3.0%-8.0%).³⁸ However, a significantly lower prevalence was estimated in South Australia (2.2% in 2004), ¹¹ compared to the current estimation for the state of South Australia (8.9%). These are in contrast to the abovementioned regional differences in the prevalence of individual forms of CAM, where the prevalence of homeopathy in South Australia, as estimated in the current survey was highest, compared to any other state or territory of Australia. Furthermore, after examining the South Australia surveys in 1993,⁹ 2000¹⁰ and 2004,¹¹ it is very surprising that the prevalence of homeopathy decreased substantially from 4.4% in 1993, and 4.3% in 2000 to the most recent estimate of 2.2% in 2004. The large discrepancy in estimated homeopathy use requires further investigation.

Variations in the prevalence of CAM practitioner visits between Australian states were somewhat less than variations in the overall prevalence of CAM use. Based on the current findings, the proportion of the Australian adults that consulted at least one CAM practitioner over a 12-month period was 44.1%. This is again much higher than the prevalence of CAM practitioner visits estimated in the 2004 South Australian study (26.5%).¹¹ Again, regional differences may, at least in part, explain the difference, although, it is also likely that different CAM modalities in the two studies are a factor. For example, importantly, massage therapy was not included in any one of the recent South Australian CAM surveys,⁹⁻¹¹ whereas a substantial proportion of the current survey participants had visited massage therapists.

Figure 4.3 presents the prevalence of visits to practitioners of eight individual forms of CAM, that were estimated by the 2004 South Australia survey,¹¹ by the current survey for the State of South Australia, and by the current survey for the whole nation. As noted above, the current survey provides a higher overall prevalence of CAM practitioner visits in South Australia, compared to the estimated prevalence in previous South Australia surveys; this seems to be the same for visits to practitioners of most individual forms of CAM, except naturopaths, herbal therapists, reflexologists and chiropractors.

For visits to reflexologists and chiropractors, the South Australia estimates and the current survey estimates for South Australia were similar. Surprisingly, based on the current survey, none of the herbal medicine users from South Australia had visited practitioners; and the South Australia survey estimated that only 1.9% had visited such practitioners.¹¹ A low prevalence of visiting herbal medicine practitioners was also observed in both studies. It is also important to note the limitations of the current study in drawing conclusions at a state level. In particular, the relatively small number of interviews that were conducted in South Australia.



Overall CAMP: visits to any one type of CAM practitioners SA: South Australia

Figure 4.3 Prevalence of visits to CAM practitioners in the current survey and a recent South Australia survey

The prevalence of visits to naturopaths estimated by the South Australia survey¹¹ (5.7%) was much higher than the current survey estimate for South Australian adults (1.3%) but was similar to the national estimation (5.9%). In the 2004 South Australia study,¹¹ however, a prevalence of 5.7% was estimated for visits to a naturopath or a natural therapist. The definition of natural therapist is generally broader than the definition of a defined naturopath, and may have contributed to the different finding.

4.2.1.2 Comparison with Overseas Studies

It is of interest to compare the current estimated prevalence of CAM use in Australia with the most recent estimates for the United States (62%),¹ Canada (50%),⁶ the United Kingdom $(28.3\%)^5$ and Japan (76.0%).⁷ The data were collected through the computer-assisted telephone interview (CATI) method, with the exceptions of the US data, which were through computer-assisted personal interview (CAPI), and the UK data, which were through postal questionnaire. Detailed data collection methods were discussed in Chapter 2.3.2.2.

The comparisons of CAM as a whole among these countries would be less meaningful without the recognition of the limitations of different forms of CAM being included in these surveys. For example, the 62% CAM prevalence as estimated in the 2002 US survey included prayer specifically for health reasons.¹ When prayer for health reasons was excluded from the definition of CAM, only 36% of US adults used CAM in the past 12 months, which is lower than a 1997 US survey conducted by the Harvard Medical School (42.1%).³

The 1998 UK study found a 28.3% CAM prevalence.⁵ However, the definition of CAM in this study included only eight forms of CAM. The number of CAM investigated was markedly less than the number investigated by most CAM surveys in other countries. In addition, in this UK study, except for over-the-counter homeopathic remedies and herbal remedies, the prevalence of other CAM was based on visits to practitioners rather than actual CAM use. These may have impacted considerably on the overall prevalence of CAM use in the UK. Surprisingly, a more recent survey conducted by the same group of researchers in 2001 found that only an estimated 10.0% of the UK population had received any one of the 23 forms of CAM from practitioners in the past 12 months.¹⁶ Again, this prevalence was for practitioner visits only; the overall prevalence of CAM use in the UK remains unclear.

With respect to the prevalence of specific forms of CAM, this would be clear only if comparisons were made with several common forms of CAM. Again, as noted above, this is due to the names and classifications of individual forms of CAM varying between different surveys in different countries and even within a country. For example, the modality of herbal medicine was not singled out in a recent US survey in 2002, thus creating a difficulty in making comparisons with other studies.¹ Nevertheless, presented in Table 4.17 are the common CAM modalities that were investigated in the current survey, the 1997 and 2002 US surveys,^{1,3} the 1998 UK survey,⁵ the 1999 Canadian survey⁶ and the 2001 Japanese survey.⁷

A high prevalence of the use of herbal medicine, massage therapy and chiropractic was observed in the current survey, the 1997 US survey and the 1999 Canadian survey. A minimum of 11% of the adult population of these countries had used these therapies during a 12-month period. On the other hand, with the exception of the extremely popular use of homeopathy as estimated in the UK in 1998 (8.6%), for most individual forms of CAM, its use among Australians was more prevalent than most other countries.

Existing evidence indicates that the practice of, and training in, acupuncture is common in the US and the UK. For example, it has been estimated that over 10,000 non-physician acupuncturists and nearly 3,000 physicians practised acupuncture in the US in 1998.¹⁵⁰ In addition, according to the Accreditation Commission for Acupuncture and Oriental Medicine, a national accrediting agency recognized by the U.S. Department of Education, there were over 50 schools and colleges of acupuncture with accredited or candidacy status.¹⁵¹ However, it is surprising to find out that less than 2% of adults from the US and the UK had used acupuncture, whereas nearly 10% of Australians had done so. In all, the biggest distinction between the prevalence of CAM use in Australia and other countries is in the use of acupuncture, aromatherapy and naturopathy.

~~~~	Prevalence (%)									
САМ	AUS 05*	<b>US 02</b> ¹	US 97 ³	UK 98 ⁵	CAN 97 ⁵	<b>JPN 01</b> ⁷				
Acupuncture	9.2	1.1	1.0	1.6 [‡]		6.7				
Chiropractic	16.1	7.5	11.0	3.6 [‡]	13.0	7.1 ^{‡‡}				
Massage therapy	28.8**	5.0	11.1		12.0	14.8				
Osteopathy	4.6			4.3 [‡]		$7.1^{\ddagger\ddagger}$				
Reflexology	4.1			$2.4^{\ddagger}$						
Herbal medicine	19.8**	***	12.1	19.8	12.0	17.2				
Homeopathy	6.0	1.7	3.4	8.6	4.0	0.3				
Aromatherapy	16.1		$5.6^{\dagger\dagger}$	3.5 [‡]	5.0	9.3				
Naturopathy	10.7	0.2	$0.7^{\dagger\dagger}$							
Any CAM	68.9	$62.0^{\dagger}$	42.1	28.3	50.0	76.0				

Table 4.17 Comparison of CAM prevalence with selected overseas CAM surveys

AUS: Australia; US: United States; UK: United Kingdom; JPN: Japan; CAN: Canada.

... Indicates prevalence was not provided in that survey

* Refers to the current survey that conducted in 2005

** Prevalence includes different forms of massage therapy or herbal medicine

*** The category of "non-vitamin, non-mineral, natural products" included common herbal medicine, such as Ginseng and Ginkgo biloba. The total prevalence was 18.9%.

† Prevalence included prayer specifically for health reasons.

†† Use of these CAM was not part of the total 42.1% CAM prevalence.

Prevalence was visits to practitioners for named CAM therapy rather than the actual use. Use of each of the CAM therapies was not estimated in any UK study.

‡‡ A combined prevalence of chiropractic and osteopathy was estimated (total, 7.1%)

Anecdotally, aromatherapy and naturopathy are commonly practiced in Australia. Thus, the current survey estimated that nearly one in six Australians had used aromatherapy in the 12 months preceding the survey. The high prevalence was also estimated in an Australian regional study.¹¹ On the other hand, only approximately 5% of Canadians and Americans reported the use of aromatherapy.^{3,5} Surprisingly, in the 1997 US study the use of aromatherapy was not counted toward the total CAM prevalence.³ Efforts to ensure the CAM modalities were comparable to a previous 1990 study conducted by the same group of researchers seemed to be the main rationale for excluding additional, newly investigated

therapies in 1997.^{2,3} Thus, the total estimated CAM prevalence would be higher in the US if the CAM definition included aromatherapy and naturopathy.

It is also of interest to compare the findings arising from the current survey with data for Japan. The prevalence of overall CAM use in Japan is much higher than most countries, including Australia (Table 4.17). It is not surprising that some forms of CAM, including Kampo and acupuncture, are widely accepted in Japan.⁷ For example, in addition to the 17.2% adults who used herbs or over-the-counter (OTC) Kampo (Table 4.17), 10% of adults also used ethical Kampo and nearly half (43.1%) of Japanese adults used a dietary supplement. However, the use of other individual forms of CAM was relatively lower in Japan than in Australia. Thus, despite the overwhelming number of licensed acupuncturists in Japan (113,000 in 2001),¹⁵² the total prevalence of using acupuncture and moxibustion in 2001 was only 6.7%, which was lower than the Australian prevalence (9.2%) in 2005.

The findings of the current study provide strong evidence that complementary and alternative medicine forms an integral part of Australia's health care system. The popularity of CAM in Australia is not only evidenced by the high prevalence of common forms of CAM, such as chiropractic and acupuncture. But also by forms of CAM that are not common in other countries, such as aromatherapy and naturopathy. Concerning the efficacy and safety of use, further rigorous research in the CAM field in Australia is necessary and critical.

# 4.2.2 Characteristics of CAM Users

Most commonly, survey studies on CAM prevalence have also investigated the use of CAM by different socio-demographic groups. Common characteristics related to a higher prevalence of CAM, such as age, educational background, and income range were first reported in a 1990 US study.² The current study suggests a higher prevalence of CAM use by females, by the better educated, by those with higher incomes, by those in employment, and by those with private health insurance cover. These findings are generally consistent with reports of similar studies from other countries^{1.3} and those of previous Australian studies.^{10,11} However, in terms of age difference, the current survey found the highest prevalence was among the youngest adults (18-34 years old), rather than by middle-aged adults. Some findings in regard to the characteristics of users may differ between studies. For example, the gender differential in the use of CAM was not observed in two US studies^{2.46} and a recent UK study.¹⁶

It is of interest to examine the ethnicity of CAM users, as the conclusion drawn from a large population survey suggested that the use of different CAM modalities for treatment versus illness varied across race and ethnicity categories.¹⁵³ Although differences in CAM use between Australian residents born in different countries were not explored, or between those in different ethnic groups, the findings indicate that, in general, individuals who were not born in Australia were less likely to visit a CAM practitioner than those born in Australia. This finding was similar to the 2004 South Australian study, which found that Australian-born residents had used CAM more than those born elsewhere.¹¹

The difference of CAM use among different ethnic groups was also commonly reported in the US surveys. However, the findings were somewhat contradictory between studies. For example, both of the Harvard surveys^{2,3} found the use of CAM was significantly less common

among African Americans (defined as black adults) than among members of other racial groups. However, a more recently published article suggested that black adults were more likely to use CAM overall, when the definition of CAM included prayer specifically for health purposes. The authors further revealed that the high prevalence among black adults was mainly in the use of the mind-body therapies, including prayer, whereas the use of biological-based therapies, such as multivitamins, was less common.¹

Limited CAM surveys have also explored the association between the use of CAM and users' status of health insurance coverage. Again, commonly, people with private health insurance were more likely to use CAM.^{1,4} This finding is not surprising. However, detailed information on what forms of CAM people with private health insurance were more likely to use was limited. The current survey has also confirmed high use among those with private health insurance. In addition, the current survey revealed that the common forms of CAM that participants had used, such as chiropractic, acupuncture and remedial massage were usually covered by private health insurance. Thus, for those who were less commonly covered by private health insurance, for CAM such as aromatherapy, a significant variation of prevalence between people with and without private health insurance was not observed. In this case, aromatherapy used by those without private health insurance was even higher than the use among those with insurance coverage.

In terms of regional differences within countries, the trend is obvious that American people who reside in the West of the country are much more likely to use CAM than residents from other regions.¹⁻⁴ In addition, in the UK, a relatively high prevalence was found in the Midlands and East Anglia, while in the North of England it was lower.¹⁶ In Canada, the highest rate was reported in British Columbia and Alberta.^{45,48}In Australia, potential regional differences had not been investigated in the past, as each of the previous surveys was

conducted in a single state. Thus, the current survey suggests that there are marked regional differences within Australia. The CAM prevalence on the East coast, including New South Wales, Victoria and Queensland was considerably higher than other states/territories.

Another frequently reported characteristic associated with CAM use is the health status of the participants, either self-rated general health or some specific health conditions. Commonly, people with chronic health conditions are more likely to use CAM.⁴⁵ More specifically, people with conditions such as back pain, allergy and fatigue are frequently found to be associated with a high prevalence of CAM use.^{3,46} In the current survey, overall CAM prevalence between people with relatively poor health and those in better health was less significant. However, for certain forms of CAM, such as acupuncture, chiropractic and energy healing, people in poor health were more likely to have used these therapies. Unfortunately, detailed health conditions associated with specific forms of CAM use in Australia's general population were not collected in the current survey, nor in any previous surveys in Australia.

In summary, the common characteristics associated with CAM use in Australia are reasonably consistent with the literature in other countries. The current survey better comprehends these general findings in the Australia health context, notably in the regional differences and the status of health insurance and CAM use among adult Australians.

# 4.2.3 Economic Considerations of CAM Use

The Australian Government's Department of Health and Ageing estimates that the average number of services provided per patient by medical practitioners in the year ending June 30, 2005 was 5.5.¹⁵⁴ The estimate from the current study is similar, being 5.53 visits over a 12-month period. For an adult population of 15.5 million (with an estimated 80.8% of them visiting medical practitioners), taking the average number of visits to a medical practitioner per annum as 5.5, the total number of consultations by the adult Australian population in a 12-month period can be estimated as 68.9 million. Again, this figure is similar to the current estimate of the total number of visits to CAM practitioners over a 12-month period (69.2 million). The similar number of practitioner visits for conventional medical services and CAM services is all the more surprising considering that generally, most of the cost of CAM practitioner visits would be borne by users, whereas medical practitioner consultations are subsidised by the national health system.

The Australian Institute of Health and Welfare (AIHW) estimated that, between 2003 and 2004, "out-of-pocket" expenditure by Australians on all over-the-counter health care products not subsidised by the national Pharmaceutical Benefits Scheme (PBS) was A\$4.05 billion.¹⁵⁵ It is of interest to compare this figure with the current estimate of A\$1.86 billion spent by Australians on CAM products in the 12-month period. It appears that the annual national expenditure on CAM products accounts for almost half the expenditure on non-subsidised health care products. With regard to practitioner services, the AIHW estimated that Australians spent over A\$1.6 billion on medical services. The current findings indicate that they spend a similar amount (A\$1.73 billion) on CAM practitioner visits.

It is of interest to note the CAM expenditure estimated in two South Australia studies.^{10,11} One, conducted in 2000, estimated that over A\$2.3 billion would had been spent by the all Australians, whereas another, more recent, South Australia study in 2004 estimated that the cost of CAM had decreased to A\$1.8 billion. The adverse publicity of the Pan Pharmaceutical crisis in 2003 could have contributed to the decline of the cost, as researchers have mentioned.¹¹ In contrast, the current survey estimated a much higher CAM cost incurred by Australians, based on the responses from participants nationwide in 2005. It was estimated that the national expenditure on CAM was 4.13 billion per annum. This was higher than that any previous Australian studies. The amount included additional expenditure related to CAM use, such as books and audio-visual materials (A\$0.54 billion); this category was not included in any of the previous Australian regional surveys. Whether the current estimation better reflects current utilisation in Australia would need to be confirmed in a future well-designed follow-up study with a consistent research protocol.

It is also worth noting that the mean expenditure on CAM practitioners was relatively low in South Australia and Western Australia, compared to the mean expenditure in New South Wales, Queensland and Victoria and to the national average. Although not statistically significant, the variations were notable. For example, the mean expenditure among South Australian adults was A\$80 less than the amount spent by New South Wales adults (A\$204.1 and A\$285.9, respectively) and about A\$60 less than the national average (A\$263.6). The average expenditure spent on CAM products in different states was reasonably comparable. As noted, the current survey suggests a lower CAM prevalence in the state of South Australia as well as a lower mean expenditure on CAM in South Australia. The above-noted discrepancy between the estimated national CAM expenditure in the current survey and the extrapolated national expenditure generated from a single state (South Australia)^{10,11} requires further consideration.

152

Internationally, out-of-pocket expenditure on CAM was also estimated in studies conducted in the US, the UK and Canada. The highest expenditure was estimated in the US in 1997, when over \$US27 billion was spent by the Americans. Notably, expenditure on herbal products alone amounted to \$US5.1 billion, and out-of-pocket expenditure on high-dose vitamins more than tripled from US\$0.9 billion in 1990 to US\$3.3 billion in 1997.³ A relatively moderate out-of-pocket expenditure on CAM was estimated in a 1997 Canadian study (3.8 billion Canadian dollars).⁶ On the other hand, Ernst and Write estimated a total of £1.6 billion was spent by UK adults in 1999, based on the average cost of a pre-categorised range of estimations.¹⁷ Thomas and her colleagues estimated the CAM cost of six "more established" therapies, and the annual out-of-pocket expenditure was estimated as £450 million in UK.⁵ Thus, the estimated CAM expenditure in the UK studies may need to be interpreted with caution, as suggested by the researchers.

In summary, the estimation of CAM expenditure is subject to a number of problems. The potential inaccuracies of recall bias (generally, for a one-year period) are the major concern for the accuracy of extrapolation. Moreover, the greater variation of cost estimations by individual CAM users restricts the precision of extrapolation from a survey population to a national level. As observed in the current survey, a small proportion of participants spent a relatively large amount on CAM, due to their specific chronic health conditions, such as trauma. The estimations made from information provided by such participants do not necessarily violate the accuracy of such cost estimation. However, the extreme values may impact substantially on the overall average. Finally, the totality of the CAM expenditure is also subject to the same weakness of current uncertainty over the definition and classification of CAM modalities in different social contexts among literature.

# CHAPTER 5. REGIONAL DIFFERENCES IN CAM USE

As mentioned earlier, regional difference in overall CAM use in Australia may contribute to the relatively high prevalence found in the current study. It is of particular interest to explore the use of specific forms of CAM and investigate the characteristics of CAM users in different states and territories in Australia. The findings and discussions are presented in this Chapter.

# 5.1 Results

## 5.1.1 Overview

When interpreting the findings presented in this Chapter, it is important to note the sample of interviews that were conducted in each state and territory. As outlined in Table 3.1, the number of interviews was allocated at a state level based on the recent Australian Bureau of Statistics data. Thus, the final completed interviews were: New South Wales (353), Victoria (264), Queensland (206), South Australia (87), Western Australia (96), other states and territories (61).

Overall, the state of New South Wales had the highest prevalence of CAM use (72.1%), closely followed by Queensland (71.0 %), then Victoria (69.8%), with South Australia having the lowest rate (60.8%). In terms of visiting a CAM practitioner, the highest prevalence was found in Victoria (48.0%), with Queensland (41.4%) and Western Australia (42.0%) having a lower rate. However, these regional differences were not statistically significant (Table 5.1 and Table 5.2).

When examining the use of each of the 17 forms of CAM specifically, the significant regional differentials were identified in several modalities between states. As shown in Table 5.2, a substantially low rate of acupuncture use was reported in Western Australia (2.7%), whereas the rate was at least three times higher in New South Wales (9.9%, p<0.01), Victoria (10.9%, p<0.01) and Queensland (11.6%, p<0.01). Similarly, significantly low rates were also found in the use of osteopathy in South Australia (1.2%) and Western Australia (2.2%), and the use of naturopathy in Queensland (6.2%) and South Australia (5.5%). In contrast, Victorian adults favoured the use of aromatherapy the most (20.9%) – more than double the use in South Australia (10.0%) and 1.5 times higher than the use in New South Wales (13.8%) and Western Australia (13.7%, p<0.05).

Anecdotally, some types of CAM practitioners are more popular in some states of Australia than in others. Thus, similar to the findings of the regional differences of CAM use in Australia, as shown in Table 5.2, it is of interest to see that visits to an acupuncturist were much higher in New South Wales (8.3%) and Victoria (10.1%) than in South Australia (3.2%, p<0.05) and Western Australia (2.7%, p<0.01). Surprisingly, in South Australia, none of the survey participants had visited practitioners for Chinese herbal medicine, Western herbal medicine or Chinese medicine dietary therapy.

	Percent (standard error) of CAM use								
Regions	Any CAM	Acupuncture	CHM	СТМ	CMDT	QGMATC	WHM	WTM	Chiropractic
NSW	72.1 (2.37)	9.9 (1.57)	6.3 (1.29)	4.8 (1.13)	2.2 (0.77)	3.7 (1.00)	16.9 (1.98)	27.1 (2.35)	17.0 (1.98)
VIC	69.8 (2.80)	10.9 (1.90)	8.8 (1.73)	8.3 (1.68)	3.3 (1.09)	7.7 (1.63)*	17.3 (2.31)	27.2 (2.71)	17.4 (2.31)
QLD	71.0 (3.15)	11.6 (2.23)	8.2 (1.91)	3.4 (1.27)	1.7 (0.89)	8.5 (1.94)*	15.0 (2.49)	32.2 (3.25)	14.9 (2.47)
SA	60.8 (5.37)	6.9 (2.78)	5.8 (2.57)	2.3 (1.66)	1.7 (1.41)	5.4 (2.48)	10.1 (3.31)	25.3 (4.78)	17.0 (4.13)
WA	66.6 (4.60)	2.7 (1.58)**	3.6 (1.80)	4.4 (2.00)	2.5 (1.52)	3.0 (1.66)	15.5 (3.53)	20.4 (3.93)	14.4 (3.42)
Regions	Osteopathy	Homeopathy	Naturopathy	Meditation	Aromatherapy	Clinical nutrition	Energy healing	Reflexology	Yoga
NSW	6.0 (1.25)	5.0 (1.16)	11.8 (1.70)	16.7 (1.97)	13.8 (1.82)	47.3 (2.63)	5.6 (1.21)	4.0 (1.04)	12.7 (1.76)
VIC	5.2 (1.36)	8.0 (1.66)	14.7 (2.16)	18.3 (2.36)	20.9 (2.48)*	43.0 (3.02)	9.1 (1.75)	5.3 (1.36)	12.4 (2.01)
QLD	3.8 (1.34)	4.2 (1.40)	6.2 (1.67)*	19.0 (2.73)	16.7 (2.60)	49.9 (3.48)	6.0 (1.66)	4.0 (1.36)	13.0 (2.34)
SA	1.2 (1.18)**	8.9 (3.13)	5.5 (2.52)*	13.0 (3.70)	10.0 (3.30)	43.6 (5.45)	5.3 (2.46)	2.0 (1.55)	8.5 (3.06)
WA	2.2 (1.43)*	6.4 (2.39)	11.4 (3.10)	17.8 (3.72)	13.7 (3.35)	46.8 (4.86)	9.7 (2.89)	3.8 (1.87)	8.6 (2.73)

Table 5.1 Regional differences of use of 17 forms of CAM

Note 1: probability of proportion being significantly different from prevalence in first category (NSW) (Z-test, *: p<0.05, **: p<0.01)

Note 2: NSW: New South Wales, VIC: Victoria, QLD: Queensland, SA: South Australia, WA: Western Australia

Note 3: CHM: Chinese herbal medicine, CTM: Chinese therapeutic massage, CMDT: Chinese medicine dietary therapy,

QGMATC: Qigong, martial art and Tai Chi, WHM: Western herbal medicine, WTM: Western therapeutic massage

	Percent (standard error) of visits to CAM practitioners								
Regions	Any CAM	Acupuncture	CHM	СТМ	CMDT	QGMATC	WHM	WTM	Chiropractic
NSW	45.0 (2.63)	8.3 (1.46)	2.4 (0.81)	3.1 (0.92)	0.5 (0.39)	2.2 (0.77)	6.0 (1.25)	21.9 (2.18)	15.0 (1.89)
VIC	48.0 (3.05)	10.1 (1.84)	3.7 (1.14)	5.0 (1.33)	1.2 (0.66)	3.9 (1.18)	6.0 (1.44)	22.0 (2.53)	15.9 (2.23)
QLD	41.4 (3.43)	7.7 (1.85)	2.0 (0.96)	2.3 (1.04)	0.4 (0.42)	5.3 (1.56)	2.1 (1.00)*	19.3 (2.75)	12.8 (2.32)
SA	44.5 (5.46)	3.2 (1.93)*	0.0 (0.00)**	2.3 (1.66)	0.0 (0.00)	1.7 (1.41)	0.0 (0.00)**	19.6 (4.37)	16.1 (4.04)
WA	42.0 (4.81)	2.7 (1.58)**	1.1 (1.03)	1.3 (1.12)	0.0 (0.00)	0.8 (0.89)	6.7 (2.44)	12.6 (3.23)*	14.4 (3.42)
Regions	Osteopathy	Homeopathy	Naturopathy	Meditation	Aromatherapy	Clinical nutrition	Energy healing	Reflexology	Yoga
NSW	4.9 (1.14)	3.0 (0.90)	7.2 (1.36)	3.6 (0.98)	2.9 (0.88)	9.2 (1.53)	4.0 (1.04)	1.5 (0.65)	5.8 (1.23)
VIC	4.3 (1.24)	3.9 (1.18)	8.7 (1.72)	3.8 (1.16)	4.5 (1.26)	7.8 (1.63)	3.1 (1.05)	3.4 (1.11)	8.8 (1.73)
QLD	2.2 (1.02)	1.5 (0.86)	2.3 (1.04)**	5.9 (1.64)	1.2 (0.76)	5.2 (1.55)	1.5 (0.86)	1.6 (0.87)	7.3 (1.81)
SA	1.2 (1.18)*	4.9 (2.37)	1.3 (1.25)**	4.4 (2.25)	2.8 (1.82)	6.6 (2.73)	4.2 (2.19)	0.8 (1.01)	5.1 (2.42)
WA	1.1 (1.00)*	1.3 (1.12)	6.8 (2.46)	4.0 (1.90)	4.0 (1.91)	6.6 (2.42)	4.2 (1.96)	2.5 (1.52)	4.4 (2.00)

Table 5.2 Regional differences of visits to practitioners for 17 forms of CAM

Note 1: probability of proportion being significantly different from prevalence in first category (NSW) (Z-test, *: p<0.05, **: p<0.01)

Note 2: NSW: New South Wales, VIC: Victoria, QLD: Queensland, SA: South Australia, WA: Western Australia

Note 3: CHM: Chinese herbal medicine, CTM: Chinese therapeutic massage, CMDT: Chinese medicine dietary therapy,

QGMATC: Qigong, martial art and Tai Chi, WHM: Western herbal medicine, WTM: Western therapeutic massage

# 5.1.2 **Profile of CAM Users in New South Wales**

More people live in New South Wales (NSW) than any other state or territory of Australia. As noted above, the proportions of CAM users and people visits to CAM practitioners in New South Wales were considerably higher than most other states/territories. A sub-group analysis on socio-demographic characteristics that related to CAM users and people visits to CAM practitioners in New South Wales is summarised in Table 5.3. It is noted noting that when conducting sub-group analyses in all states and territories, the sample sizes become smaller and, consequently, the statistical power of the test to detect differences in each region is reduced. Thus, statistically significant differences are less likely to be observed at a state level.

In New South Wales, those participants who considered their health as poor or fair were more likely to use CAM (82.8%) than those in better health (70.1%, p<0.05). Such a difference, however, was not observed in the national prevalence. In addition, New South Wales appeared to be the only state in which such a difference was found. A few national significant findings of people visits to CAM practitioners are also observed in New South Wales, such as a higher prevalence among those with a post-secondary education, who have private health insurance, and those with relatively higher income ranges (Table 5.3).

	Users		Visited practitioners		
Characteristics*	% (95% CI)	$\mathbf{p}^{\dagger}$	% (95% CI)	$\mathbf{p}^{\dagger}$	
Gender					
Female	74.6 (68.2 - 80.9)		46.0 (38.8 - 53.2)		
Male	69.5 (62.7 – 76.3)	>0.05	44.1 (36.8 - 51.4)	>0.05	
Age (year)					
18-34	75.5 (67.5 - 83.4)		45.1 (35.8 - 54.3)		
35-64	72.2 (65.7 – 78.7)	>0.05	46.2 (38.9 - 53.4)	>0.05	
65+	66.9 (55.4 - 78.5)	>0.05	42.5 (30.4 - 54.6)	>0.05	
<b>Country of Birth</b>					
Australia	72.2 (66.9 - 77.6)		46.9 (41.0 - 52.8)		
Non-Australia	71.5 (61.9 - 81.2)	>0.05	40.6 (30.1 - 51.1)	>0.05	
Self-reported health status					
Excellent/very good/good	70.1 (65.0 - 75.3)		43.4 (37.8 - 48.9)		
Fair/poor	82.8 (72.8 - 92.8)	< 0.05	54.3 (41.1 - 67.5)	>0.05	
Post-secondary education					
No	68.7 (60.8 - 76.7)		33.5 (25.5 - 41.6)		
Yes	74.2 (68.4 - 79.9)	>0.05	52.1 (45.6 - 58.6)	< 0.01	
Employment					
Employed	74.3 (68.7 - 79.8)		47.2 (40.8 - 53.5)		
Unemployed or not in					
labour force	68.4 (59.9 - 76.8)	>0.05	41.9 (32.9 - 50.8)	>0.05	
Private health insurance					
Yes	74.8 (68.6 - 80.9)		51.0 (43.9 - 58.0)		
No	69.3 (62.1 - 76.4)	>0.05	39.5 (31.9 - 47.1)	< 0.05	
Annual household income					
<a\$20,000< td=""><td>55.7 (42.2 - 69.3)</td><td></td><td>33.7 (20.8 - 46.7)</td><td></td></a\$20,000<>	55.7 (42.2 - 69.3)		33.7 (20.8 - 46.7)		
A\$20,000-A\$40,000	73.5 (62.0 – 85.1)	>0.05	53.1 (40.1 - 66.2)	< 0.05	
>A\$40,000	73.3 (67.1 – 79.5)	< 0.05	44.6 (37.7 – 51.6)	>0.05	

Table 5.3 Socio-demographic characteristic	s of	CAM	users	in	New	South	Wales
--------------------------------------------	------	-----	-------	----	-----	-------	-------

Note: significance levels were adjusted to allow for multiple comparisons

* Excludes subjects who did not provide socio-demographic characteristics

[†] Probability of proportion being significantly different from the first value in each demographic category (Z-test)

# 5.1.3 **Profile of CAM Users in Victoria**

The state of Victoria (VIC) is considered the pioneer in CAM practitioner regulation in Australia and indeed in Western countries. For example, it is currently the only state in Australia that has introduced mandatory registration of Chinese medicine practitioners. Thus, the use of acupuncture and Chinese herbal medicine (Table 5.1) and, visiting practitioners of acupuncture and Chinese herbal medicine have been found to be more prevalent in this state (Table 5.2).

Summarised in Table 5.4 is a sub-group analysis of socio-demographic characteristics that relate to CAM users and people visits to CAM practitioners in Victoria. The income characteristics relating to both CAM users and people visits to CAM practitioners in Victoria were most similar to the national figures. That is, the group of people with higher household incomes (i.e. more than \$40,000/pa) had a much higher prevalence of CAM use (79.5%) and visits to CAM practitioners (59.3%) than low-income participants (i.e. less than \$20,000/pa) (55.2% and 34.4% respectively, p<0.01). Again, similar to the overall national prevalence, the high prevalence of visits to CAM practitioners is also significantly higher for those born in Australia and among younger adults (both, p<0.05).

	Users		Visited practitioners		
Characteristics*	% (95% CI)	$\mathbf{p}^{\dagger}$	% (95% CI)	$\mathbf{p}^{\dagger}$	
Gender					
Female	74.2 (66.9 - 81.5)		53.6 (45.3 - 62.0)		
Male	65.1 (57.0 - 73.3)	>0.05	42.1 (33.7 - 50.6)	>0.05	
Age (year)					
18-34	78.3 (69.5 - 87.1)		57.9 (47.4 - 68.5)		
35-64	68.7 (60.9 - 76.5)	>0.05	46.5 (38.1 - 54.9)	>0.05	
65+	58.4 (44.3 - 72.5)	< 0.05	36.8 (23.0 - 50.6)	< 0.05	
<b>Country of Birth</b>					
Australia	71.7 (65.5 – 77.9)		52.2 (45.2 - 59.1)		
Non-Australia	63.7 (52.3 – 75.2)	>0.05	36.4 (25.0 - 47.8)	< 0.05	
Self-reported health status					
Excellent/very good/good	70.6 (64.7 - 76.5)		48.2 (41.8 - 54.7)		
Fair/poor	65.1 (50.1 - 80.0)	>0.05	46.9 (31.2 - 62.5)	>0.05	
Post-secondary education					
No	61.9 (52.9 - 71.0)		41.7 (32.5 - 50.9)		
Yes	75.2 (68.5 - 81.9)	< 0.05	52.4 (44.6 - 60.1)	>0.05	
Employment					
Employed	73.0 (66.2 - 79.7)		49.5 (41.9 - 57.2)		
Unemployed or not in labour force	63.4 (54.0 - 72.8)	>0.05	43.6 (33.9 - 53.2)	>0.05	
Private health insurance					
Yes	71.5 (64.0 - 78.9)		49.6 (41.3 - 57.8)		
No	67.7 (59.5 – 75.8)	>0.05	45.9 (37.2 - 54.7)	>0.05	
Annual household income			. , ,		
<a\$20,000< td=""><td>55.2 (40.6 - 69.7)</td><td></td><td>34.4 (20.5 - 48.2)</td><td></td></a\$20,000<>	55.2 (40.6 - 69.7)		34.4 (20.5 - 48.2)		
A\$20,000-A\$40,000	64.8 (52.2 - 77.4)	>0.05	43.7 (30.7 - 56.8)	>0.05	
>A\$40,000	79.5 (72.6 - 86.4)	< 0.01	59.3 (50.9 - 67.6)	< 0.01	

# Table 5.4 Socio-demographic characteristics of CAM users in Victoria

Note: significance levels were adjusted to allow for multiple comparisons

* Excludes subjects who did not provide socio-demographic characteristics

Probability of proportion being significantly different from the first value in each demographic category (Z-test)

# 5.1.4 Profile of CAM Users in Queensland

Surprisingly, although the prevalence of CAM use in Queensland (QLD) was considerably high, the prevalence of visits to CAM practitioners, however, was the lowest (41.4%). This raises a safety concern that a relatively large proportion (41.8%) of CAM users in Queensland self-administered CAM products and therapies, compared with a relatively lower proportion in other states (between 26.7% to 37.5%). After investigating this proportion in the use of each of the 17 forms of CAM and visits to CAM practitioners, the larger proportions of Queensland adults who self-administered Chinese herbal medicine (76.1%) and Western herbal medicine (85.8%) may partially explain the current findings (Table 5.1 and Table 5.2).

The previously mentioned large gender differential in the prevalence of CAM and visits to CAM practitioners in Australia nationwide was found to be statistically significant only in Queensland (Table 5.5) (female users vs. male users, p<0.01). Female adults in Queensland tended to be more likely to use CAM (82.0%) than male adults (59.9%). Interestingly, such a gender difference in CAM use was not observed in any of the other states. Furthermore, the proportion of CAM users among the group of Queensland young adult (82.1%) was higher than in any other state. Also, there was least variation (67.6%, 74.5% and 75.5%, respectively) in CAM use among different income earners who resided in Queensland.

	Users		Visited practitioners		
Characteristics*	% (95% CI)	$\mathbf{p}^{\dagger}$	% (95% CI)	$\mathbf{p}^{\dagger}$	
Gender					
Female	82.0 (74.6 - 89.3)		52.1 (42.5 - 61.6)		
Male	59.9 (50.5 - 69.4)	<0.01	30.5 (21.5 - 39.4)	< 0.01	
Age (year)					
18-34	82.1 (72.9 – 91.4)		48.7 (36.6 - 60.8)		
35-64	67.7 (58.8 - 76.5)	< 0.05	40.2 (30.9 - 49.5)	>0.05	
65+	60.0 (43.4 - 76.7)	< 0.05	30.5 (14.8 - 46.1)	>0.05	
<b>Country of Birth</b>					
Australia	73.5 (66.8 - 80.2)		42.4 (34.9 - 49.9)		
Non-Australia	62.6 (47.5 - 77.8)	>0.05	38.0 (22.8 - 53.2)	>0.05	
Self-reported health status					
Excellent/very good/good	71.4 (64.6 – 78.1)		40.4 (33.1 - 47.7)		
Fair/poor	68.5 (52.5 - 84.4)	>0.05	44.4 (27.4 - 61.4)	>0.05	
Post-secondary education					
No	63.6 (54.4 - 72.7)		34.6 (25.6 - 43.7)		
Yes	80.2 (72.3 - 88.2)	<0.01	49.6 (39.6 - 59.5)	< 0.05	
Employment					
Employed	74.2 (66.9 - 81.5)		43.2 (34.9 - 51.4)		
Unemployed or not in					
labour force	67.5 (56.1 - 78.9)	>0.05	39.4 (27.5 - 51.2)	>0.05	
Private health insurance					
Yes	75.6 (67.4 - 83.8)		46.5 (36.9 - 56.1)		
No	66.4 (57.2 - 75.6)	>0.05	36.1 (26.8 - 45.4)	>0.05	
Annual household income					
<a\$20,000< td=""><td>74.5 (58.6 - 90.5)</td><td></td><td>28.7 (12.1 – 45.2)</td><td></td></a\$20,000<>	74.5 (58.6 - 90.5)		28.7 (12.1 – 45.2)		
A\$20,000-A\$40,000	67.6 (54.0 - 81.1)	>0.05	33.1 (19.5 – 46.7)	>0.05	
>A\$40,000	75.5 (67.5 - 83.4)	>0.05	51.8 (42.6 - 61.1)	< 0.05	

# Table 5.5 Socio-demographic characteristics of CAM users in Queensland

Note: significance levels were adjusted to allow for multiple comparisons

* Excludes subjects who did not provide socio-demographic characteristics

Probability of proportion being significantly different from the first value in each demographic category (Z-test)

# 5.1.5 **Profile of CAM Users in South Australia**

Overall CAM use in South Australia (SA) was considered to be the lowest in the current study (excluding Australian Capital Territory, Northern Territory and Tasmania), although this was not the case when considering visits to CAM practitioners rather than CAM use (Table 5.1 and Table 5.2). Surprisingly, a high prevalence of both homeopathy users and visits to homeopaths was observed in South Australia. The proportions of these (8.9% and 4.9%, respectively) were double those proportions in Queensland (4.2% of use and 1.5% of practitioner visits respectively). Moreover, although the use of energy healing in South Australia was the lowest (5.3%), the prevalence of visits to energy healing practitioners in South Australia, however, was the highest (4.2%).

In contrast, apart from the previously mentioned low rate of use of acupuncture, Chinese herbal medicine, Chinese therapeutic massage and Western herbal medicine in South Australia, the lowest rates of naturopathy use (5.5%) and visits to naturopaths were also found in South Australia (1.3%), being about three times less than any other state except Queensland (p<0.05).

Less significant findings were found in the characteristics of CAM users in South Australia. In regard to visits to CAM practitioners, the largest variability in participant characteristics was private health insurance and household incomes. The variations were considerably higher than for any other state of Australia. For example, South Australian adults with private health insurance were two times more likely to visit a CAM practitioner than those South Australians without private health insurance (52.0% vs. 20.5%, p<0.01, Table 5.6). Finally, in contrast to the considerably higher prevalence of CAM use among females in other states (generally more than 70%), the lowest rate of CAM use by females was in South Australia (62.5%).

164

	Users		Visited practitioners		
Characteristics*	% (95% CI)	$\mathbf{p}^{\dagger}$	% (95% CI)	$\mathbf{p}^{\dagger}$	
Gender					
Female	62.5 (47.9 - 77.1)		43.9 (29.0 - 58.9)		
Male	59.0 (43.9 - 74.2)	>0.05	45.2 (29.9 - 60.5)	>0.05	
Age (year)					
18-34	66.9 (48.0 - 85.8)		45.2 (25.2 - 65.2)		
35-64	62.5 (48.0 - 77.0)	>0.05	49.7 (34.7 - 64.7)	>0.05	
65+	47.4 (23.1 - 71.7)	>0.05	30.2 (7.8 - 52.5)	>0.05	
<b>Country of Birth</b>					
Australia	60.1 (48.0 - 72.3)		44.0 (31.7 - 56.3)		
Non-Australia	58.7 (36.1 - 81.4)	>0.05	40.4 (17.8 - 62.9)	>0.05	
Self-reported health status					
Excellent/very good/good	60.3 (48.7 - 71.9)		44.4 (32.6 - 56.2)		
Fair/poor	63.3 (38.4 - 88.1)	>0.05	45.3 (19.6 - 71.0)	>0.05	
Post-secondary education					
No	54.3 (39.5 - 69.1)		38.0 (23.5 - 52.4)		
Yes	68.0 (53.4 - 82.5)	>0.05	51.8 (36.2 - 67.4)	>0.05	
Employment					
Employed	66.9 (53.7 - 80.1)		48.4 (34.4 - 62.5)		
Unemployed or not in					
labour force	52.1 (35.4 - 68.8)	>0.05	39.0 (22.7 - 55.4)	>0.05	
Private health insurance					
Yes	63.0 (51.0 - 74.9)		52.0 (39.7 - 64.4)		
No	53.9 (31.9 - 75.9)	>0.05	20.5 (2.7 - 38.4)	< 0.01	
Annual household income					
<a\$20,000< td=""><td>57.5 (25.1 - 89.8)</td><td></td><td>15.6 #</td><td></td></a\$20,000<>	57.5 (25.1 - 89.8)		15.6 #		
A\$20,000-A\$40,000	59.1 (36.5 - 81.7)	>0.05	51.3 (28.4 - 74.3)	< 0.05	
>A\$40,000	61.2 (46.5 - 76.0)	>0.05	46.1 (31.0 - 61.2)	< 0.05	

# Table 5.6 Socio-demographic characteristics of CAM users in South Australia

Note: significance levels were adjusted to allow for multiple comparisons

* Excludes subjects who did not provide socio-demographic characteristics

t Probability of proportion being significantly different from the first value in each demographic category (Z-test)

Too few responses to calculate the CI #
# 5.1.6 Profile of CAM Users in Western Australia

In contrast to the less significant findings among CAM users in South Australia, CAM users in Western Australia appeared to have a higher correlation with their socio-demographic characteristics. Most significantly, Western Australia has the highest CAM prevalence (83.2%) among people with post-secondary education than any other state. In addition, when comparing those with post-secondary education and those without such education, Western Australia is the only state in which significant differences were observed both in CAM users and in people visits to CAM practitioners (p<0.01 for CAM users and p<0.05 for people visits to CAM practitioners (p<0.01 for CAM users and p<0.05 for people visits to CAM practitioners). Also, Western Australia is the only state in which statistically significant differences were observed in both the prevalence of CAM use and visits to CAM practitioners (p<0.01), with respect to users' private health insurance coverage status (Table 5.7). In Western Australia, a large variation was also found in the country of birth of people visits to CAM practitioners, that is, people born in Australia were more likely to visit CAM practitioners (51.1%) than those born overseas (21.3%, p<0.01). This finding is similar to CAM users in Victoria.

Characteristics*	Users		Visited practitioners	
Characteristics*	% (95% CI)	$\mathbf{p}^{\dagger}$	% (95% CI)	$\mathbf{p}^{\dagger}$
Gender				
Female	74.2 (62.4 - 86.0)		51.0 (37.5 - 64.5)	
Male	58.9 (45.6 - 72.2)	>0.05	33.0 (20.3 - 45.7)	>0.05
Age (year)				
18-34	70.2 (54.7 - 85.7)		38.9 (22.4 - 55.4)	
35-64	69.6 (57.5 - 81.7)	>0.05	49.0 (35.8 - 62.1)	>0.05
65+	48.9 (24.7 - 73.1)	>0.05	25.0 (4.1 - 45.9)	>0.05
<b>Country of Birth</b>				
Australia	70.0 (59.5 - 80.5)		51.1 (39.7 - 62.6)	
Non-Australia	58.8 (41.8 - 75.8)	>0.05	21.3 (7.1 - 35.5)	< 0.01
Self-reported health status				
Excellent/very good/good	66.4 (56.7 – 76.1)		42.8 (32.6 - 53.0)	
Fair/poor	67.8 (43.5 - 92.1)	>0.05	37.1 (12.0 - 62.3)	>0.05
Post-secondary education				
No	46.8 (32.7 - 60.9)		29.6 (16.7 - 42.5)	
Yes	83.2 (73.6 - 92.9)	<0.01	52.5 (39.6 - 65.5)	< 0.05
Employment				
Employed	71.4 (61.1 – 81.7)		45.7 (34.3 – 57.1)	
Unemployed or not in				
labour force	56.0 (38.1 - 73.8)	>0.05	32.5 (15.6 - 49.3)	>0.05
Private health insurance				
Yes	75.6 (65.3 - 85.8)		51.1 (39.2 - 63.0)	
No	50.4 (34.5 - 66.4)	<0.01	25.7 (11.8 - 39.7)	< 0.01
Annual household income				
<a\$20,000< td=""><td>74.1 (52.6 – 95.7)</td><td></td><td>35.9 (12.3 – 59.5)</td><td></td></a\$20,000<>	74.1 (52.6 – 95.7)		35.9 (12.3 – 59.5)	
A\$20,000-A\$40,000	51.9 (31.1 – 72.7)	>0.05	34.7 (14.9 - 54.5)	>0.05
>A\$40,000	66.7 (54.6 - 78.9)	>0.05	41.1 (28.4 - 53.8)	>0.05

T-11. 5 7	C J.	l l	<b>4 1</b> - <b>4 6</b>	CAN	<b>!</b>	<b>XX</b> 74	A 4 12
1 able 5./	Socio-aem	ographic char	acteristics of	CAM	users in	western	Australia

Note: significance levels were adjusted to allow for multiple comparisons

* Excludes subjects who did not provide socio-demographic characteristics

Probability of proportion being significantly different from the first value in each demographic category (Z-test)

# 5.1.7 **Profile of CAM Users in Other States/Territories**

Apart from the five major states discussed above, data on CAM and visits to CAM practitioners were also collected from the state of Tasmania (TAS), the Northern Territory (NT) and the Australian Capital Territory (ACT). Sixty-one interviews were conducted in these regions (26 each in Tasmania and in the Australian Capital Territory, and nine in the Northern Territory). Thus, combined results were presented. It is important to note that when examining the prevalence of specific use between different characteristics, the sample size is too small to calculate a 95% confidence interval in several items (Table 5.8).

In general, approximately half (49.6%, 95%CI: 34.8% to 64.4%) of the adult populations of the above three regions combined, were CAM users, while less than one in three (29.4%, 95%CI: 15.9% to 42.9%) adults had visited CAM practitioners. Specifically, CAM use was higher in Northern Territory, being about 71.1% (95%CI: 34.0% to 108.1%), with 38.3% (95%CI: -1.4% to 78.1%) visiting CAM practitioners.

The prevalence of CAM use in Tasmania was lower than the national average – about 52.3% (95%CI: 32.1% to 72.5%), with 29.7% (95%CI: 11.2 % to 48.2%) visiting CAM practitioners. Lastly, in Australian Capital Territory the prevalence of CAM use was about 36.9% (95%CI: 12.3% to 61.6%), with 25.5% (95%CI: 3.2% to 47.8%) visiting CAM practitioners. Again, as noted in the Methodology section, the total number of interviews in these states/territories is not adequate to determine conclusive prevalence of CAM use.

	Users		Visited practitioners	
Characteristics*	% (95% CI)	$\mathbf{p}^{\dagger}$	% (95% CI)	$\mathbf{p}^{\dagger}$
Gender				
Female	60.4 (38.5 - 82.3)		39.0 (17.2 - 60.9)	
Male	41.3 (21.9 - 60.7)	>0.05	22.0 (5.7 - 38.3)	>0.05
Age (year)				
18-34	50.4 (24.1 - 76.6)		38.3 (12.7 - 63.8)	
35-64	62.7 (42.8 - 82.6)	>0.05	31.0 (11.9 - 50.0)	>0.05
65+	8.1 (-11.6 – 27.8)	< 0.05	8.1#	>0.05
<b>Country of Birth</b>				
Australia	49.0 (32.7 - 65.4)		26.9 (12.4 - 41.4)	
Non-Australia	57.8 (21.7 - 94.0)	>0.05	45.2 (8.8 - 81.6)	>0.05
Self-reported health status				
Excellent/very good/good	48.1 (32.1 - 64.0)		31.5 (16.7 – 46.4)	
Fair/poor	59.2 (20.2 - 98.1)	>0.05	16.4 [#]	>0.05
Post-secondary education				
No	52 (27.1 – 76.9)		30.2 (7.3 - 53.1)	
Yes	49.7 (31 - 68.3)	>0.05	29.8 (12.8 - 46.9)	>0.05
Employment				
Employed	59.0 (40.1 - 78.0)		32.5 (14.4 - 50.5)	
Unemployed or not in				
labour force	37.7 (14.8 - 60.6)	>0.05	26.2 (5.4 - 46.9)	>0.05
Private health insurance				
Yes	55.1 (35.1 – 75.1)		34.0 (15.0 – 53.0)	
No	43.1 (21.5 - 64.8)	>0.05	24.0 (5.4 - 42.7)	>0.05
Annual household income				
<a\$20,000< td=""><td>42.3 (10.1 – 74.4)</td><td></td><td>0#</td><td></td></a\$20,000<>	42.3 (10.1 – 74.4)		0#	
A\$20,000-A\$40,000	50.4 (12.2 - 88.6)	>0.05	31.5#	>0.05
>A\$40,000	53.3 (32.8 – 73.7)	>0.05	38.1 (18.2 – 58.0)	< 0.01

note:

significance levels were adjusted to allow for multiple comparisons Excludes subjects who did not provide socio-demographic characteristics *

t Probability of proportion being significantly different from the first value in each demographic category (Z-test)

Too few responses to calculate the CI #

# 5.1.8 Geographical Summary of CAM Use in Australia

The prevalence of CAM use and visits to CAM practitioners in states and territories of Australia can be geographically summarised in the map below

(Figure 5.1).



Figure 5.1 Prevalence of CAM use in Australia presented in a map

# 5.2 Discussion

Anecdotally, the practice of some forms of CAM, such as acupuncture and herbal medicine, are more common in a number of states in Australia. To a certain degree, the suggestions that people who live in New South Wales and Victoria are more likely to use CAM and visit CAM practitioners than any other states, has been confirmed in the present study. Thus, this study observed that South Australia has a much lower usage of CAM, compared to other states in Australia. Logically, the proportion (60.8%) of South Australian residents who used CAM seems consistent with previous estimations.¹¹ In addition, the proportion (72.1%) of New South Wales adults who used CAM is also consistent with previous New South Wales

As noted in Table 5.1 and Table 5.2, regional differences in the use of certain forms of CAM are also in accord with the existing literature. For example, a previous workforce survey identified a relatively small number of Western herbal medicine and naturopathy practitioners in South Australia, compared to New South Wales, Victoria, Queensland and Western Australia.¹² Consistently, based on the current survey, the lowest prevalence of both Western herbal medicine use and naturopathy use were estimated in South Australia.

Homeopathy is the only CAM modality for which the highest prevalence was observed in South Australia. A detailed rationale for the high usage is not clear. However, incidentally, there is a view that the high use of homeopathy in South Australia actually relates to the history of Germany immigration to that state. The lowest prevalence of homeopathy use was reported in Queensland. However, this seems to be in contrast with the general trend observed in the current survey that Queensland residents favoured the use of most forms of CAM. So there is an aspect of interpreting intra-Australian differences which requires further work around the social, cultural and historical developments in different states (just as these factors influence differences across countries).

At the time of conducting the survey, Victoria was still the sole Australian state in which acupuncturists and Chinese herbal medicine practitioners were subject to statutory registration. Thus, it is not surprising to see that the prevalence of acupuncture and Chinese herbal medicine use in Victoria was amongst the highest. In particular, the proportion of Victorian residents who had visited acupuncturists was three to four times higher than in South Australia and Western Australia. This implied that statutory registration could impact greatly on the public's confidence in using Chinese medicine, thus impacting on the proportion of residents from this region seeking Chinese medicine care. The general opinion of the impact on public confidence in Chinese medicine practitioner mandatory registration was also sought, and is presented in 7.1.5.2.

In Western Australia the prevalence of each individual form of CAM was generally lower than those for other states. However, the proportion of Western Australia adults who used energy healing such as reiki, and the proportion of Western Australia adults who had visited practitioners for energy healing were the highest among all states. In addition, the proportion of Western Australia adults who used meditation was higher than the prevalence in New South Wales and South Australia, but similar to the prevalence in Victoria and Queensland.

A large regional variation of CAM use was found in the prevalence of aromatherapy. The extraordinarily high prevalence in Victoria may indicate a greater aromatherapy market there than any other states. In all, there is evidence to conclude that there are regional differences in the use of certain forms of CAM in Australia. However, factors that may contribute to such differences cannot be answered by the current survey.

# CHAPTER 6. USE OF FOUR SPECIFIC FORMS OF CAM: ACUPUNCTURE, CHINESE HERBAL MEDICINE, CHIROPRACTIC AND OSTEOPATHY

As part of the study, specific questions were designed to investigate the usage and related issues on four specific forms of CAM. These are acupuncture, Chinese herbal medicine, chiropractic and osteopathy. The methodology has been described in Chapter 3.3. This Chapter provides results on these therapies and discusses implications of the findings.

# 6.1 Results

# 6.1.1 Chinese Medicine (Acupuncture and Chinese Herbal Medicine)

#### 6.1.1.1 Overview

In this section, data on overall Chinese medicine utilisation and detailed information on specific forms of Chinese medicine, for example, acupuncture and Chinese herbal medicine are presented. The definition of Chinese medicine user (CM user) refers to a person who uses any one of the five major forms of Chinese medicine: acupuncture (ACU), Chinese herbal medicine (CHM), Chinese therapeutic massage (CTM), Chinese medicine dietary therapy (CMDT), and Qigong, martial art and Tai Chi (QGMATC). This definition also applies to those who had visited any one of the five types of Chinese medicine practitioners.

Additional details on acupuncture and Chinese herbal medicine in relation to economic consideration, the rationale for use and the perceived and actual benefits and adverse events are also presented in this Chapter. Results on regulation and health insurance related to Chinese medicine are presented in relevant sections of Chapter 7.

# 6.1.1.2 Total and Specific Prevalence of Chinese Medicine Use

Overall, a total of 210 participants (weighted percentage, 19.3%, 95%CI: 16.9% – 21.7%) reported using Chinese medicine (CM) in the 12-month preceding the survey. As expected, the most common CM modality was acupuncture (9.2%, 95%CI: 7.5% - 10.9%). The use of other CM modalities were: Chinese herbal medicine (7.0%, 95%CI: 5.5% - 8.5%), Chinese therapeutic massage (5.1%, 95%CI: 3.8% - 6.4%), Chinese medicine dietary therapy (2.3%, 95%CI: 1.4% - 3.2%) and Qigong, martial art and Tai Chi (6.0%, 95%CI: 4.6% - 7.4%). There were two participants (both female from Australia and from New Zealand) who had used all these five forms of CM. A total of 16 participants (7.6% of all CM users) had used all three major forms of CM (i.e. Acupuncture, Chinese herbal medicine and Chinese herbal medicine). Among those who had used Chinese medicine dietary therapy, five out of 28 (17.9%) had used Chinese medicine dietary therapy only but had not used other forms of CM (see Table 4.4).

A total of 131 participants (11.9%, 95%CI: 10.0% to 13.9%) had visited at least one type of CM practitioner. Most acupuncture users (81.1%) had visited an acupuncturist, while the majority (75.2%) of Chinese medicine dietary therapy users did not visit a practitioner for the therapy. Among all survey participants, a total of 7.5% (95%CI: 5.9% to 9.0%) had visited an acupuncturist, while the prevalence of visits to CHM, Chinese therapeutic massage, Chinese medicine dietary therapy and Qigong, martial art and Tai Chi practitioners was 2.3% (95%CI: 1.4% - 3.2%), 3.2% (95%CI: 2.1% - 4.3%), 3.2% (95%CI: 2.1% - 4.2%) and 0.6% (95%CI: 0.1% - 1.0%), respectively (see Table 4.8).

For both prevalence of CM use and visits to CM practitioners, there were some large differences between states (Table 6.1 and Figure 6.1). The lowest prevalence was always observed in Western Australia (Table 6.1), whereas New South Wales, Queensland and Victoria appeared to have a greater prevalence of CM use and visits to CM practitioners. It is worth-noting that although 17.2% of South Australia adults had used CM, only 6.5% had visited CM practitioners. Specifically, these significant differences in visiting CM practitioners were also found in visits to both acupuncturists and Chinese herbal medicine practitioners (p<0.05, see Table 5.1 and Table 5.2).

States	Used Chine in past 12	Statistical difference		
	Yes (%)	No (%)	Chi Square	p-value
New South Wales	17.1	82.9	13.276	0.010
Victoria	22.7	77.3		
Queensland	25.1	74.9		
South Australia	17.2	82.8		
Western Australia	10.3	89.7		

Table 6.1 Regional differences of Chinese medicine use and visits to Chinese medicine practitioners

States	Visited Chine practitioners in	Statistical difference		
	Yes (%)	No (%)	Chi Square	p-value
New South Wales	11.7	88.3	9.245	0.055
Victoria	15.4	84.6		
Queensland	13.5	86.5		
South Australia	6.5	93.5		
Western Australia	6.0	94.0		



Figure 6.1 Prevalence of Chinese medicine use in Australia presented in a map

# 6.1.1.3 Frequency of Practitioner Visits and Expenditure

As noted in Table 4.9, the most frequently visited CAM practitioners were acupuncturists, with more than one visit every six weeks (8.75 visits per annum). Together with Chinese herbal medicine (CHM, 5.98 visits per annum) and Chinese therapeutic massage (5.43 visits per annum), a total of 105 people (9.5% of total survey participants) had visited practitioners for any one of three forms of Chinese medicine. This can be transformed into a total of 14.94 million visits for the adult Australian population (10.16 million, 2.12 million and 2.66 million for acupuncturists, CHM practitioners and Chinese therapeutic massage practitioners, respectively).

The average out-of-pocket expenditure of visiting a CAM practitioner, as estimated in the current survey was A\$25 (see Chapter 4.1.7.1). When extrapolating this to the estimated national visits to Chinese medicine practitioners, it suggests that a total of A\$373.5 million was paid out-of-pocket to Chinese medicine practitioners per annum. Of this, A\$254 million was spent on visiting acupuncturists, A\$53 million was spent on visiting CHM practitioners and, the remaining A\$66.5 million was on Chinese therapeutic massage therapists.

Apart from the costs related to practitioner consultation and treatment, both acupuncture and CHM users had also estimated additional costs related to their treatment. And for CHM, users had separated the costs of herbal medicine and/or other items related to CHM, such as books or equipment. A total of 17 acupuncture users spent additional costs related to acupuncture. The estimated cost from one user is considered as outlier (\$6,000), applying the same protocol in calculating CAM costs (see Chapter 4.1.7.1), thus, among those had spent additional costs related to acupuncture, the adjusted median cost was A\$88.14. Nationwide, this equals to A\$21.80 million acupuncture-related costs.

Of the total CHM users who had spent additional costs to the actual Chinese herbal medicine (71 users, or 6.4% of total participants) and CHM-related products (22 users, or 2.2% of total participants), the median cost was A\$100 and A\$60, respectively, excluding extreme values (A\$4,500 or above, two were herbs and two were CHM-related items). Thus, the national expenditure can be estimated as A\$103.11 million for Chinese herbs and A\$21.83 million for CHM related costs (see Appendix G9 for the original cost and G10 for a summary statistics of the adjusted data).

# 6.1.1.4 Acupuncture Treatment Provided by Medical Doctors

Given that there is evidence that a considerable proportion of registered medical doctors (GPs) is providing acupuncture treatment to their patients, acupuncture users were asked whether at least one of the practitioners they visited for acupuncture treatments was a medical doctor, such as a GP or a medical specialist. Of the 82 participants who had visited a practitioner for acupuncture treatment, well over one quarter (29.1%) had received acupuncture treatment from their GPs. Of these, over half (59.1%) had received acupuncture from their GPs only but not from non-GP acupuncturists.

In terms of the frequency of visits to acupuncture practitioners, either to GPs or to traditional acupuncturists, participants provided estimations for the 12-month preceding the survey. Of the total of 24 adults who had received acupuncture from their GPs, a total of 9.3 visits were estimated for the 12-month period, with a median of 4.0 visits per year (two participants mentioned they visited their GPs once a week for a spinal injury and arthritis). On the other hand, 67 participants had received acupuncture treatment from non-GP acupuncturists, including nine persons who had received treatment from both GPs and non-GP providers. Among these (67 participants), the average number of visits during the 12-month period was 7.44, and the median was 4.0. The two highest number of visits were 60 and 45, which were estimated by a patient with broken spine and by another patient with chronic fatigue syndrome.

When combining all visits to GPs and non-GP acupuncturists (considered as visiting a practitioner for acupuncture), the mean number of visits per year was 8.75, with a median of 4.0. The mean number of visits was used to estimate the national total number of visits for acupuncture treatment in Australia (see Table 4.9).

# 6.1.1.5 Characteristics of Chinese Medicine Users

The common characteristics for Chinese medicine (CM) users are consistent with most previous findings on CAM users. In general, relatively young, with a secondary education and employed contributed significantly to a relatively high prevalence of CM use (p<0.05). Furthermore, the most significant factor related to the use of CM and visits to CM practitioners was educational background. Compared to those without post-secondary education, people having a post-secondary education appeared to be more likely to use CM (22.4% vs. 15.3%) and to visit CM practitioners (14.8% vs. 8.1%) (p<0.01) (Table 6.2).

For each CM modality (see Table 4.6), the demographic statistics related to individual prevalence were different. The above noted marked educational difference in CM use was also found in the use of acupuncture and Chinese herbal medicine (any of the modalities, p<0.01), but not in other forms of CM. Surprisingly, Australian-born adults were more likely to use acupuncture (10.2%) than those born overseas (6.5%, p<0.05), but less likely to use CHM (5.5% vs. 11.1%, p<0.05) and Chinese medicine dietary therapy (1.7% vs. 4.7%, p<0.05).

A low prevalence of CHM use was also observed in those aged 65 or above (2.6% compared to 8.1% among those aged 18-34, p<0.05) and those unemployed (4.6% compared to 8.1% among employed participants, p<0.05) (Table 4.6). Moreover, for Chinese therapeutic massage, the only significant difference in use was found between younger adults (aged 18-34 5.7%) and older adults (65 and older, 2.4%, p<0.05). Interestingly, 28 participants (2.8% of total participants) had used Chinese medicine dietary therapy, all of whom were from the group of people who rated their personal health as good or better; whereas, no participants who rated themselves in poor health had used Chinese medicine dietary therapy (p<0.01). Not

unexpectedly, younger people were more likely to practice Qigong, martial art and Tai Chi (9.5%) than the middle-age group (4.7%, p<0.01) and older people (2.9%, p<0.01).

Characteristic [#]	CM Us	ers	Visited CM practitioner	
	% (SE %)	$\mathbf{p}^{\dagger}$	% (SE %)	$\mathbf{p}^{\dagger}$
Gender				
Female	21.1 (1.8)		11.8 (1.4)	
Male	17.5 (1.7)	0.1270	12.0 (1.4)	0.9447
Age (year)				
18-34	21.8 (2.3)		13.4 (1.9)	
35-64	19.3 (1.7)	0.3718	11.7 (1.4)	0.4485
65+	14.8 (2.6)	0.0451*	10.2 (2.2)	0.2623
Country of Birth				
Australia	19.2 (1.4)		12.9 (1.2)	
Non-Australia	19.3 (2.5)	0.9566	9.2 (1.8)	0.0960
Post-secondary education				
No	15.3 (1.7)		8.1 (1.3)	
Yes	22.4 (1.7)	0.0032*	14.8 (1.4)	0.0005*
Self-reported health status				
Excellent/very good/good	18.9 (1.3)		11.6 (1.1)	
Fair/poor	21.1 (3.2)	0.5306	12.9 (2.6)	0.6538
Employment				
Employed	20.9 (1.6)		12.8 (1.3)	
Unemployed or not in labour force	15.9 (1.9)	0.0417*	9.8 (1.6)	0.1298
Private Health Insurance				
Yes	19.5 (1.6)		13.3 (1.4)	
No	19.4 (1.8)	0.9574	10.3 (1.4)	0.1389
Annual household income				_
<a\$20,000< td=""><td>18.3 (3.1)</td><td></td><td>9.5 (2.3)</td><td></td></a\$20,000<>	18.3 (3.1)		9.5 (2.3)	
A\$20,000-A\$40,000	18.2 (2.7)	0.9902	10.0 (2.1)	0.8725
>A\$40,000	19.9 (1.7)	0.6455	12.9 (1.4)	0.2026

Table 6.2 Socio-demographic characteristics of users of Chinese medicine

# Excludes subjects who did not provide socio-demographic characteristics

 Probability of proportion being significantly different from the first value in each demographic category (Z-test)

* Significance level (p<0.05)

# 6.1.1.6 Chinese Medicine Utilisation–Regression Analyses

The methodology of conducting multivariate logistic regression analyses was detailed in Chapter 3.6.4.3. With an interest to produce a model that best describe the characteristics of Chinese medicine (CM) users. Separate regression models were developed. These are on overall CM use (of any one of the five modalities), on acupuncture use and on Chinese herbal medicine use. Again, unless otherwise indicated, the regression results presented are unweighted.

Variables that significantly (p<0.05) related to CM or acupuncture or CHM use were summarised in Table 6.3. These were the variables that had been entered into each individual regression model. The best-fit significant results of the multivariate analysis are presented in Table 6.4. They show post-secondary education, had consulted a GP in the preceding 12-month and coverage of CAM influenced the private insurance purchase contributed significantly to a participant's decision to use acupuncture. In addition, for the use of CHM, the factors pertaining to a participant's age ranges and country of birth were also significant predictors.

Considering the overall use of CM, similar to the findings on acupuncture users, the factors of educational background and the belief that coverage of CAM influenced the private insurance purchase were statistically significant. It is also worth noting that being resident in different regions in Australia may also impact significantly on the possibility of CM use. As shown in Table 6.4, when using Western Australia as the reference group, the Victoria adults and the Queensland adults were respectively 2.5 and 3.0 times more likely to use CM than adults in Western Australia.

Table 6.3 Predictor	variables employed in the regression models on Chinese medicine
use	

Predictors	ACU	CHM	СМ
Demographic factors (block 1)			
Age group		Yes	
Residential area			Yes
Education	Yes	Yes	Yes
Employment		Yes	Yes
Country of birth		Yes	
Health factors (block 2)			
Visited a GP in past 12 months	Yes		
Visited a GP 4 or more times in past 12 months			
External factors (block 3)			
Coverage of CAM influences health insurance purchase	Yes	Yes	Yes
CAM should be regulated as rigorously as Western medicine			Yes
Health conditions (block 4) Visited a GP for 18 specific health problems			
Condition No.1: health check-up	Yes		
Condition No.8: diabetes			Yes
Condition No.9: gastrointestinal	Yes		
Condition No.11: high blood pressure		Yes	
Condition No.18: trauma/injury	Yes		Yes

ACU: acupuncture, CHM: Chinese herbal medicine, CM: Chinese medicine

Predictors		Use of acupuncture Odds ratio & (95% CI)	Use of CHM Odds ratio & (95% CI)	Use of CM Odds ratio & (95% CI)
	No	1.0		1.0
Post-secondary education	Yes	1.678 (1.044-2.697)*		1.485 (1.02–2.146)*
	18-34		3.426 (1.282–9.152)*	
	35-64		2.685 (1.037-6.956)*	
Age	65+		1.0	
	Overseas		1.0	
Country of birth	Australia		0.434 (0.260-0.723)**	
	New South Wales			1.691 (0.787–3.633)
	Victoria			2.506 (1.161-5.410)*
	Queensland			3.012 (1.367-6.637)**
	South Australia			1.768 (0.689–4.538)
Region	Western Australia			1.0
Congultad a CD in the past 12	No	1.0		
months	Yes	2.064 (1.010-4.217)*		
Coverage of CAM influences	No	1.0		1.0
health insurance purchase	Yes	2.201 (1.366-3.546)**		2.115 (1.463-3.057)**
Omnibus Tests of Model Coefficie	ents			
Chi square (degree of freedom, p value)		23.044 (3, <0.0001)	16.539 (3, =0.001)	35.059 (6, <0.0001)
Percentage accurate in classific	ation	90.7	92.5	79.4

* p<0.05; ** p<0.01; CI: confidence interval; CM: Chinese medicine; CHM: Chinese herbal medicine

# 6.1.1.7 Rationale for Use of Acupuncture and Chinese Herbal Medicine

#### Acupuncture

A total of 97 acupuncture users responded to the question on rationale for use. The vast majority (92.0%) said it was for the purpose of treating medical conditions, including 14.1% who used it for both general health improvement and medical conditions. The remaining 8.0% used acupuncture for improving general health and well-being only. Among those who used acupuncture for a medical condition, the most common conditions were back pain and related problems (20.7%), shoulder pain and problems (15.5%), arthritis (8.5%), injury (7.0%) and knee-related problems (5.1%). The detailed medical conditions related to the use of acupuncture are presented in Table 6.5. Specifically and significantly, the majority of those who used acupuncture for treating medical conditions were aged 35-64 (97.2%) and 65 and older (92.0%), while only 80.6% users aged 18-34 used acupuncture for this purpose ( $\chi^2$ =6.258, p=0.044).

	Number		Number
Medical condition	of cases	Medical condition	of cases
Musculoskeletal related:		Chronic fatigue syndrome	2
Back pain and related	18	Gastrointestinal	2
Non-specific muscular problems	17	Stop smoking	2
Shoulder related	12	Anxiety and depression	1
Arthritis	8	Blood cholesterol	1
Injury-related	6	Energy-related problems	1
Knee-related	5	Gynaecologic problems	1
Neck-related	3	Hay fever	1
Spinal-related	3	Infertility	1
Ankle-related	1	Insomnia	1
Multiple sclerosis	1	Irregular heart beat	1
Other conditions:		Loss weight	1
Headache	3	Lung related	1
Sinus problem	3	Pregnancy	1
Asthma	2	Prostate problem	1

Table 6.5 Medical conditions related to the use of acupuncture treatment

A total of 73 CHM users responded to the question on rationale for use. About two thirds (60.4%) used it to treat medical conditions, including 29.7% who used it both for general health improvement and medical conditions. Thus, over 39.6% used CHM to improve general health and well-being only. Among those who used CHM for a medical condition, the most common conditions were colds and flu (11.6%), and problems related to energy level (8.5%). There were at least 30 different conditions for which people had used CHM (Table 6.6). Specifically, the highest prevalence of CHM use for medical conditions was among those aged 35-64 (72.0%), followed by those aged 18-34 (51.0%), while only 14.8% of those aged 65 and older used CHM for treating medical conditions ( $\chi^2$ =7.292, p<0.026). In addition, those having private health insurance were also more likely to use CHM for medical conditions (73.0%) than those without it (41.6%,  $\chi^2$ =6.935, p=0.008).

Medical condition	Number of cases	Medical condition	Number of cases
Colds/flu	6	After hysterectomy	1
Energy related problems	4	Blood cholesterol	1
Asthma	3	Chronic fatigue syndrome	1
Anxiety and depression	3	Diabetes	1
Back pain and related	3	Heart-related	1
Hay fever	3	High blood pressure	1
Gastrointestinal	3	Infertility	1
Arthritis	2	Injury related	1
Cancer	2	Leukaemia	1
Eye problems	2	Low immune system	1
Gynaecologic problems	2	Nervous condition	1
Neck-related	2	Lung-related	1
Non-specific muscular problems	2	Toothache	1
Headache	2	Sinus problem	1
Shoulder related problems	2	Skin problem	1
Ankle related problems	1	Stress relief	1

Table 6.6 Medical conditions related to the use of Chinese herbal medicine treatment

## 6.1.1.8 Outcomes of Use of Acupuncture and Chinese Herbal Medicine

In an effort to compare expectation of acupuncture or CHM users with the actual outcomes of treatment, users were asked to respond to a question of whether the treatment they used had cured the disease/solved the problem, relieved the symptoms and/or improved general health and well-being.

#### Expectations of and Actual Outcome of Using Acupuncture

Table 6.7 summarised the responses (number and percentage) from acupuncture users on their expectations and outcomes. Nearly half (48.5%) of users expected acupuncture could cure their disease. After receiving treatment, over half (59.2%) of these users considered acupuncture had cured their disease or had solved the health problem they had before the treatment. In addition, seven out of 30 (23.3%) users who did not consider acupuncture would cure their disease found acupuncture did provide such an outcome. A further six users who had been unsure whether acupuncture would cure their disease/solve their problem had also experienced a positive outcome, thus, the total positive outcome of curing diseases or solving health problems was 41.6%.

More users considered the outcome of acupuncture would be a relief of symptoms (72.3%) or that it would improve general health and well-being (71.3%) than cure the diseases (48.5%). Among those who considered acupuncture would relieve their symptoms of illnesses, the majority (89.0%) had experienced a positive outcome. A high proportion of users who did not believe the outcome would be a relief of symptoms, or who were uncertain about this outcome, also reported a positive outcome (90.0% and 77.8%, respectively). This represented a total positive outcome of 87.1% in the use of acupuncture to relieve symptoms.

187

In all, nearly four out of five (78.2%) users considered acupuncture likely to improve their general health and well-being. The percentage was higher among those who had a positive expectation of using acupuncture to improve their general health (i.e. 86.1%). When taking no consideration of users' expectations towards acupuncture before treatment, about one-third (33.7%) users considered that acupuncture had helped them achieve all three outcomes. However, six acupuncture users responded that there was no effect after treatment. Demographic information of none of these six users had been found to be particular noticeable, although these users resided in four different states other than Victoria.

Before Treatment (expectation)		After t	After treatment (outcome)			
		Yes n (%)*	No n (%)*	Not sure n (%)*		
Cure the disease/solve the problem, n (% of users)**						
Yes	49 (48.5%)	29 (59.2%)	18 (36.7%)	2 (4.1%)		
No	30 (29.7%)	7 (23.3%)	23 (76.7%)	0 (0%)		
Not sure	22 (21.8%)	6 (27.3%)	10 (45.5%)	6 (27.3%)		
Total	101	42 (41.6%)	51 (50.5%)	8 (7.9%)		
Relief of symptoms, n (% of users)**						
Yes	73 (72.3%)	65 (89.0%)	7 (9.6%)	1 (1.4%)		
No	10 (9.9%)	9 (90.0%)	1 (10.0%)	0 (0%)		
Not sure	18 9 (17.8%)	14 (77.8%)	1 (5.6%)	3 (16.7%)		
Total	101	88 (87.1%)	9 (8.9%)	4 (4.0%)		
Improve general health and well-being, n (% of users)**						
Yes	72 (71.3%)	62 (86.1%)	7 (9.7%)	3 (4.2%)		
No	16 (15.8%)	8 (50.0%)	8 (5.0%)	0 (0%)		
Not sure	13 (12.9%)	9 (69.2%)	3 (23.1%)	1 (7.7%)		
Total	101	79 (78.2%)	18 (17.8%)	4 (7.7%)		

 Table 6.7 Expectations of and actual outcomes of using acupuncture

* Percentages are of users within each row category

** Percentages are of total acupuncture users

## Expectation of and Actual Outcome of Using Chinese Herbal Medicine

Table 6.8 summarises the responses of CHM users on their expectations and outcome of using CHM. Similar to the expectation of acupuncture curing disease, just over half (53.2%) of users expected CHM would cure their disease. After receiving treatment, over two thirds (70.7%) of these users considered CHM had cured their disease or had solved the health problem they had before treatment. In addition, nearly one in five (16.7%) users who did not consider CHM would cure their disease had found CHM did provide such an outcome. A further eight users who were unsure whether CHM would cure their disease/solve their health problem had experienced a positive outcome, thus, the total positive outcome of curing diseases or solving problems was 51.9%.

More users considered the outcome of CHM would be a relief of symptoms (79.2%) or would improve general health and well-being (84.4%) than cure the diseases (53.2%). Among those who considered CHM would relieve their symptoms of illnesses, the majority (90.2%) experienced a positive outcome. A high proportion of users who had not believed a relief of symptoms was possible or who had been uncertain about such an outcome also reported a positive outcome (75.0% and 66.7%, respectively). This represented a total positive outcome of 85.7% towards relieving symptoms by CHM.

In all, nearly nine out of ten (89.6%) users considered CHM was favourable in improving their general health and well-being. The percentage was higher among those who had a positive expectation of using CHM to improve their general health (i.e. all but two users, or 96.9%). When taking no notice of users' expectations towards CHM before treatment, a total of 33 (42.9%) users considered that CHM had helped them achieve all three outcomes. In contrast, there were two CHM users who reported there was no effect after treatment. Neither of these had expected CHM would cure their disease or relieve their symptoms.

Before Treatment (expectation)		After treatment (outcome)				
		Yes, n (%)*	No, n (%)*	Not sure, n (%)*		
Cure the disease/s	olve the problem, n (% o	of users)**				
Yes	41 (53.2%)	29 (70.7%)	7 (17.1%)	5 (12.2%)		
No	18 (23.4%)	3 (16.7%)	15 (83.3%)	0 (0%)		
Not sure	18 (23.4%)	8 (44.4%)	2 (11.1%)	8 (44.4%)		
Total	77	40 (51.9%)	24 ( 1.2%)	13 (16.9%)		
Relief of symptoms, n (% of users)**						
Yes	61(79.2%)	55 (90.2%)	3 (4.9%)	3 (4.9%)		
No	4 (5.2%)	3 (75.0%)	1 (25.0%)	0 (0%)		
Not sure	12 (15.6%)	8 (66.7%)	4 (33.3%)	0 (0%)		
Total	77	66 (85.7%)	8 (10.4%)	3 (3.9%)		
Improve general health and well-being, n (% of users)**						
Yes	65 (84.4%)	63 (96.9%)	2 (3.1%)	0 (0%)		
No	4 (5.2%)	0 (0%)	3 (75.0%)	1 (25.0%)		
Not sure	8 (10.4%)	6 (75.0%)	0 (0%)	2 (25.0%)		
Total	77	69 (89.6%)	5 (6.5%)	3 (3.9%)		

#### Table 6.8 Expectations of and actual outcomes of using Chinese herbal medicine

* Numbers and percentages are of users within each row category

** Numbers and percentages are of total acupuncture users

# General Helpfulness of Acupuncture and Chinese Herbal Medicine

All users also rated their experience on the overall general helpfulness of acupuncture and CHM. A great proportion of users considered acupuncture treatment they had used was very helpful (60.0%) or somewhat helpful (30.1%), while a small proportion of users considered the outcome was not very helpful (six users, 5.1%) or not at all helpful (three users, 2.7%). These proportions were similar among the rating towards CHM. The vast majority of CHM users believed CHM treatment they had used was very helpful (55.6%) or somewhat helpful (39.3%), while two persons considered CHM was not very helpful or not at all helpful.

## 6.1.1.9 Safety Concerns of Acupuncture and Chinese Herbal Medicine Use

#### Adverse Events after Acupuncture and Chinese Medicine Treatment

Among all acupuncture users, four persons reported adverse events (patient self-perceived) after having acupuncture treatment, but one of the answers was not recorded by the interviewer. Three users referred to the pain of the needle (whether or not it should be considered as an adverse event is debatable), and one of them also mentioned having a bruise on the skin. This corresponds to an incidence rate of adverse event of 3 in 100 acupuncture users. Alternatively, among all participants who had visited acupuncturists, a total of 699 consultations had been made. This reflects an adverse event incidence rate of 4.3 per 1,000 consultations.

For Chinese herbal medicine (CHM), seven persons mentioned adverse event but one of the answers was not recorded by the interviewer. Two users experienced nausea after taking CHM; another two users considered the taste of the herbs as unpleasant or too strong. One person felt dizzy, and another user stated that "the liver cleansing power played havoc with me". Thus, this corresponds to an incidence rate of 7.8% among CHM users. More strictly, when the definition of adverse event excludes mild gastrointestinal symptoms (e.g. nausea, unpleasant taste and discomfort), the incidence rate can be estimated as approximately 2.6%. However, whether the participants who encountered CHM adverse events used herbs that had been prescribed by their CHM practitioners or were self-prescribed was not established (a total of 160 consultations had been made among all CHM users, but the rate of self-prescription was considerable high). Thus, an adverse event incidence rate per 1,000 consultations seems inappropriate for CHM users.

# 6.1.1.10 Referral Pattern and Future Use of Acupuncture

#### <u>Referral Pattern</u>

Most commonly, acupuncture users had received advice from their friends or relatives (40.8%). Advice from a medical doctor accounted for just over one fifth (20.7%) of the total responses. Some users had also received advice from a complementary medicine practitioner (18.7%), or from a newspaper, television and the Internet (4.2%). Only one user was referred by another health professional, such as a nurse.

It is of safety concern that one in six users (17.8%) did not receive advice from any source. In particular, the use of acupuncture without receiving advice was more common in those aged 35-64 (27.1%) than the younger group (18-34, 9.3%) and those aged 65 and older (none did not receive advice,  $\chi^2$ =8.230, p=0.016). In addition, middle-age users were also less likely (11.0%) to receive advice from a friend or a relative than the younger (46.3%) and older (49.2%) users, ( $\chi^2$ =6.953, p=0.031). those with private health insurance were more likely to receive advice from CAM practitioners to use acupuncture (25.7% with insurance vs. 5.7% without insurance,  $\chi^2$ =5.915, p=0.015).

#### Future Use of Acupuncture

Among all survey participants, nearly two thirds (62.4%) reported they would consider acupuncture as an option for their health care in the future, while nearly one third (30.4%) would not consider; 7.2% of users were not sure whether or not they would consider this option. More specifically, among the total of 991 participants who indicated they would consider acupuncture in the future, younger adults, those with postsecondary education, private health insurance and a higher income range were more likely to use acupuncture again (Table 6.9). In addition, compared to other states, a relatively lower proportion of South Australia residents said they would consider using acupuncture in the future, although this is not statistically significant ( $\chi^2$ =8.552, p=0.073). On the other hand, among acupuncture users, the vast majority (94.7%) indicated they would use acupuncture again. Of the three users who indicated they would not consider using acupuncture in the future, none had previously experienced side-effects after using acupuncture, but two acupuncture users found it had not been very helpful for their health conditions (i.e. arthritis and osteoarthritis).

Characteristic	Consider using acupuncture in the future		Statistical difference	
	Yes (%)	No (%)	Chi Square	р
Gender			0.216	0.642
Female	66.5	33.5		
Male	67.9	32.1		
Age (year)			6.315	0.043
18-34	69.2	30.8		
35-64	68.0	31.2		
65+	59.0	41.0		
Country of Birth			2.554	0.11
Australia	68.8	31.2		
Non-Australia	63.1	36.9		
State			8.552	0.073
New South Wales	70.6	29.4		
Victoria	67.3	32.7		
Queensland	69.0	31.0		
South Australia	54.5	45.5		
Western Australia	62.7	37.3		
Self-reported health status			0.505	0.477
Excellent/very good/good	67.6	32.4		
Fair/poor	64.6	35.4		
Post-secondary education			6.537	0.011
No	63.0	37.0		
Yes	70.7	29.3		
Employment			2.569	0.109
Employed	69.1	30.9		
Unemployed or not in labour force	64.0	36.0		
Private Health Insurance			10.176	0.001
Yes	71.6	28.4		
No	62.0	38.0		
Annual household income				
<a\$20,000< td=""><td>57.8</td><td>42.2</td><td>10.559</td><td>0.005</td></a\$20,000<>	57.8	42.2	10.559	0.005
A\$20,000-A\$40,000	69.8	30.2		
>A\$40,000	71.7	28.3		

# Table 6.9 Consideration for the use of acupuncture treatment in the future

# 6.1.1.11 Referral Pattern and Future Use of Chinese Herbal Medicine

#### <u>Referral Pattern</u>

Similar to the referral pattern for the use of acupuncture, most commonly, CHM users had received advice from their friends or relatives (45.4%). Advice from CAM practitioners accounted for just over one quarter (25.4%) of the total responses. However, only 6.5% users had received advice from a medical doctor. Receiving advice from a newspaper, television and the Internet (8.5%) and other health professionals (7.3%) constituted a relatively high proportion of responses. It is a safety concern that nearly one in 10 users (9.9%) did not receive advice from any source. In all, none of the demographic information was remarkable in relation to the referral pattern for CHM use.

#### Future Use of Chinese Herbal Medicine

Among all survey participants, over half (52.9%) said they would consider CHM as an option for their health care in the future, while over one third (35.0%) would not consider it as an option; 12.1% users were not sure whether or not they would consider this treatment. More specifically, among the total of 941 participants who had answered definitively about whether or not they would consider CHM in the future, those female, younger adults, who resided in New South Wales, Victoria and Queensland, and who were employed were more likely to use CHM again (Table 6.10). Similar to acupuncture use, the lowest rate of future projected CHM use among states was also reported in South Australia. On the other hand, among CHM users, the vast majority (97.5%) reported they would use CHM again. Of the two users who said they would not consider CHM in the future, one had previously experienced a side-effect after using CHM (dizziness). Another person had used CHM for his back pain and high blood pressure problem, but mentioned it had not been very helpful.

Characteristic	Consider using CHM in the future		Statistical difference	
	Yes (%)	No (%)	Chi Square	р
Gender			8.004	0.005
Female	64.7	35.3		
Male	55.7	44.3		
Age (year)			28.518	0.0001
18-34	66.8	33.2		
35-64	62.0	38.0		
65+	41.4	58.6		
Country of Birth			1.226	0.268
Australia	61.3	38.7		
Non-Australia	57.1	42.9		
State			17.086	0.002
New South Wales	62.4	37.6		
Victoria	66.5	33.5		
Queensland	61.6	38.4		
South Australia	42.8	57.2		
Western Australia	50.3	49.7		
Self-reported health status			2.969	0.085
Excellent/very good/good	61.3	38.7		
Fair/poor	53.7	46.3		
Post-secondary education			3.738	0.053
No	56.7	43.3		
Yes	63.0	37.0		
Employment			6.687	0.01
Employed	63.4	36.6		
Unemployed or not in labour force	54.6	45.4		
Private Health Insurance			2.905	0.088
Yes	63.2	36.8		
No	57.7	42.3		
Annual household income			4.487	0.106
<a\$20,000< td=""><td>52.9</td><td>47.1</td><td></td><td></td></a\$20,000<>	52.9	47.1		
A\$20,000-A\$40,000	63.3	36.7		
>A\$40,000	61.9	38.1		

Table 6.10 Consideration for the use of Chinese herbal medicine treatment in the future

CHM: Chinese herbal medicine

# 6.1.2 Chiropractic

#### **6.1.2.1 Overview**

As noted above, the overall prevalence of chiropractic use in Australia was 16.1% (95%CI: 13.9% - 18.3%), being one of the most popular forms of CAM in Australia. The vast majority (90.6%) of chiropractic users had visited practitioners for therapy of all forms of CAM users. Chiropractic users had the lowest proportion (9.4%) who self-administrated the therapy. This also implies that approximately one in seven adult Australians (14.6%, 95%CI: 12.4% - 16.7%) had visited a chiropractor in the 12-month preceding the survey in 2005.

In general, those born in Australia who had private health insurance and higher household incomes were more likely to use chiropractic (p<0.01, Table 4.6). Gender, age and educational background did not appear to be major factors related to chiropractic use. On the other hand, there were smaller variations in users in different states, although the prevalence in Queensland (14.4%) and Western Australia (14.9) was slightly lower than the national average (not significant) while slightly higher than national average prevalence was reported in New South Wales (17.0%), Victoria (17.4%) and South Australia (17.0%) (see Table 5.1 and Table 5.2).

In terms of the frequency of visiting chiropractors, an annual rate of approximately 8.4 times per user was estimated for all users, which approximates to 19.05 million visits per year in Australia (see Table 4.9). The average out-of-pocket expenditure of visiting a CAM practitioner, as estimated in the current survey was A\$25 (see Chapter 4.1.7.1). This suggests that a total out-of-pocket expenditure of A\$476 million on chiropractor visits. In addition, of those users, 23 had incurred additional costs such as books and equipment related to their

chiropractic treatments. The median total chiropractic-related expenditure was A\$200. Thus, nationwide, A\$68.37 million would had spent on chiropractic-related items other than costs related to practitioner visits. This suggests, Australia nationwide, the total out-of-pocket expenditure spent on chiropractic services in the 12-month period was approximately A\$544 million. (see Appendix G9 for the unadjusted expenditure and G10 for a summary statistics of the adjusted data).

# 6.1.2.2 Rationale for the Use of Chiropractic

After being identified as chiropractic users, participants were also asked about their main reasons (multiple responses) for using chiropractic. In all, well over two thirds (68.6%) used chiropractic for the relief of symptoms, with nearly half (42.5%) using chiropractic for the purpose of relieving symptoms only and not for any other purpose. Approximately one third (32.3%) used chiropractic for improving general health and well-being, with over 11.9% using it, exclusively to improve general health and well-being. Also, about one third (36.2%) of chiropractic users used it to improve the ability to undertake normal daily activities, while nearly one in 10 (9.4%) users also used chiropractic to improve their sporting performance. About 5.3% used chiropractic for all four purposes.

More specifically, cross-tabulation analyses revealed that the only gender difference in the reasons for using chiropractic was to improve sporting performance. Not surprisingly, the use of chiropractic to improve sporting performance by male users (15.7%) was much higher than female users (4.0%) ( $\chi^2$ =6.828, p=0.009). Among different age groups, the only difference in the reasons for using chiropractic was to improve the ability to undertake normal daily activities. More than half (51.9%) of those 65 years of age or older used chiropractic for this purpose, compared with 38.2% of those aged 35-64 and 24.3% of those aged 18-34 ( $\chi^2$ =6.292, p=0.043). Other factors, such as education level, employment status and personal health status did not appear to influence choosing chiropractic.

In terms of the specific symptoms for which users had sought chiropractic, nearly two thirds (65.7%) had used it for back pain or back-related problems. Other common conditions for which users had resorted to chiropractic were neck pain and related problems (20.7%), non-

specific musculoskeletal problems (14.9%), headache and migraine (9.3%), and shoulder pain and related problems (5.3%). In addition, among those had used chiropractic to relieve symptoms, near half (43.9%) did so solely to relieve back pain or back-related problems and not for any other health conditions. On the other hand, only one person reported using chiropractic for a condition/symptom unrelated to a musculoskeletal problem (a gynaecological condition).

# 6.1.2.3 Use of Chiropractic–Regression Analyses

Bivariate analyses revealed that nine predictor variables were significantly associated with chiropractic use: income range, country of birth, had consulted a GP in the preceding 12 months, consulted a GP four times or more in the preceding 12 months, covered by private health insurance coverage of CAM influences the purchase of health insurance, visited a GP for cancer, and visited a GP for diabetes. These were entered into the logistic regression model by the stepwise method.

Table 6.11 presents the results of the significant regression model for chiropractic users. It was not unexpected that those had visited a GP for back problems were 3.5 times more likely to have also used chiropractic than those who had not visited a GP for back problems. Also participants' household income status substantially influenced their use of chiropractic. Compared to the low-income participants, people with a household income A\$20,000 - A\$40,000 were 3.3 times more likely to use chiropractic. The odds ratio increased to 3.8 times more likely to use chiropractic among those with a household income higher than A\$40,000.
Predictors		Odds ratio (95% CI)	
	<a\$20,000< td=""><td>1.0</td></a\$20,000<>	1.0	
	A\$20,000-A\$40,000	3.318 (1.510–7.287)**	
Income range	>A\$40,000	3.787 (1.844–7.777)**	
	Overseas	1.0	
Country of birth	Australia	2.070 (1.241-3.454)**	
Consulted a GP in the past 12 months	No	1.0	
	Yes	1.995 (1.133–3.513)*	
Coverage of CAM influences health insurance purchase	No	1.0	
	Yes	2.325 (1.560-3.463)**	
	No	1.0	
Visited a GP for back problems	Yes	3.494 (1.447-8.437)**	
Omnibus Tests of Model Coefficients			
Chi square (degree of freedom, p value)		60.236 (6, <0.0001)	
Percentage accurate in classification		83.4	

# Table 6.11 Multivariate analyses of predictor variables for chiropractic use

* p<0.05; ** p<0.01; CI: confidence interval.

## 6.1.2.4 Outcomes and Safety Concerns of Use of Chiropractic

Users were asked the specific outcomes after their use of chiropractic. Almost three quarters (71.9%) mentioned that it relieved pain; half (50.2%) found it improved their ability to undertake normal daily activities; nearly half (45.5%) also found it improved their well-being; an additional 23.1% of users also considered chiropractic had decreased a disability. In total, nearly one in five (19.5%) users found that, after seeking chiropractic treatment, they had achieved all four outcomes.

Older people (58.2%) and those from the middle-age group (aged 34-64, 58.9%) were more likely to use chiropractic to improve their ability to undertake normal daily activities than the younger adults (aged 18-34, 30.7%) ( $\chi^2$ =11.08, p=0.004). Also, a higher proportion of users with poor or fair health reported a helpful outcome of decreased disability (38.6%) and improved normal daily activity (67.5%) than those who considered themselves to be in better health (19.5%,  $\chi^2$ =5.481, p=0.019; and 46.4%,  $\chi^2$ =4.751, p=0.029, respectively).

It may be of particular interest that there were fewer users in Victoria (58.4%) and Queensland (65.6%) who provided a positive answer to the outcome of pain relief than users in New South Wales (82.3%), South Australia (81.3%) and Western Australia (83.9%) ( $\chi^2$ =9.852, p=0.043). No additional regional difference was observed in other outcomes. On the other hand, more users with private health insurance believed chiropractic had improved their general health (51.6%) than people without private health insurance (33.0%) ( $\chi^2$ =5.303, p=0.021).

All users also rated their experience on the overall general helpfulness of chiropractic. The majority of users believed the chiropractic treatment they had used was very helpful (71.4%)

or somewhat helpful (21.6%), while a small proportion of users received a less helpful outcome (not very helpful, 2.6%, and not at all helpful, 2.4%).

Among all users, a total of nine persons reported adverse events (patient self-perceived) after having chiropractic treatment. These included pain after treatment (three persons), headache/migraine immediately after treatment (two persons), and tiredness (one person). The worst scenario was aggravation of a major back injury (treatment was provided without an xray). Two reported adverse events which could be considered as "dissatisfaction of treatment", were "it does not do any thing", and "you tend to go back and they fix it". Thus, this (7 reported adverse events) corresponds to an incidence rate of 4.0% among users. Alternatively, among all participants who had visited chiropractors, a total of 1,352 consultations were made. This reflects an adverse event incidence rate of 5.2 per 1,000 consultations.

With respect to the general opinion on the safety aspect of chiropractic, over 81.0% of all survey participants either strongly agreed (16.5%), agreed (43.6%) or partially agreed (20.9%) that chiropractic is safe. Perception of safety was even more convincing when the opinion of chiropractic users is considered. Those who strongly agreed that chiropractic is a safe form of treatment increased from 16.5% of all survey participants to 43.2% of chiropractic users. On the other hand, those who did not agree that "chiropractic is a safe form of treatment" decreased from 9.6% of all participants to approximately 1.9% (three users) of chiropractic users. In the latter category, one had encountered an adverse effect (i.e. severe pain).

#### 6.1.2.5 Referral Pattern and Future Use of Chiropractic

In terms of using chiropractic on the advice of other people, the most common response was from users' friends or relatives (43.6%). Advice from a medical doctor accounted for exactly one fifth (20.0%) of the total responses. Some users also received advice from a complementary medicine practitioner (7.2%), from a newspaper, television and the Internet (1.3%) or from another health professional such as a nurse (2.6%).

It is of safety concern that over one quarter (27.2%) of chiropractic users did not receive advice from any source. In particular, the use of chiropractic without receiving advice was much more common in those aged 65 and older (45.6%) and those aged 35-64 (31.2%) than in the younger group (18-34, 9.6%,  $\chi^2$ =13.505, p=0.001). As may have been expected, those having a better health (self-rated) were more likely to receive advice from a friend or relative (47.0% vs. 27.8%,  $\chi^2$ =4.051, p=0.044), while those having poor or fair health were more likely to receive advice from their medical doctor (36.9% vs. 16.1%,  $\chi^2$ =7.208, p=0.007). In contrast, those who had private health insurance were less likely to receive advice from their medical doctor to use chiropractic (15.6% vs. 29.0%,  $\chi^2$ =4.286, p=0.039), but more likely to use chiropractic without any advice (32.7% vs. 16.7%,  $\chi^2$ =4.957, p=0.026).

Among all survey participants, well over two thirds (68.1%) said they would consider chiropractic as an option for their health care in the future, while 26.9% would not consider it. More specifically, among the total of 1,015 participants who had made a definitive answer of whether or not they would consider chiropractic in the future, the younger, male adults, who resided in the states of New South Wales and Western Australia, were in a better health, employed and in the higher income range, were more likely to use chiropractic (Table 6.12). On the other hand, among chiropractic users, the vast majority (93.3%) said they would use

chiropractic again. Of the ten users (5.1%) who said they would not consider using chiropractic in the future, two had previously experienced side-effects after using chiropractic, including severe pain and aggravation of a major back injury.

Characteristic	Consider using aracteristic			Statistical difference	
	Yes (%)	No (%)	Chi Square	р	
Gender			5.667	0.017	
Female	68.3	31.7			
Male	75.1	24.9			
Age (year)			17.184	0.0002	
18-34	79.1	20.9			
35-64	70.2	29.8			
65+	62.0	38.0			
Country of Birth			0.001	0.993	
Australia	71.8	28.2			
Non-Australia	71.9	28.1			
State			19.637	0.001	
New South Wales	76.5	23.5			
Victoria	68.5	31.5			
Queensland	68.3	31.7			
South Australia	58.0	42.0			
Western Australia	82.7	17.3			
Self-reported health status			4.014	0.045	
Excellent/very good/good	72.8	27.2			
Fair/poor	64.9	35.1			
Post-secondary education			0.017	0.896	
No	71.9	28.1			
Yes	71.5	28.5			
Employment			9.695	0.002	
Employed	74.9	25.1			
Unemployed or not in labour force	65.6	34.4			
Private Health Insurance			0.293	0.588	
Yes	70.9	29.1			
No	72.5	27.5			
Annual household income			9.970	0.008	
<a\$20,000< td=""><td>64.3</td><td>35.7</td><td></td><td></td></a\$20,000<>	64.3	35.7			
A\$20,000-A\$40,000	68.1	31.9			
>A\$40,000	75.7	24.3			

 Table 6.12 Consideration for the use of chiropractic treatment in the future

## 6.1.3 Osteopathy

#### 6.1.3.1 **Overview**

As above noted, the overall prevalence of osteopathy use in Australia was 4.6% (95%CI: 3.3% – 5.9%, a total of 51 unweighted numbers of users). As a provider-based therapy, most osteopathy users (76.4%) had visited practitioners for treatment. Surprisingly, the proportion of self-administrating users (23.6%) was quite high. In total, approximately one in every 28 adult Australians (3.5%, 95%CI: 2.4% – 4.6%) had visited an osteopath in the 12 months preceding the survey in 2005.

There are different forms of osteopathy in clinical practice, of which the most commonly known and reported by the survey participants was as a form of massage therapy (66.1%). At least half the participants also considered that the type of osteopathy treatment they had used was the mobilisation or "popping of joints" (58.6%), stretching (58.5%), or a gentle technique (may be called functional or cranial, 51.3%). Over 57.3% also mentioned that they had received exercise advice from their osteopathy practitioners.

Consistent with the characteristics that related to overall CAM use, female (5.8%) participants were more likely to use osteopathy than male participants (3.3%, p<0.05). Interestingly, those in the middle household income range used the least osteopathy (1.3%), while those in the lower and higher income ranges used osteopathy much more (6.7% and 5.5% respectively, p<0.05) (Table 4.6). In addition, osteopathy use appeared to vary between states. Participants from New South Wales and Victoria were more likely to use osteopathy than those from any other states (p<0.05, Table 5.1). In relation to the characteristics of those who visiting an osteopath, in addition to the differences in the use of osteopathy, educational background and

private health insurance coverage also appeared to be determinants of osteopathy use (Table 5.2).

Users of osteopathy visited practitioners approximately 5.9 times per annum. Thus, a total of 3.08 million visits were estimated per annum in Australia (see Table 4.9). Using the average out-of-pocket expenditure of visiting a CAM practitioner, as estimated in the current survey (A\$25, Chapter 4.1.7.1), this equals A\$77 million spent on osteopath visits in the 12 months preceding the survey. In addition, ten osteopathy users indicated that they had spent additional money on items such as books and equipments related to their osteopathy treatments. The median cost was A\$50.0. Thus, nationwide, the total out-of-pocket expenditure on osteopathy in the 12-month period can be estimated as A\$83.47 million, with A\$6.47 million spent on items other than practitioner visits (see Appendix G9 for the unadjusted expenditure and G10 for a summary statistics of the adjusted data).

## 6.1.3.2 Rationale for the Use of Osteopathy

Osteopathy users were asked the same questions as were put to chiropractic users about their main reasons for seeking treatment (see Chapter 6.1.2.2), and they also provided multiple responses to their rationale for use. In all, over three quarters (75.9%) used osteopathy for the relief of symptoms, including nearly two fifths (38.0%) who used it solely for the purpose of relieving symptoms. Over two fifths (40.6%) used osteopathy to improve general health and well-being, including 14.5% who used it for the purpose of improving general health and well-being only, rather than for other purposes. More than two fifths (41.9%) of participants used osteopathy to improve their ability to undertake normal daily activities, while nearly one in six (17.1%) used osteopathy to improve their sporting performance.

Cross-tabulation analyses revealed an interesting finding. Those born overseas used osteopathy to improve general health and well-being much more than those born in Australia (70.6% vs. 33.5%,  $\chi^2$ =4.297, p=0.038). On the other hand, the only gender difference in the reasons for using osteopathy was to improve the ability to undertake normal daily activities. In this respect, female users were three times (54.2%) more likely to use osteopathy than male users (19.7%,  $\chi^2$ =5.443, p=0.020).

Among users who had resorted to osteopathy for the relief of symptoms, 48.4% had used it for back pain or back related problems, 35.2% had used it for non-specific musculoskeletal problems (eg. restricted movement), 22.9% had used it for shoulder problems, and 10.7% for neck problems. Osteopathy users had also used it for other conditions, such as headache, spinal or knee problems. In addition, among those had used osteopathy to relieve symptoms, nearly one quarter (24.7%) had used it solely to relieve back pain or back-related problems only.

## 6.1.3.3 Use of Osteopathy–Regression Analyses

Bivariate analyses revealed that five predictor variables were significantly associated with osteopathy use: gender, income range, coverage of CAM influences the purchase of private health insurance, visiting a GP for back problems, and visiting a GP for heart-related problems. These were entered into the logistic regression model by the stepwise method.

As presented in Table 6.13, those who had visited a GP for back problems were 4.3 times more likely to have also used osteopathy than those who without back problems. It is worth further investigating why the highest odds ratio of osteopathy use was those with the lowest household incomes. Compared to those with a household income of A\$20,000 A\$40,000, people who earned less than A\$20,000 were up to 18 time more likely to use osteopathy. The odds ratio of people with the lowest incomes using osteopathy was high, (1.6 times higher, but not statistically significant) compared to those with household incomes higher than A\$40,000 (see Table 4.6). This finding may be somewhat unreliable due to the relatively small sample size (45 osteopathy users) entered into the regression analysis.

Predictors		Odds ratio (95% CI)		
	<a\$20,000< td=""><td>1.0</td></a\$20,000<>	1.0		
	A\$20,000-A\$40,000	0.054 (0.007-0.429)**		
Income range	>A\$40,000	0.624 (0.298–1.306)		
Coverage of CAM influences health insurance purchase	No	1.0		
	Yes	4.760 (2.139–10.592)**		
	No	1.0		
Visited a GP for back problems	Yes	4.318 (1.137–16.402)*		
Omnibus Tests of Model Coefficients				
Chi square (degree of freedom,	34.817 (4, <0.0001)			
Percentage accurate in classification		95.0		

Table 6.13 Multivariate analyses of predictor variables for osteopathy use

* p<0.05; ** p<0.01; CI: confidence interval

## 6.1.3.4 Outcomes and Safety Concerns of Use of Osteopathy

In regard to the outcomes of osteopathy use, over three quarters (79.5%) of users mentioned that it relieved pain; three out of five users (60.5%) found it improved their ability to undertake normal daily activities; nearly half (45.6%) also found it improved their well-being; an additional 40.6% of users considered osteopathy had decreased a disability. In total, nearly one in three (29.7%) users found that, after seeking osteopathy treatment, they had achieved all four outcomes. None of the particular user characteristics was found significantly related to different outcomes.

All except two users rated their experience on the overall general helpfulness of using osteopathy. The majority of users believed their osteopathy treatments were very helpful (64.2%) or somewhat helpful (28.4%), while only two users reported a less helpful outcome (one did not believe it was very helpful, while another believed it was not at all helpful).

Importantly, a total of four users (an incidence rate of 7.8%) reported adverse events (patient self-perceived) after having osteopathy treatment. These included more pain, soreness, "very tired" and "neck cracks a lot more after treatment". However, it is important to note that all of these were controversial in terms of whether or not they should be considered as adverse events. The adverse event incidence rate per 1,000 consultations was not calculated against the osteopathy treatment.

## 6.1.3.5 Referral Pattern and Future Use of Osteopathy

The most common referral channel for using osteopathy was from users' friends or relatives (38.1%). Advice from complementary medicine practitioners accounted for over one fifth (21.5%) of the total responses, closely followed by advice from a medical doctor (16.1%). Two additional users were referred by other health professionals, such as a nurse (one person) and through newspapers, television and Internet (one person). However, there were six users (13.2%) did not receive advice from any source. Four of them were from New South Wales and five of them were female.

Among 32 osteopathy users who were employed at the time of the survey, only two (weighted percentage, 6.4%) had received advice from their medical doctors while six out of 18 (29.5%) unemployed osteopathy users had received advice from a medical doctor ( $\chi^2$ =4.686, p=0.030). One third (ten out of 28, 33.7%) of the users aged 35-64 had received advice from a complementary medicine practitioner, while only one user aged 65 and older and none from the group aged 18-34 had received advice from a CAM practitioner (not statistically significant,  $\chi^2$ =5.899, p=0.052).

Among all survey participants, just over half (52.0%) said they would consider osteopathy as an option for their health care in the future; 24.9% would not consider, and the remaining 23.1% were not sure about osteopathy (e.g. lack of knowledge of osteopathy and did not respond to the current question). On the other hand, among osteopathy users, five users (12.4%) said they would not consider using osteopathy again. This included one person who had experienced a side-effect (pain) after using osteopathy. Finally, among the total of 814 participants with a definite response to whether or not they would consider osteopathy in the future, there was no difference between participants' sociodemographic characteristics and their decision to use osteopathy in the future (Table 6.14).

Characteristic	Consider using osteopathy in the future		Statistical difference	
	Yes (%)	No (%)	Chi Square	р
Gender			0.039	0.843
Female	67.3	32.7		
Male	68.0	32.0		
Age (year)			4.316	0.116
18-34	69.4	30.6		
35-64	68.9	31.1		
65+	59.9	40.1		
Country of Birth			0.727	0.394
Australia	67.0	33.0		
Non-Australia	70.3	29.4		
State			8.183	0.085
New South Wales	68.2	31.8		
Victoria	73.9	26.1		
Queensland	67.5	32.5		
South Australia	56.6	43.4		
Western Australia	63.4	36.6		
Self-reported health status			1.009	0.315
Excellent/very good/good	68.3	31.7		
Fair/poor	63.6	36.4		
Post-secondary education			0.412	0.521
No	68.8	31.2		
Yes	66.7	33.3		
Employment			2.426	0.119
Employed	69.7	30.3		
Unemployed or not in labour force	64.3	35.7		
Private Health Insurance			0.404	0.525
Yes	68.7	31.3		
No	66.6	33.4		
Annual household income			0.218	0.897
<a\$20,000< td=""><td>66.6</td><td>33.4</td><td></td><td></td></a\$20,000<>	66.6	33.4		
A\$20,000-A\$40,000	69.2	30.8		
>A\$40,000	67.6	32.4		

Table 6.14 Consideration for the use of osteopathy treatment in the future

## 6.2 Discussion

## 6.2.1 Chinese Medicine (Acupuncture and Chinese Herbal Medicine)

Until the introduction of Western medicine in China in the 18th century,¹⁵⁶ traditional Chinese medicine (CM) had been the primary form of health care for several thousand years.¹⁵⁷ Even now, Chinese medicine provides 40% of health-care services to the Chinese population.¹⁵⁸ Moreover, during the past several decades, Chinese medicine has become part of the health-care systems of many other Asian regions^{159,160} and, more recently, it has been gaining acceptance in most Western countries as a therapeutic alternative to conventional Western medicine.^{3,11} In recent years, the integration of Chinese medicine in the conventional medical settings in the Western countries is not uncommon. For example, through an integrative health-care approach, the Center for East-West Medicine within the Department of Medicine at the University of California, Los Angeles believed that Chinese medicine has much to offer to patients with cancer.¹⁶¹ The increasing popularity of Chinese medicine in Western countries has been accompanied by concerns about its efficacy^{162,163} and safety,^{164,165} both when used alone and in combination with Western medical treatments.

The information available on the use of Chinese medicine in Australia comes from a landmark study completed in 1996 (and the publication of "Toward a Safer Choice – the Practice of Traditional Chinese Medicine in Australia").¹⁶⁶ It is important to note that the 1996 study was a Chinese medicine national workforce survey for which the CM participants were practitioners and educators, and the focus was on education, the risks and benefits associated with the practice of CM, and regulatory frameworks. The study provided a profile of patients who choose to use CM by asking practitioners to complete a questionnaire for every patient treated on a specific day. Since the study subjects had chosen to use CM, it was not able to

provide information on the prevalence and patterns of CM use in the general population. It also did not provide insightful information on public perceptions of CM. The study was conducted a decade ago and there would seem to have been dramatic changes since then in the public use of CM in Australia.

The use of Chinese medicine, as a defined health-care approach, was not specifically researched in most surveys in Australia and overseas, and consequently there is a lack of nationwide, population-based representative data on Chinese medicine use in Australia. Addressing this deficiency, the current study included the major modalities of Chinese medicine (acupuncture, Chinese herbal medicine, Chinese therapeutic massage, Chinese medicine dietary therapy, Qigong, martial art and Tai Chi) as the totality of CM.

With the above broader definition of CM, the current study presents an estimation of the national and regional prevalence of CM use among the general population in Australia. Nearly one in five adult Australians had used any one form of CM in the 12-month period preceding the survey, and one in nine had visited CM practitioners in the 12-month period. These findings again echo the increasing popularity of all forms of CAM in Australia. There is a lack of accurate information of CM use among the general population in other Western countries. Systematic literature searches (see Chapter 2) have identified only one similar study which investigated such CM information in Western countries. The estimated CM prevalence (17.2% visited CM practitioners in the two-year period) in New York City's Chinatown is higher than the current Australian estimation.¹⁰⁰ However, this New York study was based on randomly selected Chinatown residents.

According to the current national survey, the lowest prevalence of CM use was found in the state of Western Australia, and the lowest prevalence of visits to CM practitioners was found

in Western Australia and South Australia. Specifically, the marked differences between different states, in fact, were for the two major Chinese medicine modalities: acupuncture and Chinese herbal medicine, both in overall use and prevalence of visits to practitioners (see Table 5.1 and Table 5.2). Thus, the current survey confirmed two research hypotheses: 1) certain forms of CAM are more prevalent in some states within Australia and, 2) among all states and territories, Chinese medicine, including acupuncture and CHM, is more prevalent in New South Wales, Victoria and Queensland.

The current study also found acupuncturists were visited more frequently than practitioners of the other 16 forms of CAM. The result is not surprise, as numerous reports have suggested the popularity of acupuncture use both in Australia and overseas. ^{5,6,17,102,167,168} With the continuing increase in scientific evidence relating to the mechanism and efficacy of acupuncture in treating certain chronic diseases,¹⁶⁹⁻¹⁷² there is little doubt that acupuncture is one of the most popular forms of CAM in Western countries. It is being used by millions of patients, and is practiced by traditional acupuncturists, medical doctors,⁹⁵ and by many other health professionals, such as licensed chiropractors and nurses.

The total estimated number of visits to acupuncturists in Australia is substantial in terms of the frequency of visits and the cost to the Australian health system. As mentioned earlier, the Australian national health insurance, Medicare, only covers the cost of acupuncture services provided by registered medical practitioners. Thus, the current estimated millions of acupuncture treatments were mainly borne by the users. The cost implications and non-claimable acupuncture services provided by non-GP acupuncturists meant that in the current study, one in five people who had received acupuncture treatment, received it from GPs only. Secondary analysis of the 1996 Health Insurance Commission data reveals that there were 960, 000 Medicare claims for acupuncture in the 1996-97 financial year, totalling \$17.7

216

million in reimbursements.¹⁶⁷ It is worth noting that these approximately one million claims did not include acupuncture treatments funded by Workers Compensation and the Department of Veterans Affairs, or treatments provided in public hospitals. After 10 years of growing popularity of acupuncture use and practice in Australia, the current estimated 10.16 million visits to acupuncturists, including services provided by GPs and non-GPs, seems to be consistent with the data and reflects the increasing trend of using acupuncture.

The current survey also estimated that the combined total number of practitioner visits by adult Australians for acupuncture, CHM and Chinese therapeutic massage was about 15 million. However, the figure must be interpreted with extreme caution. Based on the nature of Chinese medicine, most CM practitioners practice both acupuncture and Chinese herbal medicine. This is also evidenced by a report of the Chinese Medicine Registration Board of Victoria in June 2006 that, nearly 60% of registered CM practitioners in Victoria are registered for both acupuncture and CHM.¹⁷³ In addition, the application of Chinese therapeutic massage in CM practitioners' practice is not unusual.

The current survey found that the characteristics of CM users are similar to the general characteristics of CAM users. However, for both CM users and people who had visited CM practitioners, there was no gender difference or difference in whether or not they had private health insurance. However, the use of acupuncture and Chinese herbal medicine between adults who were born in Australia or overseas was significant. Acupuncture appears have been more accepted than CHM by Australian-born adults. That is, a significantly higher proportion of Australian-born adults used acupuncture, but a significantly lower proportion of Australian-born adults used CHM, when compared to those born overseas. Furthermore, multivariate analyses found those born overseas were 2.3 times more likely to use CHM than the Australian-born residents.

In clinical practice, Chinese medicine treatment is typically highly individualised and based on a holistic philosophy value. It was also expected that people have used acupuncture and CHM for varied health conditions. Thus, the current survey participants mentioned that they used CM for over 30 different health conditions. What the common forms of herbs that are being used for specified health conditions were not researched. However, about three-quarters of acupuncture users used acupuncture for musculoskeletal- related conditions, notably for back pain and shoulder problems.

It is interesting to note that over one third of CHM users actually used CHM for general health and well-being. The individual names of each of the CHM modalities that were used by survey participants are beyond the scope of the current survey. However, there were a large proportion of people who self-used CHM, particularly for general health, rather than specified health conditions. This implies that a number of people may use herbal supplements to improve their health, such as Ginseng to enhance the immune system, which is based on Chinese medicine theory, (to "tonify" the Qi of the body).

Consumers are asserting their rights to choose different modes of health care. As one of the most commonly used forms of CAM, Chinese medicine is used by consumers from diverse cultural backgrounds. Since the publication of a review of Chinese medicine practice in Australia by Bensoussan, Myers *et al.*, ¹⁶⁶ CM has experienced significant development. Firstly, the introduction of the statutory regulation of Chinese medicine practitioners in Victoria, enforced by the Victorian Chinese Medicine Registration Act 2000, made Victoria the first state in Australia to register Chinese medicine practitioners. Implementation of Chinese medicine practitioner registration in Victoria is an important initiative for the profession and should lead to enhanced public safety and confidence in Chinese medicine use in Australia. Secondly, it is highly likely that other states will follow Victoria's lead in the

future. For example, the discussion documents on regulatory considerations in the states of New South Wales and Western Australia.^{174,175} Thus, the current survey has investigated the impact of the CM Registration Act on public attitudes to Chinese medicine (see regulation section, Chapter 7.1.5). In all, participants suggested that a much more stringent surveillance system in all states and territories is necessary to protect the public from unregistered and unqualified Chinese medicine practitioners.

It is of safety concern that one in six acupuncture users and one in ten CHM users did not receive advice from any source in using these therapeutic treatments. As mentioned above, there are increasing concerns about the adverse events that may relate to the use of CAM therapies and products. This safety concern extends if the CAM use was not supervised or had been used concurrently with other CAM products and/or conventional medicines. The results also suggest an educational approach should be developed by government authorities, such as the TGA and, professional registration bodies, such as the Chinese Medicine Registration Board, to improve the awareness of the potential risks among therapies and products, which are not exclusively related to any individual forms of CAM or conventional medicine.

In summary, the utilisation of different forms of CM in Australia is more prevalent among the general population than had been generally assumed. There are marked differences in the overall CM prevalence among different states. The variations between states are found mainly in the utilisation of acupuncture and Chinese herbal medicine. While more research must to be undertaken to improve the understanding of the mechanism and efficacy of CM, strategic health planning is also necessary to keep consumers and health professionals informed about the safety aspects of different forms of CM.

## 6.2.2 Chiropractic

Both chiropractic and osteopathy are manipulative therapies, for which practitioners use their hands (as well as modern technology) to diagnose and treat abnormalities of the structure and function of the musculoskeletal system of the body. ¹⁷⁶ One does not need to look very far to search for the popular use of such manipulative therapies in different countries. According to previous CAM surveys, at least 11% of Americans,³ 13% of Canadians,⁶ and 16.7% of Australians¹¹ used chiropractic in a 12-month period. The current survey revealed a similar prevalence: 16.1% of adult Australians used chiropractic.

The current survey found a higher overall CAM prevalence compared to previous regional surveys.¹¹ Individual prevalence of most forms of CAM estimated in the current survey was also relatively higher than those found previously in Australian regional studies. Perhaps, with the exception of multivitamins, that is, different definitions were used in different surveys and a comparison was not available, the only higher prevalence observed in previous surveys compared to the current survey was the use of chiropractic. In fact, the prevalence of 16.7% estimated in the South Australian study was based on the proportion of people who had visited chiropractors. This was higher than the current national estimation of people who had visited a chiropractor (14.6%). A small proportion of chiropractic users self-administrated, and details were not collected in the current survey. Again, the finding may be reasonably explained by the regional differences among different states. The current survey found that the states of South Australia, Victoria and New South Wales had a higher than national average prevalence of chiropractic use, whereas the prevalence of chiropractic in other states was slightly lower than the national average. The chiropractic prevalence in South Australia (17.0%) in the current survey seems to be consistent with a 2004 South Australia regional estimation (16.7%).¹¹

220

In all states and territories of Australia, chiropractic is subject to practitioner registration. Despite the evidence of high prevalence of use in Australia,¹¹ little is known about who the chiropractic users are in Australia, why they use it, and the perceptions and consequences of chiropractic treatment. On the other hand, these matters were investigated in the US,¹⁷⁷ Canada,¹⁷⁸ Denmark,^{179,180} the Netherlands,¹⁸¹ and in Sweden.¹⁸² Nevertheless, the profile of middle-aged Australian women who consult a chiropractor or osteopath has been published recently, based on a large-scale longitudinal study on women's health.¹⁸³ It revealed that women living in non-urban areas and women with a lower level of education were more likely to use chiropractic or osteopathy. Such findings were not supported by the current survey. However, the current survey suggests Australian-born adults are significantly more likely to use chiropractic than those born overseas (odds ratio=2.07).

Similar to the mean number (9.8) of chiropractor visits per annum in the US in 1996,³ the current survey also reveals that chiropractors are the second most-frequently visit of all forms of CAM practitioners. The average of 8.4 visits per year per chiropractic user represents over 19 million visits by all adult Australians. As noted, the costs related to chiropractor visits are still mainly borne by consumers. In addition to consultation and treatment costs, approximately one in eight chiropractic users indicated that they also spent money on books or equipment related to chiropractic treatment. Thus, the economic burden borne by chiropractic users in Australia seems substantial. It is interesting to note, based on one year health service records from over 2,000 low-back pain patients, a US study suggested that the mean costs associated with chiropractic patients were significantly higher than those patients treated by medical doctors.¹⁸⁴

The use of chiropractic for back pain and back-related problems is well documented in the literature.^{177-179,181} The current survey further confirms such a finding among the Australian

general public. Nearly two thirds of chiropractic use was for back pain or back-related problems. Thus, multivariate analysis estimated that the chance of Australian people using chiropractic was about 3.5 times higher among those with back problems, compared to those without back problems. Furthermore, the use of chiropractic treatment for a specific health condition, rather than for the musculoskeletal system, is not common in the current survey. Of 176 chiropractic users, only one person used it for a gynaecological condition. In contrast, a much higher prevalence of non-musculoskeletal complaints in chiropractic practice was found in a multinational study, which accounted for about one in ten (10.3%) of all chief complaints.¹⁸⁵

A previous survey found similar proportions of Australian adults with lower back pain had consulted GPs (22.4%) or chiropractor (19.3%), which was considerably higher than the proportion of consultations with physiotherapists (13.4%).¹⁸⁶ Thus, the researchers further suggested that visits to chiropractors may have a greater "market share" in Australia, while due to historical and other reasons, physiotherapists have been much more commonly accepted as a form of allied health in the national health system in Australia. Nevertheless, the Australian Government is increasing the exclusive ancillary position of chiropractic, including osteopathy treatment, the Australian national health system (Medicare) now provides limited rebates for people with chronic conditions and complex care needs.¹⁸⁷

In the current survey, after receiving chiropractic treatment, almost three quarters of the users considered it was very helpful, and a similar proportion of users also agreed chiropractic had relieved various forms of pain. Although nearly half of all users considered chiropractic had also improved their general health and well-being, detailed non-musculoskeletal responses to chiropractic treatment have not been investigated in the current study. A multinational survey of chiropractic patients in seven countries estimated that positive non-musculoskeletal

reactions to chiropractic were not rare, with an improvement in breathing (27%), digestion (26%) and circulation (21%) being most common.¹⁸⁸ The specific benefits of chiropractic are necessary in future chiropractic research in Australia.

It is worth mentioning that one in five chiropractic users received advice from their medical doctors to use chiropractic. In past decades, many efforts have been made to integrate conventional medicine and CAM therapies.^{95,189,190} Many health professionals are now working in a multi-disciplinary environment. For example, one in 10 chiropractors in the US is practicing as a member of a multi-disciplinary team that includes medical doctors¹⁹¹ and, most chiropractors had received a referral from medical doctors in the US.¹⁹² Such information is limited in Australia, so future studies may concentrate on a strategy to improve the communication between these two professions (i.e. chiropractor and GP). Follow-up studies should reveal the barriers and motivations of patient referral.

A moderate proportion of users (27.2%) did not receive any advice in seeking chiropractic care. This seems to be consistent with the fact that most survey participants considered chiropractic treatment to be safe and the adverse event rate reported by all users was less than three per 1,000 consultations. Limited Australian literature provides information on concerns about adverse events and safety issues relating to chiropractic. It has been suggested that adverse events were more commonly reported if a practitioner applied cervical spine manipulation rather than mobilisation.¹⁹³ On the other hand, a previous study revealed that some people experienced dizziness, nausea or other "abnormal reaction" after treatment.¹⁹⁴ In addition, although rare, serious adverse events such as stroke and vertebral artery dissection after cervical spine manipulation continue to be reported.¹⁹⁵ In the current survey, a major back injury, due to chiropractic treatment provided without performing an x-ray, was reported by one of the participants. Chiropractic practitioners should be alerted in terms of precautions.

223

## 6.2.3 Osteopathy

As noted, osteopathy is one of the major forms of manipulative CAM therapies. More popularly, a study in Scotland found 12% osteopath use in 1999.¹⁹⁶ A 1998 UK study found a 4.3% use of osteopathy in a 12-month period, and a 13% lifetime use.⁵ The use of osteopathy was found to be not very common in Canada (less than 4%).⁶ According to two UK studies,^{5,17} the prevalence of osteopathy in the UK was higher than the prevalence of chiropractic. The proportion of Australians who used osteopathy treatment, however, is far less than chiropractic, as evidenced in previous Australian regional CAM surveys (0.4% visited osteopaths and 16.7% visited chiropractors),^{10,11} and the current survey shows 4.6% visited osteopaths compared to 16.1% who visited chiropractors).

According to the second Osteopathic Survey of Health Care in the US in 2000, the prevalence of the use of osteopathy then was 7%, with a lifetime use of 16%.¹⁹⁷ It is important to note that osteopathic medicine was not included as a form of CAM in the US surveys.^{1,3,4} Practitioners of osteopathic medicine in the US are fully trained as doctors of osteopathic medicine (D.O.). They are licensed medical physicians with the same professional level as other physicians in the US health system.¹¹⁶ According to the United States Code of Federal Regulations, the practice of osteopathic medicine is included in the definition of allopathic family medicine (commonly known as Western medicine).¹⁹⁸ Thus, osteopathy practitioners in the US can do anything that a conventional medical doctor can do, in all 50 states in the US.¹⁹⁹ For historical and other reasons, osteopathy treatment may not be considered as a form of CAM in the US.²⁰⁰ However, it is still classified under one of the five NCCAM domains of CAM (i.e. the umbrella of manipulative and body-based practices).³⁶ In most countries, including Australia, osteopathy is commonly included as part of the CAM definition. This definition is used in the current study.

Unlike the small variations in the use of chiropractic between different states in Australia, the prevalence of osteopathy in different states seemed to vary significantly. The proportion of adults in the states of South Australia and Western Australia who used osteopathy was lowest among the other states. This finding, however, must be interpreted with caution, due to the small number of survey participants who had used osteopathy across all states.

Osteopathy, chiropractic and other manipulative therapies, including certain allied health professions (e.g. physiotherapy), share some common functional techniques, such as mobilisation. The current survey revealed that most commonly, about two thirds of osteopathy users considered the forms of osteopathy treatment they received were massage-like. In addition, at least half of osteopathy users also described the treatment as mobilisation, "popping of joints", strength, or cranial therapy. Thus, a variety of the techniques used by osteopaths are highly associated with techniques that deal with musculoskeletal systems. A US national survey with osteopathy practitioners revealed that the of majority practitioners used the high-velocity thrust technique, and, together with the soft tissue technique, these were the most frequently used.²⁰¹ However, high-velocity thrust, was not mentioned by osteopathy users in the current study. This may be due partially to the technical term, known by the practitioners rather than patients.

Confirming a common finding,^{183,202} the current survey also found that about half of the osteopathy users saw osteopaths for the purpose of back pain or back-related problems. The current survey estimated that those who had visited a GP for back problems were more than four times more likely to use osteopathy. On the other hand, among all demographic factors that related to osteopathy use, it is of interest to note the lowest prevalence was among those with a middle range of household income (A\$20,000-A\$40,000). However, such information

was not commonly reported in previous CAM studies, including the noted CAM study in the US,¹⁹⁷ and an Australian osteopathic study.¹⁸³

Similar to the other CAM therapies, the costs related to osteopathy use in Australia are still borne mainly by the consumers. Very little research has been published on the economic aspects of osteopathy therapy, compared to such literature on acupuncture and chiropractic. Despite the legal status of osteopathic practitioners who form part of the managed health-care system in the US, it was suggested that studies on the cost-effectiveness of osteopathy are not consistent with conventional medical literature.²⁰³

Osteopathy users in a national survey in the US considered osteopathy was more beneficial for musculoskeletal conditions than other internal conditions.²⁰⁴ The overall satisfaction of patients who attended ambulatory osteopathy clinics was found to be significantly associated with the relief of pain or discomfort.²⁰⁵ Consistently, over three-quarters of osteopathy users in the current survey considered osteopathy to be very helpful in terms of relieving pain. The surveyed osteopathy users also mentioned the usefulness of improving normal daily activities and general well-being. However, the specific benefits after osteopathy treatment are poorly understood in Australia.

In the current survey, a few osteopathy users reported they had experienced discomfort after treatment, such as pain, soreness and tiredness. As noted above, such reports deviate from the commonly discussed adverse events related to manipulative therapies. Issues relating to the safety of users are important in promoting the use of osteopathy treatment among the general population, and should be properly addressed in future integrative medical care in Australia. Again, very little literature is available.

Similar to the overall use of CAM and the use of chiropractic and acupuncture in the current survey, the most common referral channel for using osteopathy was from a user's friends or relatives. Only one in five osteopathy users had received advice from other CAM practitioners and about one in six had received advice from a GP. These data were self-reported by the survey participants. The referral pattern between osteopathy practitioners and GPs, as well as between osteopaths and other CAM practitioners, is unknown in the Australian health context. Such referral information to osteopaths, as well as to other CAM practitioners is significant, in terms of the implications for health resources planning and of safety interest to consumers, practitioners and the government.

## CHAPTER 7. FACTORS RELATED TO CAM USE

In this Chapter, matters related to CAM use, such as rationale for using CAM, benefits and risks related to CAM use, concurrent use of CAM and Western medicine, health insurance coverage of CAM and, regulatory issues on CAM use are presented.

## 7.1 Results

## 7.1.1 Rationale for CAM Use

All CAM users were asked to respond to seven pre-defined statements about complementary medicine and its comparison with Western medicine. These statements were based on previous studies on the rationale for CAM use. Survey respondents agreed/disagreed with such statements, or provided a neutral answer ("no comment").

Nearly half (44.1%) of the CAM users agreed that Western medicine would improve their medical conditions but would not cure them. This proportion was not significantly different among different socio-demographic groups of people. However, it is not surprising that those users with a relatively poor health condition tended to agree more with this statement than those in better health (55.3% vs. 42.0%,  $\chi^2$ =9.76, p=0.008).

Nearly one in nine users (11.5%) considered Western medicine would not improve their medical conditions, although over three quarters (75.4%) disagreed with this. Among those who considered Western medicine would not improve their medical conditions, there was no difference between those who had visited a medical doctor in the 12-month preceding the survey and those who had not visited one (11.1% vs. 13.3%,  $\chi^2$ =0.983, p=0.612).

Of particular concern is the proportion (24.8%) of participants who considered the services provided by their GPs as generally unsatisfactory. Typically, a relatively higher proportion was reported by those without post-secondary education (28.9% vs. 22.0%,  $\chi^2$ =7.662, p=0.022), those unemployed (30.8% vs. 22.0%,  $\chi^2$ =8.808, p=0.012), those having a lower income range (34.3%, 32.4% and 19.4%, respectively, among people with incomes of less than A\$20,000, between A\$20,000 and A\$40,000; and higher than A\$40,000,  $\chi^2$ =17.307, p=0.002), those without private health insurance (32.2% vs. 19.9%,  $\chi^2$ =15.193, p=0.001), and those with a poor health condition (35.9% vs. 22.8%,  $\chi^2$ =9.366, p=0.009).

Over half of all CAM users (56.4%) considered that Western medicine had a relatively high risk of side effects, while over three quarters (76.0%) considered CAM to be relatively safe. On the other hand, among all users, approximately 5.4% considered neither Western medicine nor CAM as safe. There was no significant difference among gender and age groups with respect to these safety statements. However, fewer old people agreed that CAM is relatively safe, while a relatively high proportion of them did not respond to this statement (p<0.05).

Over three quarters (75.9%) and over four fifths (80.5%) of CAM users considered that CAM provides a more holistic approach to health or is a more natural approach to treatment. Most significantly, the proportions of females or of younger adults who agreed with both these statements were much higher than males or older people. For example, 83% of females agreed that CAM is a holistic approach, while only 67.4% of males agreed with this statement (p<0.05). In total, 3.6% disagreed that CAM provides more holistic care or a natural approach to treatment, with the majority of them (83.2%) being male participants.

## 7.1.2 Regression Analyses on CAM Use

In Chapters 4.1.4.2 and 4.1.5.3, cross-tabulation analyses and tests of significance were carried out with each of the variables considered one at a time, in order to identify the general pattern of the overall association of participants' characteristics and their CAM utilisation status. Using the same approach (see Chapter 3.6.4.3), multivariate regression analyses on CAM use and on visits to CAM practitioners were conducted.

## 7.1.2.1 Variables that Related to the Use of CAM

In the initial step, the inter-correlations of potential predictor variables were examined (Appendix G11). In general, the inter-correlations of these variables are not highly correlated with each other, and thus fulfil the general requirement of avoiding collinearity prior to regression analysis. Similarly, the inter-correlations among other blocks of predictor variables and CAM use were also examined, with similar findings. Secondly, guided by the findings of the characteristics associated with CAM use (see Table 4.5), a bivariate analysis of all predictor variables and CAM use was conducted by cross-tabulation chi square tests (Table 7.1). It shows that 11 variables (where p<0.05) are significantly related to CAM use. This will assist the development of the regression model. Detailed statistics are presented in Appendix G12.

The development of the regression model proceeded by entering stepwise (based on Wald statistics), four blocks of predictor variables (where significant at the bivariate level): block one was demographic factors; block two, health factors; block three, external factors and block four, the 18 specific health conditions for which participants had visited GPs (Table 7.1). The best-fit significant results of the multivariate analysis are presented in Table 7.2.

230

After all demographic variables were statistically controlled, it is apparent that gender, age and educational background are significant predictors for CAM use. Female participants were 1.9 (95% CI: 1.4 to 2.6) times more likely to use CAM than male participants. Similarly, the odds of those aged 18-34 being CAM users was 1.8 (95% CI: 1.1 to 3.0) times higher than those aged 65 and older. In addition, people with post-secondary education were 2.0 (95% CI: 1.5 to 2.7) times more likely to use CAM than those without post-secondary education (Table 7.2).

After entering the second block of predictor variables, the above noted demographic results remained significant (Table 7.2). Additional predictor variables were estimated. Those participants who had consulted a GP in the preceding 12 months were 1.7 (95% CI: 1.5 to 2.8) times more likely to use CAM than those who had not visited a GP in the preceding 12 months. Consistent with this, those participants who had consulted a GP four or more times in the preceding 12 months were also 1.4 (95% CI: 1.0 to 2.1) times higher than those had visited a GP less than four times. The percentage of accurate classification (PAC) increased slightly from 70.1 to 70.7.

The results in block three further suggest that those agreed with the statement of "insurance coverage of CAM influences the decision to purchase health insurance" were 3.0 (95% CI: 2.2 to 4.2) times more likely to use CAM than those who did not agree with the statement, after all other predictor variables were statistically controlled. The Omnibus Tests, which provided an indication of how well the model executes, increased to a chi square value of 102.823 (df=7, p<0.0001). The percentage of accurate classification further increased to 72.0 (Table 7.2).

Predictors	$\chi^2$ (degree of freedom)	p value
Demographic factors (block one)		
Gender (female=0, male=1)	15.119 (1)	0.0001
Age group (18-34 year old, 35-64, 65+)	17.494 (2)	0.0002
Residential area (New South Wales, Victoria, Queensland, South Australia, Western Australia)	4.604 (4)	0.33
Education (post-secondary education: yes/no)	20.811 (1)	< 0.0001
Employment (yes/no)	10.330(1)	0.001
Income ( <a\$20,000, a\$20,000-a\$40,000,="">A\$40,000)</a\$20,000,>	10.758 (2)	0.005
Country of birth (Australia=1, overseas=0)	2.315 (1)	0.128
Health factors (block two)		
Self-rating health status (excellent/very good/good, fair/poor)	0.659 (1)	0.417
Visited a GP in past 12 months (yes/no)	10.468 (1)	0.001
Visited GP 4 or more times in past 12 months (yes/no)	6.434 (1)	0.011
External factors (block three)		
Covered by private health insurance (yes/no)	6.458 (1)	0.011
Coverage of CAM influences insurance purchase (ves/no)	84.285 (1)	< 0.0001
Believe CAM should be regulated as rigorously as Western	1.987 (1)	0.159
medicine (yes/no)		
Health conditions (block four)		
Visited GP for 18 specific health problems (dichotomous measured)	res: yes/no):	
1. Health check-up	2.657 (1)	0.103
2. Arthritis	0.451 (1)	0.502
3. Asthma	0.001 (1)	0.980
4. Back problems	2.358 (1)	0.125
5. Cancer	2.804 (1)	0.094
6. Cold/flu/fever	0.491 (1)	0.484
7. Depression	1.070 (1)	0.301
8. Diabetes	1.867 (1)	0.172
9. Gastrointestinal	3.551 (1)	0.060
10. Gynaecologic	0.349 (1)	0.555
11. High blood pressure	1.906 (1)	0.167
12. High cholesterol	0.003 (1)	0.986
13. Heart problems	0.106(1)	0.744
14. Lung problems	4.243 (1)	0.039
15. Other pains	1.157 (1)	0.282
16. Pregnancy	1.746 (1)	0.186
17. Skin problems	0.004 (1)	0.951
18. Trauma/injury	5.477 (1)	0.019

# Table 7.1 Bivariate analyses of predictor variables for CAM use

Predictors	Block 1, odds ratio & (95% CI)	Block 2, odds ratio & (95% CI)	ratio & Block 3, odds ratio & (95% CI)	
Gender				N/S
Male	1.0	1.0	1.0	
Female	1.885 (1.382-2.570)**	1.678 (1.221-2.305)**	1.612 (1.162-2.235)**	
Age (year)				
18-34	1.0	1.0	1.0	
35-64	0.726 (0.507-1.039)	0.658 (0.456-0.950)*	0.718 (0.492-1.049)	
65+	0.545 (0.336-0.884)*	0.433 (0.261-0.716)**	0.525 (0.312-0.883)*	
Post-secondary education				
No	1.0	1.0	1.0	
Yes	1.995 (1.465–2.717)**	2.058 (1.504-2.816)**	1.898 (1.372-2.618)*	
Consulted GP in the past 12 months				
No		1.0	1.0	
Yes		1.686 (1.118-2.541)*	1.586 (1.037-2.427)*	
Consulted GP $\geq$ 4 times in the past 12 months				
No		1.0	1.0	
Yes		1.447 (1.009–2.075)*	1.590 (1.096-2.306)*	
Coverage of CAM influences insurance purchase				
No			1.0	
Yes			2.991 (2.155-4.150)**	
Omnibus Tests of Model Coefficients				
Chi square (degree of freedom, p value)	42.142 (4, <0.0001)	57.912 (6, <0.0001)	102.823 (7, <0.0001)	
Percentage accurate in classification	70.1	70.7	72.0	

# Table 7.2 Multivariate analysis of predictor variables for CAM use

* p<0.05; ** p<0.01; N/S: no significant variable identified; CI: confidence interval.

## 7.1.2.2 Variables that Related to Visiting CAM Practitioners

As shown in Table 7.3, 12 predictor variables are significantly related to visiting CAM practitioners at a bivariate level. Table 7.4 presents the multivariate analysis results. Similar to the findings on CAM use, for visiting CAM practitioners, it was shown that, in addition to gender and educational background, household income ranges and participants' country of birth are significant predictors for CAM practitioner visits. Among these predictor variables, household income appeared to be the most significant factor. Participants with an income range between A\$20,000 and A\$40,000 were 2.0 (95% CI: 1.2 to 3.2) times more likely to use CAM than participants with an income of less than A\$20,000. In addition, the odds of those having an income higher than A\$40,000 being CAM users was 2.3 (95% CI: 1.5 to 3.5) times higher than those with an income of less than A\$20,000.

Again, similar to the findings on CAM users, after entering block 2 and block 3 variables, two variables were found to contribute significantly to the model for visits to CAM practitioners: consulted a GP in the past 12 months and how insurance coverage of CAM influences insurance purchase. Among demographic, health and external factors, the influence of insurance coverage of CAM on the purchase of health insurance was the most significant factor associated with visiting a CAM practitioner. Those who believed that insurance coverage of CAM would influence their purchase of health insurance were 3.5 (95% CI: 2.6 to 4.7) times more likely to visit CAM practitioners than those without such a belief.

The above regression model (Table 7.4, block four) also suggested that two additional factors were associated with CAM practitioner visits. People who had consulted a GP for back problems or for trauma/injury problems were 2.0 to 2.6 times more likely to have visited CAM practitioners than people with other specific health conditions.

Table 7.3 Bivariate analyses of predictor variables for CAM practitioner visits				
Predictors	$\chi^2$ (degree of freedom)	p-value		
Demographic factors (block one)				
Gender (female=0, male=1)	11.510(1)	0.001		
Age group (18-34 year old, 35-64, 65+)	8.749 (2)	0.013		
Residential area (New South Wales, Victoria, Queensland, South Australia Western Australia)	2.497 (4)	0.645		
Education (post-secondary education: yes/no)	24.114 (1)	< 0.0001		
Employment (yes/no)	3.700 (1)	0.054		
Income ( <a\$20,000, a\$20,000-a\$40,000,="">A\$40,000)</a\$20,000,>	17.836 (2)	0.0001		
Country of birth (Australia=1, overseas=0)	7.467 (1)	0.006		
Health factors (block two)				
Self-rating health status (excellent/very good/good, fair/poor)	0.515 (1)	0.473		
Visited a GP in past 12 months (ves/no)	17.077 (1)	< 0.0001		
Visited GP 4 or more times in past 12 months (ves/no)	7.712 (1)	0.005		
External factors (block three)				
Covered by private health insurance (ves/no)	13.616(1)	0.0002		
Coverage of CAM influences insurance purchase (ves/no)	101.357 (1)	< 0.0001		
Believe CAM should be regulated as rigorously as Western	2.313 (1)	0.128		
medicine (yes/no)	(1)	01120		
Health conditions (block four) Visited GP for 18 specific health problems (dichotomous measured)	res: yes/no):			
1. Health check-up	0.394 (1)	0.530		
2. Arthritis	0.131 (1)	0.718		
3. Asthma	0.039 (1)	0.844		
4. Back problems	8.683 (1)	0.003		
5. Cancer	6.468 (1)	0.011		
6. Cold/flu/fever	0.161 (1)	0.688		
7. Depression	0.959 (1)	0.327		
8. Diabetes	0.007 (1)	0.935		
9. Gastrointestinal	0.289 (1)	0.591		
10. Gynaecologic	0.007 (1)	0.931		
11. High blood pressure	2.935 (1)	0.087		
12. High cholesterol	1.086 (1)	0.297		
13. Heart problems	0.000(1)	0.995		
14. Lung problems	0.521 (1)	0.470		
15. Other pains	0.007 (1)	0.935		
16. Pregnancy	1.481 (1)	0.224		

17. Skin problems

18. Trauma/injury

0.770

0.001

0.088 (1)

11.761 (1)

Predictors		Block 1, odds ratio & (95% CI)	Block 2, odds ratio & (95% CI)	Block 3, odds ratio & (95% CI)	Block 4, odds ratio & (95% CI)
	Male	1.0	1.0	1.0	1.0
Gender	Female	1.82 (1.36–2.44)**	1.67 (1.25-2.25)**	1.62 (1.20-2.21)**	1.67 (1.23–2.29)**
	No	1.0	1.0	1.0	1.0
Post-secondary education	Yes	1.84 (1.36–2.49)**	1.85 (1.37-2.51)**	1.70 (1.24–2.40)**	1.72 (1.25–2.36)**
	<a\$20,000< td=""><td>1.0</td><td>1.0</td><td>1.0</td><td>1.0</td></a\$20,000<>	1.0	1.0	1.0	1.0
	A\$20,000-A\$40,000	1.98 (1.23-3.18)**	2.03 (1.26-3.27)**	1.80 (1.09–2.96)**	1.77 (1.08–2.93)*
Income	>A\$40,000	2.31 (1.53-3.50)**	2.41 (1.59-3.67)**	2.21 (1.43-3.41)**	2.21 (1.42-3.42)**
	Overseas	1.0	1.0	1.0	1.0
Country of birth	Australia	1.53 (1.08–2.16)*	1.51 (1.06–2.14)*	1.67 (1.16-2.42)**	1.68 (1.16–2.44)**
Consulted GP in the past	No		1.0	1.0	1.0
12 months	Yes		1.97 (1.34-2.90)**	1.99 (1.33-2.97)**	1.82 (1.21-2.74)**
Coverage of CAM	No			1.0	1.0
influences insurance purchase	Yes			3.50 (2.58-4.74)**	3.52 (2.59-4.79)**
	No				1.0
Back problems	Yes				2.61 (1.05-6.50)*
	No				1.0
Trauma/injury	Yes				2.05 (1.02-4.12)*
Omnibus Tests of Model Coefficients					
Chi square (degree of f	reedom, p value)	59.12 (5, <0.0001)	71.43 (6, <0.0001)	140.00 (7, <0.0001)	149.11 (9, <0.0001)
Percentage accurate in classification		60.1	62.4	67.5	68.1

# Table 7.4 Multivariate analyses of predictor variables for CAM practitioner visits

* p<0.05; ** p<0.01; CI: confidence interval

## 7.1.3 Use of CAM and Western Medicine

#### 7.1.3.1 Use of Western Medicine

Slightly over 80.8% of respondents reported that they had visited a general practitioner (GP) or medical specialist in the 12 months prior to the survey (hereafter, visits to a GP). The mean number of visits to a GP per respondents in this survey was 5.53 times in the 12-month period. When this figure was projected to the whole Australian adult population, the total estimated visits to GPs were 69.29 million times.

Among these respondents who had visited a GP, almost one fifth (19.1%) consulted a GP only for a general health check-up, to receive a prescription or immunization and/or, for travel or work-related medical examination purposes. Over six in ten respondents (60.1%) visited for a single medical problem and 20.8% for at least two medical problems. The most common medical problems for which respondents visited a GP were cold/flu or virus infection (12.2% of total visits), (high) blood pressure (10.3%), injury and trauma (6.6%), arthritis and gout (5.4%), and heart problems (5.1%).
## 7.1.3.2 Concurrent Use of CAM and Western Medicine

Almost two out of five participants (38.1%) had visited both a medical doctor and a CAM practitioner in the preceding 12-month period, whereas 42.7% had visited a medical doctor only and 6% had visited a CAM practitioner only. The remaining 13.2% of participants had not visited a medical doctor or a CAM practitioner (Figure 7.1). A relatively high proportion of people who had visited a medical doctor for back problems or trauma/injury had also visited a CAM practitioner (71% and 66% respectively).

Among all participants, those who had visited a medical doctor in the preceding 12 months were more likely to have also used any one of the 17 forms of CAM (71.1%) than those who had not visited a medical doctor (59.5%,  $\chi^2$ =10.468, p=0.001). Consistently, those participants who had visited a medical doctor in the preceding 12 months were more likely to have also visited a CAM practitioner (47.2%) than those who had not visited a medical doctor (31.2%,  $\chi^2$ =17.077, p<0.001).



Figure 7.1 Visits to CAM and Western medical practitioners

WM: Western medicine; CAM: complementary and alternative medicine

With respect to concurrent use, all identified CAM users were asked to respond to whether they had always used CAM alone, always used it together with Western medicine or sometimes used the two together. Approximately one in five (21.8%) CAM users had always used CAM and Western medicine together, only one in eight (13.2%) CAM users always used CAM alone, and over half (54.1%) had used CAM sometimes together with Western medicine (the remaining 10.9% did not respond to this question). The total proportion of CAM users who mentioned that they sometimes or always used CAM and Western medicine together (75.9%) was similar to the above noted proportion within CAM users who had also visited a medical practitioner in the preceding 12 months (71.1%).

Cross-tabulation analyses further revealed that the most significant difference of always using CAM together with Western medicine had been found between different age groups (Figure 7.2). That is, a relatively higher proportion (37.9%) of those aged 65 or older had always concurrently used both CAM and Western medicine than those younger groups ( $\chi^2$ =22.079, p<0.001).



Figure 7.2 Concurrent use of CAM and Western medicine among different age groups

Note: proportions were calculated excluding people who did not respond to this question.

## 7.1.3.3 Communication between Consumers and Practitioners

Among the 865 participants who had visited a medical doctor in the preceding 12 months, almost three quarters (71.2%) were CAM users. Of these, less than half (44.9%) indicated that they always informed their medical doctors about their use of CAM, but almost one in five (17.9%) participants never informed their medical doctors about their CAM usage. An additional one third (30.3%) had sometimes informed their GPs.

Significantly, female CAM users were more likely to discuss their CAM use with their GPs than male users were (52.2% vs. 40.4%), whereas a relatively higher proportion (27.4%) of male users than female users (14.5%) had never done this ( $\chi^2$ =18.279, p<0.001, Figure 7.3). Perhaps surprisingly, users who were unemployed at the time of the survey were more likely to always inform their GPs than those who were employed (55.0% vs. 42.9%,  $\chi^2$ =8.793, p=0.012). Surprising also, a higher proportion (59.2%) of older users (aged 65 and older) always informed their GPs than those in the younger age groups (35-64 years, 49.4% informed a GP; 18-34 years, 37.8% informed their GPs,  $\chi^2$ =18.870, p=0.001).



Figure 7.3 Gender difference in informing general practitioners about CAM use

Note: proportions were calculated excluding people who did not respond to this question.

The most common reasons given by the surveyed CAM users for discussing their use of CAM with their medical doctors were that they considered it necessary to do so (66.9%). Nearly one quarter (23.3%) of the CAM users had received advice from their GPs to use CAM, whereas an additional 13.9% informed their GPs because their GPs had asked about their CAM use. Interestingly, the only significant difference in the proportions of informing GPs among different socio-demographic groups was in users' employment status. A relatively higher proportion of those employed (16.0%) than those unemployed (9.4%) had informed their GPs because their GPs had asked them about their GPs use ( $\chi^2$ =4.093, p=0.043).

The most common reasons given by the CAM users in the survey for not discussing their use of CAM with their medical doctors were that they did not consider it necessary to do so (58.1%), and that they had not been asked by their doctors (22.4%). In addition, 10.1% of the CAM users considered that their GPs may not have been happy to know that their patients had used CAM. Nearly one in ten CAM users (9.6%) further responded that they had just forgotten to mention their CAM use to their GPs. Furthermore, there were two users whose GPs had advised them not to use CAM; the specific reason for this was not collected.

Not surprisingly, male users (70.8%) did not consider it necessary to inform their GPs compared with (47.8%,  $\chi^2$ =6.926, p=0.008) female users. Not unexpectedly, more female CAM users (21.3% compared to 3.2%,  $\chi^2$ =10.69, p=0.001, of male users) were concerned about their GPs being unhappy if they knew about their patients' use of CAM. The view that GPs may not be happy was more common among those without post-secondary education (17.1%) than those with post-secondary education (5.8%,  $\chi^2$ =4.228, p=0.040).

## 7.1.4 Health Insurance and the Use of CAM

#### 7.1.4.1 National Health Insurance Coverage of CAM

Currently, the national health care system (Medicare) in Australia does not cover the costs related to nearly all CAM products and practitioners, with the exception of acupuncture treatment provided by a registered medical doctor and very limited allied health services (including chiropractic and osteopathy treatments) for patients with chronic diseases. All survey participants were asked about their perceptions towards the possible extension of Medicare coverage to the four common provider-based and regulated CAM therapies: acupuncture, Chinese herbal medicine, chiropractic and osteopathy. Participants were informed prior to the questions that "realising there is limited amount available to fund Medicare, do you think ...?". This was an effort to minimise the possible bias of "more free coverage is always welcomed", as commented on by some researchers prior to the survey.

#### Visits to Acupuncturists

Over three quarters (82.4%) acupuncture users responded that Medicare should cover acupuncture treatment provided by acupuncturists, in addition to that provided by GPs. The view was also supported by a large proportion of survey participants (71.4%). Of this number, 11.9% said this was conditional on the acupuncturist being registered with the relevant government body. Cross-tabulation further revealed that the biggest variation in proportions of participants responding affirmatively to this question was regional difference. In general, South Australia residents did not consider that Medicare coverage be extended to acupuncture (60.3% compared to 76.1% in New South Wales, 79.7% in Victoria, 84.5% in Queensland, 78.5% in Western Australia,  $\chi^2$ =18.898, p=0.001). However, among acupuncture users, this regional difference was not found ( $\chi^2$ =3.479, p=0.481). No other additional marked difference was found in other demographic information.

## Visits to Chinese Herbal Medicine Practitioners

With respect to Medicare coverage of Chinese herbal medicine (CHM) practitioners, over half of participants (57.4%) considered Medicare should also cover visits to CHM practitioners; 8.4% of this number believed this should be conditional on the CHM practitioner being registered with the relevant government body. It is worth noting that the Medicare coverage to CHM practitioners was demanded by vast majority of the current CHM users (90.2%). Crosstabulation analysis further revealed that several demographic characteristics were significantly related to the participants' response to this question. Notably, those (69.1%) without postsecondary education were more likely to suggest coverage than those with post-secondary education (58.7%,  $\chi^2$ =11.103, p=0.001). In contrast, the proportion (57.6%) of participants with household income more than A\$40,000 who expressed a positive coverage was less than the proportion of those with a lower household income (A\$20,000-A\$40,000, 71.5%, and <A\$20,000, 67.0%,  $\chi^2$ =12.83, p=0.002).

Among different age and gender groups, the proportions of participants responding to the question on national health insurance coverage of CHM were significantly different. Female adults (66.1%) were more in favour of extending the coverage to CHM than male adults (60.0%,  $\chi^2$ =3.856, p=0.05). In contrast, the younger adults (aged 18-34, 56.0%) were less in favour of extending the coverage to CHM than the older people (35-64, 66.0% and 65+, 67.6%,  $\chi^2$ =9.830, p=0.007). Among CHM users, a marked difference was not found, or the sample size was too small, to draw a significance finding on cross-tabulation analysis.

As with acupuncture, a relatively high proportion of participants considered it necessary for chiropractic treatment to be covered by Medicare (73.6% among all participants, or 87.1% among current chiropractic users). As could be expected, a higher proportion of positive suggestions came from those without private health insurance coverage (83.2% compared to those with private health insurance, 73.9%,  $\chi^2$ =12.287, p=0.0005), and among those with a poor or fair health condition (84.8% compared to those with better health, 76.9%,  $\chi^2$ =4.658, p=0.031). Furthermore, those without post-secondary education were more likely to be in favour of Medicare coverage of chiropractic than those with post-secondary education (83.7% vs. 74.0%,  $\chi^2$ =13.705, p=0.0002). In general, there was no regional difference between the proportion of people who agreed and disagreed with Medicare coverage for chiropractic. However, among those who had used chiropractic, no such marked socio-demographic differences were identified.

#### Visits to Osteopaths

Regarding Medicare coverage of osteopathy treatment, compared to the above three CAM therapies, a relatively lower proportion (56.7%) of survey participants considered it should be covered. Among users, the proportion was increased to 85.8%. Marked differences were found in participants' responses towards osteopathy treatment. Similar to covering chiropractic treatment, a higher proportion of positive suggestions came from those without private health insurance (74.7% compared to those with private health insurance, 64.6%,  $\chi^2$ =10.42, p=0.001), and among those without post-secondary education (74.5%) than those with such education (65.1%,  $\chi^2$ =8.884, p=0.003). When it came to the monetary aspect, it was not unexpected that the proportion (81.3%) of lower-income participants (<A\$20,000) who were positive about Medicare coverage was much higher than those among the higher-income participants (A\$20,000-A\$40,000, 73.0%, and >A\$40,000, 64.1%,  $\chi^2$ =15.979, p=0.0003).

Similar to acupuncture, there is a significant regional difference of the proportions of people who considered Medicare should cover osteopathy. In general, South Australia and Western Australia residents were more reluctant to suggest that Medicare coverage should extend to osteopathy (57.3% and 61.4%, respectively, compared to 70.2% in New South Wales, 73.2% in Victoria and 72.2% in Queensland,  $\chi^2$ =9.770, p=0.044). For cross-tabulation analysis among osteopathy users, the sample size is relatively small to draw a significant conclusion.

## 7.1.4.2 Private Health Insurance Coverage of CAM

Information on participants' private health insurance coverage (PHI) status, has been analysed earlier as one of the key demographic characteristics in making comparisons of a series of questions on the possible correlation of private health insurance and the utilisation on CAM. Subgroup analyses on private health insurance among users of 17 specific forms of CAM, as well as on those who had visited relevant CAM practitioners were presented in Table 4.5, Table 4.6, Table 4.10 and Table 4.11. In addition, participants' insurance claim history on CAM, as well as detailed opinions on private health insurance coverage of four regulated forms of CAM: acupuncture, Chinese herbal medicine, chiropractic and osteopathy are presented.

Overall, more than half (55.7%) of the survey respondents had private health insurance cover in the 12 months preceding the survey. A much smaller proportion (28.1% among all participants, or 50.4% among respondents with private health insurance cover) indicated that their health insurance provided rebates on their expenditure on CAM, and approximately one quarter (25.4%) of the participants did not remember or were not sure about CAM coverage. Among all participants, less than one in seven (13.9%, or 24.9% among participants with private health insurance cover) actually received some form of CAM rebate from their insurance companies in the preceding 12 months.

Among the 150 participants who had received health insurance CAM rebates, nearly half (49.5%) were for visits to chiropractors, one in three (35.2%) were for Western massage therapy. Visits to acupuncturists were the third (18.8%) most popular form of complementary therapies for which CAM users claimed a health insurance rebate, closely followed by claims for visits to naturopaths (12.2%) and osteopaths (11.2%). A number of CAM users had also

claimed for other forms of CAM or visits to CAM practitioners. These included Chinese herbal medicine (5 people), homeopathy (5), aromatherapy (5), Western herbal medicine (4), Chinese therapeutic massage (3), clinical nutrition (3), yoga (2), meditation (1) and energy healing (1). In addition, three persons had claimed for other forms of CAM, rather than the 17 investigated.

Among those who had received a CAM rebate, nearly one third (31.0%) had claimed for two or more forms of CAM in the preceding year. However, among those who had received at least one rebate from their private health funds, little variation between different demographic characteristics was found. An exception is those with post-secondary education, who were more likely to have claimed for CAM (30.9%) than those without post-secondary education (16.3%,  $\chi^2$ =14.203, p=0.0002).

Consistent with the high frequencies of people receiving rebates for chiropractic and massage therapy, people having private health insurance were more likely to choose chiropractic treatment (19.0%) than those without (12.4%,  $\chi^2$ =8.234, p=0.004), and were more likely to use massage therapy (32.5%) than those without (20.9%,  $\chi^2$ =17.625, p<0.0001).

Participants' attitudes toward the possible influencing factors that may have attracted them to purchase insurance are important considerations for the health insurance industry. Participants were asked: "If a particular private health insurance company offered coverage of CAM for no or little additional cost, would that influence you to buy insurance from that company?" Among all participants, nearly half (45.0%) agreed that coverage of CAM would influence their decision on purchasing health insurance; a further 10.4% said it would probably influence their decision, whereas the remaining 44.5% would not consider this to be an influencing factor. When examining specific characteristics of all participants, it is clear that

the younger adults with post-secondary education, employed, with private health insurance and with a higher household income would be more likely to consider that this would influence their purchase of health insurance (Table 7.5). In addition, those identified as CAM users in the current study appeared to be more influenced than non-CAM users (Table 7.5).

Characteristic	CAM coverage influence the purchase of insurance in the future		Statistical difference	
	Yes (%)	No (%)	Chi Square	р
Age (year)			35.708	< 0.0001
18-34	60.2	39.8		
35-64	50.6	49.4		
65+	31.4	68.6		
Post-secondary education			14.783	0.0001
Yes	55.9	44.1		
No	43.4	56.6		
Employment			28.817	< 0.0001
Employed	56.9	43.1		
Unemployed or not in labour force	39.2	60.8		
Private health insurance			20.681	< 0.0001
Yes	56.8	43.2		
No	42.0	58.0		
Annual household income			12.279	0.002
<a\$20,000< td=""><td>38.6</td><td>61.4</td><td></td><td></td></a\$20,000<>	38.6	61.4		
A\$20,000-A\$40,000	55.2	44.8		
>A\$40,000	54.2	45.8		
Current CAM user			84.016	< 0.0001
Yes	60.2	29.8		
No	28.1	71.9		

Table 7.5 Will CAM coverage influence the purchase of private health insurance?

Concerning insurance coverage on the four regulated forms of CAM in Australia (i.e. acupuncture, Chinese herbal medicine, chiropractic and osteopathy), all participants were asked: "Even if you have to pay a slightly higher premium, do you think visits to [name of one of the four CAM] should be covered by private health insurance, with no gap to pay?"

#### Acupuncture

A total of 70.5% of the participants considered visits to an acupuncturist should be covered by private health insurance even they had to pay a slightly higher premium; this figure includes 6.7% who advocated such coverage only if the acupuncturists were registered. However, nearly one in five (18.9%) participants believed it did not need to be covered by private health insurance. The proportion who agreed acupuncture should be covered by private health insurance increased to 81.7% among current acupuncture users, while the proportion (13.0%) who believed acupuncture should not be covered remained relatively high. Among all participants and among all acupuncture users, no marked difference was observed between different characteristics in responding to whether or not acupuncture should be covered by private health insurance.

#### Chinese Herbal Medicine (CHM)

Nearly two thirds (64.9%) of all participants considered visits to CHM practitioners should be covered by private health insurance, even they had to pay a slightly higher premium. This number included 4.1% who were in favour of such coverage only if the practitioners were registered. However, more than one quarter (25.6%) of participants believed CHM did not need to be covered by private health insurance. The proportion of participants who agreed that visits to CHM practitioners should be covered by private health insurance increased to 81.0%

among current CHM users, while the proportion (15.0%) who considered visits to CHM practitioners should not be covered was also relatively high.

Among all participants, a few marked differences were observed in the characteristics in responding to whether visits to CHM practitioners should be covered by private health insurance. Surprisingly, the proportion (75.9%) of participants without private health insurance who believed that visiting CHM practitioners should be covered was higher than the proportion among those currently having private health insurance (68.8%,  $\chi^2$ =5.887, p=0.015). On the other hand, CHM users were more likely to suggest private health insurance coverage for CHM (84.3%) than non-CHM users (70.7%,  $\chi^2$ =6.018, p=0.014).

#### **Chiropractic**

A high proportion (79.9%) of all participants considered visits to chiropractors should be covered by private health insurance, even they had to pay a slightly higher premium. However, the proportion of participants (14.1%) who believed chiropractic did not need to be covered was also relatively high. The proportion who agreed chiropractic should be covered increased to 90.4% among chiropractic users, whereas the proportion of people who believed chiropractic should not be covered decreased to 6.4%. Consistent with this, chiropractic users in the current study appeared to request chiropractic to be covered by private health insurance more than non-chiropractic users (93.4% vs. 83.4%,  $\chi^2$ =10.987, p=0.001).

Specifically, the younger adults without post-secondary education and without private health insurance were more likely to agree that chiropractic should be covered by private health insurance (Table 7.6). When examining the attitude towards chiropractic coverage among chiropractic users only, these marked differences however, did not exist.

Characteristics	Should be covered by private health insurance?		Statistical difference	
	Yes (%)	No (%)	Chi Square	р
Age (year)			6.017	0.049
18-34	88.2	11.8		
35-64	84.9	15.1		
65+	79.9	20.1		
Post-secondary education			6.797	0.009
Yes	82.5	17.5		
No	88.4	11.6		
Private Health Insurance			6.085	0.014
Yes	82.9	17.1		
No	88.4	11.6		

Table 7.6 Should visits to chiropractors be covered by private health insurance?

#### **Osteopathy**

Nearly two thirds (65.7%) of all participants believed visits to osteopaths should be covered by private health insurance, even they had to pay a slightly higher premium; 17.0% believed it did not need to be covered. The proportion of participants who agreed osteopathy should be covered by private health insurance increased to 84.2% among current osteopathy users, while one in ten (9.3%) believed osteopathy should not be covered.

The percentage of those osteopathy users in the current study who considered that osteopathy be covered by private health insurance was greater than for non-osteopathy users (90.0% vs. 78.9%,  $\chi^2$ =3.301, p=0.069, not statistically different). On the other hand, a higher proportion of participants without private health insurance (83.0%) considered that visits to osteopaths should be covered by private health insurance than those currently having private health insurance (77.2%,  $\chi^2$ =4.568, p=0.033). Among all participants and among all osteopathy users, no additional marked difference was observed between different characteristics in responding to whether or not osteopathy should be covered by private health insurance.

## 7.1.5 Regulation and the Use of CAM

#### 7.1.5.1 CAM Products and Government Regulation

Currently in Australia, the regulation of therapeutic products, including CAM products, is separated from CAM practitioner regulation (mandatory registration).

All survey participants were asked whether CAM products should be subjected to rigorous regulation, similar to the standards set for Western medicine. A total of 944 participants (unweighted figure) responded to this question, while 123 participants did not respond. Among all participants, nearly three quarters (73.6%) considered that CAM products should be regulated as rigorously as Western medicine products, whereas 15.3% did not consider this necessary. Of the 944 participants, a higher proportion of female adults (86.2%) than male participants (79.3%,  $\chi^2$ =7.977, p=0.005) were in favour of such regulation. Age and other demographic factors seemed to have little influence on the proportion of those in favour of regulation of CAM products.

Among CAM users, a very similar proportion (74.0%) agreed that CAM products should be regulated in the same way as Western medicine products. A proportion of CAM users (16.6%) did not agree with this statement, which was slightly higher than the proportion of all participants (15.3%). The above noted gender differential between people who agreed and disagreed with the regulation statement remains significant among CAM users, although such variation is smaller (84.8% among female CAM users compared to 78.0% among male users,  $\chi^2$ =5.189, p=0.023).

## 7.1.5.2 CAM Practitioner Registration

## Acupuncturists

Participants were asked to respond to a question of relatively recent mandatory registration of two types of CAM practitioners: acupuncturists and Chinese herbal medicine practitioners in Victoria. These two types of practitioners are only subjected to government registration in Victoria, however, such registration will be implemented in the states of Western Australia and New South Wales in the near future. Public attitudes towards the implementation of registration in Victoria and the necessity for implementing it in all states and territories around Australia were sought from all participants.

Well over four fifths (86.0%) believed that the mandatory registration of acupuncturists in Victoria provided for greater public safety and confidence in acupuncture. When excluding those who did not respond to this question, the biggest disagreements were between female and male participants and between people with and without private health insurance. A higher proportion (95.8%) of female participants agreed with the public safeguards provided by government regulation than male participants (91.4%,  $\chi^2$ =7.924, p=0.005), and people with private health insurance appear to be more likely to agree with government regulation (95.8%) than those without private health insurance (90.8%,  $\chi^2$ =9.898, p=0.002).

When examining the responses of acupuncture users only, there is no gender difference, whereas there is a marked difference between people having and not having private health insurance. That is, all but one acupuncture user with private health insurance (98.6%) agreed with the safety viewpoint, while only 84.6% of acupuncture users without private health insurance agreed ( $\chi^2$ =6.920, p=0.009).

Given the existence of regulation in Victoria, it is of particular interest to examine the possible regional difference in response to the question on acupuncture regulation provides public safety. However, for all participants, there is no marked difference among states; the proportion (93.8%) of Victoria participants who agreed with the statement ranges in the middle among all states (from 91.9% to 97.4%,  $\chi^2$ =3.38, p=0.496). In addition, a similar trend is also observed among acupuncture users only.

Similar to the proportion of people who believe that mandatory registration of acupuncturists in Victoria would provide greater public safety, a high proportion (86.4%) of participants believed that all states and territories should require government regulation of acupuncturists. While there was less variance of gender difference, a statistically significant difference was observed between different age groups. A relatively higher proportion (95.3%) of middle-aged (35-64) participants agreed acupuncturists in all states and territories should be regulated than younger adults (18-34, 91.1%) and older adults (65+, 91.5%,  $\chi^2$ =6.602, p=0.037). The age differential, however, did not exist among acupuncture users only.

The above noted difference between people with and without private health insurance towards the statement on public safety and confidence is also observed in the statement on the implications of government regulation on acupuncture in all states and territories. Among those participants with private health insurance, 95% responded positively, while only 90.6% of those without private health insurance ( $\chi^2$ =9.134, p=0.003) responded positively. Among acupuncture users, the positive response was 98.6% among those with private health insurance but only 84.3% among those without private health insurance ( $\chi^2$ =6.970, p=0.008). The regulation of Chinese herbal medicine (CHM) practitioners in Australia is in the same situation as that relating to acupuncturists, that is, the registration of CHM practitioners is mandatory only in Victoria. In general, visits to acupuncturists are more common than visits to Chinese herbal medicine practitioners (Table 4.8). In an attempt to validate possible similarities and/or differences in attitudes towards the regulation of acupuncturists and CHM practitioners, all participants were also asked to respond to regulatory questions on CHM practitioners, similar to that regulatory question on acupuncturists, discussed above.

Similar to the findings on the regulation of acupuncturists, well over four fifths (82.7%) believed the mandatory registration of CHM practitioners in Victoria provided for greater public safety in CHM. When excluding those who did not respond to the regulatory question, the biggest variation in responses was between female and male participants and between people with different income ranges. A higher proportion (94.3%) of female participants agreed with the public safeguard on CHM than male participants (88.9%,  $\chi^2$ =8.832, p=0.003), and those participants with middle-range household income (A\$20,000-A\$40,000) were also more likely to appreciate government regulation (96.0%) than those with an income of less than A\$20,000 (88.5%) and those with an income higher than A\$40,000 (91.9%,  $\chi^2$ =6.395, p=0.041). When examining the responses of CHM users only, none of these differences was observed.

In terms of the possible regional difference in responding to current regulation on CHM practitioners in Victoria, both among all participants and among CHM users, there was no significant regional difference. However, surprisingly, the proportion (81.8%) of CHM users in New South Wales who believed that regulation of CHM practitioners would enhance public

safety was considerable lower than the proportions in other states (95.0% to 100.0%), although it is not statistically significant ( $\chi^2$ =4.505, p=0.342).

With respect to the implementation of CHM regulation in all states and territories, a high proportion (85.0%) of participants believed all states/territories should require government regulation of CHM practitioners. However, both among all participants and among CHM users, there was no marked difference between different demographic information. Nevertheless, the proportions of CHM users in New South Wales and South Australia (77.7% and 85.4%, respectively) who considered all states/ territories should regulate CHM practitioners were considerable lower than the proportions in other states (94.5% to 100.0%), although it is not statistically significant ( $\chi^2$ =5.584, p=0.232).

# 7.1.6 Lifetime Use of CAM

In addition to the 737 (weighted percentage 68.9%) participants who have been identified as CAM users in the current study, 42 (4.1%) used prayer (when considered as a form of CAM) and three (0.34%) used other forms of CAM only. Thus, when these are included, a higher proportion (73.4%) of CAM prevalence is estimated by the current survey.

The remaining 285 participants who did not use any form of CAM (this also excluded prayer and other forms of CAM) in the 12 months preceding the survey were asked about their lifetime use of CAM. Of these, a further 39.3% had used CAM at some time. Thus, among all participants in the current survey, 83.7% had used CAM products or had visited CAM practitioners at some time in their lives. On the other hand, among the lifetime-only CAM users (i.e. not a current CAM user, 110 in total), the demographic characteristics are similar to the above-noted characteristics among those CAM users (Table 7.7).

	Lifetime CAM user		Statistical difference	
Characteristic	Yes (%)	No (%)	Chi Square	р
Gender			10.644	0.001
Female	52.2	47.8		
Male	32.4	67.6		
Age (year)			15.041	0.001
18-34	43.4	56.6		
35-64	47.1	52.9		
65+	18.0	82.0		
Country of Birth			2.026	0.155
Australia	42.9	57.1		
Non-Australia	33.4	66.6		
State			9.864	0.043
New South Wales	41.8	58.2		
Victoria	34.2	65.8		
Queensland	43.7	56.3		
South Australia	21.0	79.0		
Western Australia	57.4	42.6		
Self-reported health status			7.536	0.006
Excellent/very good/good	43.0	57.0		
Fair/poor	18.6	81.4		
Post-secondary education			0.376	0.540
No	38.5	61.5		
Yes	42.1	57.9		
Employment			8.589	0.003
Employed	47.2	52.8		
Unemployed or not in labour force	29.2	70.8		
Private Health Insurance			0.852	0.356
Yes	42.8	57.2		
No	37.3	62.7		
Annual household income			4.965	0.084
<a\$20,000< td=""><td>27.2</td><td>72.8</td><td></td><td></td></a\$20,000<>	27.2	72.8		
A\$20,000-A\$40,000	43.6	56.4		
>A\$40,000	45.8	54.2		

Table 7.7 Socio-demographic characteristics of lifetime-only C	'AM use	ers
----------------------------------------------------------------	---------	-----

# 7.1.7 Future Use of CAM

Among all survey participants, over four fifths (81.0%) reported that they would consider using CAM for their health care in the future, while a relatively small proportion (11.2%) would not consider using it; the remaining 7.7% preferred not to respond to this question or were unsure about their future use of CAM. More specifically, of the 988 participants who made a definitive answer as to whether or not they would consider using CAM in the future, those who were female, younger adults, in a better health condition, employed and with higher income range were more likely to use CAM in the future (Table 7.8).

On the other hand, among CAM users, the vast majority (93.4%) indicated they would use CAM again, whereas 3.4% users would not consider using CAM in the future. Of these 25 CAM users who would not consider using CAM in the future, none, surprisingly, had previously experienced side-effects after using acupuncture, Chinese herbal medicine, chiropractic or osteopathy. Furthermore, the proportion of people who would consider using CAM in the future increased to 95.4% of those who had visited a CAM practitioner in the previous 12 months.

Characteristic	Consider using CAM in the future		Statistical difference	
	Yes (%)	No (%)	Chi Square	р
Gender			10.717	0.001
Female	91.3	8.9		
Male	84.3	15.7		
Age (year)			35.067	0.0001
18-34	94.1	5.9		
35-64	87.8	12.2		
65+	75.3	24.7		
Country of Birth			2.670	0.102
Australia	88.8	11.2		
Non-Australia	84.8	15.2		
State			8.062	0.089
New South Wales	88.3	11.7		
Victoria	89.6	10.4		
Queensland	89.8	10.2		
South Australia	78.2	21.8		
Western Australia	87.9	12.1		
Self-reported health status			18.924	0.0001
Excellent/very good/good	89.7	10.3		
Fair/poor	77.0	23.0		
Post-secondary education			3.183	0.074
No	85.6	14.4		
Yes	89.4	10.6		
Employment			19.666	0.0001
Employed	91.3	8.7		
Unemployed or not in labour force	81.6	18.4		
Private Health Insurance			2.794	0.095
Yes	89.3	10.7		
No	85.7	14.3		
Annual household income			14.317	0.001
<a\$20,000< td=""><td>79.2</td><td>20.8</td><td></td><td></td></a\$20,000<>	79.2	20.8		
A\$20,000-A\$40,000	91.4	8.6		
>A\$40,000	90.0	10.0		

# Table 7.8 Consideration for CAM use in the future

# 7.2 Discussion

# 7.2.1 Rationale for CAM Use

Apart from the characteristics discussed above relating to CAM use, there are factors that are often reported to be associated with a high prevalence of CAM, such as "perceived holistic approach of dealing with general health and diseases by CAM products and therapies", "perceived unmet health-care needs", "perceived higher risk of conventional medicine, or perceived lower risk of non-conventional medicine", and "belief in the efficacy of conventional medicine and/or the actual experience of the efficacy of the therapies".⁴⁶

This survey asked all CAM users about their attitudes towards some common beliefs and attitudes to CAM use. However, in the survey, data on such beliefs and attitudes were not collected from those who were not CAM users. This consequently limits the extent to which the current survey can interpret the attitudes towards the use of CAM and conventional medicine. A comprehensive rationale for CAM use, in particular the rationale for not using CAM in Australia, remains unclear at this stage. Nevertheless, over half of CAM users believed Western medicine had a relatively high risk of adverse events, while less than a quarter of CAM users did not agree CAM is relatively safe. On the other hand, a relatively high proportion of CAM users (well over three quarters) agreed that CAM is a holistic or a natural approach to treatment.

When it comes to medical/health services, there is evidence of people using CAM because they have a poor perception of other health services.²⁰⁶⁻²⁰⁸ A small proportion of the current study's participants also expressed this view. Nearly one in four CAM users in this survey believed that the service provided by their GPs was generally unsatisfactory. However, again, whether such a viewpoint differed between CAM users and non-CAM users is unknown, as non-CAM users were not asked whether or not they were satisfied with medical services. The association between the attitudes towards, and experience of, Western medicine and their use of CAM is non-conclusive in the literature. Astin's US study found that negative attitudes towards, and experience of, conventional medicine were not predictive of CAM use.⁴⁶ In contrast, some earlier studies, based mainly on clinical patients, found a positive association towards CAM use.²⁰⁹⁻²¹² Thus, in a recent study, it was suggested that although being "very satisfied" with overall care was significantly associated with much less CAM use, satisfaction with doctor-patient interaction was not associated with CAM use.²¹³

There is ample evidence to suggest that those who are female and in a relatively higher social class are more likely to use CAM. This was also confirmed in the current survey. However, these characteristics of CAM users should not be confused with the rationale for CAM use. In addition, it would be more appropriate to describe attitudes and perceptions towards CAM use as factors that may relate to CAM use, rather than the "rationale" or "why" people use CAM.

Grzywacz *et al.* suggested that adults of different ages and races are using different forms of CAM to treat illnesses or prevent them. ¹⁵³ The demographic characteristics related to CAM use, such as age range and income range, refer, in fact, to individuals' spending power (income), and the fact that the current middle-age adults were highly exposed to CAM during its rapid proliferation during the past two decades.¹⁵³ Thus, rather than concluding that a group of factors may have contributed to overall CAM use, there may have been numerous factors that could have contributed to different CAM modalities. CAM therapies aimed primarily at treatment, such as alternative-medicine systems (e.g. Chinese medicine), may be better explained by a health-behaviour model, guided by scientific evidence of efficacy and safety, and the health conditions of the users. Whereas, CAM therapies that are generally used

for the promotion of health (e.g. meditation and other mind-body therapies) may be better explained by a health-belief model, guided by a holistic approach and perceptions of use. ¹⁵³

As noted above, CAM utilisation studies are commonly conducted with cancer patients. Studies on rationale or on why patients choose CAM are also more commonly conducted with cancer patients. This has been emphasised in a recent published systematic review of 52 cancer studies.²¹⁴ The review revealed that such results are more promising in terms of clinical implications of CAM use. It found that frequently reported factors influencing CAM use were the belief that conventional medical treatment (e.g. chemotherapy) had harmful side-effects,²¹⁵ that CAM could control the side-effects believed to be caused by Western medicine²¹⁶⁻²¹⁹ and that CAM was a non-toxic or less harmful therapy than Western medicine.^{63,220-224} However, it is still important to note that the definition of overall CAM is still a major barrier to a generalisation of the above findings. Thus, perhaps the rationale for CAM use may be particularly important for health-care professionals in clinical practice, if such a model of the rationale for using CAM was established on the basis of a single cohort of clinical patients.

Finally, given the increasing emphasis on scientific evidence for both CAM and Western medicine, it is important to evaluate the resources where people receive information (particularly evidence of efficacy), and what types of resources or health interventions would influence the choices of CAM therapies. Ideally, these should be based on the use of CAM for specific health conditions, to better comprehend the complex trend of health-seeking behaviours.

# 7.2.2 Regression Analyses on CAM Use

Participants' characteristics of CAM use discussed above were based on statistical analyses at a bivariate level. Therefore, such findings have not been adjusted for potential confounders that may influence the associations of CAM with some particular characteristics. Such an approach is commonly used in the early stages of exploring the factors that may relate to CAM use, or as confirmatory analyses of previous findings, which is consistent with the objectives of the current survey. On the other hand, multivariate regression analyses were conducted to further confirm previous findings in Australia and overseas, and to discover the likelihood of individual factors that were related to CAM use among adult Australians.

In terms of demographic information, gender, age and educational background were found to be statistically associated with CAM use among current survey participants. In this respect, the odds ratio of females versus males using CAM was estimated as 1.6 in the current survey (the final predictive model, see Table 7.2), which is the same as the 2000 South Australian survey,¹⁰ is very similar to an earlier 1993 South Australia study (odds ratio=1.7),⁹ and is slightly higher than a 1999 Canadian study (odds ratio=1.3).⁴⁸

The current survey also estimated that educational background is highly associated with CAM use. That is, those people with a post-secondary education are 1.9 times more likely to use CAM than those without a post-secondary education. This significant finding was also reported in the 2000 South Australia study, although the odds ratio was less significant.¹⁰ It is of interest to note that in an in-depth US study on why patients use CAM, education also emerged as the only significant socio-demographic variable that will predict CAM use.⁴⁶

In the current survey, age also emerged as another significant factor that predicts CAM use. While this was not reported in the 2000 South Australia study, ¹⁰ it had been previously estimated in a 1993 South Australia study⁹ conducted by the same group of researchers. The estimated odds ratios were similar. The oldest age range (55+ in the South Australia study or 65+ in the current survey) appeared to have the smallest odds ratio (i.e. less likely to use CAM). Again, this was consistent with the 1999 Canadian study.⁴⁸

The health behavioural model (HBM)²²⁵ is one of the conventional models used to study health behaviour as a predictor of health care use. Thus, this may be particularly relevant in predicting the use of CAM services. Some researchers have taken this model into consideration.²²⁶ It has been hypothesised and confirmed that individual believes the philosophic of CAM use related to holistic and natural characterises are more likely to use CAM.⁴⁶ In addition, individual with "perceived unmet health care needs" was also found to be 1.5 times more likely to be CAM users than those without such a perception. ⁴⁸ Unfortunately, in the current study, data related to health behaviour was collected from CAM users only, and not from all surveyed participants. Therefore, it is not possible to include such data into a predictive model for this study. Nevertheless, other perceptions that are possibly associated with the use of CAM by adult Australians were collected, and emerged from the regression analysis. Notably, the view of health insurance coverage of CAM influences the decision to purchase of such insurance was frequently reported only among those CAM users.

Finally, although specific health conditions were not found significantly as predictor variables in the model of CAM use, back problems and trauma/injury were found to be significantly associated with those who had visited CAM practitioners over a 12-month period. The odds ratio (2.6) of people with back problems seeking CAM practitioners in the current survey was similar to a previous US study (2.3).⁴⁶

# 7.2.3 Impact of Health Insurance on CAM Use

#### National Health Insurance

CAM has become a vast resource for health care internationally. The increasing popularity of CAM is also echoed by the increasing involvement of health insurance companies. In the US, based on a national survey,²²⁷ Wolsko reported that full or partial coverage of CAM by health insurance providers is strongly associated with high-frequency use of CAM services. In the United Kingdom, there is a growing trend for the National Health Service to cover complementary medicine services.²²⁸

As CAM becomes widely available and acceptable in Australia, there is greater public interest in CAM being included under the cover provided by the national health insurance system (Medicare). For example, a study conducted over 10 years ago found that acupuncture services in Australia have risen steadily over the past two decades, and 15.1% of GPs in Australia have claimed Medicare rebates for acupuncture treatment.¹⁶⁷ Certainly, with the increasing evidence of a large proportion of GPs in Australia now practicing acupuncture and other CAM therapies,⁹⁵ the number of Medicare claims for acupuncture services performed by GPs is considerable. In 2002-2003, as estimated by the Australian Institute of Health and Welfare, the Australian Health Insurance Commission (HIC) provided A\$13.5 million for 595,000 acupuncture Medicare procedures performed by medical practitioners.¹⁴⁷

However, research on national health insurance and private health insurance and CAM use in Australia is extremely limited. Previous health surveys and surveys on CAM in Australia failed to investigate the possible correlation between insurance and CAM use, and did not explore the perceptions of, and attitudes towards, insurance and CAM use among Australians. Thus, it is important to note from the current survey that approximately three quarters of participants who considered visits to practitioners of acupuncture and chiropractic should be covered by the Australian national health care; even though the surveyed participants were asked to take into account that there is limited money to fund Medicare for all Australians. Well over half of all participating adults considered Medicare should also cover visits to practitioners of Chinese herbal medicine and osteopathy, regardless of their status as current CAM users.

A recent endorsement from the Australian Government is promising for the future integration of CAM in the national health system. Thus, from 1 July 2005, new items on the Medicare Benefits Schedule (MBS) will make it easier for GPs to manage the health care of patients with chronic medical conditions, including patients needing multidisciplinary care. Eligible Australian patients are now allowed to receive up to five Medicare rebates from certain allied health practitioners, including chiropractors and osteopaths.¹⁸⁷

As Pelletier and Astin pointed out, an increasing number of health providers, including hospitals, in the US are incorporating CAM to meet the high demand from consumers; however, the uncertainty about the profitability of CAM is a primary obstacle to such integration.²²⁹ Thus, a representative national survey in Australia should be carried out to assess the true consumer demand of CAM coverage and reimbursement from Medicare. Optimistically, it will derive a set of commonly demanded CAM to be included in the Medicare system, which is rapidly growing in use and which is supported by growing rigorous scientific evidence.

#### Private Health Insurance

The "legitimisation" of a number of forms of CAM practice through statutory regulation in states or nationwide in Australia, for example, chiropractic, osteopathy, Chinese herbal medicine and acupuncture, and approval by the Therapeutic Goods Administration to market certain complementary medicines, raises considerations of relative access and equity for CAM and mainstream health care. As noted, in Australia, CAM treatments do not qualify for Medicare rebates, with the exception of limited referral services for chiropractic and osteopathic treatments, and acupuncture delivered by medical practitioners accepted by the Health Insurance Commission. Nevertheless, at present, some forms of CAM have attracted limited coverage from most major private health insurance funds in Australia.

In Australia, in March 2006, nearly half (43.1%) the Australian population were covered by private health insurance.²³⁰ Most health insurance providers in Australia, including the Hospitals Contribution Fund of Australia (HCF), Newcastle Industrial Benefits (NIB) private health funds, Medibank Private Limited and the Medical Benefits Fund of Australia Limited (MBF) had benefit packages, which included rebates for some forms of CAM (generally chiropractic, remedial massage, osteopathy and acupuncture). However, detailed availability and public demand for private health insurance rebates for CAM in Australia is unknown.

In the current study, although over half (55.7%) of survey participants had private health insurance, less than one in seven (13.9%) had received a rebate from their insurance companies for CAM services in the previous 12 months. It appears that the cost of using CAM is mainly borne by individuals. With some 1.87 million Australians being recipients of the age pension (low household incomes),¹⁴⁹ the socio-economic implications of high levels of CAM use in the elderly needs to be considered.

The use of chiropractic remains one of the most substantial CAM services in Australia and overseas.^{11,231} A considerably high prevalence of chiropractic has also been found in this study, and it was for the most popular forms of CAM, together with remedial massage and acupuncture, that CAM users had lodged health insurance claims. In particular, nearly half of those people who had received private health insurance rebates on CAM were for chiropractic services. This may be partially explained by the fact that, the practice of chiropractic in Australia, which has a history of over 100 years ago,²³² and which has been subjected to mandatory registration with regulatory bodies in all states and territories of Australia, is one of the earliest and most common forms of CAM services covered by private health insurance in Australia. Perhaps, it is not surprising that, in the US, a large telephone interview study revealed that virtually all the insurance providers in three states in the northeast of the US cover chiropractic services in some form, whereas only less than half of the insurance companies cover acupuncture or massage therapy.²³³

Perhaps one can also expect that people with private health insurance are more likely to choose chiropractic and remedial massage, as at least part of these services are currently covered by private health insurance in Australia. The present study has confirmed this supposition. Similarly, it is clear that a much higher proportion of people with private health insurance are likely to choose acupuncture services than those without such coverage. Given that the mandatory registration of acupuncture practitioners has been in place in Victoria, and is soon to be introduced in some other states in Australia, the association of private health insurance and acupuncture use also needs further investigation. Among all participants in this survey, nearly half agreed that insurance coverage of CAM would influence their decision on purchasing health insurance. It is of interest to note that a four-year retrospective data analysis on insurance claims suggested that people with chiropractic insurance coverage had lower annual health-care expenditure, compared to those without such coverage.²³⁴

269

Based on existing literature, there is strong evidence that many cancer patients will seek CAM therapies if insurance covers the cost.²³⁵ More specifically, the use of naturopathy and acupuncture were more common, while the use of chiropractic was less common for cancer patients than people without cancer.²³⁵ Detailed information on which forms of CAM have been claimed for which specific health conditions were not researched in this survey. Australian literature in this area has not been reported. Nevertheless, the majority of the current survey participants suggested private health insurance should cover the common forms of CAM, such as acupuncture and chiropractic. Thus, private health insurances in Australia cover certain common forms of CAM to a certain extent (approximately A\$400-A\$600 per year for a single person). However, it is not comprehensively understood whether consumers are fully utilising such coverage and what their attitudes towards the extent of the coverage are.

Finally, the consequences (particularly the economic aspects for the consumers and insurance companies) of including CAM in private health insurance policy have been poorly researched in Australia and overseas. This suggests the need for a representative national survey in the near future to meet the demands of rapidly growing CAM use and the potential requirements of insurance coverage in Australia.

# 7.2.4 Impact of Regulation on CAM Use

A regional Australian study in 2000 suggested that there has been a steady increase in the use of CAM by the South Australian population in the previous 10 years.¹⁰ However, researchers also suggested a decline in the CAM market in 2004,¹¹ which could be attributed to consumer response to the adverse publicity generated by the TGA recalling Pan Pharmaceutical products from the market in 2003.²³⁶ In contrast to this finding, the current survey found an increase in CAM use, when compared to these South Australia studies in 1993, 2000 and 2004.⁹⁻¹¹ However, the CAM classification used in these surveys was inconsistent, the targeted population were divergent (regional and national), and importantly, the current survey suggests the lowest CAM prevalence was, in fact, found in the state where the regional studies were conducted (South Australia). ⁹⁻¹¹ Thus, whether the Pan Pharmaceutical product recall would have impacted on the popularity of CAM use in Australia is unclear. As a cross-sectional survey, the current survey is not able to confirm the trend of CAM use over time.

To strengthen the framework of CAM regulation in Australia in 2005 the Australian Government accepted most of the recommendations made by the Expert Committee on Complementary Medicines in the Australian Health System.²⁶ It is essential that a safety guide be established for the Australian general public to address continuing concerns of CAM use. Therefore, importantly, the Therapeutic Goods Administration has recently published a series of Australian regulatory guidelines for complementary medicines (ARGCM) to ensure that all therapeutic goods, including complementary medicines use and practice in Australia, are subject to rigorous standard.²³⁷⁻²⁴¹

Whether such regulatory guidelines will contribute to the future increased use of CAM is unclear and will be hard to determine. As Weir suggest, there are many factors that may contribute to the increasing use of CAM, including the higher cost of conventional medicine and the inability of science to provide a cure for all illnesses with a single form of medicine.²⁴² From a sociological perspective, Coulter and Willis also suggested that the "baby boomers" in Australia will age with a variety of chronic diseases, which are often considered to amenable to treatment by some forms of CAM.²⁴³ The increasing availability of health insurance coverage for CAM will also increase the likelihood of people choosing a CAM service.²⁴³

Some three quarters of participants in the current survey considered CAM products should be regulated as rigorously as Western medicine. This figure includes those who did not identify themselves as CAM users. This, perhaps, further reflects a demand from the general public to have safer CAM products in Australia. In addition, this also echoes the findings of a recent South Australia survey that about half the survey participants assumed that a government agency tested CAM products. ¹¹

Currently, the practice of Chinese medicine in Australia, outside the jurisdiction of Victoria, is not subject to statutory registration. Nationwide, with the exception of chiropractic and osteopathy, there is no statutory registration requirement for other CAM professions. Over the past decades, some CAM professions in Australia have developed a self-regulatory system. Such regulatory mechanisms have been established in the absence of, or to complement legislation requiring mandatory registration, and rely on voluntary compliance by members of the relevant professions and industry organisations.²⁴⁴ In Australia, the Australian Self-Medication Industry (ASMI) is the highest profile organisation representing industries involved in the manufacture and distribution of non-prescription consumer health-care products, such as over-the-counter medicines and complementary medicines.

272

In the interests of comprehending better the potential impact of the introduction of the statutory registration of Chinese medicine in a state of Australia (Victoria), this study specifically asked participants' about their perceptions toward this regulation. This study found that well over four fifths of survey participants considered that the statutory registration of Chinese medicine practitioners, including acupuncturists and Chinese herbal medicine practitioners, provided for greater public safety and confidence in these forms of CAM. Significantly, there were similar proportions of survey participants from other states and territories. Furthermore, when asked whether acupuncturists and Chinese herbal medicine practitioners should be subject to registration with government authorities in all states and territories of Australia, again, well over four fifths of all survey participants considered it necessary. Such a proportion was observed both among the Victorian participants and participants from any other states/territories. Thus, as Carlton and Bensoussan commented, the Victorian Chinese Medicine Registration Act provides a model for regulating those CAM professions where the practices are potentially intrusive and dangerous.²⁴⁴

A previous study in Melbourne in 2002 found a large proportion of survey participants would consult a Chinese medicine practitioner who was not registered, and only 20% considered that such a registration system would make them more confident about consulting Chinese medicine practitioners.⁶² As noted above, a large proportion of participants from this survey considered that statutory registration of Chinese medicine professions was important. Thus, it appears that Chinese medicine has been valued in Victoria since its introduction. Unfortunately, whether people are still prepared to consult an unregistered Chinese medicine practitioner was not asked in current study.
## 7.2.5 Risks Associated with CAM Use

The risks associated with the use of CAM are growing concerns for consumers, health professionals and governments. There is a widespread and misguided belief amongst consumers of many types of CAM products that since they are derived from natural sources they must be safe. However, herbal therapies are frequently complex mixtures of potentially biologically active ingredients. Such remedies may have multiple actions and may carry a high risk of both direct adverse effects and of adverse effects arising from interactions between their constituents and concurrent conventional therapeutic agents. It is of particular importance to note that nearly half of a survey population in Australia wrongly believed that CAM had been independently tested by a government agency before being sold.¹¹ Surprisingly, those younger than 35 years old, who were better educated and had a relatively high income comprised a higher proportion of people who held this belief.¹¹

Additional concerns relate to the quality/purity of CAM products, the training of CAM practitioners, and the possibility that patients may discontinue beneficial conventional treatments. These concerns are particularly relevant to the elderly and to victims of chronic illness, due to factors such as the high prevalence of chronic diseases of the elderly, greater use of conventional medication, and their likely greater susceptibility to adverse effects and to therapeutic interactions. These issues may be compounded by poor communication between patients, CAM practitioners and conventional health-care practitioners, and indeed, by a lack of understanding by patients and practitioners of the potential adverse effects of CAM use.

#### Adverse Events of CAM Use

The potential risks associated with the use of medicinal products are not completely avoidable. These apply to all forms of medicines and therapies, including both Western medicine and CAM. Adverse events may occur with the use of products that have problems of contamination, toxicity, lack of quality control and inappropriate labelling, ¹¹ as well as problems related to the advice provided by health professionals (it could be negligence). In addition, reports on interactions between CAM and Western medicine are not unusual.²⁴⁵⁻²⁴⁸

It is important that consumers should be aware of the potential negative effects of all medicines or treatments they are taking. Current survey participants were asked about their experiences of encountering adverse events after using acupuncture, Chinese herbal medicine, chiropractic and osteopathy (only if such a participant used one of these four forms of CAM). As noted-above, four participants claimed they experienced adverse events after acupuncture treatment. However, when the definition of adverse event relating to acupuncture was specified according to standards developed in collaboration with the British Medical Acupuncture Society,²⁴⁹ there was only one case which qualified as an adverse event (bruise).

An adverse event rate per 1,000 or per 10,000 practitioner consultations have been reported.²⁵⁰⁻²⁵² Survey participants in the current study provided an estimated incidence of 14 adverse events per 10,000 acupuncture consultations, which is much lower than the adverse event rate estimated by a prospective survey of 32,000 acupuncture consultations in the UK (i.e. an incidence of 684 per 10,000 consultations).²⁴⁹ However, the current estimation was based on self-reported cases from those surveyed adults, whereas the UK study was based on reports from acupuncture practitioners over a 21-month period. The data are not directly comparable.

In the current survey, users of CHM, chiropractic and osteopathy also provided information on adverse events that may have been related to such CAM use. For the reasons mentioned in Chapters 6.1.1.9 and 6.1.3.4, the commonly estimated incidence rate per 1,000 consultations was not available for CHM and osteopathy. The current survey participants reported a relatively high incidence (2.96 per 1,000 consultations) of adverse events related to chiropractic treatment. Comparable data reported among Australian chiropractors were not identified at the time of writing this thesis. Nevertheless, it is important to point out that safety issues have attracted continuing safety concerns in previous studies in Australia. Thus, it has been commented that the practices of forms of CAM are not risk-free, and fatalities have occurred.^{12,164,253} The incidence reported by Chinese medicine practitioners (including acupuncture and CHM) was estimated at 1.6 per 1,000 consultations.¹⁶⁴ Adverse events associated with herbal medicines, nutritional substances and homeopathic medicines was estimated at 2.3 per 1,000 consultations.¹²

In Australia, reports of suspected adverse reactions to drugs/medicines are sought from health professionals and are also received from consumers. Section 29A of the Therapeutic Goods Act 1989 requires sponsors of medicines registered or listed in Australia to report adverse reactions.²³⁷ However, as stated in the report submitted by the Expert Committee on Complementary Medicines in the Health System to the Minister for Health and Ageing, Australian Commonwealth Government, "the current adverse reaction reporting system may be biased away from complementary medicines, because complementary medicines are usually self-prescribed and problems may not be reported, should they arise".¹⁴ Recently, a special "1800" telephone number has been established within the Therapeutic Goods Administration to facilitate reporting of adverse reactions²⁶ in an effort to strengthen the collection of adverse event.¹²

276

Some consumers are so committed to CAM products and therapies that they actively selfprescribe or self-administer CAM. In the current study a high proportion of participants used CAM and conventional medical services concurrently, and that many CAM users did not inform their medical doctors (and the medical doctors did not ask) about their use of CAM. Clearly, these findings are of concern, given the increasing reports of side-effects of CAM products²⁵⁴ and other problems, such as contamination of CAM products.²⁵⁵

A Canadian study conducted over a decade ago found users of chiropractic tended to also be concurrent users of conventional health care.²⁵⁶ However, most studies in the current literature failed to investigate the concurrent use of specific forms of CAM and Western medicine.^{11,47,257-259} Perhaps, one of the biggest limitations of the current literature on the concurrent use of CAM and Western medicine is that it has failed to collect detailed information on which specific forms of CAM have been commonly used concurrently with Western medicine, and to what extent (e.g. used on the same day, for the whole duration of the illness, or only used intermittently). Addressing these limitations, in a recently published South Australia study,¹¹ CAM and Western medicine use on the same day was investigated, and the study found that about half of CAM users had concurrently used Western medicine. Due to the time and resource limitations, the current survey was not designed to collect such data.

It is worth noting that the pattern of concurrent use has been found to be related to some specific characteristics, such as gender and age.²⁶⁰ In addition, some studies have also suggested that the greater the value the participants attached to spirituality, the greater the likelihood of users to use both treatments.^{46,210,260} A national survey in the US further revealed

that adults who used both CAM and Western medicine appeared to value both, and were less concerned about their medical doctor's disapproval than about the their doctor's inability to understand or incorporate CAM therapy use within the context of their medical management.²⁵⁹ Supporting this finding, a relative lower proportion of CAM users in the current survey was concerned that their medical doctors may not approve of their use of CAM.

The concurrent use of CAM and Western medicine among those over 65 years old is becoming a matter of concern, as the proportion of concurrent was about 80%, being the highest among all ages.¹¹ Similarly, in the current survey, a significantly higher proportion of participants aged 65 and above had concurrently used both CAM and Western medicine than those of younger ages. The rationale of users for such concurrent use certainly warrants investigation in further studies. The high prevalence of concurrent use also suggests the need to educate older people to understand better the potential risks of concurrent use, and increase the awareness of voluntary communication of such use between older people and the health professionals.

The concurrent use of both CAM and Western medicine may increase risk, as the pharmacological action and therapeutic mechanism of most forms of CAM products and therapies are inadequately understood to allow health professionals to identify possible interactions. The use of CAM does not, in general, suggest dissatisfaction with conventional medicine.^{259,261} Thus, more recently, researchers have addressed the communication gap between conventional and alternative medicine.²⁶²⁻²⁶⁶ In an interest to keep patients, CAM practitioners and medical doctors informed, and to provide better guidance to patients regarding the safety and efficacy of the concurrent use of CAM and Western medicine, the disclosure of CAM use to health professionals seems to be a critical step in future integrative health-care. Thus, comprehensive information to advise patients to use CAM is particular welcome.²⁶⁷

The reported non-disclosure rates are significantly high. About half of the current survey participants had only sometimes, or never informed their medical doctors about their CAM use. A national US survey estimated that nearly two thirds (63%) of CAM users did not disclose at least one of their CAM therapies.²⁵⁹ Furthermore, qualitative studies on clinical patients also discovered a high proportion of non-disclosure. For example, a study on women with breast cancer found that nearly half (46%) did non disclose their use of CAM,²⁶⁶ while another study on cancer patients revealed 34% non-disclosure.²⁶⁴ The lowest disclosure rate, in the literature, was 23% among older, rural women.²⁶⁸

Although, as noted above, a significantly high proportion of participants aged 65 and older concurrently used both CAM and Western medicine than those in the younger age groups, these older CAM users, in fact, had a significantly higher proportion of always informed their

medical doctors of their CAM use than younger CAM users. This may be partially explained by the increasing popularity of CAM which has occurred in the past two decades, and most forms of CAM are relatively new to older people. While they are sensitive to CAM treatment to improve their health conditions, older people may be more proactive in informing their GPs, to optimise their health-care outcomes.

In the current study, the main reasons for non-disclosure volunteered by survey participants were, "medical doctors did not need to know", and the "doctor did not ask". Similar findings were commonly reported,²⁶⁹⁻²⁷¹ and have also been summarised in a recent review.⁹⁸ However, such "don't know" or "don't ask" results are no longer acceptable in modern health care, and a proactive management system is needed to approach individualised personal health.²⁶¹

It is of interest to examine the perspectives of conventional health professionals about the communication of CAM use with their patients. A study conducted in the state of Ohio, in the US, found that none of the physicians would terminate the physician-patient relationship if a patient used CAM, in addition to the prescribed conventional medicines.²⁶³ This should not be unexpected, as the use of CAM by conventional medical doctors themselves was quite high,²⁶⁵ and there are increasing debates about the benefits of conventional doctors proactively asking patients about their use of CAM.^{261,267,272}

As a greater understanding develops about the question of the interaction of CAM with conventional medicines and the matter of full disclosure of such use, various ethical and legal issues are raised.^{273,274} Even more contentious than ignoring patients' use of CAM, is the possibility that conventional medical doctors may be at risk of potential malpractice liability.²⁷⁵ For example, as scientific evidence of CAM is increasingly acceptable to the

conventional medical system, there is a strong argument that medical doctors should advise patients about the CAM treatment options, in addition to other medical treatment options, to satisfy the duty of care of a medical doctor.²⁷⁶ On the other hand, on the basis of existing US laws, "a physician referral to a CAM provider could be considered a "negligent" decision if it delays or eliminates the patient's opportunity to obtain necessary medical treatment".²⁷⁷ However, the risk of litigation for medical doctors' failure to provide adequate information on CAM to patients is probably low at this stage.²⁷⁸

In Australia, according to the General Guidelines for Medical Practitioners on Providing Information to Patients, published by the National Health and Medical Research Council, a "medical doctor should encourage patients to reflect on opinions, ask more questions, and should be assist patients in seeking other medical opinion where this is requested".²⁷⁹ Thus, in 2002, the Australian Medical Council posted a position statement (Section 4.4): "the medical practitioners should specifically ask patients about their use of complementary medicine and take account of this in their management of conditions. Medical practitioners should be sufficiently well informed about complementary medicine to be able to provide advice to patients when appropriate."²⁸⁰

In summary, the fact of concurrent use of CAM and Western medicine, and possibly other unnamed forms of medicines or therapies, is no longer uncommon in today's health-care system. The trend of such use may increase over time. A better understanding of the benefits of complementary and alternative medicine, and the potential risks of using these medicines is critical to improve the communication of such use between patients and health professionals.

### 7.2.6 Lifetime Use and Future Use of CAM

In addition to investigating the prevalence of CAM use over a 12-month period, the current survey has also investigated the lifetime usage of CAM, that is, its use at least once at any time previously. Approximately five out of six adult Australians have used CAM at some time in the past. The prevalence is comparable to a previous regional study in New South Wales, where three-quarters of the adults reported on a lifetime use of CAM.³⁸ In addition, consistent with noted regional differences of CAM use found in the current survey, the prevalence of lifetime use only (precluding current users) varied among different states. For example, the prevalence of lifetime use only in South Australia was found to be significantly lower – about half the prevalence found in New South Wales, Queensland and Western Australia. However, there may be a recall bias for lifetime CAM prevalence, as participants may have used non-Western medicine years ago, and not able to recall the details.

Lifetime use of CAM is occasionally reported among CAM surveys. In general, a significantly higher prevalence of lifetime use was reported, compared to the prevalence over a 12-month period. This is evident in several frequently cited CAM surveys. For example, the prevalence rose from 42.1% for a 12-month period to 67.6% for a lifetime period in the 1997 US national survey.²⁸¹ The prevalence of lifetime use also increased from 50% to 73% in the 1997 Canadian national survey,⁶ and from 28.3% to 46.6% in the UK survey.⁵ Furthermore, the UK and Canadian studies also provided detailed information on the lifetime use of each of the individual forms of CAM. The lifetime use of these therapies was generally consistent with the popularity of use for each modality in a 12-month period.^{5,6} Limited information of such lifetime CAM use is available in the Australian literature, and has not been investigated in the current survey.

It is apparent that the use of complementary medicine is more popular in Australia and in many Western countries than it was a decade ago. It is not hard to find reasons why this type of medicine is becoming more popular. The increasing scientific evidence of the efficacy and safety of certain CAM therapies, and perhaps the recognition of the cost-effectiveness of some specific forms of CAM would partially explain its popularity. Thus, for various reasons (not investigated in the current survey), a large proportion of the surveyed participants indicated that they would use CAM in the future, and a vast majority of the current users will use CAM again.

# CHAPTER 8. GENERAL DISCUSSION, CONCLUSIONS AND POLICY IMPLICATIONS

## 8.1 Overview of the CAM Survey

The detailed outcomes of conducting the survey are presented in Chapter 4.1. In brief, a total of 1,067 interviews were conducted with a representative cohort of adult Australians. By employing a random-digit telephone dialling method to contact potential participants, and by applying a national quota to ensure representativeness, the study population can be regarded as representative of the national Australian adult population. In addition, the representativeness of the study population is further confirmed though a series of statistical analyses (see Chapter 4.1.3).

Using a predefined quality assurance protocol, the survey information was captured during interviews and recorded in a computer-assisted telephone interview system. Thus, missing data were minimised. The validity of the data was extended after scale reliability analyses were carried out for different sections of the instrument, which confirmed its internal consistency of items measured by the scale. Finally, after all data were collected, a data-weighting process was applied before generating the survey results and performing consequent analyses. Thus, all data were adjusted to represent the national population.

Although the current study was conducted with an appropriate sample size and a national representative population, it is also important to note the study limitations when interpreting the results, in particular those concerning state differences. These are discussed below.

## 8.2 Limitations of the CAM Survey

The response rate is always a major concern for survey researchers, and is dependent on what is considered to be a completed interview. The rate can vary significantly according to some experienced survey companies. The response rate of the current survey is lower than for some other population surveys of its kind and could be considered as a significant study shortcoming. It seems likely that the employment of national quotas for gender and age-range during the recruitment stage limited voluntary participation and impacted significantly on the final response rate. This is further evidenced by the fact that when a particular quota category was filled (at an advanced stage of the recruitment process), a large number of potential participants was rejected (see Figure 4.1).

The increasing use of telemarketing from call centres making it harder for non-commercial telephone health surveys, such as the present survey. In addition, the short period of survey may have influenced the response rate. Hence, the number of call-backs was limited, and there was not enough opportunity for optimising follow-ups. A large number of unresolved call-back numbers (76 appointments in total) would also have impacted on the final response rate. Finally, recognised government authorities administer most population health surveys, and people may have a greater tendency to participate in these. The low response rate may have introduced a non-response bias. If those refused to participate were less likely to use CAM or had negative attitudes towards CAM, this could result in a decrease of the current overall estimation of CAM prevalence.

The relatively small sample size in a number of states and territories (i.e. Australian Capital Territory, Northern Territory and Tasmania) may have limited the results that can be generated at a state level. With a larger sample size, additional information would have been

generated. The most important information would be the definitive differences on regional use of CAM, and on the use of each of the 17 forms of CAM. The current survey suggests marked regional differences among states. However, the sample size was too small to draw conclusions from Tasmania, Northern Territory and Australian Capital Territory. Similarly, some questions relating to specific CAM therapy (in particular, osteopathy) were administered to a relatively small group of users and restricted the reliability of the findings.

In terms of outcome measures, perhaps the biggest limitation was that data on the specific medical conditions treated by each form of CAM were not collected. Such specific questions on each of the 17 CAM therapies would have been time-consuming, and it was impossible to collect adequate relevant data in the relatively short interview time available. Information about the medical conditions for which CAM, as a whole, is being used is of general interest but more specific data about each individual form of CAM would have been of greater interest. On the other hand, data were collected on medical conditions that resulted participants visiting a GP. Analyses were then performed to examine the association between participants having such a medical condition and visiting a CAM practitioner or using CAM. In addition, data were also collected on which acupuncture, Chinese herbal medicine, chiropractic and osteopathy were used to treat specific medical conditions.

Due to the limitation on interview times, it was not possible to obtain detailed data on the perceived benefits and risks of using each of the 17 forms of CAM. This information should be particularly valuable for regulatory bodies and policy makers. Nevertheless, general opinions on the overall helpfulness of CAM and the actual benefits and risks after using the four regulated forms of CAM were collected.

Several common problems encountered in similar CAM studies remain unsolved. Previous researchers have commented on the fact that discrepancies in CAM classification have led to discrepancies in the data relating to CAM prevalence even within a specific country.¹⁸ This creates great difficulties in interpreting survey results and comparing data from different studies.²⁸² There is no general agreement on what constitutes the totality of CAM therapy. In particular, the matter of precisely classifying different forms of vitamins and supplements as complementary medicine products or non-CAM products is critical and impacts on the overall prevalence significantly. The argument being that most studies found that the highest prevalence of a single form of CAM was generally in the use of vitamins and minerals. For example, the current study found 45.8% participants used multivitamins, and the majority (86.4%) of these users had not visited a CAM practitioner. If self-administered multivitamins are excluded, a lower overall CAM prevalence will certainly be estimated.

This situation becomes more complicated when considering the modality of multivitamins included in the definitions of CAM in previous studies. For example, people may purchase supplements over the counter on the recommendation of friends or relatives who had originally received prescriptions from their medical doctors or CAM practitioners. The indirect "prescription" is contradictive with the definition being used in the 1996 US CAM study –CAM was defined with the exclusion of vitamins prescribed by a medical doctor and daily vitamins.³ Another US study employed an alternative definition of self-measurement of vitamins in terms of "overdose in excess of the Recommended Daily Allowance".¹ The issues about vitamins and supplements also presented dilemma for the current survey in regard to participants making a decision about what constitutes CAM and what doesn't. This implies that arriving at a precise and acceptable definition of each CAM modality requires further research.

There are substantial differences between countries and regions in the relative popularity of many therapies usually considered as CAM, presumably reflecting different social and health care contexts. A pilot study of the current study to determine the types of CAM used most commonly in Australia revealed the relatively high popularity of aromatherapy (subsequently confirmed by the main survey), which concurred with previous Australian regional surveys⁹⁻¹¹. Surprisingly, aromatherapy was not included in the total CAM prevalence of relevant US national health surveys.^{1,4}

Through this study, other issues that need to be addressed in future studies are identified. These problems and limitations are consistent with existing literature on population surveys²⁸³ (e.g. the difficulty of accessing mobile phones). This may be justified, based on the fact that fixed telephones are used by over 97.5% of households in Australia.²⁸⁴ In this respect, gaining access to mobile phone has only been incorporated into limited numbers of Australian population surveys. Another complexity is that the use of multi-phone lines within households. This is relevant, because more than one line in a household, will increase the chances of that household being selected.

It was determined that English would be used as the language for the interviews, therefore people from ethnic groups that do not speak English have not been included in this study. Fortunately, throughout the interviews very few people were excluded due to language barriers. To date, no study has been conducted in Australia to examine CAM use in the indigenous population. Future follow up studies should address the above matters, and possibly include participants from diverse social culture backgrounds.

## 8.3 General Conclusions

Prior to the conduct of the survey, three systematic reviews were completed to summarise the current overall utilisation of CAM, and the specific use of Chinese medicine in general populations around the world. CAM use in different regional and clinical populations in Australia has also been systematically evaluated. As noted above, the uncertainties of CAM definitions and classifications may affect estimates of the true prevalence of CAM. Nevertheless, there is evidence that the prevalence of most forms of CAM (e.g. herbal medicine and acupuncture) in many countries has increased steadily.

This study took an important step in initiating a national population-based survey in Australia to provide critical data on the current use of CAM in Australia. The findings indicate that the use of CAM in Australia is considerably higher than that extrapolated from previous regional studies. National expenditure on CAM appears to account for almost half the total expenditure on non-subsidised health-care products. Moreover, it was estimated that Australians made almost the same number of visits over a 12-month period to CAM practitioners as they did to conventional medical practitioners. This study again raises serious concerns about the concurrent use of Western and CAM therapies, and the lack of communication between patients and medical practitioners with regard to CAM use.

In addition to estimations of the utilisation of CAM from a national perspective, this study also explores possible regional differences, which shows that there were marked differences between states, for example, South Australia has lower overall prevalence of CAM use when compared with other states. In addition, the present survey also suggested that regional differences in the use of several specific forms of CAM were apparent. Thus, extrapolation of national CAM use based on regional studies should be undertaken with caution.

289

As Straus (2005) stated, understanding who uses CAM and why they do so informs the research goals, initiatives and collaborations of the NCCAM.⁸ Thus, the current study provides a useful national resource base for further research, including intervention studies. The outcomes of this investigation will be relevant to a broad range of health-care professionals and researchers, therapeutic regulators, and educators in the areas of public health, health policy, general medicine, and complementary medicine. In all, in Australia, what was known previously in the field of this study and what this study adds can be summarised in the table below (Table 8.1).

#### Table 8.1 What was known previously and what this study adds

#### What was known previously on this topic

- There has been no previous national Australian study on the use of CAM. National estimates of CAM use have been extrapolated from a regional study in 2004.
- There is evidence from studies in the US and the UK of increasing CAM use. Previous studies have highlighted major concerns about the concurrent use of CAM and conventional medical treatments.

#### What this study adds

- A reliable estimate of CAM use by the Australian population.
- Evidence of marked regional differences in CAM use in Australia.
- Evidence of significant regional differences in the use of specific forms of CAM
- Expenditure on CAM products by Australians is estimated to account for almost half of their "out-of-pocket" expenditure on non-subsidised health-care products.
- Comparison of national expenditure on CAM and conventional medical services.
- Specific benefits and risks perceived by the users of common forms of CAM
- Specific benefits and adverse events that users of common forms of CAM considered they have experienced.
- The discussion of CAM use between CAM users and their medical doctors.
- The extent, the demand for and consequences of health insurance cover for CAM.
- Public attitudes towards statutory CAM product and practitioners regulation.

# 8.4 Policy Implications

The findings of the current study have a number of implications for ongoing health policy development. Firstly, given ongoing shortages in the medical workforce, particularly general medical practitioners, and the increasing public and private costs of medical services, the increasing popularity of certain CAM therapies may offer opportunities to reduce the burden of heath-care services. This could be done by identifying CAM modalities which, based on evidence of their safety and effectiveness, could effectively contribute to health service delivery, in concert with conventional health-care services. To achieve such outcomes would require understanding by practitioners of the relevant CAM and conventional therapies of how the two systems could be effectively harmonised to achieve agreed health-economic outcomes. Such "harmonisation" could probably best be achieved through the curricula of CAM and conventional health practitioner training programs, and would need to be updated and strengthened by ongoing professional educational programs. In some cases, specific forms of CAM might prove be more cost-effective, either alone or in combination with conventional therapies, than conventional treatments alone. However, little is known about the economic implications of CAM from a macro and/or a micro level. Cost-effectiveness studies on CAM should be part of a future research program.

Secondly, the increasing popularity of CAM therapies offers alternative avenues to patients who have not been effectively managed by conventional medicine. However, there is inadequate research that produces critical data concerning the quality, efficacy and safety of these CAM therapies. And critically, the current funding model of National Health and Medical Research Council does not provide a mechanism to support such research. In addition, there is no national policy that recognises the role of CAM in the health care system and as a result, these therapies were not managed in the same manner as western medicines with respect to quality, safety and efficacy. Consequently, patients who use these therapies may be exposed to additional risks due to the inadequate information available. The contribution by consumers to these therapies is substantial, thus, it should be the government's role to ensure research in CAM therapies are supported to ensure public safety.

Thirdly, comprehensive knowledge of the forms of CAM that are most commonly used within the community, the purposes for which they are being used, and the demographics of the users are essential for effective health service planning. It is also important to understand how much consumers know about the efficacy and safety of the CAM products and therapies they choose to use, particularly in regard to the concurrent use of multiple therapies, irrespective of whether these are CAM or conventional. Based on this information, appropriate strategies can be developed to educate CAM users about the appropriate use of the therapies, their benefits and risks, and the importance of discussing the use of concurrent therapies with qualified practitioners.

Fourthly, quality of CAM practitioner training needs to incorporate sufficient Western medical knowledge and clinical assessment skills to ensure timely identification of risks and proper referral is made. In addition, practitioner training should also incorporate recent research findings. Furthermore, compulsory continuing professional education is needed to ensure that practitioners are providing the best possible care to their patients by applying new research results into practice and to ensure public safety.

Finally, the issue of health insurance coverage of CAM products and services requires further consideration. It is clear from the current study and a number of previous studies in Australia and overseas, that there is a high demand for insurance cover of CAM from users, both through the national health insurance system and private health insurance. Such demand is

likely to increase with increasing evidence of efficacy of certain forms of CAM and the introduction of regulation of both CAM products and services by governments. However, increasing access to insurance cover for CAM will certainly have major economic implications which will need to be fully considered in the context of overall public and private health expenditure.

# 8.5 Recommendations and Future Studies

In relation to further regulatory control of CAM therapies, products and practitioners, a large proportion of the participants in the current survey considered that regulation of CAM should be as rigorously as that for Western medicine and other conventional health services. Whilst several CAM therapies, are already subject to statutory regulation, in the interest of public safety and confidence in CAM, it is desirable that regulation be extended to other CAM modalities, particularly substance-based therapies such as naturopathy and types of herbal medicine in addition to Chinese herbal medicine.

Considerable efforts were to make the current study on CAM use by the Australian population as comprehensive as possible and to ensure that the validity of the data obtained. However, some issues related to CAM use could not be investigated in the current survey, such as the specific medical conditions for which each individual form of CAM was being used, and the outcomes of the treatments. To address the safety and efficacy concerns, these are perhaps the next aspects of CAM use that need to be investigated. In extending the study in this way it would be appropriate to give particular consideration to the use of specific CAM therapies for the treatment of chronic diseases, particularly in the elderly. It is often proposed that conventional medical treatments are not coping well with chronic illnesses and that forms of CAM for which treatment is based "holistic" principles may achieve better outcomes. At this time, rigorous evidence for this contention is lacking.

Prevalence of use data for most forms of CAM is patchy, even within specific countries and comparisons between surveys and between countries are difficult and often not meaningful. Future studies might focus on a group of common forms of CAM (e.g. the current 17 forms of

CAM investigated in the current study) in a particular region or country. Rather than exploring general demographic characteristics related to CAM use, such a study could focus on the factors that influence people's choice and use of CAM, such as, access to health insurance cover, regulatory controls, referrals from health professionals, and user's knowledge of evidence of efficacy, and safety.

Future Australian studies should also address the current lack of information about the use of CAM by population sub-groups, such as older people, children, adolescents, and the indigenous population. It would be important to establish an Australian national resource database to better comprehend matters surrounding CAM use, such as perceived and actual benefits and risks, as experienced by consumers. Ideally, these should be on an individual CAM modality basis, rather than overall CAM. From this aspect, consumer knowledge and beliefs of CAM is another area for further investigation.

## REFERENCES

- 1. Barnes PM, Powell-Griner E, McFann K, Nahin RL. Complementary and alternative medicine use among adults: United States, 2002. *Advance Data: From Vital and Health Statistics*. 2004;**No. 343.**
- Eisenberg DM, Kessler RC, Foster C, Norlock FE, Calkins DR, Delbanco TL. Unconventional medicine in the United States. Prevalence, costs, and patterns of use. *N Engl J Med* 1993;**328**(4):246 - 52.
- Eisenberg DM, Davis RB, Ettner SL, et al. Trends in alternative medicine use in the United States, 1990-1997: results of a follow-up national survey. JAMA 1998;280(18):1569 -75.
- 4. Ni H, Simile C, Hardy AM. Utilization of complementary and alternative medicine by United States adults: Results from the 1999 National Health Interview Survey. *Med Care* 2002;**40**(4):353-358.
- 5. Thomas KJ, Nicholl JP, Coleman P. Use and expenditure on complementary medicine in England: a population based survey. *Complement Ther Med* 2001;9(1):2-11.
- 6. Ramsay C, Walker M, Alexander J. Alternative medicine in Canada: Use and public attitudes. *Public Policy Sources* 1999;**21:**1-31.
- 7. Yamashita H, Tsukayama H, Sugishita C. Popularity of complementary and alternative medicine in Japan: a telephone survey. *Complement Ther Med* 2002;**10**(2):84-93.
- Straus SE. National Center for Complementary and Alternative Medicine: Fiscal Year 2006 Budget. <u>http://nccam.nih.gov/about/offices/od/directortestimony/0305.htm</u> accessed December 10, 2005.
- 9. MacLennan AH, Wilson DH, Taylor AW. Prevalence and cost of alternative medicine in Australia. *Lancet* 1996;**347**(9001):569-573.
- 10. MacLennan AH, Wilson DH, Taylor AW. The escalating cost and prevalence of alternative medicine. *Prev Med* 2002;**35**(2):166-173.
- 11. MacLennan AH, Myers SP, Taylor AW. The continuing use of complementary and alternative medicine in South Australia: costs and beliefs in 2004. *Med J Aust* 2006;**184**(1):27-31.
- 12. Bensoussan A, Myers SP, Wu SM, O'Connor K. Naturopathic and Western herbal medicine practice in Australia-a workforce survey. *Complement Ther Med* 2004;**12**(1):17-27.
- 13. Chinese Medicine Registration Board of Victoria. The Victorian Chinese Medicine Registration Act 2000. Act No.18/2000 Victoria, Australia., 2000.
- 14. Therapeutic Goods Administration (TGA). Expert Committee Report: Complementary Medicines in the Australian Health System. Canberra, 2003.
- 15. Paramore LC. Use of alternative therapies: Estimates from The 1994 Robert Wood Johnson Foundation National Access to Care Survey. *J Pain Symptom Manage* 1997;**13**(2):83-89.
- 16. Thomas K, Coleman P. Use of complementary or alternative medicine in a general population in Great Britain. Results from the National Omnibus Survey. *J Public Health.* 2004;**26**(2):152-7.
- 17. Ernst E, White A. The BBC survey of complementary medicine use in the UK. *Complement Ther Med* 2000;**8**(1):32-36.
- 18. Ernst E. Prevalence of use of complementary/alternative medicine: a systematic review. *Bull World Health Organ* 2000;**78**(2):252-7.
- 19. Harris P, Rees R. The prevalence of complementary and alternative medicine use among the general population: a systematic review of the literature. *Complement Ther Med* 2000;**8**(2):88-96.

- 20. Little CV, Parsons T. Herbal therapy for treating rheumatoid arthritis. *The Cochrane Database of Systematic Reviews* 2000;**4**:CD002948.
- 21. Garrow D, Egede LE. Association between complementary and alternative medicine use, preventive care practices, and use of conventional medical services among adults with diabetes. *Diabetes Care* 2006;**29**(1):15-9.
- Davis PA, Gold EB, Hackman RM, Stern JS, Gershwin ME. The use of complementary/alternative medicine for the treatment of asthma in the United States. J Investig Allergol Clin Immunol 1998;8(2):73-7.
- 23. Meydani SN, Leka LS, Fine BC, et al. Vitamin E and respiratory tract infections in elderly nursing home residents: a randomized controlled trial. *JAMA* 2004;**292**(7):828-36.
- 24. Furlan AD, Brosseau L, Imamura M, Irvin E. Massage for low-back pain. *The Cochrane Database of Systematic Reviews* 2002;**2:**CD001929.
- 25. Berman BM, Lao L, Langenberg P, Lee WL, Gilpin AM, Hochberg MC. Effectiveness of acupuncture as adjunctive therapy in osteoarthritis of the knee: a randomized, controlled trial. *Ann Intern Med* 2004;**141**(12):901-10.
- 26. Therapeutic Goods Administration (TGA). Government Response to the Recommendations of the Expert Committee on Complementary Medicines in the Health System. Canberra, 2005.
- 27. Ernst E. Complementary medicine-A definition. Br J of Gen Prac 1995(5):506.
- Hui KK, Yu JL, Zylowska L. The progress of Chinese Medicine in the USA. In: Chan K, Lee H, eds. The Way Forward for Chinese Medicine: Taylor and Francis, New York, 2002.
- 29. Wootton JC. Classifying and defining complementary and alternative medicine. *J Altern Complement Med* 2005;**11**(5):777-8.
- British Medical Association. Complementary medicine: New approaches to good practice. Oxford: Oxford University Press, 1993.
- 31. WHO. General guidelines for methodologies on research and evaluation of traditional medicine. WHO/EDM/TRM/2000.1. Geneva, 2000.
- 32. NCCAM. What is Complementary and Alternative Medicine? National Center for Complementary and Alternative Medicine, USA, 2004.
- 33. Panel on Definition and Description: CAM Research Methodology Conference. Defining and describing complementary and alternative medicine. *Altern Ther Health Med* 1997;**3**(2):49-57.
- 34. Rosner AL. Unconventional medicine: Correspondence. N Engl J Med 1993;**329**(16):1201.
- 35. Friedman R, Zuttermeister P, Benson H. Unconventional medicine: Correspondence. *N Engl J Med* 1993;**329**(16):1201.
- 36. NCCAM. What are the major types of complementary and alternative medicine?: National Center for Complementary and Alternative Medicine, USA, 2004.
- 37. Wilkinson JM, Simpson MD. High use of complementary therapies in a New South Wales rural community. *Aust J Rural Health* 2001;9(4):166-71.
- 38. Kermode S, Myers S, Ramsay L. Natural and complementary therapy utilisation on the north coast of NSW. *Aust J Holist Nurs* 1998;**5**(2):7-13.
- 39. Wootton JC, Sparber A. Surveys of complementary and alternative medicine: part I. General trends and demographic groups. *J Altern Complement Med* 2001;7(2):195-208.
- 40. Murphy LS, Reinsch S, Najm WI, et al. Searching biomedical databases on complementary medicine: the use of controlled vocabulary among authors, indexers and investigators. *BMC Complement Altern Med* 2003;**3**:3.
- 41. Landmark Healthcare. The Landmark Report on Public Perceptions of Alternative Care. Sacramento, USA. Landmark Healthcare, 1998.

- 42. Nielsen MW, Hansen EH, Rasmussen NK. Use of natural medicines in the Danish population: a national cross-sectional survey. *Ann Pharmacother* 2005;**39**(9):1534-8.
- 43. Menniti-Ippolito F, Gargiulo L, Bologna E, Forcella E, Raschetti R. Use of unconventional medicine in Italy: a nation-wide survey. *Eur J Clin Pharmacol* 2002;**58**(1):61-4.
- 44. Thomas KJ, Fall M, Nicholl J, Williams B. Methodological study to investigate the feasibility of conducting a population-based survey of the use of complementary health care. Final reprot to the council for complementary medicine.: University of Sheffield. UK, 1993.
- 45. Millar WJ. Use of alternative health care practitioners by Canadians. *Can J Public Health* 1997;**88**(3):154-8.
- 46. Astin JA. Why patients use alternative medicine: results of a national study. *JAMA* 1998;**279**(19):1548-53.
- 47. Druss BG, Rosenheck RA. Association between use of unconventional therapies and conventional medical services. *JAMA* 1999;**282**(7):651-6.
- 48. Millar WJ. Patterns of use--alternative health care practitioners. *Health Rep* 2001;**13**(1):9-21.
- 49. Bernstein JH, Shuval JT. Nonconventional medicine in Israel: Consultation patterns of the Israeli population and attitudes of primary care physicians. *Soc Sci Med* 1997;**44**(9):1341-1348.
- 50. Burg MA, Hatch RL, Neims AH. Lifetime use of alternative therapy: a study of Frorida residents. *South Med J* 1998;**91:**1126-1131.
- 51. Chi C, Lee JL, Lai JS, Chen SC, Chen CY, Chang SK. Utilization of Chinese medicine in Taiwan. Altern Ther Health Med 1997;3(4):40-53.
- 52. Emslie M, Campbell M, Walker K. Complementary therapies in a local healthcare setting. Part I: Is there real public demand? *Complement Ther Med* 1996;**4**(1):39-42.
- 53. Haidinger G, Gredler B. Degree of knowledge, frequency of application, and success of alternative medical treatment among the general population. *Off Gesundheitswes* 1988;**50**(1):9-12.
- 54. Haussermann D. Increased confidence in natural therapies [in German]. *Deutsches* Arzteblatt 1997;**94:**1857-1858.
- 55. Vaskilampi T, Merilainen P, Sinkkonen S. The use of alternative treatment in the Finnish adult population. In: Lewith GT, Aldridge D, eds. Clinical research methodology for complementary therapies. London: Hodder & Stoughton, 1993: 204-229.
- 56. Verhoef MJ, Russell ML, Love EJ. Alternative medicine use in rural alberta. *Can J Public Health* 1994;**85**(5):308-9.
- 57. Vickers A. Use of complementary therapies. *BMJ* 1994;**309**(6962):1161a.
- 58. Yung B, Lewis P, Charny M, Farrow S. Complementary medicine: some population-based data. *Compl med Res* 1988;**3**:23-28.
- 59. Hanssen B, Grimsgaard S, Launso L, Fonnebo V, Falkenberg T, Rasmussen NK. Use of complementary and alternative medicine in the Scandinavian countries. *Scand J Prim Health Care* 2005;**23**(1):57-62.
- 60. Fisher P, Ward A. Medicine in Europe: Complementary medicine in Europe. *BMJ* 1994;**309**(6947):107-111.
- 61. Sherwood P. Patterns of use of complementary health services in the south-west of Western Australia. *Aust J Rural Health* 2000;**8**(4):194-200.
- 62. Xue CCL, Lee C, Karagiannis J, et al. Public attitudes towards Chinese medicine in Melbourne, Australia. *J Complement Integr Med* 2005;**2**(1):Article 8.
- 63. Begbie SD, Kerestes ZL, Bell DR. Patterns of alternative medicine use by cancer patients. *Med J Aust* 1996;**165**(10):545-8.

- 64. Correa-Velez I, Clavarino A, Barnett AG, Eastwood H. Use of complementary and alternative medicine and quality of life: changes at the end of life. *Palliat Med* 2003;**17**(8):695-703.
- 65. McGorm K, Rieger N. Complementary and alternative therapies used by patients who have colorectal cancer. *Au J Cancer Nurs* 2004;**5**(1):25-28.
- 66. Miller M, Boyer MJ, Butow PN, Gattellari M, Dunn SM, Childs A. The use of unproven methods of treatment by cancer patients. Frequency, expectations and cost. Support Care Cancer 1998;6(4):337-47.
- 67. Salminen E, Bishop M, Poussa T, Drummond R, Salminen S. Dietary attitudes and changes as well as use of supplements and complementary therapies by Australian and Finnish women following the diagnosis of breast cancer. *Eur J Clin Nutr* 2004;**58**(1):137-144.
- 68. Sawyer MG, Gannoni AF, Toogood IR, Antoniou G, Rice M. The use of alternative therapies by children with cancer. *Med J Aus* 1994;**160**:320-322.
- 69. Sibbritt D, Adams J, Easthope G, Young A. Complementary and alternative medicine (CAM) use among elderly Australian women who have cancer. *Support Care Cancer* 2003;**11**(8):548-50.
- Byrne MJ, Semple SJ, Coulthard KP. Complementary medicine use during pregnancy. Interviews with 48 women in a hospital antenatal ward. *Aust Pharm* 2002;**21**(12):954-959.
- 71. Day AS. Use of complementary and alternative therapies and probiotic agents by children attending gastroenterology outpatient clinics. *J Paediatr Child Health* 2002;**38**(4):343-6.
- 72. Fong DP, Fong LK. Usage of complementary medicine among children. *Aust Fam Physician* 2002;**31**(4):388-91.
- 73. Shenfield G, Lim E, Allen H. Survey of the use of complementary medicines and therapies in children with asthma. *J Paediatr Child Health* 2002;**38**(3):252-7.
- 74. Adams J, Sibbritt DW, Easthope G, Young AF. The profile of women who consult alternative health practitioners in Australia. *Med J Aust* 2003;**179**(6):297-300.
- 75. Biddle RA, Wilkinson JM, Simpson MD. Do women in Australia use complementary medicines for reproductive conditions? *Aust Pharm* 2003; **22**(1):52-56.
- 76. Gollschewski S, Anderson D, Skerman H, Lyons-Wall P. The use of complementary and alternative medications by menopausal women in South East Queensland. Women's Health Issues 2004;14(5):165-171.
- 77. Sibbritt D, Adams J, Young AF. A longitudinal analysis of mid-age women's use of complementary and alternative medicine (CAM) in Australia, 1996-1998. Women & Health 2004;40(4):41-56.
- 78. Kestin M, Miller L, Littlejohn G, Wahlqvist M. The use of unproven remedies for rheumatoid arthritis in Australia. *Med J Aust* 1985;**143**(11):516-8.
- 79. Kristoffersen SS, Atkin PA, Shenfield GM. Use of alternative medicine on Sydney's North Shore. *Aust J Hosp Pharm* 1997;**27:**367-372.
- 80. Visser R, Ezzy D, Bartos M. Alternative or complementary? Nonallopathic therapies for HIV/AIDS. *Altern Ther Health Med* 2000;**6**(5):44-52.
- 81. Welch SA. The use of complementary medicines by inpatients at St Vincent's Hospital Sydney. *Aust J Hosp Pharm.* 2001;**31:**111-3.
- Buchbinder R, Gingold M, Hall S, Cohen M. Non-prescription complementary treatments used by rheumatoid arthritis patients attending a community-based rheumatology practice. *Intern Med J* 2002;**32**(5-6):208-14.
- 83. Burrows JB, J. Bell, SC. Prevalence of complementary and alternative medicine use by adults with cystic fibrosis. *J Pharm Pract Res* 2002;**32**(4):320-323.

- 84. Visser R, Grierson J. Use of alternative therapies by people living with HIV/AIDS in Australia. *AIDS Care* 2002;**14**(5):599-606.
- 85. Alderman CP, Kiepfer B. Complementary medicine use by psychiatry patients of an Australian hospital. *Ann Pharmacother* 2003;**37**(12):1779-84.
- Clifford R, Batty K, Davis W, Davis TME. Prevalence and predictors of complementary medicine usage in diabetes: Fremantle Diabetes Study. J Pharm Prac Res 2003;33:260-264.
- 87. George J, Ioannides-Demos LL, Santamaria NM, Kong DC, Stewart K. Use of complementary and alternative medicines by patients with chronic obstructive pulmonary disease. *Med J Aust* 2004;**181**(5):248-51.
- 88. Parslow RA, Jorm AF. Use of prescription medications and complementary and alternative medicines to treat depressive and anxiety symptoms: results from a community sample. *J Affect Disord* 2004;**82**(1):77-84.
- 89. Zochling J, March L, Lapsley H, Cross M, Tribe K, Brooks P. Use of complementary medicines for osteoarthritis--a prospective study. *Ann Rheum Dis* 2004;**63**(5):549-54.
- 90. Hall K, Giles-Corti B. Complementary therapies and the general practitioner. A survey of Perth GPs. *Aust Fam Physician* 2000;**29**(6):602-6.
- 91. Pirotta MV, Cohen MM, Kotsirilos V, Farish SJ. Complementary therapies: have they become accepted in general practice? *Med J Aust* 2000;**172**(3):105-9.
- 92. Wilkinson JM, Simpson MD. Complementary therapy use by nursing, pharmacy and biomedical science students. *Nurs Health Sci* 2001;**3**(1):19-27.
- 93. Wilkinson JM, Simpson MD. Personal and professional use of complementary therapies by nurses in NSW, Australia. *Complement Ther Nurs Midwifery* 2002;**8**(3):142-7.
- 94. Gaffney L, Smith C. Use of complementary therapies in pregnancy: The perceptions of obstetricians and midwives in South Australia. Aust N Z J Obstet Gynaecol. . 2004;44(1):24-9.
- 95. Cohen MM, Penman S, Pirotta M, Da Costa C. The integration of complementary therapies in Australian general practice: results of a national survey. *J Altern Complement Med* 2005;**11**(6):995-1004.
- 96. Ernst E. Prevalence of complementary/alternative medicine for children: a systematic review. *Eur J Pediatr* 1999;**158**(1):7-11.
- 97. Ernst E, Cassileth BR. The prevalence of complementary/alternative medicine in cancer: a systematic review. *Cancer* 1998;**83**(4):777-782.
- 98. Robinson A, McGrail MR. Disclosure of CAM use to medical practitioners: a review of qualitative and quantitative studies. *Complement Ther Med* 2004;**12**(2-3):90-98.
- 99. VIP Information. VIP Scientific Literature System [In Chinese]: http://newweb.cqvip.com/index.asp?Cookies=True, 2005.
- 100. Chan CW, Chang JK. The role of Chinese medicine in New York City's Chinatown. Part 1. *Am J Chin Med* 1976;**4**(1):31-45.
- 101. Wong TW, Wong SL, Donnan SP. Prevalence and determinants of the use of traditional Chinese medicine in Hong Kong. *Asia Pac J Public Health* 1995;**8**(3):167-70.
- 102. Chou P, Lai MY, Chung C, Chen JM, Chen CF. Acupuncture utilization in Taiwan. *Zhonghua Yi Xue Za Zhi (Taipei)* 1998;**61**(3):151-8.
- 103. Lau JT, Leung EM, Tsui HY. Predicting traditional Chinese medicine's use and the marginalization of medical care in Hong Kong. *Am J Chin Med* 2001;**29**(3-4):547-58.
- 104. Wong TW, Yu TS, Liu JLY, Lee NL, Lloyd OL. Factors associated with the utilisation of traditional Chinese medicine in a small town in Hong Kong. *Am J Chin Med* 1997;**25**(3-4):367-373.
- 105. Cassidy CM. Chinese medicine users in the United States. Part I: Utilization, satisfaction, medical plurality. *J Altern Complement Med* 1998;4(1):17-27.

- 106. Wong LK, Jue P, Lam A, Yeung W, Cham-Wah Y, Birtwhistle R. Chinese herbal medicine and acupuncture. How do patients who consult family physicians use these therapies? *Can Fam Physician* 1998;**44**:1009-15.
- 107. Lau JTF, Yu A. The choice between Chinese medicine and Western medicine practitioners by Hong Kong adolescents. *Am J Chin Med.* 2000;**28**(1):131-139.
- 108. Hon EK, Lee K, Tse HM, et al. A survey of attitudes to traditional Chinese medicine in Hong Kong pharmacy students. *Complement Ther Med* 2004;**12**(1):51-6.
- 109. Ng TP, Tan CH, Kua EH. The use of Chinese herbal medicines and their correlates in Chinese older adults: the Singapore Chinese Longitudinal Aging Study. Age Ageing. 2004; 33(2):135-42.
- 110. Wong VW-S, Law M-Y, Hui AY, et al. A hospital clinic-based survey on traditional Chinese medicine usage among chronic hepatitis B patients. *Complement Ther Med* 2005;**13**(3):175-182.
- 111. Wong TW, Wong SL, Donnan SPB. Prevalence and determinants of the use of traditional Chinese medicine in Hong Kong. Asia Pac J Public Health 1995;8(3):167-70.
- 112. MacPherson H, Sinclair-Lian N, Thomas K. Patients seeking care from acupuncture practitioners in the UK: A national survey. *Complement Ther Med* 2006;**14**(1):20-30.
- 113. State Administration of Traditional Chinese Medicine P. R. China. National Statistics Summary of Chinese Medicine Resources. Available at <u>http://www.satcm.gov.cn/lanmu/zonghe_xinxi/2004html.htm:</u> Accessed 10th April 2006, 2004.
- 114. Bodeker G, Ong CK, Grundy C, Burford G, Shein K. WHO Global Atlas of Traditional, Complementary and Alternative Medicine. Kope: World Health Oranization, WHO Centre for Health Development, 2005.
- 115. Kasilo OM, Soumbey-Alley E, Wambebe C, Chatora R. Regional Overview: African Region. In: Bodeker G, Ong CK, Grundy C, Burford G, Shein K, eds. WHO Global Atlas of Traditional, Complementary and Alternative Medicine. Kope: World Health Oranization, WHO Centre for Health Development, 2005.
- 116. WHO. Legal Status of Traditional Medicine and Complementary/ Alternative Medicine: A Worldwide Review. WHO/EDM/TRM/2001.2. Geneva, 2001.
- 117. Gupta MP. Chapter 6: Regional Overview: Region of the Americas. In: Bodeker G, Ong CK, Grundy C, Burford G, Shein K, eds. WHO Global Atlas of Traditional, Complementary and Alternative Medicine. Kobe: World Health Oranization, WHO Centre for Health Development, 2005.
- 118. Gaitonde BB, Kurup PN. Chapter 10: Regional Overview: South-East Asia Region. In: Bodeker G, Ong CK, Grundy C, Burford G, Shein K, eds. WHO Global Atlas of Traditional, Complementary and Alternative Medicine. Kobe: World Health Oranization, WHO Centre for Health Development, 2005.
- 119. El-Gendy AR. Chapter 20: Regional Overview: Eastern mediterranean Region. In: Bodeker G, Ong CK, Grundy C, Burford G, Shein K, eds. WHO Global Atlas of Traditional, Complementary and Alternative Medicine. Kobe: World Health Oranization, WHO Centre for Health Development, 2005.
- 120. Ong C-K, Hog E, Bodeker G, Burford G. Chapter 15: Regional Overview: European Region. In: Bodeker G, Ong CK, Grundy C, Burford G, Shein K, eds. WHO Global Atlas of Traditional, Complementary and Alternative Medicine. Kobe: World Health Oranization, WHO Centre for Health Development, 2005.
- 121. Roh P-U. Chapter 25: Regional Overview: Western Pacific Region. In: Bodeker G, Ong CK, Grundy C, Burford G, Shein K, eds. WHO Global Atlas of Traditional, Complementary and Alternative Medicine. Kobe: World Health Oranization, WHO Centre for Health Development, 2005.

- 122. Australian Bureau of Statistics. Population by Age and Sex, Australian States and Territories: June 2005: ABS Cat No. 3201.0, 2005.
- 123. Centers for Disease Control and Prevention. Epi Info[™] 6.0 User's Manual. Atlanta: Available: <u>http://www.cdc.gov/epiinfo/</u> Accessed 20th January 2005, 2005.
- 124. Australian Bureau of Statistics. Population by Age and Sex, Australian States and Territories: December 2004: ABS Cat No. 3201.0, 2004.
- 125. Couper MP, Barker RP, Bethlehem J, Clark CZF. Computer Assisted Survey Information Collection. NY: John Wiley & Sons, 1998.
- 126. Australian Bureau of Statistics. Australian Housing Survey, 1999: ABS Cat. No. 4182.0, 2001.
- 127. Sahai H, Khurshid A. A bibliography on telephone survey methodology. *J Off Stat* 1995;**11**:325-367.
- 128. Kinsey SH, Jewell DM. A systematic approached to instrument development in CATI. In: Couper M, Bethlehem R, Clark C, eds. Computer Assisted Survey Information Collection. NY: John Wiley & Sons., 1998.
- 129. Frankel MR, Frankel LR. Some recent developments in sample survey design. J Mark Res 1977;**3**:280-293.
- 130. SPSS. Statistical Package for Social Sciences 13.0 for Window. Chicago: SPSS Inc, 2005.
- 131. Korn EL, Garaubard BL. Analysis of health surveys. New York: Wiley, 1999.
- 132. SPSS. Statistical Package for Social Sciences 14.0 for Window: Data Validation Specifications. Chicago: SPSS Inc, 2006.
- 133. Mislevy R. Can there be reliability without reliability? *J Educ Behav Stat* 2004;**29:**241-244.
- 134. Moss PA. Can there be validity without reliability? Educ Res 1994;23:5-12.
- 135. Messick S. Test validity: A matter of consequence. Soc Indic Res 1998;45:35-44.
- 136. De Vaus D. Analyzing Social Science Data: 50 Key Problems in Data Analysis. London: SAGE Publications Ltd, 2002.
- 137. Briggs SR, Cheek JM. The role of factor analysis in the development and evaluation of personality scales. *J Pers* 1986;**54**:106 -148.
- 138. De Vaus D. Surveys in Social Research. 5th ed. New South Wales, Australia: Allen & Unwin, 2002.
- 139. Menard. Applied logistic regression analysis. 2nd ed: Thousand Oaks, Calif: Sage Publications, 2002.
- 140. Hosmer DW, Lemeshow S. Applied Logistic Regression. 2nd ed. New York: John Wiley & Sons, Inc, 2000.
- 141. Peduzzi P, Concato E, Kemper T, Holford R, Feinstein A. A stimulation of the number of events per variable in logistic regression analysis. *J Clin Epidemiol* 1996;**99:**1373-1379.
- 142. Pallant JF. SPSS survival manual : a step by step guide to data analysis using SPSS. 2nd ed. Crows Nest, N.S.W: Allen & Unwin, 2004.
- 143. Statistical Package for Social Sciences. SPSS Regression Models 12.0 Manual. Chicago: SPSS Inc, 2003.
- 144. Tabachnick BG, Fidell LS. Using multivariate statistics. 4th ed. Boston, Mass: Allyn and Bacon, 2001.
- 145. Winship C, Radbill L. Sampling weights and regression analysis. *Sociol Methods & Res* 1994;**23**(2):230-257.
- 146. The American Association for Public Opinion Research. Standard Definitions: Final Dispositions of Case Codes and Outcome Rates for Surveys 4th edition. Lenexa, Kansas: AAPOR, 2006.

- 147. Australian Institute of Health and Welfare. Australia's Health 2004. AIHW Cat No AUS 44, Canberra, 2004.
- 148. Australian Bureau of Statistics. Australia Year Book 2005: ABS Cat. 1301.0, 2005.
- 149. Australian Bureau of Statistics. Australian Social Trend 2005: ABS Cat. 4102.0, 2005.
- 150. Ulett GA, Han J, Han S. Traditional and evidence-based acupuncture: history, mechanisms, and present status. *South Med J* 1998;**91**(12):1115-20.
- 151. The Accreditation Commission for Acupuncture and Oriental Medicine (ACAOM). http://www.acaom.org/ Accessed 01 June 2006., 2006.
- 152. Tanzawa S. Acupuncture and moxibustion in Japan. *The Japan Society of Acupuncture and Moxibustion News* 2004(<u>http://www.jsam.jp/english/index.htm</u> Accessed 01 June 2006).
- 153. Grzywacz JG, Lang W, Suerken C, Quandt SA, Bell RA, Arcury TA. Age, race, and ethnicity in the use of complementary and alternative medicine for health self-management: evidence from the 2002 National Health Interview Survey. *J Aging Health* 2005;**17**(5):547-72.
- 154. Australian Department of Health and Ageing. Medicare Statistical data for the June Quarter 2005. Available from: <u>http://www.health.gov.au/</u>, 2005.
- 155. Australian Institute of Health and Welfare. Health Expenditure 2003-04. AIHW Cat No HWE 32. Canberra, 2005.
- 156. Hong FF. History of medicine in China: when medicine took an alternative path. *McGill Journal of Medicine* 2004;**8**(1):79-84.
- 157. Lucas R. Secrets of the Chinese Herbalists. Revised Edition ed. New York: Parker Publishing Company, Inc., 1987.
- 158. Scheid V. The globalization of Chinese medicine. Lancet 1999;354(Suppl):SIV10.
- 159. Lau JT, Leung EM, Tsui HY. Predicting traditional Chinese medicine's use and the marginalization of medical care in Hong Kong. *Am J Chin Med* 2001;**29**(3-4):547-58.
- 160. Chi C, Lee JL, Lai JS, Chen SC, Chen CY, Chang SK. Utilization of Chinese medicine in Taiwan. *Altern Ther Health Med* 1997;**3**(4):40-53.
- 161. Hui KK, Hui EK, Johnston MF. The potential of a person-centered approach in caring for patients with cancer: a perspective from the UCLA center for East-West medicine. *Integr Cancer Ther* 2006;5(1):56-62.
- 162. Ito H, Shimura K. Effects of a blended Chinese medicine, xiao-chai-hu-tang, on Lewis lung carcinoma growth and inhibition of lung metastasis, with special reference to macrophage activation. *Jpn J Pharmacol* 1986;41(3):307-14.
- 163. Lin C, Lin X, Yang J. An observation on combined use of chemotherapy and traditional Chinese medicine to relieve cancer pain. *J Tradit Chin Med* 1996;**16**(4):267-9.
- 164. Bensoussan A, Myers SP, Carlton AL. Risks associated with the practice of traditional Chinese medicine: an Australian study. *Arch Fam Med* 2000;**9**(10):1071-8.
- 165. Ko RJ. Causes, epidemiology, and clinical evaluation of suspected herbal poisoning. J Toxicol Clin Toxicol 1999;**37**(6):697-708.
- 166. Bensoussan A, Myers SP. Towards a safer choice: the practice of traditional Chinese medicine in Australia: Faculty of Health, University of Western Sydney, Macarthur, Australia, 1996.
- 167. Easthope G, Beilby JJ, Gill GF, Tranter BK. Acupuncture in Australian general practice: practitioner characteristics. *Med J Aust* 1998;**169**(4):197-200.
- 168. Easthope G, Gill GF, Beilby JJ, Tranter BK. Acupuncture in Australian general practice: patient characteristics. *Med J Aust* 1999;**170**(6):259-62.
- 169. Linde K, Vickers A, Hondras M, et al. Systematic reviews of complementary therapies an annotated bibliography. Part 1: acupuncture. *BMC Complement Altern Med* 2001;**1**:3.

- 170. Manheimer E, White A, Berman B, Forys K, Ernst E. Meta-analysis: acupuncture for low back pain. *Ann Intern Med* 2005;**142**(8):651-63.
- 171. Xue C, English R, Zhang J, Da Costa C, Li C. Effect of acupuncture in the treatment of seasonal allergic rhinitis: a randomized controlled clinical trial. *Am J Chin Med* 2002;**30**(1):1 11.
- 172. Xue C, Dong L, Polus B, et al. Electroacupuncture for tension-type headache on distal acupoints only: a randomized, controlled, crossover trial. *Headache* 2004;**44**(4):333 41.
- 173. Chinese Medicine Registration Board of Victoria. The Victorian Chinese Medicine Registration Board Website. Victoria, Australia, 2006.
- 174. Parliament of New South Wales. Joint Committee on the Health Care Complaints Commission Report: Inquiry into Traditional Chinese Medicine, Report No. 9/53. Sydney, 2005.
- 175. Western Australian Government Department of Health. Regulation of Practitioners of Chinese Medicine in Western Australia. Perth, 2005.
- 176. Vickers A, Zollman C. ABC of complementary medicine. The manipulative therapies: osteopathy and chiropractic. *BMJ* 1999;**319**(7218):1176-9.
- 177. Coulter ID, Hurwitz EL, Adams AH, Genovese BJ, Hays R, Shekelle PG. Patients using chiropractors in North America: who are they, and why are they in chiropractic care? *Spine* 2002;**27**(3):291-6; discussion 297-8.
- 178. Hurwitz EL, Chiang LM. A comparative analysis of chiropractic and general practitioner patients in North America: findings from the joint Canada/United States Survey of Health, 2002-03. *BMC Health Serv Res* 2006;**6**:49.
- 179. Hartvigsen J, Sorensen LP, Graesborg K, Grunnet-Nilsson N. Chiropractic patients in Denmark: a short description of basic characteristics. *J Manipulative Physiol Ther* 2002;**25**(3):162-7.
- 180. Hartvigsen J, Bolding-Jensen O, Hviid H, Grunnet-Nilsson N. Danish chiropractic patients then and now--a comparison between 1962 and 1999. *J Manipulative Physiol Ther* 2003;**26**(2):65-9.
- 181. Rubinstein S, Pfeifle CE, van Tulder MW, Assendelft WJ. Chiropractic patients in the Netherlands: a descriptive study. *J Manipulative Physiol Ther* 2000;**23**(8):557-63.
- 182. Leboeuf-Yde C, Hennius B, Rudberg E, Leufvenmark P, Thunman M. Chiropractic in Sweden: a short description of patients and treatment. J Manipulative Physiol Ther 1997;20(8):507-10.
- 183. Sibbritt D, Adams J, Young AF. A profile of middle-aged women who consult a chiropractor or osteopath: findings from a survey of 11,143 Australian women. J *Manipulative Physiol Ther* 2006;**29**(5):349-53.
- 184. Stano M, Haas M, Goldberg B, Traub PM, Nyiendo J. Chiropractic and medical care costs of low back care: results from a practice-based observational study. *Am J Manag Care* 2002;**8**(9):802-9.
- 185. Hawk C, Long CR, Boulanger KT. Prevalence of nonmusculoskeletal complaints in chiropractic practice: report from a practice-based research program. *J Manipulative Physiol Ther* 2001;**24**(3):157-69.
- 186. Walker BF, Muller R, Grant WD. Low back pain in Australian adults. health provider utilization and care seeking. *J Manipulative Physiol Ther* 2004;**27**(5):327-35.
- 187. Australian Department of Health and Ageing. Medicare items for allied health services for people with chronic conditions and complex care needs. Canberra. DoHA, 2006.
- 188. Leboeuf-Yde C, Pedersen EN, Bryner P, et al. Self-reported nonmusculoskeletal responses to chiropractic intervention: a multination survey. *J Manipulative Physiol Ther* 2005;**28**(5):294-302; discussion 365-6.

- 189. Zhang X. Integration of traditional and complementary medicine into national health care systems. *J Manipulative Physiol Ther* 2000;**23**(2):139-40.
- 190. Meeker WC. Public demand and the integration of complementary and alternative medicine in the US health care system. *J Manipulative Physiol Ther* 2000;**23**(2):123-6.
- 191. Stultz T. The Fourth Annual Expense and Salary Report. *Chiropractic Economics* 2001;**43**(5):34-48.
- 192. Jackson P. Summary of the 2000 ACA Professional Survey on Chiropractic Practice. J Am Chiropr Assoc 2001;38(2):27-30.
- 193. Hurwitz EL, Morgenstern H, Vassilaki M, Chiang LM. Adverse reactions to chiropractic treatment and their effects on satisfaction and clinical outcomes among patients enrolled in the UCLA Neck Pain Study. *J Manipulative Physiol Ther* 2004;**27**(1):16-25.
- 194. Leboeuf-Yde C, Hennius B, Rudberg E, Leufvenmark P, Thunman M. Side effects of chiropractic treatment: a prospective study. *J Manipulative Physiol Ther* 1997;**20**(8):511-5.
- 195. Ernst E. Manipulation of the cervical spine: a systematic review of case reports of serious adverse events, 1995-2001. *Med J Aust* 2002;**176**(8):376-80.
- 196. Emslie MJ, Campbell MK, Walker KA. Changes in public awareness of, attitudes to, and use of complementary therapy in North East Scotland: surveys in 1993 and 1999. *Complement Ther Med* 2002;**10**(3):148-153.
- 197. Licciardone JC. Awareness and use of osteopathic physicians in the United States: results of the Second Osteopathic Survey of Health Care in America (OSTEOSURV-II). *J Am Osteopath Assoc* 2003;**103**(6):281-9.
- 198. National Archives and Records Administration. The Federal Register (FR). http://www.gpoaccess.gov/fr/index.html: The United States, 2006.
- 199. Howell JD. The paradox of osteopathy. *N Engl J Med* 1999;**341**(19):1465-8.
- 200. American Academy of Osteopathy. What is osteopathy?: <u>http://www.academyofosteopathy.org/whatis.cfm</u>, 2006.
- 201. Johnson SM, Kurtz ME. Osteopathic manipulative treatment techniques preferred by contemporary osteopathic physicians. *J Am Osteopath Assoc* 2003;**103**(5):219-24.
- 202. Johnson SM, Kurtz ME. Conditions and diagnoses for which osteopathic primary care physicians and specialists use osteopathic manipulative treatment. *J Am Osteopath Assoc* 2002;**102**(10):527-32, 537-40.
- 203. Gamber R, Holland S, Russo DP, Cruser A, Hilsenrath PE. Cost-effective osteopathic manipulative medicine: a literature review of cost-effectiveness analyses for osteopathic manipulative treatment. *J Am Osteopath Assoc* 2005;**105**(8):357-67.
- 204. Licciardone JC, Herron KM. Characteristics, satisfaction, and perceptions of patients receiving ambulatory healthcare from osteopathic physicians: a comparative national survey. J Am Osteopath Assoc 2001;101(7):374-85.
- 205. Licciardone J, Gamber R, Cardarelli K. Patient satisfaction and clinical outcomes associated with osteopathic manipulative treatment. J Am Osteopath Assoc 2002;102(1):13-20.
- 206. Lengacher CA, Bennett MP, Kip KE, et al. Frequency of use of complementary and alternative medicine in women with breast cancer. *Oncol Nurs Forum* 2002;**29**(10):1445-52.
- 207. Ceylan S, Hamzaoglu O, Komurcu S, Beyan C, Yalcin A. Survey of the use of complementary and alternative medicine among Turkish cancer patients. *Complement Ther Med* 2002;**10**(2):94-9.
- 208. Verhoef MJ, Hagen N, Pelletier G, Forsyth P. Alternative therapy use in neurologic diseases: use in brain tumor patients. *Neurology* 1999;**52**(3):617-22.

- 209. Furnham A, Bhagrath R. A comparison of health beliefs and behaviours of clients of orthodox and complementary medicine. *Br J Clin Psychol* 1993;**32** ( **Pt 2**):237-46.
- 210. Furnham A, Forey J. The attitudes, behaviors and beliefs of patients of conventional vs. complementary (alternative) medicine. *J Clin Psychol* 1994;**50**(3):458-69.
- 211. Sutherland LR, Verhoef MJ. Why do patients seek a second opinion or alternative medicine? *J Clin Gastroenterol* 1994;**19**(3):194-7.
- 212. Dimmock S, Troughton PR, Bird HA. Factors predisposing to the resort of complementary therapies in patients with fibromyalgia. *Clin Rheumatol* 1996;**15**(5):478-82.
- 213. Lee GB, Charn TC, Chew ZH, Ng TP. Complementary and alternative medicine use in patients with chronic diseases in primary care is associated with perceived quality of care and cultural beliefs. *Fam Pract* 2004;**21**(6):654-60.
- 214. Verhoef MJ, Balneaves LG, Boon HS, Vroegindewey A. Reasons for and characteristics associated with complementary and alternative medicine use among adult cancer patients: a systematic review. *Integr Cancer Ther* 2005;4(4):274-86.
- 215. Shumay DM, Maskarinec G, Kakai H, Gotay CC. Why some cancer patients choose complementary and alternative medicine instead of conventional treatment. *J Fam Pract* 2001;**50**(12):1067.
- 216. Shen J, Andersen R, Albert PS, et al. Use of complementary/alternative therapies by women with advanced-stage breast cancer. *BMC Complement Altern Med* 2002;**2**:8.
- 217. Eng J, Ramsum D, Verhoef M, Guns E, Davison J, Gallagher R. A population-based survey of complementary and alternative medicine use in men recently diagnosed with prostate cancer. *Integr Cancer Ther* 2003;**2**(3):212-6.
- 218. Hall JD, Bissonette EA, Boyd JC, Theodorescu D. Motivations and influences on the use of complementary medicine in patients with localized prostate cancer treated with curative intent: results of a pilot study. *BJU Int* 2003;**91**(7):603-7.
- 219. Navo MA, Phan J, Vaughan C, et al. An assessment of the utilization of complementary and alternative medication in women with gynecologic or breast malignancies. *J Clin Oncol* 2004;**22**(4):671-7.
- 220. Downer SM, Cody MM, McCluskey P, et al. Pursuit and practice of complementary therapies by cancer patients receiving conventional treatment. *BMJ* 1994;**309**(6947):86-9.
- 221. Richardson MA, Sanders T, Palmer JL, Greisinger A, Singletary SE. Complementary/alternative medicine use in a comprehensive cancer center and the implications for oncology. *J Clin Oncol* 2000;**18**(13):2505-14.
- 222. Moschen R, Kemmler G, Schweigkofler H, et al. Use of alternative/complementary therapy in breast cancer patients--a psychological perspective. *Support Care Cancer* 2001;9(4):267-74.
- 223. Tough SC, Johnston DW, Verhoef MJ, Arthur K, Bryant H. Complementary and alternative medicine use among colorectal cancer patients in Alberta, Canada. *Altern Ther Health Med* 2002;**8**(2):54-6, 58-60, 62-4.
- 224. Chrystal K, Allan S, Forgeson G, Isaacs R. The use of complementary/alternative medicine by cancer patients in a New Zealand regional cancer treatment centre. *N Z Med J* 2003;**116**(1168):U296.
- 225. Andersen RM. Revisiting the behavioral model and access to medical care: does it matter? *J Health Soc Behav* 1995;**36**(1):1-10.
- 226. Sirois FM, Gick ML. An investigation of the health beliefs and motivations of complementary medicine clients. *Soc Sci Med* 2002;**55**(6):1025-37.
- 227. Wolsko P, Eisenberg D, Davis R, Ettner S, Phillips R. Insurance coverage, medical conditions, and visits to alternative medicine providers: results of a national survey. *Arch Intern Med* 2002;**162**(3):281 7.

- 228. House of Lords. Science and Technology Sixth Report. House of Lords Committee on Science and Technology: <u>http://www.parliament.the-stationery-office.co.uk/pa/ld199900/ldselect/ldsctech/123/12301.htm</u>, 2000.
- 229. Pelletier KR, Astin JA. Integration and reimbursement of complementary and alternative medicine by managed care and insurance providers: 2000 update and cohort analysis. *Altern Ther Health Med* 2002;**8**(1):38-9.
- 230. PHIAC. Percentage of Australians covered by Private Health Insurance. Private Health Insurance Administration Council of Australia: Available: <u>http://www.phiac.gov.au/</u>, 2006.
- 231. Wolsko P, Eisenberg D, Davis R, Kessler R, Phillips R. Patterns and perceptions of care for treatment of back and neck pain: results of a national survey. *Spine* 2003;28(3):292 7; discussion 298.
- 232. Peters RE, Chance MA. Chiropractic in Australia: the first forty years (1905-1945). *Chiropr Hist* 1996;**16**(1):29-38.
- 233. Cleary-Guida MB, Okvat HA, Oz MC, Ting W. A regional survey of health insurance coverage for complementary and alternative medicine: current status and future ramifications. *J Altern Complement Med* 2001;**7**(3):269-73.
- 234. Legorreta AP, Metz RD, Nelson CF, Ray S, Chernicoff HO, Dinubile NA. Comparative analysis of individuals with and without chiropractic coverage: patient characteristics, utilization, and costs. *Arch Intern Med* 2004;**164**(18):1985-92.
- 235. Lafferty WE, Bellas A, Corage Baden A, Tyree PT, Standish LJ, Patterson R. The use of complementary and alternative medical providers by insured cancer patients in Washington State. *Cancer* 2004;**100**(7):1522-30.
- 236. Therapeutic Goods Administration (TGA). TGA reminds Australians of the potential danger of Pan Pharmaceuticals. Media release 24 August 2003, Canberra, 2003.
- Therapeutic Goods Administration (TGA). Australian Regularoty Guidelines for Complementary Medicines (ARGCM) Part I: Registration of Complementary Medicines. Canberra, 2005.
- 238. Therapeutic Goods Administration (TGA). Australian Regularoty Guidelines for Complementary Medicines (ARGCM) Part II: Listed Complementary Medicines. Canberra, 2005.
- 239. Therapeutic Goods Administration (TGA). Australian Regularoty Guidelines for Complementary Medicines (ARGCM) Part III: Evaluation of Complementary Medicine Substances. Canberra, 2005.
- 240. Therapeutic Goods Administration (TGA). Australian Regularoty Guidelines for Complementary Medicines (ARGCM) Part IV: General guidance. Canberra, 2005.
- 241. Therapeutic Goods Administration (TGA). Australian Regularoty Guidelines for Complementary Medicines (ARGCM) Part V: Policy documents and guidelines. Canberra, 2005.
- 242. Weir M. Alternative Medicine: A New Regulatory Model. Melbourne: Australian Scholarly Publishing Pty Ltd., 2005.
- 243. Coulter ID, Willis EM. The rise and rise of complementary and alternative medicine: a sociological perspective. *Med J Aust* 2004;**180**(11):587-9.
- 244. Carlton AL, Bensoussan A. Regulation of complementary medicine practitioners in Australia: Chinese medicine as a case example. *Complement Ther Med* 2002;**10**(1):20-6.
- 245. D'Arcy PF. Adverse reactions and interactions with herbal medicines. Part 1. Adverse reactions. *Adverse Drug React Toxicol Rev* 1991;**10**(4):189-208.
- 246. D'Arcy PF. Adverse reactions and interactions with herbal medicines. Part 2. Drug interactions. *Adverse Drug React Toxicol Rev* 1993;12(3):147-62.

- 247. Coon JT, Ernst E. Panax ginseng: a systematic review of adverse effects and drug interactions. *Drug Saf* 2002;**25**(5):323-44.
- 248. Izzo AA, Ernst E. Interactions between herbal medicines and prescribed drugs: a systematic review. *Drugs* 2001;**61**(15):2163-75.
- 249. White A, Hayhoe S, Hart A, et al. Survey of adverse events following acupuncture (SAFA): a prospective study of 32,000 consultations. *Acupunct Med* 2001;**19**(2):84-92.
- 250. MacPherson H, Thomas K, Walters S, Fitter M. A prospective survey of adverse events and treatment reactions following 34,000 consultations with professional acupuncturists. *Acupunct Med* 2001;**19**(2):93-102.
- 251. MacPherson H, Thomas K, Walters S, Fitter M. The York acupuncture safety study: prospective survey of 34 000 treatments by traditional acupuncturists. *BMJ* 2001;**323**(7311):486-7.
- 252. White A, Hayhoe S, Hart A, Ernst E. Adverse events following acupuncture: prospective survey of 32 000 consultations with doctors and physiotherapists. *BMJ* 2001;**323**(7311):485-6.
- 253. Myers SP, Cheras PA. The other side of the coin: safety of complementary and alternative medicine. *Med J Aust* 2004;**181**(4):222-5.
- 254. Webb AN, Hardikar W, Cranswick NE, Somers GR. Probable herbal medication induced fulminant hepatic failure. *J Paediatr Child Health* 2005;**41**(9-10):530-1.
- 255. Cheung T, Xue CCL, Leung K, Chan K, Li CG. Aristolochic acids detected in some raw Chinese medicinal herbs and manufactured herbal products a consequence of inappropriate nomenclature and imprecise labelling? *Clin Toxicol* 2006;**44**:371-378.
- 256. Northcott HC, Bachynsky JA. Concurrent utilization of chiropractic, prescription medicines, nonprescription medicines and alternative health care. *Soc Sci Med* 1993;**37**(3):431-5.
- 257. Shenfield GM, Atkin PA, Kristoffersen SS. Alternative medicine: an expanding health industry. *Med J Aust* 1997;**166**(10):516-7.
- 258. Blais R, Maiga A, Aboubacar A. How different are users and non-users of alternative medicine? *Can J Public Health* 1997;**88**(3):159-62.
- 259. Eisenberg DM, Kessler RC, Van Rompay MI, et al. Perceptions about complementary therapies relative to conventional therapies among adults who use both: results from a national survey. *Ann Intern Med* 2001;**135**(5):344-51.
- 260. Muhajarine N, Neudorf C, Martin K. Concurrent consultations with physicians and providers of alternative care: results from a population-based study. *Can J Public Health* 2000;**91**(6):449-53.
- 261. Haller CA. We should always ask our patients about unconventional treatments. *West J Med* 2001;**175**(3):164.
- 262. Wynia MK, Eisenberg DM, Wilson IB. Physician-patient communication about complementary and alternative medical therapies: a survey of physicians caring for patients with human immunodeficiency virus infection. *J Altern Complement Med* 1999;**5**(5):447-56.
- 263. Crock RD, Jarjoura D, Polen A, Rutecki GW. Confronting the communication gap between conventional and alternative medicine: a survey of physicians' attitudes. *Altern Ther Health Med* 1999;**5**(2):61-6.
- 264. Tasaki K, Maskarinec G, Shumay DM, Tatsumura Y, Kakai H. Communication between physicians and cancer patients about complementary and alternative medicine: exploring patients' perspectives. *Psychooncology* 2002;**11**(3):212-20.
- 265. Flannery MA, Love MM, Pearce KA, Luan JY, Elder WG. Communication about complementary and alternative medicine: Perspectives of primary care clinicians. *Altern Ther Health Med* 2006;**12**(1):56-63.

- 266. Adler SR, Fosket JR. Disclosing complementary and alternative medicine use in the medical encounter: a qualitative study in women with breast cancer. *J Fam Pract* 1999;**48**(6):453-8.
- 267. Eisenberg DM. Advising patients who seek alternative medical therapies. Ann Intern Med 1997;127(1):61 9.
- 268. Johnson JE. Older rural women and the use of complementary therapies. *J Community Health Nurs* 1999;**16**(4):223-32.
- 269. Rao JK, Mihaliak K, Kroenke K, Bradley J, Tierney WM, Weinberger M. Use of complementary therapies for arthritis among patients of rheumatologists. *Ann Intern Med* 1999;131(6):409-16.
- 270. Hilsden RJ, Scott CM, Verhoef MJ. Complementary medicine use by patients with inflammatory bowel disease. *Am J Gastroenterol* 1998;**93**(5):697-701.
- 271. Lazar JS, O'Connor BB. Talking with patients about their use of alternative therapies. *Prim Care* 1997;**24**(4):699-714.
- 272. Smith M, Carber L. Chiropractic health care in health professional shortage areas in the United States. *Am J Public Health* 2002;**92**(12):2001-9.
- 273. Kerridge IH, McPhee JR. Ethical and legal issues at the interface of complementary and conventional medicine. *Med J Aust* 2004;**181**(3):164-6.
- 274. Cohen MH. Legal and ethical issues in complementary medicine: a United States perspective. *Med J Aust* 2004;**181**(3):168-9.
- 275. Cohen M, Eisenberg D. Potential physician malpractice liability associated with complementary and integrative medical therapies. *Ann Intern Med* 2002;**136**(8):596 603.
- 276. Weir M. Obligation to advise of options for treatment--medical doctors and complementary and alternative medicine practitioners. *J Law Med* 2003;**10**(3):296-307.
- 277. Studdert D, Eisenberg D, Miller F, Curto D, Kaptchuk T, Brennan T. Medical malpractice implications of alternative medicine. *JAMA* 1998;**280**(18):1610 5.
- 278. Brophy E. Does a doctor have a duty to provide information and advice about complementary and alternative medicine? *J Law Med* 2003;**10**(3):271-84.
- 279. National Health and Medical Research Council. General Guidelines for Medical Practitioners on providing information to patients. Canberra, Australia, 2004.
- 280. Australian Medical Council. AMA Position Statement: Complementary Medicine, 2002.
- 281. Kessler R, Davis R, Foster D, et al. Long-term trends in the use of complementary and alternative medical therapies in the United States. Ann Intern Med 2001;135(4):262 -8.
- 282. Spence M, Ribeaux P. Complementary and alternative medicine: consumers in search of wellness or an expression of need by the sick? *Psychol Mark* 2004;**21**(2):113-139.
- 283. Barnett V. Sample survey: principles & methods. 3rd ed. London: Oxford University Press, 2002.
- 284. ABS. Australian Bureau of Statistics. Australian Housing Survey, 1999 (ABS Cat. no. 4182.0), 2001.
# **APPENDICES**

Appendix A Brief explanations of the 17 forms of CAM that used in the survey	308
Appendix B Survey questionnaire (sample)	310
Appendix C Survey computer programming scripts (sample)	312
Appendix D Initial telephone contact (with plain language statement)	314
Appendix E Plain language statement (print version)	315
Appendix F Human ethics study approval form	317
Appendix G Additional statistical outputs	318
Appendix G1 Variables employed in the survey	318
Appendix G2 Frequency of visits to CAM practitioner (unadjusted data)	322
Appendix G3 Statistics of frequency of CAM practitioner visits (adjusted data)	323
Appendix G4 Expenditure on CAM (unadjusted data)	324
Appendix G5 Expenditure on CAM practitioner (unadjusted data)	325
Appendix G6 Expenditure on CAM-related items (unadjusted data)	326
Appendix G7 Statistics of expenditure on CAM (unadjusted data)	327
Appendix G8 Statistics of expenditure on CAM (adjusted data)	328
Appendix G9 Individual expenses on four forms of CAM (unadjusted data)	329
Appendix G10 Statistics of expenses on four forms of CAM (adjusted data)	330
Appendix G11 Inter-correlations of CAM use and potential predictor variables	331
Appendix G12 Statistics of the regression model on CAM use	333

# Appendix A Brief explanations of the 17 forms of CAM that used in the survey

[Explain to survey participants by the interviewers, if asked] Modified from the NCCAM and other sources.

### 1. Acupuncture

A Chinese medical system, involves penetrating the skin with needles in acupuncture points

### 2. Chinese herbal medicine

A Chinese medical system, using different types of herbs including plants, minerals and animal substances

### 3. Chinese therapeutic massage

A Chinese medical system, based on Chinese medicine theory, it manipulates muscle and connective tissue to promote relaxation and well-being.

## 4. Chinese medicine dietary therapy

A Chinese medical system, the idea is: "food is also medicine", practitioners will consider the food's colour, taste, texture and shape.

## 5. Qigong ("Chee-Gong"), martial art or Tai Chi ("Tai-Chee")

Qigong---Combines movement, meditation, and regulation of breathing to enhance the flow of qi-energy of a human body.

Martial art---In Chinese, it means "Wu Shu ("Wu-Su")" also known as "Flighting system". But it is different from body balance and body combat.

Tai Chi---A traditional Chinese exercise, a series of movements normally done very slowly to move the body's qi.

### 6. Western herbal medicine not including Chinese herbal medicine

Western herbalists use a mixture of herbs in their preparations eg. Garlic and St John's Wort, but generally do not use animal or mineral materials.

### 7. Massage therapy not including Chinese therapeutic massage

A manipulate therapy, for the purpose of this study, it distinguishes to Chinese therapeutic massage. The common ones include Thai massage, Swedish massage and Japanese massage.

### 8. Chiropractic ("kie-roh-PRAC-tic")

A manipulation therapy, it focuses on the relationship between body structure (primarily that of the spine) and function.

### 9. **Osteopathy** ("*ahs-tee-oh-PATH*")

A full-body manipulation therapy, it believes that diseases arising in the musculoskeletal system.

### 10. Homeopathy ("home-ee-oh-PATH")

A medical system bases on the belief that "*like cures like*," or "*The Law of Similar*". A medicine is homeopathic only if it is taken based upon the similar nature of the medicine to the illness.

### 11. Naturopathy ("nay-chur-o-PATH")

A therapy by natural approaches using medicinal plants, nutrition, dietary supplements, lifestyle and exercise etc.

### 12. Meditation

A variety of techniques to enhance people's mind, such as stress relief, pain management etc.

### 13. Aromatherapy

Aromatherapy involves the use of essential oils, such as rose oil, peppermint oil, often apply through massage.

### 14. Clinical nutrition eg. taking multivitamins or minerals

Diseases and health that may benefit from dietary counselling and using multivitamins and minerals.

### 15. Energy healing such as reiki ("*RAY-kee*") & Therapeutic Touch

Reiki---is a Japanese word representing universal life energy. Therapeutic Touch---A technique called "laying-on of hands", therapist passing their hands over the patient, healers can identify energy imbalances.

### 16. **Reflexology**

Foot therapy, believes foot is a corresponding part of the body.

### 17. Yoga

Mind body therapy, using different techniques, for example, coordination of breath and movement, and holding the postures.

# **Appendix B Survey questionnaire (sample)**

## Complementary Medicine Utilisation in Australia: A National Study (2005)

### Section B: Use of and expenditure on complementary medicine

People use different types of complementary medicine for their health conditions or for improving general health. In order to obtain accurate information, I will need to read to you a list of types of complementary medicine.

QB1. In the last 12 months, have you used any one of the following types of complementary medicine? Please say "yes" or "no" when it is named, I may also ask you whether or not you have visited a practitioner and the number of visits in the last 12 months [*for that complementary medicine*]. Did you use:

Type of complementary medicine	Use in points	past 12	Visite 12 mo	Visited practitioners in past 12 months				
	Yes	No	Yes	No	No. of visits			
1. Acupuncture								
2. Chinese herbal medicine								
3. Chinese therapeutic massage								
4. Chinese medicine dietary therapy								
5. Qigong, martial art or Tai Chi								
6. Western herbal medicine not including Chinese herbal medicine								
7. Massage therapy not including Chinese therapeutic massage								
8. Chiropractic								
9. Osteopathy								
10. Homeopathy								
11. Naturopathy								
12. Meditation								
13. Aromatherapy								
14. Clinical nutrition eg. taking multivitamins or minerals								
15. Energy healing such as reiki and therapeutic touch								
16. Reflexology								
17. Yoga								
18. Prayer [Prevalence only]								
66. Other forms of CAM (Specify)								
[Name of other forms of CAM]:								

QB2. Consider the total amount you spent on complementary medicine products over the last 12 months, <u>not</u> including the cost of visiting practitioners, how much would you estimate this to be? [*Examples of complementary medicine products are herbs, multivitamins, homeopathic and naturopathic medicines etc.* **Ignore** any rebates subsequently received, if there is any, and do <u>not</u> include the cost of visiting practitioners]

1. Enter amount: A\$:
2. Do not remember
99. Refuse to answer

Now, I would like to ask you about your expenditure on visits to complementary medicine practitioners.

QB3. Considering the total amount your complementary medicine practitioners charged you for consultation fees and treatments fees, how much was your personal out-of-pocket expenditure in the last 12 months? [*Do not* include the cost of complementary medicine products, either prescribed by practitioners, or self-prescribed.] [Confirm it is out-of-pocket expenditure, that is, after deducting rebates received from any sources such as insurance or employer.][If participant mentions the use of direct billing system from complementary medicine practitioners, tells participant that we are interested in their out-of-pocket expenditure only]

1. Enter amount: A\$:	
2. Do not remember	
99. Refuse to answer	

QB4. Over the last 12 months, did you spend any additional money on things like courses, books, equipment or any other items related to complementary medicine? If yes, how much would you estimate this to be? [*Do not include the cost answered in QB2 and QB3, that is, expenditure on complimentary medicine products and consultation fee*]

2. [Yes] Do not remember	1. [Yes] Enter amount: A\$:
	2. [Yes] Do not remember
[ 3. [No]	3. [No]
99. Refuse to answer	99. Refuse to answer

# Appendix C Computer programming scripts for the interview (sample)

COMMENT: COMPLEMENTARY MEDICINE UTILISATION IN AUSTRALIA COMMENT: MAY 2005 - WG2932 - FINAL VERSION

COMMENT: Screening section

comment set for live project with sample and quotas

set intype='Live'

set projname='wg2932'

# COMMENT: SECTION B: USE OF AND EXPENDITURE ON COMPLEMENTARY MEDICINE

PREBTXT display '@

People use different types of complementary medicine for their health

@ conditions or for improving general health. In order to obtain accurate

@ information, I will need to read to you a list of types of complementary

@ medicine. My apologies for the long list but this is the only one in this@ study.

@ @ [If interviewee asks WHAT ARE SOME EXAMPLES OF COMPLEMENTARY MEDICINE.

- @ Answer: Examples include acupuncture, herbal medicine, chiropractic and so
- @ on, but DO NOT include allied health such as physiotherapy, occupational

@ therapy, podiatry and speech pathology]

@ @ [If interviewee asks WHAT ARE EXAMPLES OF A COMPLEMENTARY MEDICINE PRODUCT.

@ Answer: Examples include herbs, multivitamins, homeopathic and naturopathic

@ medicines etc.]

@ @ [If interviewee asks WHAT ARE EXAMPLES OF COMPLEMENTARY MEDICINE PRACTITIONER

@ Answer: Examples include acupuncturist, Chinese medicine practitioner,

@ chiropractor, herbalist and Osteopath]@'

pause

qb1list define [NAMES OF ALL FORMS OF CAM]

qb1alst define

'CHINESE HERBAL MEDICINE'/ 'CHINESE THERAPEUTIC MASSAGE'/ 'WESTERN HERBAL MEDICINE NOT INCLUDING CHINESE HERBAL MEDICINE'/ 'MASSAGE THERAPY NOT INCLUDING CHINESE THERAPEUTIC MASSAGE'/ 'CHIROPRACTIC'/ 'OSTEOPATHY'/ 'HOMEOPATHY'/ 'NATUROPATHY'/ 'AROMATHERAPY' qb1blst define

'CHINESE MEDICINE DIETARY THERAPY'/ 'QIGONG, MARTIAL ART OR TAI CHI'/ 'MEDITATION'/ 'CLINICAL NUTRITION EG. TAKING MULTIVITAMINS OR MINERALS'/ 'ENERGY HEALING SUCH AS REIKI AND THERAPEUTIC TOUCH'/ 'REFLEXOLOGY'/ 'YOGA'/

COMMENT : questions for acupuncture only [Additional questions are required]

QB1AA	ask '@
QB1AB	ask '@
QB1AC	ask '@
QB1AD	ask '@

QB1TXT display '@

QB1. In the last 12 months, have you used any one of the following types of @ complementary medicine? Please say "yes" or "no" when it is named, I may @ also ask you whether or not you have visited a practitioner and the number @ of visits in the last 12 months for that complementary medicine.@'

pause

set user=0 set major=0 set minor=0 set cmp=0

unset qb1adum unset qb1bdum

- QB1ADUM dummyask 'Major CAM used' resp mp qb1alst
- QB1BDUM dummyask 'Minor CAM used' resp mp qb1blst
- QB1TDUM dummyask 'All CAM used' resp mp qb1list
  - for QB1L = qb1list ran

set i=iteration

QB1 ask '@ QB1 Did you use ... @@ '+QB1L+'?@' resp sp 'Yes'/ 'No'

# Appendix D Initial telephone contact (with plain language statement)

 $\Rightarrow$  Good [morning/afternoon/evening], my name is [Researcher name] from [Survey research company name]. I am conducting a national study for RMIT University on complementary medicine use.

Your telephone number was randomly selected. I need to conduct the survey with a member of your household over the age of 18. If more than one person over 18, I would like to speak with the person over 18 whose birthday occurs next. Would that member of your household be available?

[NOT PREPARED TO PARTICIPATE: RECORD REFUSAL]

[THE NOMINATED HOUSEHOLD MEMBER IS NOT AVAILABLE

Would it be possible to arrange a time when I could speak with the person that you have identified?]

[IF THIS IS AGREED, ASK FOR THE FIRST NAME OF THE NOMINATED PERSON (IF NOT ALREADY KNOWN)

 $\Rightarrow$ Thank you, I will call at (time, day) to speak with (name).]

[IF THE PERSON NOMINATED IS AVAILABLE AND IS A DIFFERENT PERSON TO THE ONE WHO ANSWERED, <u>REPEAT PARAGRAPH 1, ABOVE BEFORE</u> <u>PROCEEDING</u>]

 $\Rightarrow$  The information you provide will be treated in the strictest confidence and anonymously. No personal identifying information will be collected. You are free to withdraw at any time during the interview, in which case your responses will be deleted.

The study has been approved by the Human Research Ethics Committee of RMIT University. If, you have any questions about the study, please feel free to telephone the Research Team on 03-9925 7002 at RMIT University, or if have any complaints, you may contact the Ethics Committee on 03-9925 1745.

 $\Rightarrow$  The interview will take about 15 minutes, are you prepared to participate in this survey now?

[NOT NOW

Could I call you at another time?] [ARRANGE APPOINTMENT]

[NO – REFUSE TO PARTICIPATE] ⇒[RECORD REFUSAL] [YES – AGREE TO BE INTERVIEWED NOW Thank you very much.] [START THE INTERVIEW]

# **Appendix E Plain language statement (print version)**

# A National Study on Complementary & Alternative Medicine Utilisation

You are being invited to participate in a research project called "Complementary and Alternative Medicine National Study". Thanks for your consideration in take part in this important study, which is a joint initiative of the Division of Chinese Medicine, RMIT University and School of Public Health, LaTrobe University.

## 1. The purpose of the study

This study is the first Australian nationwide study to investigate the use of complementary and alternative medicine (CAM). The purpose of this study is to obtain accurate information on the current prevalence, rationale and public expenditure on the use of various forms of CAM by the Australian population.

## 2. **Study Investigators**

The principal investigator is Mr. Lin Zhang, PhD candidate, RMIT University. The other members of the investigating team include Professor Charlie Xue, Head, Division of Chinese Medicine, RMIT University; Professor Vivian Lin, Head, School of Public Health, LaTrobe University; Professor David Story, Professor in Therapeutic Sciences, RMIT University; Prof. Marc Cohen, Professor in Complementary Medicine, RMIT and, Ms Anne-Louise Carlton, Senior Policy Analyst, Victorian Department of Human Services.

Wallis Consulting Group has been contracted to conduct the telephone interviews for the study, if you have any questions regarding the procedure of the interview, please contact them on 03-96211066.

## **3.** Benefits of the study

The findings of this study are intended to provide reliable and comprehensive information on the prevalence, socio-economic implications and rationales for the use of CAM in Australia. It will also research attitudes to, and the influence of health insurance cover for CAM treatments. The outcomes are intended to provide reliable information of relevance to governments, health professional organisations, government, regulatory bodies and training institutions. This study will make a significant contribution to ensure the public confidence in the effective and safe health care choices of complementary medicine.

### 4. The selection of interview participants

Your telephone number was randomly selected along with other households across Australia by a computer system. As you requested additional information of the project and provided your contact details, we send this information sheet to you. No record of this information will be retained after sending you the requested information. Therefore your privacy will be preserved.

When the interviewer calls, you will be asked a few questions to select an individual within your household. Our target population is adult who is 18 years of age or older. People who have used CAM service and who have never used CAM service are both eligible to participate. When there is more than one eligible person in the household, the methodology used to select the participant will be the "next birthday" technique.

## 5. The procedure of the interview

We are using an interviewing system called computer assisted telephone interview (CATI), the information you provided to the interviewer will be entered into a computer database directly and anonymously. When a "call back" is requested, you will be asked to provide your first name in order to facilitate the re-contact. No record of this information will be retained after the purpose for which it was collected has been satisfied.

Immediately on completion of an interview, the data collected will be incorporated into the aggregate data file and the temporary file deleted to ensure anonymity. The telephone interview should take approximately 15 minutes to complete. In general, you just need to answer questions relevant to your experience and opinion of using CAM services. You are welcome to skip among questions, but we are very much appreciating if you can answer as many questions as possible.

## 6. **Confidentiality**

Your participation is completely voluntary, and you may withdraw from the study at any time during the interview. The information you provide to us will be treated in the strictest confidence and stored in a computer database anonymously. The only people who will have access to the information will be the research team of the study. No personal identifying information will be collected in this study. Publication and presentation of this study result will not identify you as a participant.

## 7. Ethical issues

This study has been reviewed and approved by the Human Research Ethics Committee of RMIT University. If you have any complaints about your participation in this study, you may contact directly to the Secretary, RMIT Human Research Ethics Committee, University Secretariat, RMIT, GPO Box 2476V, Melbourne, 3001 or by telephone on (03) 9925 1745.

## 8. **Questions**

The principal investigator, Mr Lin Zhang will be happy to answer any questions at any time, and he may be contacted in the School of Health Sciences, RMIT University on 03-99257002 or on Mobile 0412030771.

We appreciate your time to take part in this important national study. Thank you very much.

# Appendix F Human ethics study approval form

2nd February 2005

Lin Zhang Division of Chinese Medicine School of Health Sciences RMIT University Bundoora Campus

Dear Lin,

# **FLSAPP 30 – 04 ZHANG Prevalence, expenditure and trends of complementary and alternative medicine usage in Australia: a national population-based study**

Thank you for submitting your amended application for review.

I am pleased to inform you that the committee has approved your application for a period of  $\underline{3}$  years to <u>February 2008</u> and your research may now proceed.

The committee would like to remind you that annual reports are due during December for all research projects that have been approved by the Human Research Ethics Sub-Committee.

The necessary form can be found at: <u>http://www.rmit.edu.au/browse;ID=sp7y1u3kp66w;STATUS=A?QRY=ethics&STYPE=ENT</u> IRE

Yours faithfully,

Diana Donohue, Deputy Chair, Science, Engineering & Technology Portfolio HREC Sub-committee Life Sciences

cc: Charlie Xue Barbara Polus

# Appendix G Additional statistical outputs

Id	IDENTIFICATION
Phone	PHONE PREFIX
Pcode	POSTCODE
Region	STATE OR TERRITORY
f1	OFI AGE
f2	OE CENDER
12 f2	QE2 OENDER
13	OF4 ENDLOYMENT STATUS
14	QF4 EMPLOT MENT STATUS
15	QES ANNUAL HOUSEHOLD INCOME
10	
a1 o2	AT FERSONAL HEALTH KATINO
a2	A2 MEDICAL CONDITIONS VISITED MD EOD
a.3	A NUMBED OF VISITS TO MD
a4 D1@1	
DI@I D1@2	DI USED ACOFONCIORE
D1@2	DI USED CHINESE THERADELITIC MARSAGE
B1@3	BI USED CHINESE HERAPEUTIC MASSAGE
B1@4	BI USED CHINESE MEDICINE DIETAK Y THEKAPY
	DI USED WEGTEDN HEDDAL MEDICINE NOT INCLUDING CUM
D1@0	DI USED WESTEKIN HEKBAL MEDICINE NOT INCLUDING CHM DI USED MASSACE THEDADY NOT INCLUDING CTM
B1@8	
B1@10	
B1@10	
D1@11	
D1@12	
D1@13	DI USED AROMATILERAFI DI USED CUNICAL NUTDITION
B1@14	BI USED ENERGY HEALING
B1@15	BI USED EFE EVOLOGY
B1@10	BI USED VOCA
D1@17	
D1@10	
B1ba@1	BLAR VISITED DRACTITIONED FOR ACUDINCTURE IN LAST 12 MONTHS
Diba@1	DIAD VISITED TRACTITIONER FOR ACCIONCTORE IN LAST 12 MONTHS
Blac	BLAC TIMES VISITED A MD FOR A CUDINICTURE TREATMENT
Blad	BIAC TIMES VISITED NON MD FOR ACUDINCTURE TREATMENT
B1ba@2	BIRA VISITED A DRACTITIONED FOR ACUI ONCLORE HERRAL MEDICINE
B1ba@3	BIBA VISITED A PRACTITIONER FOR CHINESE THERAPEUTIC MASSAGE
B1ba@4	BIBA VISITED A PRACTITIONER FOR CHINESE MEDICINE DIETARY THERAPY
B1ba@5	BIBA VISITED A PRACTITIONER FOR OLGONG MARTIAL ART OR TALCHI
B1ba@6	BIBA VISITED A PRACTITIONER FOR WESTERN HERBAL MEDICINE
B1ba@7	BIBA VISITED A PRACTITIONER FOR MASSAGE THERAPY
B1ba@8	BIBA VISITED A PRACTITIONER FOR CHIROPRACTIC
B1ba@9	BIBA VISITED A PRACTITIONER FOR OSTEOPATHY
B1ba@10	BIBA VISITED A PRACTITIONER FOR HOMEOPATHY
B1ba@11	BIBA VISITED A PRACTITIONER FOR NATUROPATHY
B1ba@12	B1BA VISITED A PRACTITIONER FOR MEDITATION
B1ba@13	BIBA VISITED A PRACTITIONER FOR AROMATHERAPY
B1ba@14	BIBA VISITED A PRACTITIONER FOR CLINICAL NUTRITION
B1ba@15	B1BA VISITED A PRACTITIONER FOR ENERGY HEALING
B1ba@16	BIBA VISITED A PRACTITIONER FOR REFLEXOLOGY
B1ba@17	BIBA VISITED A PRACTITIONER FOR YOGA
b1ba@18	B1BA VISITED A PRACTITIONER FOR PRAYER
b1bb@1	B1BB TIMES VISITED A PRACTITIONER FOR ACUPUNCTURE

# Appendix G1 Variables employed in the survey

b1bb@2	B1BB TIMES VISITED A PRACTITIONER FOR CHINESE HERBAL MEDICINE
b1bb@3	B1BB TIMES VISITED A PRACTITIONER FOR CHINESE THERAPEUTIC MASSAGE
b1bb@6	B1BB TIMES VISITED A PRACTITIONER FOR WESTERN HERBAL MEDICINE
b1bb@7	B1BB TIMES VISITED A PRACTITIONER FOR MASSAGE THERAPY
b1bb@8	B1BB TIMES VISITED A PRACTITIONER FOR CHIROPRACTIC
b1bb@9	B1BB TIMES VISITED A PRACTITIONER FOR OSTEOPATHY
b1bb@10	B1BB TIMES VISITED A PRACTITIONER FOR HOMEOPATHY
b1bb@11	B1BB TIMES VISITED A PRACTITIONER FOR NATUROPATHY
b1bb@13	B1BB TIMES VISITED A PRACTITIONER FOR AROMATHERAPY
b2	B2 AMOUNT SPENT ON CAM
b3	B3 AMOUNT SPENT ON CAM PRACTITIONERS
b4	B4 SPENT ON RELATED CAM YES/NO
b4x	B4 SPENT ON RELATED CAM
b5	QB5. EVER USED ANY FORM OF CAM
c1@1	Q1C WESTERN MEDICINE WOULD NOT IMPROVE YOUR MEDICAL CONDITIONS
c1@2	Q1C WM WOULD IMPROVE MEDICAL CONDITIONS BUT NOT CURE THEM
c1@3	Q1C WM HAS A RELATIVELY HIGHER RISK OF SIDE EFFECTS
c1@4	Q1C SERVICE PROVISION BY WMP IS GENERALLY UNSATISFACTORY
c1@5	Q1C COMPLEMENTARY MEDICINE IS RELATIVELY SAFE
c1@6	Q1C CAM PROVIDES A MORE HOLISTIC APPROACH TO HEALTH
c1@7	Q1C CAM IS A MORE NATURAL APPROACH TO TREATMENT
c2	QC2. USED CAM TOGETHER WITH WM
c3	QC3. TELL DOCTOR WHEN USE COMPLEMENTARY MEDICINE
@c41	QC4. REASONS TELL DOCTOR ABOUT CAM USAGE (1)
@c42	QC4. REASONS TELL DOCTOR ABOUT CAM USAGE (2)
@c43	QC4. REASONS TELL DOCTOR ABOUT CAM USAGE (3)
@c44	QC4. REASONS TELL DOCTOR ABOUT CAM USAGE (NONE)
@c51	QC5. REASON DO NOT TELL DOCTOR ABOUT CAM USAGE (1)
@c52	QC5. REASON DO NOT TELL DOCTOR ABOUT CAM USAGE (2)
@c53	QC5. REASON DO NOT TELL DOCTOR ABOUT CAM USAGE (3)
@c54	QC5. REASON DO NOT TELL DOCTOR ABOUT CAM USAGE (4)
@c55	QC5. REASON DO NOT TELL DOCTOR ABOUT CAM USAGE (5)
@c56	QC5. REASON DO NOT TELL DOCTOR ABOUT CAM USAGE (NONE)
<u>c6</u>	QC6. CONSIDER USING CAM IN THE FUTURE
C/	QC7. BELIEVE CAM SHOULD BE REGULATED AS RIGOROUSLY AS WM
d1	OD2 DDIVATE HEALTH INSUDANCE COVED COST OF CAM
d2	OD3 OBTAINED A REBATE FOR CAM FROM DHI IN LAST 12 MONTHS
u3 @d41	QDS ODTAINED A REDATE FOR CAM FROM FILLIN LAST 12 MONTHS
@d41	OD/ FORM OF CAM RECEIVED A REBATE FROM PHI (ACO)
@d42	OD/ FORM OF CAM RECEIVED A REBATE FROM PHI (CTM)
@d43	OD/ FORM OF CAM RECEIVED A REBATE FROM PHI (CHM)
@d45	OD4 FORM OF CAM RECEIVED A REBATE FROM PHI (OIGONG)
@d46	OD4 FORM OF CAM RECEIVED A REBATE FROM PHI (WHM)
@d47	OD4 FORM OF CAM RECEIVED A REBATE FROM PHI (WASSAGE)
@d48	OD4 FORM OF CAM RECEIVED A REBATE FROM PHI (CHIROPRACTIC)
@d49	OD4 FORM OF CAM RECEIVED A REBATE FROM PHI (OSTEOPATHY)
@d410	OD4 FORM OF CAM RECEIVED A REBATE FROM PHI (HOMEOPATHY)
@d411	QD4 FORM OF CAM RECEIVED A REBATE FROM PHI (NATUROPATHY)
@d412	QD4 FORM OF CAM RECEIVED A REBATE FROM PHI (MEDITATION)
@d413	QD4 FORM OF CAM RECEIVED A REBATE FROM PHI (AROMATHERAPY)
@d414	QD4 FORM OF CAM RECEIVED A REBATE FROM PHI (CLINICAL NUTRITION)
@d415	QD4 FORM OF CAM RECEIVED A REBATE FROM PHI (ENERGY HEALING)
@d416	QD4 FORM OF CAM RECEIVED A REBATE FROM PHI (REFLEXOLOGY)
@d417	QD4 FORM OF CAM RECEIVED A REBATE FROM PHI (YOGA)
@d418	QD4 FORM OF CAM RECEIVED A REBATE FROM PHI (PRAYER)
@d419	QD4 FORM OF CAM RECEIVED A REBATE FROM PHI (OTHER)
@d420	QD4 FORM OF CAM RECEIVED A REBATE FROM PHI (DON'T KNOW)
@d421	QD4 FORM OF CAM RECEIVED A REBATE FROM PHI (REFUSED ANSWER)
d5	QD5 COVERAGE OF CAM PRODUCTS INFLUENCE INSURANCE PURCHASE?

e1	E1 AMOUNT SPENT ON ITEMS RELATED TO ACUPUNCTURE
e2	QE2 REASONS USED ACUPUNCTURE TREATMENT
e3@1	QE3 CURE THE DISEASE/ SOLVE THE PROBLEM (BEFORE ACU)
e3@2	OE3 RELIEVE THE SYMPTOMS (BEFORE ACU)
e3@3	QE3 IMPROVE YOUR GENERAL HEALTH AND WELL-BEING (BEFORE ACU)
e4@1	OE4 CURE THE DISEASE/ SOLVE THE PROBLEM (AFTER ACU)
e4@2	OE4 RELIEVE THE SYMPTOMS (AFTER ACU)
e4@3	OF4 IMPROVE YOUR GENERAL HEALTH AND WELL-BEING (AFTER ACU)
e4x	OF4X DID THE ACLIPLINCTURE HAVE NO FEFECT
e5	OF5 ENCOUNTERED ANY ADVERSE EFFECTS OR SIDE EFFECTS (ACU)
e6	OF6 HOW HELPEUL IN IMPROVING GENERAL HEALTH (ACU)
@e71	OF7 LISED ACLIPLINCTURE ON ADVICE FROM A MD?
@e72	OF7 LISED A CURLINCTURE ON A DVICE FROM A CAM PRACTITIONER?
@e73	OF7 LISED ACUPUNCTURE ON ADVICE FROM FRIENDS/RELATIVES?
@e74	OF7 USED ACUPUNCTURE ON ADVICE FROM NEWS/TV/INTERNET
@075	OF7 USED A CUDUNCTUDE ON A DVICE FROM OTHED HEAT TH DDOFESSIONALS
@e75	QE7 USED ACUPUNCTURE ON ADVICE FROM OTHER HEALTH FROMESSIONALS
@277	QE7 USED ACUPUNCTURE ON ADVICE EDOM (DON'T KNOW)
@e//	QE7 USED ACUPUNCTURE ON ADVICE FROM (DON T KNOW)
20	VEO CONSIDER USING ACUPUNCTURE IN THE FUTURE?
210	VE7 REQUEATION FROVIDES FOR OREATER FUBLIC SAFETT & CONFIDENCE (ACU)
011	VETVALE STATES/TEKKTTOKIES SHOULD KEVUIKE GOVEKIMENT KEGULATION (ACU)
e11 a12	VELT MEDICARE SHOULD COVER TREATMENT PROVIDED BY ACUPUNCTURISTS
e12	VE12 VIGHTS TO ALL ACUPUNCTURISTS SHOULD BE COVERED BY PHI
e13	E13 AMOUNT SPENT ON ITEMS FOR CHINESE HERD AL MEDICINE
e14	E14 AMOUNT SPENT ON TIEMS FOR CHINESE HERBAL MEDICINE
e15	QEIS REASONS USED CHINESE HERBAL MEDICINE TREATMENT
e16@1	QE16 CUKE THE DISEASE/ SOLVE THE PROBLEM (BEFORE CHM)
e16@2	QE16 RELIEVE THE SYMPTOMS (BEFORE CHM)
e16@3	QE16 IMPROVE YOUR GENERAL HEALTH AND WELL-BEING (BEFORE CHM)
el7@1	QE17 CURE THE DISEASE/ SOLVE THE PROBLEM (AFTER CHM)
el7@2	QE17 RELIEVE THE SYMPTOMS (AFTER CHM)
el7@3	QE17 IMPROVE YOUR GENERAL HEALTH AND WELL-BEING (AFTER CHM)
el7x	QE17X DID THE CHINESE HERBAL MEDICINE HAVE NO EFFECT
e18	QE18 ENCOUNTERED ANY ADVERSE EFFECTS OR SIDE EFFECTS (CHM)
e19	QE19 HOW HELPFUL IN IMPROVING GENERAL HEALTH (CHM)
@e201	QE20 USED CHM ON ADVICE FROM MD
@e202	QE20 USED CHM ON ADVICE FROM CAM PRACTITIONERS
@e203	QE20 USED CHM ON ADVICE FROM FRIENDS/RELATIVES
@e204	QE20 USED CHM ON ADVICE FROM NEWS/TV/INTERNET
@e205	QE20 USED CHM ON ADVICE FROM OTHER HEATH PROFESSIONALS
@e206	QE20 USED CHM WITHOUT ADVICE
@e207	QE20 USED CHM ON ADVICE FROM (DON'T KNOW)
e21	QE21 CONSIDER USING CHINESE HERBAL MEDICINE IN THE FUTURE
e22	QE22 REGULATION PROVIDES FOR GREATER PUBLIC SAFETY & CONFIDENCE (CHM)
e23	VE23 ALL STATES/TERKITOKIES SHOULD REQUIRE GOVERNMENT REGULATION (CHM)
e24	VE24 MEDICAKE SHOULD COVER TREATMENT PROVIDED BY CHM PRACTITIONERS
e25	VE72 A 19112 TO CHIM LKYCTHIONERS 2HOULD RE COARRED RA LHI
@e261	VE20 KEASUNS FOR VISITING CHIROPRACTOR (1)
@e262	VE20 KEASUNS FUK VISITING CHIROPRACTOR (2)
@e263	QE26 REASONS FOR VISITING CHIROPRACTOR (3)
@e264	VE20 KEASUNS FUK VISITING CHIROPRACTOR (4)
@e265	QE26 REASONS FOR VISITING CHIROPRACTOR (NONE OF ABOVE)
e27	E27 AMOUNT SPENT ON CHIROPRACTIC TIEMS IN LAST 12 MONTHS
e28	QE28 ENCOUNTERED ANY ADVERSE EFFECTS OR SIDE EFFECTS (CHIRO)
@e291	QE29 OUTCOME OF THE CHIROPRACTIC TREATMENT (1)
@e292	QE29 OUTCOME OF THE CHIROPRACTIC TREATMENT (2)
@e293	QE29 OUTCOME OF THE CHIROPRACTIC TREATMENT (3)
@e294	QE29 OUTCOME OF THE CHIROPRACTIC TREATMENT (4)
@e295	QE29 OUTCOME OF THE CHIROPRACTIC TREATMENT (NONE OF ABOVE)
e30	QE30 HOW HELPFUL IN IMPROVING GENERAL HEALTH (CHIRO)
@e311	QE31 USED CHIRO ON ADVICE FROM MD
@e312	QE31 USED CHIRO ON ADVICE FROM CAM PRACTITIONER

@e313	QE31 USED CHIRO ON ADVICE FROM FRIENDS/RELATIVES
@e314	QE31 USED CHIRO ON ADVICE FROM NEWS/TV/INTERNET
@e315	QE31 USED CHIRO ON ADVICE FROM OTHER HEATH PROFESSIONALS
@e316	QE31 USED CHIRO WITHOUT ADVICE
@e317	QE31 USED CHIRO ON ADVICE FROM (DON'T KNOW)
e32	OE32 CONSIDER USING CHIROPRACTOR IN THE FUTURE
e33	OE33 MEDICARE SHOULD COVER TREATMENT PROVIDED BY CHIROPRACTOR
e34	OE34 VISITS TO CHIROPRACTOR SHOULD BE COVERED BY PHI
e35	OE35 AGREE CHIROPRACTIC CARE IS A SAFE FORM OF TREATMENT
@e361	OE36 REASONS FOR USING OSTEOPATHY (1)
@e362	OE36 REASONS FOR USING OSTEOPATHY (2)
@e363	OE36 REASONS FOR USING OSTEOPATHY (3)
@e364	QE36 REASONS FOR USING OSTEOPATHY (4)
@e365	QE36 REASONS FOR USING OSTEOPATHY (NONE OF ABOVE)
e37	E37 AMOUNT SPENT ON ITEMS RELATED TO OSTEOPATHY
@e381	QE38 TYPE OF TREATMENT THE OSTEOPATH USED (1)
@e382	OE38 TYPE OF TREATMENT THE OSTEOPATH USED (2)
@e383	OE38 TYPE OF TREATMENT THE OSTEOPATH USED (3)
@e384	OE38 TYPE OF TREATMENT THE OSTEOPATH USED (4)
@e385	OE38 TYPE OF TREATMENT THE OSTEOPATH USED (1)
@e386	OE38 TYPE OF TREATMENT THE OSTEOPATH USED (NONE OF ABOVE)
@e391	OE39 OUTCOME OF THE OSTEOPATHY TREATMENT (1)
@e392	OE39 OUTCOME OF THE OSTEOPATHY TREATMENT (2)
@e393	OE39 OUTCOME OF THE OSTEOPATHY TREATMENT (3)
@e394	OE39 OUTCOME OF THE OSTEOPATHY TREATMENT (4)
@e395	OE39 OUTCOME OF THE OSTEOPATHY TREATMENT (NONE OF ABOVE)
e40	OE40 ENCOUNTERED ANY ADVERSE EFFECTS OR SIDE EFFECTS (OSTEO)
e41	OE41 HOW HELPFUL IN IMPROVING GENERAL HEALTH (OSTEO)
@e421	OE42 USED OSTEOPATH ON ADVICE FROM MD
@e422	OE42 USED OSTEOPATH ON ADVICE FROM CAM PRACTITIONERS
@e423	OE42 USED OSTEOPATH ON ADVICE FROM FRIENDS/RELATIVES
@e424	OE42 USED OSTEOPATH ON ADVICE FROM NEWS/TV/INTERNET
@e425	OE42 USED OSTEOPATH ON ADVICE FROM OTHER HEALTH PROFESSIONALS
@e426	OE42 USED OSTEOPATH WITHOUT ADVICE
@e427	QE42 USED OSTEOPATH ON ADVICE FROM (DON'T KNOW)
e43	QE43 CONSIDER USING OSTEOPATH IN THE FUTURE
e44	QE44 MEDICARE SHOULD COVER TREATMENT PROVIDED BY OSTEOPATH
e45	QE45 VISITS TO OSTEOPATH SHOULD BE COVERED BY PHI
QA3@0	A3 MEDICAL CONDITIONS VISITED MD FOR
QA3@1	A3 MEDICAL CONDITION (1) VISITED MD FOR
QA3@2	A3 MEDICAL CONDITION (2) VISITED MD FOR
QA3@3	A3 MEDICAL CONDITION (3) VISITED MD FOR
QA3@4	A3 MEDICAL CONDITION (4) VISITED MD FOR
QA3@5	A3 MEDICAL CONDITION (5) VISITED MD FOR
QE2X@0	QE2 REASONS USED ACUPUNCTURE TREATMENT
QE2X@1	QE2 REASON (1) USED ACUPUNCTURE TREATMENT
QE2X@2	QE2 REASON (2) USED ACUPUNCTURE TREATMENT
QE15X@0	QE15 REASONS USED CHINESE HERBAL MEDICINE TREATMENT
QE15X@1	QE15 REASON (1) USED CHINESE HERBAL MEDICINE TREATMENT
QE15X@2	QE15 REASON (2) USED CHINESE HERBAL MEDICINE TREATMENT
QE26X@0	QE26 SYMPTOMS FOR VISITING CHIROPRACTOR
QE26X@1	QE26 SYMPTOM (1) FOR VISITING CHIROPRACTOR
QE26X@2	QE26 SYMPTOM (2) FOR VISITING CHIROPRACTOR
QE26X@3	QE26 SYMPTOM (3) FOR VISITING CHIROPRACTOR
QE36X@0	QE36 SYMPTOMS FOR VISITING OSTEOPATH
QE36X@1	QE36 SYMPTOM (1) FOR VISITING OSTEOPATH
QE36X@2	QE36 SYMPTOM (2) FOR VISITING OSTEOPATH
QE36X@3	QE36 SYMPTOM (3) FOR VISITING OSTEOPATH
QE36X@4	QE36 SYMPTOM (4) FOR VISITING OSTEOPATH

Acupuncture		СНМ		CHM CTM		WHM WTM		Chiropractic		Osteopathy		Homeopathy		Naturopathy		Aromatherapy			
No. of	Frequ	No. of	Frequ	No. of	Frequ	No. of	Frequ	No. of	Frequ	No. of	Frequ	No. of	Frequ	No. of	Frequ	No. of	Frequ	No. of	Frequ
visits	ency	visits	ency	visits	ency	visits	ency	visits	ency	visits	ency	visits	ency	visits	ency	visits	ency	visits	ency
1	5	1	4	1	7	1	13	1	42	1	22	1	7	1	8	1	18	1	10
2	11	2	4	2	4	2	12	2	39	2	24	2	5	2	9	2	13	2	10
3	18	3	4	3	8	3	9	3	30	3	14	3	6	3	2	3	7	3	3
4	8	4	2	4	6	4	5	4	24	4	10	4	7	4	3	4	6	4	3
5	9	5	4	5	4	5	1	5	8	5	10	6	6	5	1	5	2	5	2
6	9	6	4	6	1	6	4	6	18	6	15	8	1	6	3	6	6	6	1
8	4	10	1	8	1	8	1	7	3	7	1	10	1	12	2	8	1	12	2
10	2	12	1	10	1	10	3	8	4	8	8	12	2	52	2	10	2	30	1
12	7	16	1	12	1	12	1	9	1	9	1	14	1	DK	1	12	5	300	1
16	1	20	1	24	1	20	1	10	8	10	9	15	1	Total	31	52	2	365	1
20	2	26	1	25	1	DK	2	12	19	11	1	20	1			DK	1	Total	34
26	1	Total	27	26	1	Total	52	14	1	12	25	24	1			Total	63		
30	1			Total	36			15	1	14	3	25	1						
32	1							19	1	15	6	Total	40						
45	1							20	4	20	1								
50	1							24	2	24	4								
112	1							30	2	26	1								
Total	82							35	1	35	1								
								40	1	40	1								
								50	2	50	1								
								52	1	60	1								
								55	1	120	1								
								DK	2	Total	160								
								Total	215										

# Appendix G2 Frequency of visits to CAM practitioners (unadjusted data)

DK: Don't know, CHM: Chinese herbal medicine, CTM: Chinese therapeutic massage, WHM: Western herbal medicine, WTM: Western therapeutic massage

Statistics/CAN	I therapies	Acupuncture	CHM	СТМ	WHM	WTM	Chiropractic	Osteopathy	Homeopathy	Naturopathy	Aromatherapy
Ν	Valid	80	24	34	48	213	155	37	30	62	31
	Missing/										
	outliers	0	0	0	2	2	0	0	1	1	2
Mean		8.75	5.98	5.43	3.75	6.29	8.44	5.69	6.65	5.32	3.98
Std. Error of Mea	an	1.82	1.27	1.14	0.53	0.60	1.00	0.97	2.44	1.21	1.12
Median		4.00	4.00	3.00	2.06	3.00	5.00	4.00	2.00	2.00	2.00
Mode		3.00	3.00	3.00	1.00	1.00	2.00	1.00	2.00	1.00	1.00
Std. Deviation		16.32	6.25	6.60	3.65	8.70	12.53	5.92	13.26	9.55	6.24
Variance		266.19	39.06	43.50	13.34	75.72	156.88	35.03	175.82	91.22	38.97
Skewness		5.05	2.12	2.53	2.41	3.48	5.75	1.97	3.22	4.28	3.69
Std. Error of Ske	wness	0.27	0.47	0.40	0.34	0.17	0.19	0.39	0.43	0.30	0.42
Kurtosis		29.08	4.43	5.56	7.46	14.26	44.62	3.62	9.51	19.25	14.18
Std. Error of Kur	tosis	0.53	0.91	0.79	0.67	0.33	0.39	0.76	0.84	0.60	0.82
Minimum		1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00
Maximum		112.00	26.00	26.00	20.00	55.00	120.00	25.00	52.00	52.00	30.00
Sum		699.86	145.92	182.57	179.98	1337.59	1311.43	211.49	196.85	331.11	123.22
Percentiles	95	33.11	24.84	25.34	10.84	20.00	24.00	23.30	52.00	12.61	26.57

# Appendix G3 Statistics of frequency of CAM practitioner visits (adjusted data)

CHM: Chinese herbal medicine, CTM: Chinese therapeutic massage, WHM: Western herbal medicine, WTM: Western therapeutic massage

<b>D</b> 11		~	Valid	Cumulative	D 11		4	Valid	Cumulative	D 11	r.	e e e e e e e e e e e e e e e e e e e	Valid	Cumulative
Dollar	Frequency	%	(%)	(%)	Dollar	Frequency	%	(%)	(%)	Dollar	Frequency	%	(%)	(%)
0	115	15.6	16.4	16.4	80	13	1.8	1.9	45.9	500	31	4.2	4.4	90.5
1	1	.1	.1	16.5	90	2	.3	.3	46.2	600	10	1.4	1.4	91.9
5	2	.3	.3	16.8	99	1	.1	.1	46.3	650	1	.1	.1	92.0
6	1	.1	.1	17.0	100	75	10.2	10.7	57.0	700	2	.3	.3	92.3
8	1	.1	.1	17.1	110	2	.3	.3	57.3	750	3	.4	.4	92.7
10	5	.7	.7	17.8	120	5	.7	.7	58.0	800	3	.4	.4	93.2
12	1	.1	.1	17.9	130	1	.1	.1	58.1	850	1	.1	.1	93.3
15	2	.3	3	18.2	134	1	.1	.1	58.3	1000	23	3.1	3.3	96.6
16	2	3	3	18.5	140	2	3	3	58.5	1040	l T	.1	.1	96.7
18	1	.5	.5	18.7	150	31	.5		63.0	1200	5	./	./	97.4
20	26	.1	.1	22.9	160	1	4.2	4.4	62.1	1400	1	.1	.1	97.0
20	50	4.9	5.1	23.8	170	1	.1	.1	03.1	1050	3	./	./	98.5
25	6	.8	.9	24.6	170	1	.1	.1	63.2	2000	1	.1	.1	98.4
20	1	.1	.1	24.8	180	1	.1	.1	63.4	3000	1	.1	.1	98.0
30	27	3.7	3.8	28.6	200	79	10.7	11.3	74.6	3120	3	.4	.4	99.0
35	2	.3	.3	28.9	230	1	.1	.1	74.8	3500	1	.1	.1	99.1
36	1	.1	.1	29.1	240	2	.3	.3	75.1	4500	1	.1	.1	99.4
40	9	1.2	1.3	30.3	250	14	1.9	2.0	77.1	5000	2	.1	.1	99. <del>7</del>
45	2	.3	.3	30.6	275	1	.1	.1	77.2	6000	1	.5	.5	99.9
50	58	7.9	8.3	38.9	300	30	4.1	4.3	81.5	10000	1	.1	.1	100.0
55	1	.1	.1	39.0	350	8	1.1	1.1	82.6	Sub-		0.5.0	100.0	10010
60	24	3.3	3.4	42.5	360	3	.4	.4	83.0	total	702	95.3	100.0	
65	1	.1	.1	42.6	400	15	2.0	2.1	85.2	Don't Know	35	4.7		
70	7	.9	1.0	43.6	450	5	.7	.7	85.9	Know	55	,		
75	3	.4	.4	44.0	480	1	.1	.1	86.0	Total	737	100.0		

# Appendix G4 Expenditure on CAM products (unadjusted data)

			Valid	Cumulative				Valid	Cumulative				Valid	Cumulative
Dollar	Frequency	%	(%)	(%)	Dollar	Frequency	%	(%)	(%)	Dollar	Frequency	%	(%)	(%)
0	93	19.7	20.6	20.6	137	1	.2	.2	47.1	550	1	.2	.2	83.4
5	2	.4	.4	21.0	140	2	.4	.4	47.6	575	1	.2	.2	83.6
8	1	.2	.2	21.2	150	11	2.3	2.4	50.0	600	15	3.2	3.3	86.9
10	2	.4	.4	21.7	160	3	.6	.7	50.7	630	2	.4	.4	87.4
13	1	.2	.2	21.9	170	1	.2	.2	50.9	640	1	.2	.2	87.6
15	1	.2	.2	22.1	180	3	.6	.7	51.5	650	1	.2	.2	87.8
20	3	.6	.7	22.8	192	1	.2	.2	51.8	700	8	1.7	1.8	89.6
30	9	1.9	2.0	24.8	200	45	9.5	10.0	61.7	720	1	.2	.2	89.8
32	2	.4	.4	25.2	220	2	.4	.4	62.2	750	2	.4	.4	90.3
35	2	.4	.4	25.7	230	1	.2	.2	62.4	800	1	.2	.2	90.5
37	1	.2	.2	25.9	239	1	.2	.2	62.6	840	1	.2	.2	90.7
40	6	1.3	1.3	27.2	240	4	.8	.9	63.5	900	1	.2	.2	90.9
45	1	.2	.2	27.4	250	8	1.7	1.8	65.3	1000	19	4.0	4.2	95.1
48	2	.4	.4	27.9	280	1	.2	.2	65.5	1200	1	.2	.2	95.4
50	11	2.3	2.4	30.3	300	23	4.9	5.1	70.6	1205	1	.2	.2	95.6
55	1	.2	.2	30.5	312	1	.2	.2	70.8	1250	1	.2	.2	95.8
60	8	1.7	1.8	32.3	320	1	.2	.2	71.0	1300	1	.2	.2	96.0
65	2	.4	.4	32.7	330	1	.2	.2	71.2	1500	6	1.3	1.3	97.3
70	4	.8	.9	33.6	350	6	1.3	1.3	72.6	2000	8	1.7	1.8	99.1
80	9	1.9	2.0	35.6	360	1	.2	.2	72.8	3000	1	.2	.2	99.3
88	1	.2	.2	35.8	361	1	.2	.2	73.0	6500	1	.2	.2	99.6
90	5	1.1	1.1	36.9	380	1	.2	.2	73.2	7000	1	.2	.2	99.8
99	1	.2	.2	37.2	385	1	.2	.2	73.5	10000	1	.2	.2	100.0
100	33	7.0	7.3	44.5	400	24	5.1	5.3	78.8	Total	452	95.6	100.0	
102	1	.2	.2	44.7	440	1	.2	.2	79.0	Don't	21	1 1		
108	1	.2	.2	44.9	450	1	.2	.2	79.2	Know	21	4.4		
120	9	1.9	2.0	46.9	500	18	3.8	4.0	83.2	Total	473	100.0		

# Appendix G5 Expenditure on CAM practitioners (unadjusted data)

Dollar	Frequency	0%	Valid	Cumulative	Dollar	Frequency	0%	Valid	Cumulative
2	frequency	10	(70)	(10)	176	requency	10	(70)	(1)
3	1	.0	.0	.0	170	1	.0	.0	64.0
10	1	.0	.0	1.1	180	2	1.1	1.1	64.0
10	2	1.1	1.1	2.3	190	1	.0	.0	64.6
12	1	.6	.6	2.9	195	1	.0	.6	65.1
15	1	.6	.6	3.4	200	21	11.7	12.0	77.1
20	9	5.0	5.1	8.6	240	1	.6	.6	77.7
25	3	1.7	1.7	10.3	250	2	1.1	1.1	78.9
30	12	6.7	6.9	17.1	300	6	3.4	3.4	82.3
35	2	1.1	1.1	18.3	400	6	3.4	3.4	85.7
40	5	2.8	2.9	21.1	500	4	2.2	2.3	88.0
45	2	1.1	1.1	22.3	600	4	2.2	2.3	90.3
50	17	9.5	9.7	32.0	650	1	.6	.6	90.9
60	5	2.8	2.9	34.9	800	2	1.1	1.1	92.0
70	2	1.1	1.1	36.0	1000	5	2.8	2.9	94.9
75	3	1.7	1.7	37.7	1500	3	1.7	1.7	96.6
80	4	2.2	2.3	40.0	1800	1	.6	.6	97.1
90	1	.6	.6	40.6	2000	1	.6	.6	97.7
100	22	12.3	12.6	53.1	3000	2	1.1	1.1	98.9
110	1	.6	.6	53.7	5000	1	.6	.6	99.4
120	4	2.2	2.3	56.0	6000	1	.6	.6	100.0
140	2	1.1	1.1	57.1	Total	175	97.8	100.0	
150	6	3.4	3.4	60.6	Denk	175	27.0	100.0	
160	1	.6	.6	61.1	Know	4	2.2		
170	2	1 1	1.1	62.2		4	2.2		
170	2	1.1	1.1	02.3	Total	179	100.0		

Appendix G6 Expenditure on CAM-related items (unadjusted data)

# Appendix G7 Statistics of expenditure on CAM (unadjusted data)

### **B2 AMOUNT SPENT ON CAM**

### **B3 AMOUNT SPENT ON CAM PRACTITIONER**

### **B4 AMOUNT SPENT ON RELATED CAM**

Ν	Valid	702				
	Missing	35				
Mean		260.54				
Std. Error of Mean		24.322				
Median	100.00					
Mode	0					
Std. Deviation	644.430					
Variance		415290.2				
Skewness		8.305				
Std. Error of Skewness		.092				
Kurtosis		95.470				
Std. Error of Kurtosis		.184				
Range		10000				
Minimum	Minimum					
Maximum	10000					
Sum		182902				

Ν	Valid	452				
	Missing	21				
Mean		341.63				
Std. Error of Mean		35.020				
Median		155.00				
Mode	0					
Std. Deviation	744.529					
Variance	554322.9					
Skewness		8.082				
Std. Error of Skewness		.115				
Kurtosis		86.456				
Std. Error of Kurtosis		.229				
Range		10000				
Minimum	Minimum					
Maximum	10000					
Sum	154416					

Ν	Valid	175
	Missing	4
Mean		311.07
Std. Error of Mean		53.939
Median	100.00	
Mode	100	
Std. Deviation	713.540	
Variance	509139.7	
Skewness		5.434
Std. Error of Skewness		.184
Kurtosis		34.953
Std. Error of Kurtosis		.365
Range		5997
Minimum		3
Maximum	6000	
Sum		54438

# Appendix G8 Statistics of expenditure on CAM (adjusted data)

### **B2 AMOUNT SPENT ON CAM**

Ν	Valid	665		
	Missing	34		
Mean		182.30		
Std. Error of Mean		9.253		
Median		100.00		
Mode	0			
Std. Deviation	238.672			
Variance	56964.515			
Skewness		2.290		
Std. Error of Skewnes	S	.095		
Kurtosis		5.641		
Std. Error of Kurtosis		.189		
Range		1400		
Minimum		0		
Maximum	1400			
Sum	121298			
Percentiles	95	700.00		

Ν	Valid	427
	Missing	19
Mean		263.64
Std. Error of Mean	15.389	
Median	150.58	
Mode		0
Std. Deviation		317.813
Variance	101004.956	
Skewness		2.000
Std. Error of Skewness		.118
Kurtosis		4.643
Std. Error of Kurtosis		.236
Range		2000
Minimum		0
Maximum	2000	
Sum	112445	
Percentiles	95	1000.00

**B3 AMOUNT SPENT ON CAM PRACTITIONER** 

### **B4 AMOUNT SPENT ON RELATED CAM**

Ν	Valid	168		
	Missing	4		
Mean		212.33		
Std. Error of Mean		23.601		
Median	100.00			
Mode	100			
Std. Deviation	305.879			
Variance	93562.038			
Skewness		3.222		
Std. Error of Skewness		.187		
Kurtosis		12.034		
Std. Error of Kurtosis		.373		
Range		1988		
Minimum		12		
Maximum	2000			
Sum		35665		
Percentiles	95	1000.00		

Acupu	incture	CHM (actual herbs)		CHM rel	ated items	Chiro	practic	Osteopathy		
Dollar (\$)	Frequency	Dollar (\$)	Frequency	Dollar (\$)	Frequency	Dollar (\$)	Frequency	Dollar (\$)	Frequency	
1	1	1	3	2	1	50	2	20	3	
20	1	10	4	15	1	55	1	50	3	
25	1	12	1	20	1	100	7	100	1	
50	2	15	2	25	1	120	1	200	1	
60	1	30	3	30	1	150	1	700	1	
70	1	40	1	50	6	200	3	2000	1	
85	1	50	9	60	1	240	1	Total	10	
100	1	60	7	100	4	300	2			
150	1	70	1	150	1	500	1			
200	1	80	1	180	1	800	1			
220	1	90	1	250	1	1000	3			
300	2	100	16	400	1	Total	23			
800	1	150	4	500	1					
1000	1	160	1	1000	1					
6000	1	200	6	6000	1					
Total	17	240	1	10000	1					
		300	2	Total	24					
		500	3							
		1000	4							
		1040	1							
		4500	1							
		10000	1							
		Total	73							

# Appendix G9 Individual expenses on four forms of CAM (unadjusted data)

CHM: Chinese herbal medicine

Statistics		Acupuncture	CHM (actual herbs)	CHM (related items)	Chiropractic	Osteopathy	
Ν	Valid	16	68	23	24	9	
	Outliers	1	2	2	0	0	
Mean		203.4229	167.56	145.97	298.65	429.57	
Std. Error of Mean		70.17481	30.151	43.233	64.154	242.280	
Median		88.13868	100.00	60.00	200.00	50.00	
Mode		50	100	50	100	20	
Std. Deviation		281.7954	248.832	208.984	311.020	722.760	
Variance		79408.65	61917.351	43674.132	96733.714	522381.856	
Skewness		2.266923	2.690	3.076	1.611	1.999	
Std. Error of Skewness		0.56242	.291	.478	.477	.720	
Kurtosis		4.676604	6.594	11.000	1.321	3.331	
Std. Error of Kurtosis		1.08719	.574	.928	.926	1.407	
Range		999	1039	998	950	1980	
Minimum		1	1	2	50	20	
Maximum		1000	1040	1000	1000	2000	
Sum		3280.237	11412	3411	7019	3823	
Percentile	95	1000.0	1000.00	869.63	1000.00	2000.00	

# Appendix G10 Statistics of expenses on four forms of CAM (adjusted data)

CHM: Chinese herbal medicine

Kendall's Tau_b		Use CAM	1	2	3	4	5	6	7	8	9	10	11	12	13
Use CAM	Correlation Coefficient	1.000													
	Sig. (2-tailed)														
	N	1067													
1. Gender	Correlation Coefficient	115(**)	1.000												
	Sig. (2-tailed)	.000													
	N	1067	1067												
2. Age	Correlation Coefficient	104(**)	038	1.000											
e	Sig. (2-tailed)	.000	.191												
	N	1064	1064	1064											
3. Region	Correlation Coefficient	036	050	002	1.000										
5	Sig. (2-tailed)	.214	.081	.950											
	N	1006	1006	1003	1006										
4. Post-secondary	Correlation Coefficient	.123(**)	.097(**)	098(**)	093(**)	1.000									
education	Sig. (2-tailed)	.000	.002	.001	.001										
	Ν	1067	1067	1064	1006	1067									
5. Employment	Correlation Coefficient	.074(*)	.170(**)	268(**)	.004	.205(**)	1.000								
	Sig. (2-tailed)	.016	.000	.000	.902	.000									
	Ν	1067	1067	1064	1006	1067	1067								
6. Income	Correlation Coefficient	.098(**)	.113(**)	203(**)	012	.277(**)	.470(**)	1.000							
	Sig. (2-tailed)	.002	.000	.000	.694	.000	.000								
	Ν	925	925	924	873	925	925	925							
7. Country of birth	Correlation Coefficient	.044	097(**)	044	002	109(**)	023	034	1.000						
	Sig. (2-tailed)	.148	.002	.131	.940	.000	.455	.277							
	Ν	1060	1060	1057	1000	1060	1060	922	1060						
8. Personal health rating	Correlation Coefficient	.021	.004	.148(**)	003	138(**)	170(**)	229(**)	.019	1.000					
	Sig. (2-tailed)	.489	.906	.000	.917	.000	.000	.000	.536						
	N	1066	1066	1063	1005	1066	1066	924	1059	1066					
9. Consulted GP in past	Correlation Coefficient	.111(**)	134(**)	.172(**)	.022	001	093(**)	069(*)	.058	.153(**)	1.000				
12 months	Sig. (2-tailed)	.000	.000	.000	.450	.972	.002	.028	.060	.000					
	Ν	1067	1067	1064	1006	1067	1067	925	1060	1066	1067				
10. Consulted GP 4	Correlation Coefficient	.080(**)	130(**)	.195(**)	.022	116(**)	224(**)	192(**)	.002	.328(**)	.394(**)	1.000			
times or more in past 12	Sig. (2-tailed)	010	000	000	449	000	000	000	946	000	000				
months		1010	10000	10000	,	1000	1000	.000	.,	1000	1000	•			
	N	1056	1056	1053	995	1056	1056	917	1049	1055	1056	1056			
11. Covered by private	Correlation Coefficient	.096(**)	039	.085(**)	.054	.185(**)	.147(**)	.300(**)	.061(*)	119(**)	.106(**)	021	1.000		
health insurance	Sig. (2-tailed)	.002	.208	.004	.058	.000	.000	.000	.048	.000	.001	.500	•		
	Ν	1061	1061	1058	1000	1061	1061	922	1055	1060	1061	1050	1061		
12. Coverage of CAM	Correlation Coefficient	.274(**)	046	164(**)	.013	.118(**)	.135(**)	.089(**)	001	031	.028	075(*)	.171(**)	1.000	
influences insurance	Sig. (2-tailed)	.000	.156	.000	.680	.000	.000	.007	.985	.338	.394	.021	.000		
purchase	N	954	954	952	899	954	954	835	948	953	954	946	951	954	
13. CAM should be	Correlation Coefficient	020	089(**)	.039	.055	057	013	037	.067(*)	.026	.044	.044	.028	034	1.000
regulated as rigorously as Western medicine	Sig. (2-tailed)	.535	.006	.208	.071	.077	.689	.268	.041	.419	.178	.176	.392	.321	
as western medicine	Ν	944	944	941	895	944	944	834	939	943	944	937	940	852	944

# Appendix G11 Inter-correlations of CAM use and potential predictor variables (blocks one, two and three)

** Correlation is significant at the 0.01 level (2-tailed).* Correlation is significant at the 0.05 level (2-tailed).

Unc.         Second         Second </th <th>Keno</th> <th>dall's Tau_b</th> <th>Use CAM</th> <th>1</th> <th>2</th> <th>3</th> <th>4</th> <th>5</th> <th>6</th> <th>7</th> <th>8</th> <th>9</th> <th>10</th> <th>11</th> <th>12</th> <th>13</th> <th>14</th> <th>15</th> <th>16</th> <th>17</th> <th>18</th>	Keno	dall's Tau_b	Use CAM	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18
No.4 with the sector of the sector	Use CAM	Correlation Coefficient	1.000																		
1. Handback         0. Map         1.		Sig. (2-tailed)																			
Interm         No         No        No         No <th< td=""><td>1 II. M. Araba</td><td>N Gambatian Gamfiniant</td><td>1067</td><td>1.000</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></th<>	1 II. M. Araba	N Gambatian Gamfiniant	1067	1.000																	
N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N	1. Health check-up	Sig (2-tailed)	024	1.000																	
1 Mathem         Norm		N	1067	1067																	
Channel         Big Long	2 6 11/6 /6	Correlation Coefficient	.016	-	1.000																
N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N	2. Cold/flu/fever	Sig. (2-tailed)	.597	.001																	
Sheak		N	1067	1067	1067																
mean         sign of the sign of	3. High blood	Correlation Coefficient	.022	045	069(*)	1.000															
Namember         No         No        No         No <t< td=""><td>pressure</td><td>Sig. (2-tailed)</td><td>.464</td><td>.138</td><td>.025</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>	pressure	Sig. (2-tailed)	.464	.138	.025																
4. Transmip     Condition Configue     Mom     - mode		N	1067	1067	1067	1067															
Name         Name <th< td=""><td><ol><li>Trauma/injury</li></ol></td><td>Correlation Coefficient</td><td>.067(*)</td><td>-</td><td>060(*)</td><td>006</td><td>1.000</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></th<>	<ol><li>Trauma/injury</li></ol>	Correlation Coefficient	.067(*)	-	060(*)	006	1.000														
5. Antimic Orticity of all		Sig. (2-tailed)	.030	.007	.049	.835															
Normal         Normal<	5 Arthritic	IN Correlation Coefficient	- 008	1007	- 068(*)	002(**)	- 050	1.000													
Name         Nom         Nom </td <td>J. Artifitis</td> <td>Sig (2-tailed)</td> <td>801</td> <td>003</td> <td>028</td> <td>.092( )</td> <td>103</td> <td>1.000</td> <td></td>	J. Artifitis	Sig (2-tailed)	801	003	028	.092( )	103	1.000													
6.4erg notes     9.4erg notes     9.4erg notes     9.4erg notes     9.4erg notes     9.4erg noteg		N	1067	1067	1067	1067	1067	1067													
Net     Net </td <td>6. Heart problems</td> <td>Correlation Coefficient</td> <td>021</td> <td>050</td> <td>048</td> <td>.030</td> <td>048</td> <td>.147(**)</td> <td>1.000</td> <td></td>	6. Heart problems	Correlation Coefficient	021	050	048	.030	048	.147(**)	1.000												
Name         No         No        No         No         No		Sig. (2-tailed)	.494	.104	.119	.322	.120	.000													
7. Garcionic Oefficien       9.02       9.01       9.09       9.09       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       9.00       <		Ν	1067	1067	1067	1067	1067	1067	1067												
No.         No. <td><ol><li>Gastrointestinal</li></ol></td> <td>Correlation Coefficient</td> <td>.052</td> <td>017</td> <td>061(*)</td> <td>.000</td> <td>023</td> <td>.009</td> <td>.013</td> <td>1.000</td> <td></td>	<ol><li>Gastrointestinal</li></ol>	Correlation Coefficient	.052	017	061(*)	.000	023	.009	.013	1.000											
Nome         No         No <td></td> <td>Sig. (2-tailed)</td> <td>.090</td> <td>.586</td> <td>.046</td> <td>.996</td> <td>.462</td> <td>.769</td> <td>.669</td> <td></td>		Sig. (2-tailed)	.090	.586	.046	.996	.462	.769	.669												
a. buttomedication         b. b. b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b. b.         b.	9 Dishatas	N Completion Coefficient	1067	1067	1067	106/	1067	1067	1067	1067	1 000										
bit         bit <td>8. Diabetes</td> <td>Sig (2 toiled)</td> <td>055</td> <td>044</td> <td>062(*)</td> <td>.147(***)</td> <td>040</td> <td>.057</td> <td>.063(*)</td> <td>.045</td> <td>1.000</td> <td></td>	8. Diabetes	Sig (2 toiled)	055	044	062(*)	.147(***)	040	.057	.063(*)	.045	1.000										
9.8.kin problems         Oracla ison Coefficient         0.11         -0.30         -0.41         -0.36         -0.36         -0.40         1.00         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01         1.01 <th1.01< th="">         1.01         1.01<td></td><td>N</td><td>1067</td><td>1067</td><td>1067</td><td>1067</td><td>1067</td><td>1067</td><td>1067</td><td>1067</td><td>1067</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></th1.01<>		N	1067	1067	1067	1067	1067	1067	1067	1067	1067										
Name         Sig 2-tailed         728         728         729         77         747         747         740         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720         720 <t< td=""><td>9. Skin problems</td><td>Correlation Coefficient</td><td>.011</td><td>030</td><td>037</td><td>011</td><td>041</td><td>010</td><td>036</td><td>004</td><td>034</td><td>1.000</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>	9. Skin problems	Correlation Coefficient	.011	030	037	011	041	010	036	004	034	1.000									
Name         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067 <th< td=""><td>· · · · · · · · · · · · · · · · · · ·</td><td>Sig. (2-tailed)</td><td>.728</td><td>.328</td><td>.228</td><td>.729</td><td>.177</td><td>.737</td><td>.245</td><td>.892</td><td>.263</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></th<>	· · · · · · · · · · · · · · · · · · ·	Sig. (2-tailed)	.728	.328	.228	.729	.177	.737	.245	.892	.263										
10. Ashma         Correlation Configuing         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -0.00         -		N	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067									
Sig. Catalak)         73         194         375         105         635         789         436         286         924         232         .         I         I         I         I         I         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         <	10. Asthma	Correlation Coefficient	009	040	.005	050	015	008	.024	033	003	.037	1.000								
N         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         10		Sig. (2-tailed)	.773	.194	.875	.105	.633	.789	.436	.286	.924	.232									
		N	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	4 000							
Base (2-tailed)         Base         Jor         Jac         Jor         Jac	11. Depression	Correlation Coefficient	.040	054	.024	.034	040	.048	.024	002	.027	.003	029	1.000							
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		Sig. (2-tailed)	.189	.077	.425	.200	.191	.120	.430	.940	.3/3	.915	.345								
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	12 Back problems	Correlation Coefficient	049	- 052	- 053	015	038	- 035	- 004	- 001	- 033	004	076(*)	006	1.000						
No.         No.         No.7         N	12. Back problems	Sig. (2-tailed)	.106	.088	.083	.616	.212	.247	.891	.973	.288	.886	.013	.834	1.000						
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		N	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067						
Sig. (2-tailed)       096       0.24       957       1.76       9.20       3.26       3.49       3.74       3.68       4.16       4.31       5.85       4.39        L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L       L <thl< th="">       L       L</thl<>	13. Pregnancy	Correlation Coefficient	.051	069(*)	.002	041	003	030	029	027	028	025	024	.017	024	1.000					
N         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         107         107         107         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         107         107         107         107         1067         1067         1067         1067         1067         1067         1067         1067         107         107         107         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087         1087		Sig. (2-tailed)	.096	.024	.957	.176	.920	.326	.349	.374	.368	.416	.431	.585	.439						
14. Cancer       Correlation Coefficient Sig. (2-tailed)       .040 $071^{(*)}$ .000 $018$ $005$ .002 $.072^{(*)}$ $028$ $026$ $025$ .015 $021$ $1.000$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $016$ $1.00$ $1.00$ $016$ $1.00$ $1.00$ $1.00$ $1.00$ $1.00$ $1.00$ $1.00$ $1.00$ $1.00$ $1.00$ $1.00$ $1.00$ $1.00$ $1.00$ $1.00$ $1.00$ $1.00$ $1.00$ $1.00$ $1$		N	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067					
Sig. (2-tailed)         J.91         J.02         J.88         J.58         J.81         J.95         J.01         J.95         J.81         J.95         J.05         J.06         J.05         J.067         J.07         J.07 <thj.07< th="">         J.07         <thj.07< th="">         J.07         J.07<td>14. Cancer</td><td>Correlation Coefficient</td><td>.040</td><td>071(*)</td><td>.000</td><td>018</td><td>005</td><td>.002</td><td>.072(*)</td><td>028</td><td>028</td><td>026</td><td>025</td><td>.015</td><td>024</td><td>021</td><td>1.000</td><td></td><td></td><td></td><td></td></thj.07<></thj.07<>	14. Cancer	Correlation Coefficient	.040	071(*)	.000	018	005	.002	.072(*)	028	028	026	025	.015	024	021	1.000				
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		Sig. (2-tailed)	.191	.021	.988	.558	.881	.956	.018	.303	.356	.405	.420	.619	.428	.502					
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	15 Currencelogia	N Correlation Coofficient	1067	1067	022	042	034	1067	1067	1067	1067	012	025	015	024	021	025	1.000			
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	15. Gynaecologic	Sig (2-tailed)	401	0/1(*)	023	042	034	956	338	028	028	.013	420	619	024 428	5021	408	1.000			
		N	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067			
Sig. (2-tailed)       .286       .291       .151       .630       .288       .338       .361       .117       .746       .428       .550       .443       .043       .523       .513       .513            17. High cholesteri       N       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067	16. Other pains	Correlation Coefficient	.033	032	044	015	033	029	028	.048	.010	024	.018	024	.062(*)	020	020	020	1.000		
N         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         1067         10	· ·	Sig. (2-tailed)	.286	.291	.151	.630	.288	.338	.361	.117	.746	.428	.550	.443	.043	.523	.513	.513			
17. High cholesterol       Correlation Coefficient      018      021      039       .107(**)      029       .012       .015      024       .099(**)      022       .026       .026       .021      017      018       .0.18       .040       1.000         Sig. (2-tailed)       .567       .492       .200       .000       .343       .701       .632       .439       .001       .479       .402       .501       .568       .559       .559       .194       .         N       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067       1067 <th< td=""><td></td><td>Ν</td><td>1067</td><td>1067</td><td>1067</td><td>1067</td><td>1067</td><td>1067</td><td>1067</td><td>1067</td><td>1067</td><td>1067</td><td>1067</td><td>1067</td><td>1067</td><td>1067</td><td>1067</td><td>1067</td><td>1067</td><td></td><td></td></th<>		Ν	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067		
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	17. High cholesterol	Correlation Coefficient	018	021	039	.107(**)	029	.012	.015	024	.099(**)	022	.026	.026	021	017	018	018	.040	1.000	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		Sig. (2-tailed)	.567	.492	.200	.000	.343	.701	.632	.439	.001	.479	.402	.402	.501	.568	.559	.559	.194		
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	10 I	N Or sealed in a Confficient	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1 000
$Sig_{1,2}$ (2-tanou) $U^{T1}$ (2-17 1.54 2.54 $U^{U1}$ (2.54 $U^{U2}$ (4.50 (4.54) $U^{H2}$ (4.54 $U^{H2}$ (4.54 $U^{H2}$ (4.55 $U^{H2}$ (4.56) $U^{H2}$ (4.56) $U^{H2}$ (4.56) $U^{H2}$ (4.57) $U^{H2}$ (4.	16. Lung problems	Sig (2-tailed)	.063(*)	038	010	035	.008	025	024	025	025	.020	.070(*)	020	.029	017	017	017	.101(***)	015	1.000
N 1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067   1067		N	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067	1067

# Appendix G11 Inter-correlations of CAM use and potential predictor variables (continued, block four)

* Correlation is significant at the 0.05 level (2-tailed). ** Correlation is significant at the 0.01 level (2-tailed).

### Appendix G12 Statistics of a regression model on CAM use

(based on unweighted data)

**Case Processing Summary** 

Unwei	ghted Cases(a)	Ν	Percent
Selected Cases Included in Analysis		819	76.8
	Missing Cases	248	23.2
	Total	1067	100.0
Unselected Cases		0	.0
Total		1067	100.0

a If weight is in effect, see classification table for the total number of cases.

### Dependent Variable Encoding

Original Value	Internal Value
No	0
Yes	1

#### Categorical Variables Codings

			Paramete	er coding
		Frequency	(1)	(2)
6. Income	Less than \$20,000	148	.000	.000
	\$20,000-\$40,000	171	1.000	.000
	Over \$40,000	500	.000	1.000
2. Age	18 - 34 years	249	.000	.000
	35 - 64 years	451	1.000	.000
	65 years+	119	.000	1.000

### **Block 0: Beginning Block**

Iteration History (a,b,c)

		Coefficients
Iteration	-2 Log likelihood	Constant
Step 0 1	1006.467	.784
2	1006.125	.828
3	1006.125	.828

a Constant is included in the model.

b Initial -2 Log Likelihood: 1006.125

c Estimation terminated at iteration number 3 because parameter estimates changed by less than .001.

Classification Table (a,b)

Observed				Predicted					
			Use CAM			Percentage Correct			
			No		Yes				
Step 0	Use CAM	No		0	249	.0			
		Yes		0	570	100.0			
	Overall Percentage					69.6			

a Constant is included in the model. b The cut value is .500

Variables in the Equation

		В	S.E.	Wald	df	Sig.	Exp(B)
Step 0	Constant	.828	.076	118.862	1	.000	2.289

#### Block 1: Method = Forward Stepwise (Wald)

Iteration History (a,b,c,d)

-2 Log			Coefficients							
neration		likelinood	Coefficients							
				Postsec						
			Constant	ondary	Gender	Age(1)	Age(2)			
Step 1	1	986.932	.454	.585						
	2	985.843	.462	.686						
	3	985.842	.462	.689						
	4	985.842	.462	.689						
Step 2	1	972.382	.190	.609	.497					
	2	970.442	.151	.728	.605					
	3	970.439	.149	.733	.610					
	4	970.439	.149	.733	.610					
Step 3	1	966.398	.411	.569	.510	241	492			
	2	963.989	.432	.685	.627	315	600			
	3	963.983	.434	.691	.634	320	607			
	4	963.983	.434	.691	.634	320	607			

a Method: Forward Stepwise (Wald); b Constant is included in the model.

c Initial -2 Log Likelihood: 1006.125; d Estimation terminated at iteration number 4 because parameter estimates changed by less than .001.

#### **Omnibus Tests of Model Coefficients**

		Chi-square	df	Sig.
Step 1	Step	20.282	1	.000
	Block	20.282	1	.000
	Model	20.282	1	.000
Step 2	Step	15.403	1	.000
	Block	35.686	2	.000
	Model	35.686	2	.000
Step 3	Step	6.456	2	.040
	Block	42.142	4	.000
	Model	42.142	4	.000

Model Summary

Step	-2 Log likelihood	Cox & Snell R Square	Nagelkerke R Square
1	985.842(a)	.024	.035
2	970.439(a)	.043	.060
3	963.983(a)	.050	.071

a Estimation terminated at iteration number 4 because parameter estimates changed by less than .001.

Hosmer and Lemeshow Test

Step	Chi-square	df	Sig.
1	.000	0	
2	1.384	2	.501
3	2.897	6	.822

Contingency Table for Hosmer and Lemeshow Test

			Use CAM = No		Use CAM = Yes		
		Observed	Expected	Observed	Expected	Total	
Step 1	1	138	138.000	219	219.000	357	
_	2	111	111.000	351	351.000	462	
Step 2	1	74	77.749	94	90.251	168	
	2	64	60.251	125	128.749	189	
	3	74	70.251	166	169.749	240	
	4	37	40.749	185	181.251	222	
Step 3	1	56	58.127	63	60.873	119	
_	2	33	35.124	57	54.876	90	
	3	42	41.960	85	85.040	127	
	4	49	43.272	91	96.728	140	
	5	32	30.122	89	90.878	121	
	6	8	6.724	20	21.276	28	
	7	18	22.060	97	92.940	115	
	8	11	11.611	68	67.389	79	

	Classification Table (a)						
	Observed			Predicted	d		
			Use	CAM	Percentage Correct		
			No	Yes			
Step 1	Use CAM	No	0	249	.0		
		Yes	0	570	100.0		
	Overall Percentage				69.6		
Step 2	Use CAM	No	0	249	.0		
		Yes	0	570	100.0		
	Overall Percentage				69.6		
Step 3	Use CAM	No	16	233	6.4		
		Yes	12	558	97.9		
	Overall Percentage				70.1		

a: The cut value is .500

	Variables in the Equation								
		В	S.E.	Wald	df	Sig.	Exp(B)	95.0% C.I.	for EXP(B)
								Lower	Upper
Step 1(a)	Postsecondary	.689	.154	20.081	1	.000	1.993	1.474	2.694
	Constant	.462	.109	18.055	1	.000	1.587		
Step 2(b)	Gender	.610	.157	15.129	1	.000	1.841	1.354	2.504
	Postsecondary	.733	.156	22.072	1	.000	2.082	1.533	2.826
	Constant	.149	.135	1.227	1	.268	1.161		
Step 3(c)	Gender	.634	.158	16.062	1	.000	1.885	1.382	2.570
	Age			6.392	2	.041			
	Age(1)	320	.183	3.074	1	.080	.726	.507	1.039
	Age(2)	607	.247	6.047	1	.014	.545	.336	.884
	Postsecondary	.691	.158	19.228	1	.000	1.995	1.465	2.717
	Constant	.434	.188	5.329	1	.021	1.543		

a Variable(s) entered on step 1: Postsecondary. b Variable(s) entered on step 2: Gender. c Variable(s) entered on step 3: Age.

### Block 2: Method = Forward Stepwise (Wald)

### Omnibus Tests of Model Coefficients

		Chi-square	df	Sig.
Step 1	Step	11.686	1	.001
	Block	11.686	1	.001
	Model	53.828	5	.000
Step 2	Step	4.085	1	.043
	Block	15.771	2	.000
	Model	57.912	6	.000

Model Summary

Step	-2 Log likelihood	Cox & Snell R Square	Nagelkerke R Square
1	952.297(a)	.064	.090
2	948.212(a)	.068	.097

a Estimation terminated at iteration number 4 because parameter estimates changed by less than .001.

### Hosmer and Lemeshow Test

Step	Chi-square	df	Sig.
1	5.582	8	.694
2	7.380	8	.496

### Contingency Table for Hosmer and Lemeshow Test

		Use CA	e CAM = No Use CAM = Yes			
		Observed	Expected	Observed	Expected	Total
Step 1	1	40	41.660	37	35.340	77
	2	33	35.177	48	45.823	81
	3	32	29.311	41	43.689	73
	4	27	26.288	51	51.712	78
	5	29	27.993	63	64.007	92
	6	33	33.658	89	88.342	122
	7	25	18.830	57	63.170	82
	8	6	9.628	42	38.372	48
	9	16	18.296	86	83.704	102
	10	8	8.159	56	55.841	64
Step 2	1	40	41.798	37	35.202	77
	2	41	39.094	46	47.906	87
	3	26	26.926	45	44.074	71
	4	28	25.543	45	47.457	73
	5	15	17.063	39	36.937	54
	6	23	25.044	61	58.956	84
	7	34	25.093	66	74.907	100
	8	12	14.668	55	52.332	67
	9	12	15.873	67	63.127	79
	10	18	17.898	109	109.102	127

	Classification Table (a)						
Observed			Predicted				
		Use	CAM	Percentage Correct			
			No	Yes			
Step 1	Use CAM	No	31	218	12.4		
		Yes	24	546	95.8		
	Overall Percentage				70.5		
Step 2	Use CAM	No	24	225	9.6		
		Yes	15	555	97.4		
	Overall Percentage				70.7		

#### Classification Table (a)

a: The cut value is .500

	variables in the Equation								
		В	S.E.	Wald	df	Sig.	Exp(B)	95.0% C.I.	for EXP(B)
								Lower	Upper
Step 1(a)	Gender	.541	.161	11.221	1	.001	1.717	1.252	2.357
	Age			9.560	2	.008			
	Age(1)	403	.187	4.657	1	.031	.668	.464	.964
	Age(2)	764	.253	9.108	1	.003	.466	.283	.765
	Postsecondary	.694	.159	19.067	1	.000	2.001	1.466	2.732
	ConsultedGPinpast12months	.676	.196	11.876	1	.001	1.966	1.338	2.887
	Constant	.013	.224	.003	1	.953	1.013		
Step 2(b)	Gender	.517	.162	10.180	1	.001	1.678	1.221	2.305
	Age			10.998	2	.004			
	Age(1)	418	.187	4.997	1	.025	.658	.456	.950
	Age(2)	838	.257	10.616	1	.001	.433	.261	.716
	Postsecondary	.722	.160	20.338	1	.000	2.058	1.504	2.816
	ConsultedGPinpast12months	.522	.209	6.213	1	.013	1.686	1.118	2.541
	ConsultedGP4timesormoreinpast12months	.370	.184	4.045	1	.044	1.447	1.009	2.075
	Constant	016	225	005	1	943	1 016		

#### Variables in the Equation

a Variable (s) entered on step 1: ConsultedGPinpast12months. b Variable (s) entered on step 2: ConsultedGP4timesormoreinpast12months.

### **Block 3: Method = Forward Stepwise (Wald)**

Omnibus Tests of Model Coefficients

		Chi-square	df	Sig.
Step 1	Step	44.911	1	.000
	Block	44.911	1	.000
	Model	102.823	7	.000

Model Summary

Step	-2 Log likelihood	Cox & Snell R Square	Nagelkerke R Square
1	903.302(a)	.118	.167

a Estimation terminated at iteration number 4

because parameter estimates changed by less than .001.

Hosmer and Lemeshow Test

Step	Chi-square	df	Sig.	
1	4.228	8	.836	

Classification Table (a)

	Observed			Predicted			
			Use	Percentage Correct			
			No	Yes			
Step 1	Use CAM	No	66	183	26.5		
		Yes	46	524	91.9		
	Overall Percentage	2			72.0		

a The cut value is .500

				-					
		В	S.E.	Wald	df	Sig.	Exp(B)	95.0% C.I. for EXP(B)	
								Lower	Upper
Step 1(a)	Gender	.477	.167	8.181	1	.004	1.612	1.162	2.235
	Age			6.160	2	.046			
	Age(1)	331	.193	2.930	1	.087	.718	.492	1.049
	Age(2)	644	.265	5.902	1	.015	.525	.312	.883
	Postsecondary	.640	.165	15.064	1	.000	1.896	1.372	2.618
	ConsultedGPinpast12months	.461	.217	4.521	1	.033	1.586	1.037	2.427
	ConsultedGP4timesormoreinpast12months	.464	.190	5.976	1	.014	1.590	1.096	2.306
	Coverage of CAM influences insurance	1.096	.167	42.955	1	.000	2.991	2.155	4.150
	Constant	492	.244	4.066	1	.044	.611		

Variables in the Equation

a Variable(s) entered on step 1: Coverage of CAM influences insurance.

	Selected	Observed		Predicted	Temporary Variable		
Case	Status(a)	Use CAM	Predicted	Group	Resid	ZResid	
5	S	N**	.899	Y	899	-2.977	
94	S	N**	.899	Y	899	-2.977	
176	S	N**	.899	Y	899	-2.977	
177	S	N**	.864	Y	864	-2.523	
302	S	N**	.934	Y	934	-3.753	
377	S	N**	.881	Y	881	-2.726	
547	S	N**	.864	Y	864	-2.523	
574	S	N**	.899	Y	899	-2.977	
585	S	N**	.881	Y	881	-2.726	
683	S	N**	.881	Y	881	-2.726	
691	S	N**	.899	Y	899	-2.977	
987	S	N**	.897	Y	897	-2.957	
1005	S	N**	.897	Y	897	-2.957	

Block 4: Method = Forward Stepwise (Wald): none entered Casewise List (b)

a: S = Selected, U = Unselected cases, and ** = Misclassified cases. b: Cases with studentised residuals greater than 2.000 are listed.

Residual and scatter plot of the predictive probability – Regression analysis



Histogram

Normal P-P Plot of Standardised Residual

Scatter plot